

Technische Universität München TUM School of Engineering and Design

To Wait or Not to Wait? Redirecting Passengers and Reallocating Capacities During Incidents in Public Transport Systems

Frederik Reinhard Bachmann

Vollständiger Abdruck der von der TUM School of Engineering and Design der Technischen Universität München zur Erlangung eines *Doktors der Ingenieurwissenschaften (Dr.-Ing.)* genehmigten Dissertation

Vorsitz: Prof. Dr.-Ing. Klaus Bogenberger

Prüfende der Dissertation:

- 1. Prof. Dr.-Ing. Fritz Busch
- 2. Associate Prof. Zhiwei (David) Wang, Ph.D.
- 3. Prof. Dr.-Ing. Gebhard Wulfhorst

Die Dissertation wurde am 25.05.2023 bei der Technischen Universität München eingereicht und durch die *TUM School of Engineering and Design* am 08.10.2023 angenommen.

"If you get tired, learn to rest, not to quit." – Bansky

Acknowledgements

In the course of the past five years during which I developed this dissertation, a lot of people talked to me about it, gave me feedback, gave me advice, comforted me, and motivated me. I am thankful for all the support of any kind I got and would like to thank some people in particular in the following.

First and foremost, I would like to express my gratitude to my first supervisor, Fritz Busch, for agreeing to supervise my doctorate, for his wise and kind feedback, for supporting me, and motivating me to achieve the doctoral degree. Moreover, I want to thank him for opening the door to my Singapore adventure. The possibility of conducting my research for more than two years in Singapore was an unforgettable experience professionally and personally.

Then, I want to thank my second supervisor for accepting the supervision of my thesis and by that enabling me to participate in the joint Ph.D. programme of the Technical University of Munich (TUM) and the Nanyang Technological University (NTU) of Singapore. Additionally, I am also thankful for his support during the application process of the public transport user survey conducted in Singapore.

In addition, I would like to thank my two mentors: Avishai Ceder for his advice and feedback whenever I needed it, and Antonios Tsakarestos for always having an open ear for me and listening to issues I was facing concerning my thesis or other topics. And, for motivating me to challenge my ideas and pointing out when I was wrong in a very kind manner, as well as for his support to develop and shape our literature review paper.

Furthermore, I want to thank my former team lead at TUMCREATE, Andreas Rau, for motivating me to accept Fritz Busch's invitation to Singapore, and for his feedback and kind attitude.

Moreover, thank you to my superior, Klaus Bogenberger, for giving me the opportunity to rejoin the Chair of Traffic Engineering and Control at TUM and for being the chairman of my doctoral defence.

Next, thank you to Gebhard Wulfhorst, for accepting to join and complete the examination board of my thesis.

Additionally, I want to thank some of my co-authors for their support and feedback, Lars Briem, Ganesh Sethuraman, Florian Dandl, Arslan Ali Syed, and Roman Engelhardt, I learned a lot from them, and publishing together helped me to improve my skills in research and writing.

Also, I want to thank my former colleague, Penelope Kong, for her cooperation in conducting the public transport user survey.

Further, thank you to the proofreaders of my thesis, Margarete Bachmann, Patrick Malcolm, Patrick Paap Peretzki, Gabriel Tilg, Yannik Süß, Lars Briem, Madhur Jain, Johanna Vonderhecken, Fabian Fehn, Sebastian Stadler, and Mario Ilic for giving me valuable feedback which helped to improve this work.

Then, I want to thank my former and current colleagues at the chair at TUM for making the time more fun as well as their support whenever it was needed. A special thank you goes to Gabriel Tilg, Mario Ilic, Philipp Stüger, Sasan Amini, and Fabian Fehn. Also, I especially thank Matthias Spangler, who lured me to the chair shortly before I gained my master's degree and thereby kicked off my doctoral journey. More importantly for his open ear at the beginning of this journey when I was struggling with my decision to pursue the doctoral degree.

Moreover, I am very thankful for my former colleagues at TUMCREATE in Singapore, especially for the "usual suspects", who are rather friends than colleagues and made my Singapore adventure what it was. A blast!

To broaden our understanding of incident management in public transport, Lars Briem and I visited the operations control centre of several operators (BVG, KVV, MVG, VAG, SBSTransit, and Tower Transit) in Germany and Singapore. Therefore, I want to thank the managing staff for taking the time for our interviews, as well as the dispatchers for allowing us to observe them during their work and answering our questions. A special thank you goes to Daniela Linke (BVG) for her support in setting up the platform for the public transport user survey conducted in Berlin and Hamburg.

Furthermore, I would like to thank the city administration of Kassel, the Kasseler transport company (KVG), and the Gertz Gutsche Rümenapp GbR for providing me with the transport model of Kassel, Germany.

In addition, the research for this thesis started at TUMCREATE in Singapore and was supported in part by the National Research Foundation (NRF) Singapore under its Campus for Research Excellence and Technological Enterprise (CREATE) programme.

Writing a doctoral thesis can be very frustrating at times and caters for long working hours, especially during the final stage. No matter which mood I was in and for how long I was sitting at the desk, Laurene Scheurell, my partner in life, love, and silliness, kept telling me that I can do this and that she is very proud of me while she kept my back from the usual everyday tasks. I am very grateful to have her by my side.

Most importantly I want to express my deep gratitude towards my parents, Margarete and Dietrich Bachmann, as well as my sister, Johanna Vonderhecken, for their unconditional support from day one, through my maturation, school, university, and doctorate. It is extremely re-ensuring to know that no matter what happens there is a solid safety net, which is always ready to catch me when I fall and motivate me to get up again.

Lastly, I want to use this opportunity to express my gratefulness towards this planet we all live on, Earth, which provides us every day with all we need. I sincerely hope that we come to realise how lucky we are to be here and that our way of life will give evidence of our responsibility to the planet we live on very soon.

Frederik Bachmann

Executive Summary

Humanity's greatest challenge is to adjust its way of life in a way that planet Earth stays habitable for us as well as for other living beings. Consequently, the emissions we produce need to be drastically reduced in various sectors of our daily lives. In the passenger transport sector, this can, besides avoiding trips, be accomplished by a change in our transport habits, for example, by a shift towards more environmentally friendly modes of transport such as public transport (PT). An improvement in the reliability of PT services raises its attractiveness and can motivate people to use it more often. Incidents (e.g. traffic accidents, technical failures, passenger falls, etc.) occur every day in PT operation and lower its reliability.

In literature, there are many investigations focussing on *supply-centric* incident management which refer to methods that are used for readjusting the PT supply following an incident to mitigate its negative effects such as passenger delays. Significantly less has been done in the area of *passenger-centric* incident management, which is understood here as methods that include public transport (PT) users in the reduction of their delays, for instance, by redirecting them through the provision of path advice.

This thesis introduces a novel contribution to this research area. In addition to an extensive literature review about incident management in PT systems, findings from visits to operations control centres, complement the knowledge gained through the literature. Moreover, a PT user survey concerning passengers' perception of incidents and incident management was conducted to better understand the passenger side during incidents.

Most importantly, the gathered knowledge and gained insight in the practice of incident management and the perception of PT users during incidents have been used to develop the model introduced here. This model redirects affected passengers onto available alternative paths during incidents considering their remaining capacities. This reduces passenger delays. Furthermore, the model includes the reallocation of capacities in the forms of buses during incidents to support passenger redirection with extra capacities on the suggested alternative paths. The model is tested in the Mandl network considering a PT bus system as well as an autonomous PT system and in a more complex real network of the city of Kassel, Germany.

The results show the great potential of this model to reduce the relative delay of affected passengers through capacity-aware redirection of passengers. The effectiveness of the model grows with an increase in the severity and duration of an incident as well as with the share of travellers who follow path advice. However, further testing and development are needed as limitations of the model have been identified and room for further research has been recognised.

On the whole, this work represents a new step towards implementing passenger redirection and capacity reallocation in practice and is a contribution to close the gap between research and practice in *passenger-centric* incident management in PT systems.

Kurzfassung

Die größte Herausforderung der Menschheit besteht darin, ihre Lebensweise so zu gestalten, dass der Planet Erde für uns und andere Lebewesen bewohnbar bleibt. Daher müssen unter anderem die von uns verursachten Emissionen in verschiedenen Bereichen unseres täglichen Lebens drastisch reduziert werden. Im Bereich des Personenverkehrs kann dies, neben der Vermeidung von Fahrten, durch eine Veränderung unserer Mobilitätsgewohnheiten erreicht werden. Zum Beispiel durch eine Verlagerung auf umweltfreundlichere Verkehrsmittel wie den öffentlichen Verkehr (ÖV). Eine Verbesserung der Zuverlässigkeit des ÖV erhöht dessen Attraktivität und kann Leute dazu motivieren diesen öfter zu nutzen. Störungen wie Verkehrsunfälle, technische Ausfälle und Fahrgaststürze treten täglich im ÖV-Betrieb auf und verringern dessen Zuverlässigkeit.

In der Literatur finden sich zahlreiche Untersuchungen, die sich mit dem *betriebszentrierten* Störungsmanagement befassen. Damit sind Methoden zur Anpassung des ÖV-Angebots an eine Störung gemeint, die deren negative Auswirkungen wie Verspätungen abmildern. Deutlich weniger geforscht wurde im Bereich des *fahrgastzentrierten* Störungsmanagements. Darunter werden Methoden verstanden, die betroffene Fahrgäste in die Reduzierung ihrer Verspätungen einbeziehen. Zum Beispiel durch deren Umleitung mit Hilfe von Routenempfehlungen.

Die vorliegende Arbeit leistet einen Beitrag zu diesem Forschungsgebiet. Neben einer umfassenden Literaturrecherche zum Thema Störungsmanagement im ÖV, ergänzen Erkenntnisse aus Besuchen in Leitzentralen die durch die Literatur gewonnenen Erkenntnisse. Darüber hinaus wurde eine Umfrage unter ÖV-Nutzern zur Wahrnehmung von Störungen und Störungsmanagement durchgeführt, um ein besseres Verständnis der Fahrgastseite zu erhalten.

Der Kernbeitrag dieser Arbeit ist ein Modell, in dessen Entwicklung das gesammelte Wissen und die gewonnenen Erkenntnisse geflossen sind. Das Modell leitet Fahrgäste während einer Störung auf Alternativrouten, unter Berücksichtigung deren Kapazitäten, um. Außerdem beinhaltet das Modell die Neuzuweisung von Kapazitäten, um die Fahrgastumleitung zu unterstützen. Das Modell wird im Mandl-Netz unter Berücksichtigung eines Bussystems sowie eines autonomen ÖV-Systems und in einem komplexeren realen Netz der Stadt Kassel getestet.

Die Ergebnisse zeigen das große Potenzial dieses Modells die relative Verspätung der betroffenen Fahrgäste durch kapazitätsberücksichtigende Umleitung der Fahrgäste zu reduzieren. Die Wirksamkeit des Modells nimmt mit der Schwere und Dauer einer Störung sowie mit dem Anteil der Fahrgäste, die den angebotenen Routenempfehlungen folgen, zu. Es sind jedoch weitere Tests und Entwicklungen erforderlich, da auch Grenzen des Modells identifiziert und Raum für weitere Forschungsansätze erkannt wurden.

Insgesamt stellt diese Arbeit einen neuen Schritt zur Einführung der Umleitung von Fahrgästen und der Neuzuweisung von Kapazitäten in die Praxis dar und ist ein Beitrag zum Schließen der Lücke zwischen Forschung und Praxis im Bereich des *fahrgastzentrierten* Störungsmanagements im ÖV.

Contents

1	Intr	oduction	1				
	1.1	Motivation	1				
	1.2	Goals, Scope and Research Questions	2				
	1.3	Overall Approach and Thesis Structure	4				
2	Stat	te of the Art Analysis	7				
	2.1	Background	7				
		2.1.1 Passenger Demand Assignment	7				
		2.1.2 Shortest Path Search and Path Choice	8				
		2.1.3 Incidents in Public Transport	0				
		2.1.4 Passengers' Reactions to Incidents	2				
		2.1.5 Passenger Information	4				
	2.2	Incident Management in Public Transport	6				
		2.2.1 Supply-centric Incident Management	6				
		222 Passenger-centric Incident Management 2	3				
	23	Summary and Research Gaps 4	6				
			Ũ				
3	Pas	senger Perceptions of Incidents and Incident Management 4	7				
	3.1	Passenger Survey Setup	7				
	3.2	Survey Results	8				
		3.2.1 Sample Description	9				
		3.2.2 Experienced Incidents	2				
		3.2.3 Passengers' Preferences	6				
		3.2.4 Perceived Emotions	0				
		3.2.5 Correlations	5				
Δ	Μο	del Formulation 7	1				
•	<i>A</i> 1	Real Life Story 7	∎ 1				
	4.1 4.2	Problem Statement 7	י ז				
	т. <u>∠</u> // З	Network Graph	7				
	т.5 Л Л	Situation Analysis					
	т.т 1 Б	Passanger Redirection	1				
	4.5		1 1				
		$4.5.1 \text{Intuition Approach} \qquad \qquad$	丘				
	16	$4.3.2 \text{Optimisation Approach} \dots \dots \dots \dots \dots \dots \dots \dots \dots $	5				
	4.0 17	Information Dissemination	1				
	4.1	Information Dissemination	Т				

5	Case 5.1 5.2	Se Studies and Model Evaluation Model Implementation Implementation in the Mandl Network 5.2.1 Scenarios and Cases 5.2.2 Results of the Case Study in the Mandl Network 5.2.3 Implementation in an Autonomous PT System	93 93 96 98 102 143
	5.3 5.4	Implementation in the Network of the City of Kassel	148 150 151 154
6	Disc 6.1 6.2 6.3	cussion Model Limitations Contribution and Comparison Implementation in Practice	157 157 160 165
7	Con 7.1 7.2 7.3	nclusions Summary	171 171 173 174
Bi	bliog	graphy	177
Lis	st of	Abbreviations	197
Lis	st of	Symbols	201
Lis	st of	Figures	205
Lis	st of	Tables	209
Ar	inexe	e	211
	A B C	Incident Examples in Detail	211 212 212 217 219 220 220 222 224

Chapter 1

Introduction

The first chapter of this thesis introduces the reader to the topic of the investigation conducted, describes the motivation which has led to the goals and research questions of this work, as well as explains the applied methodical approach and structure of the thesis.

1.1 Motivation

The greatest challenge of our time is to adapt the human way of life to a sustainable way of living so that planet Earth remains habitable for our kind and for other living beings. Besides the exploitation of the limited resources of planet Earth [MEADOWS, 1972; GLOBAL FOOTPRINT NETWORK, 2022], is the anthropogenic warming of the global temperature, called climate change, a key problem. Especially through the combustion of fossil fuels such as oil, gas, and coal, mankind artificially increases the concentration of greenhouse gases like carbon dioxide and others in the atmosphere, which boosts the greenhouse effect on the planet [HOFMANN et al., 2006].

Over a century ago, scientists and society already started to discuss and warn about the influences of the combustion of fossil fuels on the climate [ARRHENIUS, 1896; CALLEN-DAR, 1938; REVKIN, 2016]. In the scientific world, there is a consensus of over 97% on the statement that humans cause global warming [COOK et al., 2013]. As climate change causes devastating consequences such as extreme weather events, dramatic loss of biodiversity, destruction of whole biotopes, as well as severe negative effects on our health, the Intergov-ernmental Panel on Climate Change (IPCC) recommends drastic changes in various sectors of our way of living to mitigate these consequences [MCMICHAEL, 2003; CEBALLOS et al., 2015; HUGHES et al., 2017; IPCC, 2018].

Transportation is one of these sectors which play a significant role in the emission of greenhouse gases. Therefore, a change in our mobility behaviour is necessary to reduce the emissions in this sector [WATERSON et al., 2003; CHAPMAN, 2007; EUROPEEAN ENVIRONMENT AGENCY, 2022]. To achieve this goal a reduction of travel as well as a switch to means of transport which are more compatible with the limits of Earth and our budget for the emission of greenhouse gases is inevitable.

In passenger transport, more of people's trips, need to be conducted with more environmentally friendly means of transport. For short distances, active mobility such as walking and cycling are the most obvious choices. For longer distances, PT, supported by shared mobility services, is the most obvious alternative [WATERSON et al., 2003; CHAPMAN, 2007; SCHRÖDER et al., 2022]. SCHRÖDER et al. [2022] point out that apart from the ecological costs, also the economic costs for society can be reduced by shifting from motorized private transport to PT and active mobility.

So-called push and pull strategies can support a change in the mobility behaviour of people. Push strategies refer to measures that lower the attractiveness of a transport system such as applying restrictions and reducing investments in the corresponding infrastructure and thus "pushing" travellers away from certain means of transport; pull strategies refer to measures which increase the attractiveness of a system by for instance lowering fares or improving infrastructure and thus "pulling" travellers towards a certain means of transport [HRELJA and RYE, 2022; SOMMER et al., 2022]. Push strategies, for instance, restrictions on motorized private transport can trigger anger and frustration in the society as it contradicts with people's mobility habits [BUEHLER et al., 2017]. Pull strategies such as improving the attractiveness of PT systems, however, give people incentives to change their mobility habits. Among other options, it is therefore expedient to improve the attractiveness of the aforementioned means of passenger transport like PT as well as shared and active mobility such as cycling and walking.

1.2 Goals, Scope and Research Questions

This thesis focuses on urban mass PT, such as bus, tram, metro, and comparable services. The attractiveness of a PT system rely on several aspects such as its flexibility, fares, and reliability. A PT system's attractiveness can be raised by lower fares or improve its flexibility by raising the services' frequency and extend its network [BAUM, 1973; CATS, SUSILO et al., 2017; VAN LIEROP et al., 2018; MIETZSCH, 2019; ANDOR et al., 2021].

Another possibility is to improve its reliability. The term reliability can be interpreted in different ways, in the context of PT. On the one hand, there is the perspective of the PT operator who understands reliability as maintaining and operating the PT system as scheduled [BRIEM, 2023]. On the other hand, there is the perspective of the PT users on which this thesis focuses.

For travellers, reliability was stated as one of the most important characteristics of transport systems [PRASHKER, 1979; OUTWATER et al., 2011; CARREL, HALVORSEN et al., 2013; ALLEN et al., 2019; HELFERS et al., 2022]. From a PT users perspective there are various aspects to the reliability of a PT system. ALLEN et al. [2019] for example list access to transport, availability (time interval), speed, reliability of arrival time, and easiness of transfer. PRASHKER [1979] considers a higher level of detail and differs between the reliability of different parts of travel time (i.e. walking, waiting, in-vehicle time) which the author names under travel-time reliability. Furthermore, mechanical reliability referring to that a vehicle is mechanically fit to serve, performance reliability referring to a transport service performing as estimated, weather influence, and others are taken into account. Reliability can therefore also be understood in the context of a system being fit to serve as planned.

CARREL, HALVORSEN et al. [2013] states that a clear definition in the context of passenger transport is yet missing in literature. In the following, it is focussed on the part of reliability that refers to reliable travel time and arrival time of PT users.

It is also mentioned by CARREL, HALVORSEN et al. [2013] that some forms of unreliability such as delays due to congestion during peak hours might not be perceived as unreliable by daily commuters as they expect those during peak hours. For people who are not used to the dynamics in PT, also to be expected delays during peak hours are probably perceived as unreliable. However, if unreliability is caused by unplanned and unpredicted events such as incidents, it is perceived as such by both, the passengers as well as the operators [CARREL, HALVORSEN et al., 2013; BACHMANN, BRIEM et al., 2022; BRIEM, 2023].

Incidents are events such as traffic accidents, deployment of emergency forces, technical failures, falling passengers, etc. which disrupt the PT service. Further examples and explanations of incidents are given in Section 2.1.3. They occur daily in PT operations and lower its reliability and thus its attractiveness. Therefore, it is desirable to keep their negative effects on PT services, such as delays and cancellations, as low as possible.

The field of research which focuses on the dissolution of incidents and the mitigation of their negative effects is called incident management. It is the umbrella goal of the method introduced here to improve incident management in PT and thus to improve the reliability of PT systems. In more detail, to include the passengers affected by an incident in the incident management and the reduction of delays.

As Section 2.2.1 shows, a lot of research has been done in the area of incident management regarding the operator or supply side, respectively. This research finds answers to the question how the service can be temporally adjusted to the disturbances caused by an incident? Hence, this is referred to as *supply-centric* incident management in the following. Less has been done on the passenger side.

This leads to another goal of this thesis which is the understanding of the passengers' perspective during incidents. The general idea of the model presented in Chapter 4 is to redirect passengers affected by an incident by providing them with concrete path advice. This advice guides them through the PT network during incidents and reduces their overall delay. The area of research which includes affected passengers in incident management and the reduction of their delays by for example redirecting them is understood here as *passenger-centric* incident management (PCIM).

In addition, one hypothesis is that by a complementary reallocation of capacities, in the forms of PT vehicles, the positive effects of passenger redirection can be further improved. To better understand the effectiveness of passenger redirection and capacity reallocation during incidents in PT systems, this thesis aims to answer the following research questions:

- 1. To what extent does path advice lower the overall passenger delay during incidents?
- 2. Under which circumstances is the redirection of passengers reasonable?
- 3. What are factors that influence the effectiveness of passenger redirection?
- 4. Does capacity reallocation support passenger redirection?

The development of this thesis started as part of the project "Towards the Ultimate Public Transport System" conducted by the research company TUMCREATE Limited, in which an

autonomous PT system, the Dynamic Autonomous Road Transit (DART) system, has been investigated [RAU, TIAN et al., 2018; RAU, JAIN et al., 2019; TUMCREATE, 2021]. Therefore, it is a subsidiary aim of the work presented here to understand incidents and incident management in the context of autonomous PT systems.

1.3 Overall Approach and Thesis Structure

To accomplish the aforementioned goals and to answer the listed research questions in the previous section the methodology presented here has been developed.

Figure 1.1 shows the structure of this thesis and the connections of its elements. The shades of grey indicate of the thematic depth of a particular element. After this introductory Chapter 1, Section 2.1 gives insight into relevant adjacent research terms and topics as background. Furthermore, Section 2.2.1 explains the state-of-the-art of *supply-centric* incident management through a corresponding literature review as well as findings through visits to the operations control centre (OCC) of several PT operators. This helps to better understand the supply side during incidents. The same section also describes the state-of-the-art in PCIM, which is the research area this thesis belongs to. As it is indicated by Figure 1.1, PCIM connects the supply side with the passenger side.

To better understand the passenger side during incidents, a PT user survey was conducted of which setup and results are presented in Chapter 3.

Based on the knowledge gathered through Chapters 2 and 3, a model for passenger redirection and capacity reallocation has been developed which is formulated in Chapter 4 and evaluated in Chapter 5.

Chapter 6 discusses the limitations of the model and compares it to other works in the field of PCIM.

Chapter 7 concludes the work and gives incentives for future research.

To avoid misunderstandings, the term path refers to the way of a PT user, whereas the term route is used in the context of vehicles. The terms traveller, passenger and PT user are understood as synonyms.



Figure 1.1: Thesis structure

Chapter 2

State of the Art Analysis

This chapter is meant to equip the readers with the necessary background knowledge to better understand the methodology introduced here and to put it into the context of already existing investigations.

2.1 Background

The following subsections provide the readers with a variety of terms and topics which are not directly tackled by this thesis but are connected to it and relevant for the understanding of the methodology presented here. It gives a short insight in the state of the art of these.

2.1.1 Passenger Demand Assignment

Passenger demand assignment (PDA) models have been developed to predict and analyse the distribution of passenger demand onto a PT network, meaning, to associate certain passenger flows with certain PT lines and vehicles. In this way, it is determined which PT lines and vehicles are used by how many passengers in a certain period. These models have mainly been used in PT planning processes, for example, to solve the PT network design and frequency setting problem, to provide the passenger demand of a city with an appropriate PT service [UL ABEDIN et al., 2018]. Generally speaking, one has to distinguish between the following model aspects: frequency- (or headway-) and schedule- (or timetable-) based, deterministic and stochastic, referring to the path-choice model within PDA, dynamic or static, and congested or non-congested demand assignment models [LAM and BELL, 2007, pp. 93-198; FU et al., 2012; GENTILE and NÖKEL, 2016, pp. 287-484; UL ABEDIN et al., 2018]. The contrast between frequency- and schedule-based models is described in Table 2.1 by UL ABEDIN [2019].

Frequency- or headway-based models are faster in computing results whereas schedule- or timetable-based models need more data and computational effort. However, the latter provide a higher level of detail. Usually, time-dependent origin-destination (OD) matrices serve as demand input for PDA models. There are several sources from which predictions of the demand can be derived, namely: sample censuses, ticket sales, automatic vehicle location (AVL), automatic fare collection (AFC), smartphone data, or automatic passenger count (APC) [BACHIR et al., 2019; GE, SARHANI et al., 2021; HUSSAIN et al., 2021; MO, KOUTSOPOULOS and ZHAO, 2022].

The deterministic and stochastic characteristic of a PDA model refers to its path choice component. In any PDA model there is a path choice model which clarifies the passenger's decision about which path to take from the origin to the destination. In literature, the term path is also referred to as a strategy that describes a list of subsequent decisions which enables a passenger to get from an origin to a destination within a given PT network [SPIESS and FLORIAN, 1989]. The topic of path choice is further discussed in Section 2.1.2.

The adjectives dynamic and static describe the kind of input in terms of supply and demand data that is used for a PDA model. Especially macroscopic models work with a static demand and supply for the period of the simulation. Dynamic PDA models however take certain variations in supply and demand for a day or even a week or month into account [GENTILE and NÖKEL, 2016, p. 312].

The characteristic mentioned last of PDA models refers to the question of whether the models are applied to a congested or uncongested network. Uncongested networks which do not take capacity constraints of PT services into account are simpler and need less data, whereas congested networks consider capacity constraints, are more complex, and computationally more expensive. Since it is a complex problem to consider "fail-to-board" or even "fail-to-sit" factors due to overcrowding, capacity constraints have not always been considered in research and practice [DE CEA and FERNÁNDEZ, 1993; SCHMÖCKER, BELL et al., 2008; SCHMÖCKER, FONZONE et al., 2011; FU et al., 2012; BRIEM, BUCK, EBHART et al., 2017].

However, in the last two decades, more and more PDA models for congested models have been developed. $\rm Fu$ et al. [2012] give a comprehensive overview of the literature about PDA models.

Assignment type	Advantages	Disadvantages
Headway-based	Ideal for the strategic planning Minimum data requirement Computationally efficient	Unable to identify peak load Not suitable for low frequency PT network Lacks detailed attributes of PT service
Schedule-based	Ideal for detailed spatio-temporal planning Suitable for low frequency PTnetwork Coordination of timetable possible	Requires extensive information Computationally expensive

Table 2.1: Headway and schedule-based assignment comparison [UL ABEDIN, 2019]

2.1.2 Shortest Path Search and Path Choice

As mentioned in the previous section, path choice models are used in PDA models to predict the passengers' choices about which path they are likely to take among all their options to reach their destination. Their options and the corresponding travel times of these are examined by shortest path search algorithms. These algorithms analyse a network to calculate the shortest path, meaning the path with the lowest costs according to certain criteria in a network between an origin and a destination. It is thereby also referred to as the cheapest path. However, in most cases, the costs of time are used as criteria.

The first algorithms to solve the shortest pathfinding problem were published in the late 1950s by Bellman [1958] and DIJKSTRA [1959]. Most of the algorithms that have been developed since then can be considered as extensions or modifications of these two. For example, the A*-algorithm that builds upon the Dijkstra algorithm [HART et al., 1968; BAST et al., 2016].

Another extension is the k-shortest paths search algorithm which does not look for just one shortest path between an origin and destination, but a certain number of shortest paths defined by the variable k. This is determined by certain criteria or just by a given deterministic value [HERSHBERGER et al., 2007].

CASEY et al. [2014] compare the performance of nine different shortest path search algorithms solving the dynamic multi-modal pathfinding problem. They describe that there are quite significant differences in finding the fastest paths in a road or a PT network due to the restrictions which come along with timetable dependencies in PT networks.

Furthermore, dynamic variations of originally static algorithms such as the Dijkstra or the A*-algorithm are explained. It is differentiated between static and dynamic path searches. As already mentioned in the previous section, this refers to whether the inputs for the link travel times are defined statically or dynamically, meaning time-dependent. All these algorithms have the purpose to find the fastest or the fastest and additionally subsequent fastest paths from a certain origin to a certain destination. A comprehensive overview of path search algorithms and the underlying approaches for route planning is given by BAST et al. [2016].

Whether a path is more or less convenient for a traveller is calculated by taking certain factors such as the travel time or crowding into account. A distinction is to be made between the actual travel time of a path and the perceived travel time. The perceived travel time considers the differences in perceiving waiting, walking, and in-vehicle time or transfers, whereas the actual travel time represents the actually needed time to travel a certain path [CATS, KOUTSOPOULOS et al., 2011; KIM et al., 2015].

Equation 2.1 shows a utility function representing the perceived travel time by CATS, KOUT-SOPOULOS et al. [2011]. Each part of a travel time (i.e. waiting, walking, and in-vehicle time) is multiplied by a coefficient β which represents the differences in the perception of the different parts. For instance, waiting for a certain period is perceived as longer compared to riding for the same amount of time. Furthermore, the number of transfers is multiplied with a penalty time, because transfers are inconvenient for travellers.

CEDER and JIANG [2019] introduce a path search procedure that takes into account personal preferences of individual travellers over a variety of path characteristics to offer travellers the most satisfactory path via a smartphone application (also known as app).

$$U_i = \beta_{wait} \cdot t^i_{wait} + \beta_{ivt} \cdot t^i_{ivt} + \beta_{walk} \cdot t^i_{walk} + \beta_{trans} \cdot trans_i + \eta_i$$
(2.1)

Where:

 $\begin{array}{ll} U_i & = \mbox{Utility of path i,} \\ t^i_{component} & = \mbox{time needed for path component (waiting, walking, in-vehicle) of path i,} \\ \beta_{component} & = \mbox{corresponding coefficient of path component,} \\ trans_i & = \mbox{number of transfers on path i,} \\ \eta_i & = \mbox{error term of path i.} \end{array}$

The goal of path choice models is to reflect the passengers' decisions on which path to take as realistically as possible. Therefore, for such models, the perceived travel time is more relevant. Generally, travel time is a prerequisite for path choice models. Once a set of attractive paths or hyper-paths, as it is referred to by NGUYEN and PALLOTTINO [1988], is determined, the set can be assessed according to the paths' features or even the traveller's preferences. Then, it can be decided on a path within the given set.

Moreover, it is to be differentiated between deterministic and stochastic path choice models. A deterministic path choice model, on the one hand, is simpler in a way as it assumes certain rules after which it is decided which path passengers take according to their origin, destination, available attractive lines, and the next approaching PT vehicle. It determines one particular path for each passenger or respective passenger group. A stochastic path choice model, on the other hand, assigns certain probabilities to the different factors which influence the path choice of passengers, meaning the solution is not one particular path but several paths with certain probabilities for being chosen. Therefore, the latter is more realistic but at the same time more complex which is why in practice deterministic path choice models are still in use [GENTILE and NÖKEL, 2016, p. 311].

Y. LIU et al. [2010] present a review on path choice models for PT users. Besides the choice of the path, first, the transport mode among a traveller's options is chosen [SOHN and YUN, 2009]. RAHIMI SIEGRIST, BÜCHEL et al. [2022], for example, show that passenger information (PI) and measures such as increasing the capacity of roads or PT lines can influence the mode choice during an incident. However, as this thesis focuses on PT systems, mode choice is not further discussed here.

2.1.3 Incidents in Public Transport

As pointed out in Chapter 1, this thesis belongs to the research area of incident management. Incident management starts with incidents occurring daily in PT systems. In literature, there are different definitions and understandings of various terms such as delay, incident, disruption, disturbance, reliability problems and so on. Nevertheless, they are all in the same premise of understanding [VUCHIC, 1969; CARREL, 2009; ZHU and GOVERDE, 2019a; BRIEM, BUCK, MAGDOLEN et al., 2020]. CAREY [1999] for instance differentiates between exogenous and knock-on delays. Exogenous delays are caused by failure of equipment such as vehicles or infrastructure, lateness of crew, passenger boarding and alighting, etc. Knock-on delays are understood as delays which are caused by exogenous delays but can be attributed to the planning process and timetable. These are delays that could be prevented by changing the

headway of the PT service for example. ABKOWITZ et al. [1978] distinguish in environmental and inherent reliability problems in PT. Environmental problems are problems such as traffic conditions, traffic signals, crew and vehicle availability, and deviations in passenger demand. Inherent problems include the distribution of transfer times, boarding and alighting times, travel times, and waiting times, so basically, problems that are perceived by the passengers.

In the thesis presented here the term incident is understood as any event which disturbs or even disrupts the PT services. Disturbances are understood as the consequences of incidents. Typical disturbances are delays, cancellations, or any other deviation of the planned PT timetable. Severe disturbances, such as complete service cancellations over several hours or delays of more than one hour are also referred to as disruptions.

It is further distinguished between external and internal incidents. The origin of external incidents lies outside the PT system, typical examples are traffic accidents and congestion, deployments of emergency forces, and passenger-related incidents such as falling passengers and vandalism of passengers or other people, whereas internal incidents have their origin within the PT system, such as technical failures, PT vehicle breakdowns, and PT driver-related issues.

Disturbances are subdivided into primary and secondary ones. Primary disturbances are direct consequences of an incident, whereas secondary disturbances can be a result of primary disturbances. A famous example of a secondary disturbance, which is thoroughly investigated in literature is bus bunching [VUCHIC, 1969; STRATHMAN et al., 2001; GERSHENSON and PINEDA, 2009; WEINSZIEHR et al., 2023]. Due to an initial incident, a bus is delayed (primary disturbance). The delay results in passengers who actually want to take the following bus, but are already at the stop when the delayed bus arrives, boarding the delayed bus. Hence, a constantly rising number of passengers get on the delayed bus, which prolongs its dwelling time due to more people boarding and alighting, which increases the bus's delay even further. At the same time, fewer travellers board the following bus, which reduces its dwelling time, consequently, it keeps catching up with the delayed bus and causes bunching (secondary disturbance) [BERREBI et al., 2018]. Table 2.2 gives an overview of incident examples, categorised by the related subject and type. Detailed descriptions of further incident examples can be found in Annexe A.

Additional Causes of Incidents in Autonomous PT Systems: Autonomous mobility in PT is a constantly growing topic in research and practice and several projects with theoretical investigations and practical field tests have already been carried out [BVG AND CHARITÉ, 2017; RAU, JAIN et al., 2019; VOLVO AB, 2019; S. WANG and ZHAO, 2019; HAMBURGER HOCHBAHN AG, 2021; PIGEON et al., 2021; POINSIGNON et al., 2022] As autonomous vehicles are operated differently, they also lead to changes in incident management in PT. For such autonomous PT services, the listed incidents in Table 2.2 are applicable as well. The only exception is the PT driver-related incidents since autonomous vehicles are driverless. However, the deployment of autonomous vehicles raises three additional categories of incidents [TAEIHAGH and LIM, 2019]. These new categories of incidents shown in Table 2.3 also have the potential to cause disturbances and lower the reliability of PT systems. However, the remainder of this thesis focuses on incident management as it is today.

Category	Туре	Examples
PT operations-related	Internal	Missing (replacement) driver, missing PT vehicle, OCC is understaffed, etc.
Infrastructure-related	Internal	Signal disruption, track switch failure, railway barrier malfunction, etc.
PT vehicle-related	Internal	Flat tyre, engine failure, charging issues, door malfunction, etc.
PT driver-related	Internal	Driving wrong route, health issues, late for work, etc.
Traffic-related	External	Illegal parking, congestion, accident blocking tracks or roads, etc.
Emergency-related	External	Ambulance, police, or fire department deployments blocking stops, tracks or roads, passenger needs medical attention, etc.
Passenger-related	External	Falling passenger, high passenger demand, passengers blocking doors, overcrowding, etc.

Table 2.2: Incident categories in PT

Category	Туре	Examples
	Internal	Difficulties to adjust to rarely occurring incidents
System maturation-related		and driving manoeuvres, etc.
		[S. WANG and ZHAO, 2019]
	External	Hacking attacks on autonomous vehicles or
Cyber security-related		corresponding infrastructure, etc.
		[SHEEHAN et al., 2019]
	External	Acceptance and proper interaction with
Society-related		autonomous vehicles, etc. [SALONEN, 2018],
		[PIGEON et al., 2021; YUEN et al., 2022]

Table 2.3: Incident categories in autonomous PT services

2.1.4 Passengers' Reactions to Incidents

Incidents, such as the ones mentioned in the previous section, cause PT service cancellations and delays which lead to passenger delays or even trip cancellations. This section gives a short insight into the common reactions of PT users to incidents.

It is to be differentiated between sudden, unplanned incidents such as the ones described in the previous section and planned incidents which are known to the PT users in advance, such as construction sites and maintenance work. In the latter case, passengers have more time to change their trip plans in accordance with the present incident situation. The focus of this thesis lies on the first kind of suddenly occurring, unplanned incidents.

Table 2.4 by LIN, SHALABY et al. [2016] shows typical reactions of PT users categorized

by the nature of the decision, namely: intermediate action, which is taken pre-trip or en-route, as a response to suddenly occurring incidents; pre-planned intentions, which can be short-term or long-term, as a reaction to planned incidents, and gradual adaptions according to changes in the PT service, either short- or long-term.

Common reactions to sudden incidents are waiting for the dissolution of the incident, changing the departure time of the trip, and changing the chosen path within the PT network, which can also go along with a shift in the PT mode, a shift of the transport mode towards a private car, a taxi, cycling or walking for example, and cancellation of the trip.

As pointed out by E. RAHIMI et al. [2020], many factors influence the reaction to incidents of passengers, namely: location and duration of the incident, weather, the remaining distance to the destination, car ownership, availability of information about the incident, alternatives in the PT network, as well as other modes of transport. In addition, the waiting tolerance of PT users during incidents is influenced by many factors, which have been summarised by E. RAHIMI et al. [2019]. Arrival time flexibility, the purpose of the trip, distance, experience with ride-hailing and bike-sharing services, availability of alternative paths through the PT network, the density of the walking links, as well as socio-demographics are named as such factors.

LIN, SRIKUKENTHIRAN et al. [2018] present the results of a combined revealed and stated preference survey. Its results show that the remaining distance of travellers to their destination strongly affects the likelihood of travellers deciding on walking the distance instead of waiting for an alternative ride on PT facilities. Furthermore, the authors compare the modal split of travellers with and without PI, showing the available alternative options in the PT system during the incident. A significant increase in the share of people using an alternative path through the PT network is observed. PI is thereby an important factor when it comes to passengers' reactions to incidents.

The importance of adequate PI is also emphasized by PASSENGER FOCUS [2011]. Rail riders have been asked about their information preferences during incidents. An estimation of the delay, the reason for the incident, and alternative paths through the PT network have been stated as most important.

ADELÉ et al. [2019] show that there is a relation between passengers' attitude towards PI and their likelihood of changing the chosen path during incidents. Another result is that the longer the duration of the incident, the fewer passengers wait and the more choose an alternative path. Further, BAI and KATTAN [2014] show that PI in combination with other factors such as weather, car ownership, and others, influence the reaction of passengers to service disturbances such as delays.

Moreover, WILKE [2023] presents a model to predict the reaction of passengers to suggestions for alternative connections in long-distance railway services during disruption. The author differentiates between internal and external factors which influence the willingness to follow given suggestions. Internal factors are related to the passenger such as age, travel habits, etc. and external factors are referring to factors concerning the circumstances of the incident as well as properties of the used PT service. The results show that a seat guarantee on the proposed connection, as well as the possibility of continuing the journey in first class, positively influences the willingness to take the proposed connection, especially in case of a service cancellation. In situations in which an alternative connection is suggested to prevent overcrowding, the influencing factors are less successful. These are factors that are hard or not at all transferable to urban PT services which do not provide for seat reservations or different classes. Nevertheless, another finding is that monetary incentives such as vouchers for drinks or a fare reduction influence the willingness to follow connection suggestions positively. The latter in particular could also be applicable in urban PT services. Apart from that, a survey of railway service customers reveals that about 57% would follow connection suggestions without an incentive in the situation of the service cancellation.

The topic of passenger reactions to incidents is further discussed in Chapter 3, which presents the results of a PT user survey tailored to this thesis.

Intermediate action	Pre-planned intention	Gradual adaption
Pre-trip and en-route	Short-term and long-term	Short-term and long-term
Wait until service restored (no change)	No change (long-term)	No change
Departure time change only (pre-trip)	Path change only	Path change only
Path change only	Transit mode shift	Transit mode shift
Transit mode shift	Mode shift	Mode shift
Mode shift	Destination change	Destination change
Trip cancellation	Trip cancellation	Trip cancellation

Table 2.4: Most common alternatives at different stages [of a passenger's trip] [LIN, SHALABY et al., 2016]

2.1.5 Passenger Information

The previous section already implies the importance of PI, especially during incidents. Furthermore, as pointed out by the research questions stated in Section 1.2, one of the goals of the methodology presented in this thesis is to include passengers affected by an incident actively in the mitigation of the delays caused by an incident. Consequently, it is necessary to communicate with PT users in general and especially in the event of an incident. PI is the connection between PT operators and PT users. Its potential and benefits are presented in this section.

PI describes the kind of information PT users are provided with to plan their trips in a PT system. Typical aspects are the fare of a PT service and its timetable. Based on this information, users can plan their trips as well as compare the service with other modes of transport, e.g. taxi, private car, or cycling. Today, there are many information channels available that are used to inform travellers about the PT service: Speaker announcements, dynamic PI displays at stops and in PT vehicles, the operators' representation on websites or social media channels, as well as online trip planners on websites or as part of smartphone apps of operators or third parties (e.g. GoogleMaps, Oeffi or CityMapper) [BRAKEWOOD and K. WATKINS, 2019; BACHMANN, BRIEM et al., 2022; BACHMANN, TSAKARESTOS et al., 2023].

The direction of development of PI is implied by CHOW et al. [2016]. The authors introduce the investigation of online PT information connected with the user's smartphone and wearable device (i.e. smartwatch). By comparing the timetable information with the global positioning system (GPS) information of the user and the remaining distance to the stop, feedback can be given to the user if an adjustment of the walking speed is necessary to catch the desired PT service. Trip planners on websites and as part of smartphone apps are connected via an application programming interface (API), such as the general transit feed specification (GTFS), to enable them to provide the timetable of a PT system [MOBILITY DATA, 2022]. With these information technologies, it is also possible to provide customers with real-time information, for example about the actual arrival times of PT vehicles, in which also delays of vehicles due to incidents or other reasons can be considered.

DZIEKAN and KOTTENHOFF [2007] show that dynamic PI displays at stops, providing PT users with real-time departures, have a positive effect on the travellers' perception of waiting time, their certainty and trust in the service, as well as their general satisfaction with the PT service. Even stronger positive effects are shown by FERRIS et al. [2010], who introduce a cross-channel PI system, including a smartphone app, a website, a phone service, and a short message service (SMS). This system called OneBusAway provides the users with real-time arrival information as well as locations of stops and further information. K. E. WATKINS et al. [2011] also use OneBusAway and investigate the waiting times of passengers. The actual and the perceived waiting time was reduced through mobile real-time PI.

Through a simulation study, CATS, KOUTSOPOULOS et al. [2011] reveal that real-time PI has the potential to reduce travel times and cause shifts in the travellers' path choice. Furthermore, DRABICKI et al. [2023] complements a discrete path choice model with the findings through a PT user survey concerning the willingness of travellers to wait for the next PT service to avoid in-vehicle crowding, when provided with real-time crowding information.

Moreover, CHAN and SCHOFER [2014] and PENDER et al. [2014] investigate the potential of social media during disruption in PT systems and emphasize the importance of real-time PI to passengers in general. A comprehensive review of investigations about PI, especially real-time PI, is given by BRAKEWOOD and K. WATKINS [2019]. The authors summarize and compare the existing literature about PI as well as the corresponding findings. The main benefits of real-time PI are reduction of waiting time (perceived and actual) and total travel time, increase in usage of and satisfaction about PT, as well as the perceived security of PT users.

In the following, PI refers to real-time passenger information in PT. BACHMANN, TSAKARES-TOS et al. [2023] bring the topic of PI into context with incident management. A summary of investigations about the effects of capacity-aware passenger redirection during incidents is given by the authors. As this is the research area of this thesis it is comprehensively discussed in Section 2.2.2.

2.2 Incident Management in Public Transport

As the term already gives away, incident management describes the management of incidents as well as the disturbances they cause. In practice, dispatchers in OCCs are the backbone of incident management. They monitor the PT service and take disposition measures to mitigate the negative effects of incidents on the PT system and return the PT service to normal operations. In the following, different aspects and perspectives of incident management will be examined. The topic of incident management is divided into two parts: *supply-centric* incident management that focuses on disposition measures to adjust the PT service to an incident situation, and PCIM that focuses on the PT users' perspective during incidents and their involvement in the reduction of passenger delays. PCIM is understood here as an enhancement or extension of *supply-centric* incident management.

2.2.1 Supply-centric Incident Management

Many disposition measures in incident management have been thoroughly investigated, this section will provide for both an insight in studies which have been done about *supply-centric* disposition measures as well as an insight in the state of practice.

Literature Review

Supply-centric measures focus on the supply side of incident management, meaning that the focus lies on the measures taken by dispatchers to rearrange the PT supply in accordance with an incident.

 $\rm EBERLEIN$ [1995] categorises these measures into three groups, namely: station control, inter-station control, and other strategies. For a more general expression, it is referred to stops rather than stations in the following.

Stop control strategies, include all measures which take place at a stop or temporarily alter the stops served by a PT line. Holding and stop(s) skipping measures such as expressing and deadheading are typical measures of this category.

Inter stop controls refer to strategies such as speed control and traffic signal prioritisation, these measures influence the course of PT vehicles while they are between two stops.

All other measures fall under the category of other strategies. These are for instance dispatching additional vehicles or extending and splitting trains or platoons of PT vehicles. All of the other strategies are related to a change in the capacity of the PT service and could therefore be also titled capacity-related strategies.

The list below shows an overview of *supply-centric* disposition measures with a short description, and corresponding studies, organised by the mentioned categories. References which appear under more than one measure investigate the differences or the combination of the respective measures.

Stop control strategies

Holding

A PT vehicle's dwelling time is extended to increase the headway to the preceding vehicle and/or reduce the headway to the following vehicle. This is done to readjust the vehicle's run in accordance with a PT schedule or to the desired headway.

Studies: [EBERLEIN, 1995; EBERLEIN, WILSON et al., 1999; EBERLEIN, WILSON et al., 2001; CATS, LARIJANI, KOUTSOPOULOS et al., 2011; DAGANZO and PI-LACHOWSKI, 2011; CATS, LARIJANI, ÓLAFSDÓTTIR et al., 2012; DELGADO et al., 2012; SÁEZ et al., 2012; CEDER, HADAS et al., 2013; NESHELI and CEDER, 2014; BERREBI et al., 2018; RODRIGUEZ et al., 2023]

Stop(s) skipping (Expressing, deadheading)

A delayed PT vehicle skips one or several stops to reduce its delay. Expressing describes the skipping of stops along the planned route, while deadheading means that all passengers alight the vehicle at the next stop and it moves directly to the terminal to start the next run on schedule or at least with less delay.

Studies: [EBERLEIN, 1995; EBERLEIN, WILSON et al., 1999; EBERLEIN, WILSON et al., 2001; SÁEZ et al., 2012; CEDER, HADAS et al., 2013; NESHELI and CEDER, 2014; RODRIGUEZ et al., 2023]

Rerouting PT vehicles

This measure is often used when the planned route of a PT line is affected by any kind of blockage, for example, due to a traffic accident. A temporarily altered route is established. Due to the rerouting, some of the stops of the affected line might be temporarily not served.

Studies: [MANDL, 1980; CATS, KOPPENOL et al., 2017; FRIEDRICH et al., 2018; YAP et al., 2018]

Short-/Long-turning, line-splitting

At the terminal of a route, PT vehicles turn around to serve the opposite direction of a route after a short break, which is called layover time. Short-turning describes the disposition measure through which the turning around is done before reaching the terminal of a route, for example, to avoid an incident at the end of the route or supporting the opposite direction earlier than planned. If this is done on both sides of an unserved section of a PT line, for example, due to an incident, it is called line-splitting. Longturning is rarely used. It extends a route beyond its planned end and the turning around at the end of a route is done later. This is used to support another PT line.

Studies: [CEDER, 1989; TIRACHINI et al., 2011; CANCA et al., 2016; GKIOTSALITIS, Z. WU et al., 2019]

Inter stop control strategies

Speed control

PT drivers who are off schedule are told to speed up or slow down to readjust their position in accordance with the schedule. This is solely done within legal boundaries. **Studies:** [DAGANZO and PILACHOWSKI, 2011; T. LIU et al., 2014]

Traffic light prioritisation

PT vehicles in mixed traffic are preferred at signalised intersections to reduce their delay due to waiting at red lights. With the change in the level of prioritisation, this can also be used to speed up or slow down a PT vehicle. This measure is mainly a matter of planning, however, it is also referred to incident management in literature. **Studies:** [CHANG et al., 2003; LEE et al., 2005]

Other / Capacity related strategies

Dispatch standby buses

Some PT operators reserve buses to be deployed spontaneously when needed. These buses are positioned at strategically reasonable locations. Whenever a PT vehicle breaks down or a PT line faces unusual high and overwhelming demand, a standby bus can replace a troubled PT vehicle or support an over-demanded PT line.

Studies: [CATS and JENELIUS, 2015; PETIT et al., 2018; MORALES et al., 2019]

Dispatch on-demand services

Similar to the previous measure of dispatching standby buses, this measure describes the dispatching of on-demand services such as taxis to replace broken down PT vehicles or support an over-demanded PT line with extra capacity. **Studies:** [CEBECAUER et al., 2021]

Rail replacement / bus-bridging service

Breakdowns and failures in railway services can have very severe consequences. Furthermore, it takes often a lot of time to fix the problem which leaves the PT users without service for a long time. If a rail-bound PT line needs to be split or replaced completely, the unserved section is temporarily bridged by a rail replacement bus service.

Studies: [H. HU et al., 2016; BOJIC et al., 2021; ITANI and SHALABY, 2021; LUO and XU, 2021; J. WANG et al., 2021]

Train- / Platoon-splitting and -extending

In PT modes with a modular setup, for example, tram or metro trains, it is thinkable that one PT line with dispensable capacity helps out another over-demanded one, by reallocating some vehicle modules such as train carriages or half of the train in cases of two or more coupled trains, or dispatch corresponding extra vehicles. In the future, this might be also possible with platoons of autonomous road-bound bus modules such as the DART system [RAU, TIAN et al., 2018].

Studies: [SCHMÖCKER, COOPER et al., 2005; BACHMANN, RAU et al., 2021]

As the description above clearly shows, most *supply-centric* disposition measures in PT incident management are well investigated. For a more comprehensive and detailed overview of investigations about *supply-centric* incident management measures, it is referred to CEDER [2016, pp. 309–332], NESHELI [2016] and GKIOTSALITIS and CATS [2021] and GE, VOSS et al. [2022].

State of Practice – Findings Through Visits to Operations Control Centres

The previous section enumerates *supply-centric* disposition measures and lists several corresponding studies. However, significantly fewer studies have focused on the state of practice in incident management. In practice, incident management in PT systems is done by dispatchers in OCCs. They are the main actors of incident management in PT systems, monitoring the PT service with the help of an intermodal transport control system (ITCS) and taking disposition measures as soon as deviations from the schedule occur. PANGILINAN et al. [2008] investigate such an ITCS and describe it thoroughly. With the help of AVL of buses through the GPS and in rail-bound services through signalling, it is possible to monitor PT vehicles' location in real-time and compare it to the schedule. In this way, the dispatchers notice in real-time if a PT vehicle is off schedule [SAEZ et al., 2012].

CARREL, MISHALANI et al. [2010] describe the processes and challenges in the OCC of a metro line. The authors conclude that staff management is a crucial factor in incident management. Missing staff and unavailability of replacement staff have the potential to worsen every kind of incident.

BRIEM, BUCK, MAGDOLEN et al. [2020] have interviewed dispatchers of the OCC in Karlsruhe, Germany, to understand their perspective on incident management. The authors categorise the dispatchers' answers into three categories, namely: influences and limitations, described dispatching procedures, and other topics besides incident management. They conclude that passengers themselves and staff management restrictions, such as the legal driving time of drivers, are the most predominant influences and limitations on dispatching decisions. Moreover, they point out that rerouting PT vehicles is the most preferred disposition measure to mitigate an incident's negative impact on the PT service. This is often followed by shortand long-turning measures to return the PT service to its schedule. Furthermore, the authors provide insight into the motivation of dispatchers by asking them why they chose their jobs.

BRIEM [2023] comprehensively analyses vehicle data of an ITCS to identify deviations from the PT schedule and compares it with actually taken disposition measures. The author notes that not for every incident, disposition measures are taken and proposes an assessment model for PT schedules to verify their scope for disposition measures.

BACHMANN, BRIEM et al. [2022] present the finding through visits to six OCCs in Germany and Singapore, they interviewed OCC directors and trainers and observed dispatchers during their work. Whereas the findings through interviews cater for a comprehensive overview of the processes and procedures in an OCC, the observations of dispatchers during their work add valuable and detailed findings about the dynamics in OCCs taking place while dispatchers handle incidents. This is especially true for communication between the involved parties, such as PT drivers, dispatchers, emergency forces (i.e. police, ambulance, fire department), PT users, and towing services. This work was an important first piece of the puzzle to this thesis, as it helps to understand the operators' view on incidents.

As mentioned before, dispatchers play the main role in PT incident management, by monitoring the service and take disposition measures whenever a deviation from the timetable occurs. To do so, they talk to the drivers via radio and to emergency forces and ground staff (i.e. transport wardens, mechanics) via phone to manage incidents adequately. In most cases, the information about an incident is provided by the PT drivers who send a call signal to the OCC in accordance with the present situation, once they encountered an incident.

Figure 2.1 shows the step-by-step procedure taking place in an OCC when an incident occurs. In most cases, it is a PT driver who encounters the incident first, as the drivers are out in the field and directly affected by such (Step 1). Right away, the driver notifies the dispatchers in the OCC about the encountered incident (Step 2). There are four different call signals to initialize a call corresponding to the present situation: assault, accident, missing replacement, or wish-to-talk call. The priority of these calls decreases from the first to the last mentioned. As per the call signal, the dispatcher follows certain procedures to handle the situation (Step 3). Flow charts and explanations of these procedures are given in detail by BACHMANN, BRIEM et al. [2022].

In Step 3, two things happen simultaneously, the dispatchers instruct the drivers of the disturbed PT line and calls the emergency forces, if necessary. In OCCs the staff work closely together and support each other when an incident occurs in order to be able to handle this stressful event. If the situation requires the deployment of a transport warden, one is dispatched in Step 4. A transport warden is a field dispatcher. In Step 5 the emergency forces and the transport warden arrive on site. The transport warden talks to the emergency forces and coordinates the dissolution of the incident on site. Moreover, some situations, such as an assault on the driver or an accident can be quite burdening for PT drivers, hence, the transport warden is also on-site to help the driver in any way necessary. Once a transport warden got a clear picture of the situation, the OCC is provided with more detailed information about the present situation. In this way, the dispatchers can readjust their taken measures according to the additional information, if needed, and give the drivers updated instructions in Step 7. At this point, all decisions needed to resolve the incident have been made. Therefore, in Step 8, the dispatchers monitor the incident situation and let the subsequent PT drivers inform them about the development of the situation. Once the situation is cleared in Step 9, the dispatchers revoke all given instructions.

Even though there have been many investigations in incident management as pointed out in the previous section, the authors identified a gap between research and practice. Many findings of scientific studies have not yet been implemented in practice. This is one of the goals of the here described work by BACHMANN, BRIEM et al. [2022], to let practice benefit more from research. Another goal is to identify potentials for improvement. As pointed out before, the most important tasks of dispatchers are monitoring the service and handling incidents by communicating with the other involved parties. This is mostly done verbally via radio and phone. Already today PT vehicles are equipped with AVL to enable the dispatchers to monitor their delay. However, when a troubled PT vehicle needs assistance from emergency forces, because it is involved in an accident, for example, its location has to be forwarded verbally via the dispatcher to the emergency forces and the ground staff. All information is going via the dispatchers who are thereby a bottleneck in the incident management system. Although dispatchers do an outstanding job to keep a certain quality of service and resolve incidents as quickly as possible, especially during peak hours, when traffic and passenger demand are high, it can easily come to overwhelming situations. When dispatchers are overwhelmed, disposition measures are stalled and thereby incidents' dissolutions as well. This caters for additional delays for PT services and consequently their users.

Therefore, it is desirable to reduce this bottleneck in the system and establish more direct and automated communication between PT vehicles and emergency forces, by automatically transmitting the troubled PT vehicle's GPS location to the emergency forces, for example. Another possibility is to automatically match the location of a broken down PT vehicle with the location of the operator's mobile mechanics and transport wardens. The closest ground staff get a request on their mobile device with the location and situation of the troubled vehicle. Then, they have several seconds to answer the request, otherwise, it is forwarded to the second closest personnel and so on. This kind of procedure is already used in ride-hailing services such as Uber or Lyft to match customers with drivers [ÖZKAN and WARD, 2020; UBER TECHNOLOGIES INC., 2022].

In addition, misunderstandings, especially through language barriers are an issue in the communication between drivers and dispatchers. The better the communication, the better the incident management. Due to the lack of drivers, PT operators continuously more often hire drivers from neighbouring countries. In many cases, this comes along with language barriers which could be solved by language courses for the drivers and/or dispatchers. However, that would be rather demanding. Another thinkable solution would be translations. Both, the dispatchers and the drivers are already equipped with computers or board computers. Today's translation programmes are capable of translating spoken words into written words in realtime. Such translations could be displayed on the monitors to avoid further misunderstandings between dispatchers and drivers. It goes without saying that in the case of the driver's monitor, this is only done when the vehicle is standing still. Otherwise, a verbal output by the computer is also technically feasible.

With autonomous vehicles emerging, a higher level of automation in OCCs is generally desirable. To install the infrastructure for direct communication not only between the involved people but also between people and machines as well as between machines to speed up communication, reduce misunderstandings and support dispatchers. In this way, dispatchers can focus on their most important job, to decide on adequate disposition measures to dissolve an incident and return the PT service to normal operations as quickly as possible. This particular investigation by BACHMANN, BRIEM et al. [2022] also helps to understand how the PCIM method presented here could be implemented in practice. This is discussed in Section 6.3. Furthermore, it helped with the development of the aforementioned work by BRIEM [2023].



Figure 2.1: Incident management procedure of an OCC [BACHMANN, BRIEM et al., 2022]

2.2.2 Passenger-centric Incident Management

Even though the studies in Section 2.2.1 focus on *supply-centric* measures, most of them are *passenger-oriented* in a way that they try to adjust the PT service to a present incident situation to reduce passenger delays. There is also a large variety of works in literature concerning *passenger-oriented* disruption management in railway operations. These works focus mainly on the rescheduling of trains and cancellation decisions during severe incidents in regional and long-distance railway services. One goal is to also provide dispatchers with decision support [RÜCKERT et al., 2017; VEELENTURF et al., 2017; LONG et al., 2021].

This chapter introduces investigations that are not only *passenger-oriented* but *passenger-centric*. This means they do not merely reduce passenger waiting and travel times during incident situations but they also include the passengers as part of incident management and the delay reduction. This is done by providing them with adequate PI about the incident or even available alternative paths and services, on which basis they can re-plan their disturbed trip. Thereby, this section introduces the literature of the research area this thesis is associated with, *passenger-centric* incident management (PCIM). The orientation for this section is based on BACHMANN, TSAKARESTOS et al. [2023], the authors summarise, compare, and discuss the to-date conducted investigations about capacity-aware passenger redirection (CAPR). As in the mentioned reference, this section is subdivided into academic investigations and approaches in practice.

Another research area about passenger redirection is to be found in air traffic. In case of a service disruption, airline customers are redirected onto other scheduled flights to ensure that they reach their destination with a minimum of delay [SINCLAIR et al., 2014; Y. HU et al., 2015; HASSAN et al., 2021]. However, there are many differences between the natures of passenger air transport and land PT when it comes to operations and incident management. In air transportation, each passenger is known by name, origin and destination, and seat number. Furthermore, the headway between certain connections, the number of airports, and the vessels are very different than in land PT especially in urban PT in which the headways are very small, the passengers are not known, and the vessels are smaller and there is a bigger variety among them (i.e. buses, trams, metros, etc.). Therefore, the research area of passenger redirection in air transportation is not further considered here.

The reviewed work is very diverse in their assumptions, considerations, and approaches. Therefore, several criteria of comparison have been derived from the literature review, which are used in the Tables 2.5 and 2.6:

Firstly, the focus and goal of each work is presented. Even though most of them aim to reduce passenger delays or inconvenience, respectively, the focus of the works differentiates.

Secondly, as CAPR is a crucial aspect of PCIM, the consideration of capacity constraints is another important criterion of comparison, since secondary disturbances through overcrowding are to be avoided.

Thirdly, the meaning of PI has already been explained in Section 2.1.5. Since it is one of the core ideas of PCIM to involve passengers actively in the reduction of their delays, communicating with them is vital. Three different aspects of PI have been included, namely:

its content, which refers to the information which is given to travellers; its schemes, explaining the used timings and location of the PI, and its channels, clarifying which PI channels have been considered in the corresponding work.

Fourthly, as passengers are the main focus of PCIM, criteria of comparison about their representation and handling in the presented approaches are included. Furthermore, used compliance rates, which describe the share of affected travellers, who would follow path advice, if given during an incident, and, if applicable, how the effect of PI during incidents on a path choice model is considered.

Fifthly, the resources that have been used to conduct the corresponding investigation, such as used tools and algorithms, as well as if an optimisation is taken into account, are criteria of comparison.

Sixthly, the conducted case studies are compared. Here, the criteria of comparison are the used networks, the characteristics of the considered incidents (i.e. duration, location), the demand input, and the transportation modes that have been taken into account.

Lastly, the considered *supply-centric* disposition measures as well as if and how the integration of the corresponding PCIM method into an OCC are used as criteria for comparison.

For describing some of the criteria of the presented works the following three terms are used in the Tables 2.5 and 2.6: *considered*, which expresses that the corresponding feature is considered in the developed method of the corresponding reference; *mentioned* means that the corresponding feature is mentioned but not considered and lastly *not mentioned* for which the corresponding feature is neither mentioned in the publication nor considered in the corresponding PCIM method.

Academic Investigations

Table 2.5 by BACHMANN, TSAKARESTOS et al. [2023] presents and compares twelve relevant academic contributions to the research field of PCIM, which are introduced in the following.

Literature Review: CATS and JENELIUS [2014] investigate the betweenness centrality and vulnerability of PT networks. To do so, a stochastic and dynamic notion of a PT system has been developed. Their case study includes five different incident scenarios in the PT network of Stockholm, Sweden. The incident takes place in the metro part of the network. In each scenario, four different PI schemes have been tested, namely: no PI, PI at stops, clustered PI at transfer hubs, and network-wide PI. The PI broadcast contains the updated arrival times of PT services in accordance with the changes through the incident. This PI has then a corresponding influence on the path choices of the simulated agents and thus onto the passenger assignment in the simulation. The scenarios with PI are then compared with the scenario without PI. The results show the positive influence of PI on passenger delays. In all but one cases, the passenger delay has been reduced. The phenomenon that the delay increased through PI in this one case is explained by overcrowding effects on the alternative paths. These could not provide sufficient remaining capacity for the additional demand of passengers transferring onto these because of the incident. A danger of such effects is specifically present when the disrupted PT service has a significantly higher capacity than the services of

the alternative paths. For example, passengers transferring from a disrupted metro line onto bus lines. The authors conclude that such secondary incidents could be avoided by employing effective PI schemes as well as proactive fleet management. Moreover, they point out that the importance of a link in a PT network also depends on the available real-time PI and not only on the betweenness centrality.

VAN DER HURK et al. [2018] present an optimisation model to reduce passenger delays and operators' costs by providing travellers, affected by an incident, with concrete path advice and rescheduling trains. The developed model builds upon KROON et al. [2015]. Severe disruptions of several hours at different locations are simulated in the Dutch railway network. Passengers who require path advice are clustered into indivisible groups in accordance with their OD relation. Furthermore, capacity constraints are considered for the path advice to avoid secondary incidents as previously mentioned.

Moreover, a compliance rate is introduced to account for passengers who receive path advice but do not follow it. Especially travellers who know the PT network well, such as commuters, might decide to redirect themselves rather than follow path advice given by the operator. Several compliance rates have been tested by the authors, deterministic values between 0% and 100% compliance as well as logarithmic calculated compliance rates. The logarithmic values take the difference in travel time between the fastest and the suggested alternative path into account. If the difference is larger than ten minutes, the willingness to follow path advice decreases rapidly.

It is concluded that rescheduling rolling stock and providing path advice can significantly reduce the passenger overall delay and operational costs. A further finding is that the location of the incident plays a crucial role. The closer to the centre of the PT network the incident location is, the more severe the negative effects of the incident, however, also the more effective is path advice, as more alternative paths are available. Interestingly, the best results are to be seen in scenarios with less than complete compliance. Additionally, the better the prediction of the demand assignment, the better the effects of the rescheduling model, as capacity bottlenecks appear more rarely.

Passenger redirection during severe disruption in a long-distance railway network is investigated by MÜLLER-HANNEMANN et al. [2019] in cooperation with the German railway operator German rail [Translated from German: Deutsche Bahn] (DB). Through a multi-commodity flow model with an event-activity network, optimal alternative paths are found. This work also considers capacity constraints and even introduces a soft (number of seats) and hard (total number of allowed passengers) constraint. Furthermore, the authors take a penalty for crowding into account which increases rapidly once the soft constraint is exceeded. This work also handles the passengers as indivisible groups, corresponding to their ODs, which is justified by the fact that it is impossible to know who is travelling with whom.

The idea of the authors is that their PCIM method can support dispatchers in their decisions in delay management [DOLLEVOET et al., 2012] while considering the passengers' perspective and providing real-time PI. Several different approaches are tested, namely: nocap, in which the passengers are redirected without consideration of the remaining capacity; greedy, which redirects passengers by groups, meaning that already assigned groups influence the redirection

options for the groups to be redirected and two approaches using Pareto fronts are performed [KNOWLES and CORNE, 1999]. The results of the study show that the developed PCIM method reduces passenger delays and lowers the number of passengers without an alternative path. The authors further test their approach with an increase in demand by 20% and show that the capacity-aware redirection of passengers becomes more important with an increase in demand. Furthermore, it is pointed out that searching for available alternative paths is the computationally most expensive part of their method.

Another piece of research in the field of PCIM is contributed by ZHU and GOVERDE [2019a]. With the help of an event-activity network, passengers, who are affected by train delays and cancellations, are dynamically reassigned. The disruption timetable which reflects the service offered during the disruption is considered. By comparing the passengers' paths with the disruption timetable, affected passengers can be identified. Re-planning decisions based on the disruption timetable are done for each traveller individually. If the updated arrival time of a certain passenger exceeds a set threshold, the passenger is assumed to leave the PT system.

In the simulated scenarios, the incident duration and location do not differ (one hour of full section blockage) but the location and content of the given PI. It is given nowhere and at stations and/or on trains. Its content includes the operating services and in some scenarios also the level of train congestion. Furthermore, different thresholds at what point in time the PI interventions are triggered are tested. The scenarios are tested in the Dutch railway network.

Findings include that the publication of the given PI reduces passenger delays. The more locations the information is broadcast, the greater the drop in delays. The higher the acceptance of delays, the fewer people leave the system. The information about the level of occupancy of trains helps to lower the delays further. Moreover, the incident is more severe for people who are already on the trip, compared to the ones starting their trip after the occurrence of the incident, as they have more options to re-plan.

The works by ZHU and GOVERDE [2019a] and by ZHU and GOVERDE [2019b] – the latter is about a timetable rescheduling model – are combined and further developed by ZHU and GOVERDE [2020] to a *passenger-oriented* timetable rescheduling model in which the dynamic reassignment of passengers during disruption is considered by the timetable rescheduling model to improve the overall situation even further than the two models do separately. For each disruption, several rescheduling timetables can be found, and each of these is combined with a corresponding dynamic reassignment of passengers to find the best solution. The results show a significant drop in passenger delays. In Table 2.5, ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] share a column as the latter builds upon the first work and the authors are the same. As ZHU and GOVERDE [2019a] focuses more on PCIM than ZHU and GOVERDE [2020] it is more relevant to the following discussions.

Moreover, the four publications by LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021] share the same column in Table 2.5 and are referred to as *Corman group*, named after the common last author.

LENG and CORMAN [2020] focus on the effects of different information schemes during incident situations. Three different schemes are tested, namely: no information, in which
affected passengers notice the disruption of the PT services when they reach the corresponding stop to take the disrupted service; timely information which refers to the dissemination of PI at the time of the incident's occurrence and advanced information which covers cases in which the incident is already known in advance to the passengers such as planned construction sites.

The authors use the agent-based tool Multi-Agent Transport Simulation (MATSim) [HORNI et al., 2016], to model disruption scenarios in the railway network of Zurich, Switzerland. A within-day planning procedure is implemented to equip agents with the option to adjust their daily plans during the day, for example, re-plan their trips in accordance with an incident. An agent's day plan exists of several activities and trips. Their results show that the advanced information scheme caters for almost no delays, whereas the timely information scheme reduces the delays less but still by a significant amount compared to the no information scheme.

LENG, LIAO et al. [2020] present an extension to the previously introduced paper. The authors conclude that by combining disposition measures with PI passenger delays can be further reduced. The authors investigate the mutual influence of PI availability with disposition measures such as rescheduling rolling stock. A mixed integer programming optimisation model is developed and implemented in the framework, which is introduced by LENG and CORMAN [2020]. It is used to find the optimal disposition measures in combination with the three aforementioned PI schemes.

Additional findings are gained by M. RAHIMI et al. [2021], by combining PI with measures such as increasing frequency or increasing the capacity of PT vehicles on lines which are relevant to the passengers who are re-planning their day due to the incident.

In addition, RAHIMI SIEGRIST and CORMAN [2021] used the same framework as the other works of the *Corman group* to cast light on the direct and indirect effects of the three PI schemes during the service disruption. Furthermore, the authors introduce two more information schemes: one at which the passengers receive the PI at the beginning of their next trip and a special form of advanced information in which the passengers are also aware of the decisions of other travellers.

As an overall conclusion of the works of the *Corman group*, it can be summarised that PI about an occurred disruption helps to significantly reduce passenger delays. Furthermore, the earlier the passengers are informed, the stronger its positive effect. Additional delay reductions can be gained by combining the information schemes with disposition measures such as rescheduling rolling stock or increasing the capacity or frequency on the relevant alternative paths. Another finding by LENG, LIAO et al. [2020] is that partially cancelled railway services have a much better effect on passenger delays than fully cancelled services. Furthermore, M. RAHIMI et al. [2021] found out that picking specific PT lines for an increase in capacity is most effective, among their tested disposition measures. Moreover, RAHIMI SIEGRIST and CORMAN [2021] point out that the negative effects of an incident spread through a network and affect passengers who are not directly disturbed by an incident, as well.

Further research in the field of PCIM is done by BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023]. The two publications show earlier stages of the PCIM method which is also presented in this thesis and are summarised in the same column in Table 2.5.

The method by BACHMANN, RAU et al. [2021] redirects affected passengers during incid-

ents by assuming that these passengers are provided with concrete path advice. Here, the redirection duration is introduced which states for how long it is reasonable to redirect passengers during an incident. This, of course, is only beneficial as long as the alternative path caters for an earlier arrival at the destination than the original path of a passenger. Travellers are grouped in accordance with the stop at which they are redirected, which is referred to as the redirection point and their destination.

A numerical case study has been conducted in a simple artificial network with assumed demand. Three different incident scenarios are tested: 1) disrupted PT services are rerouted, 2) additionally, passengers are redirected, and 3) in addition, dipensable capacities are reallocated. The scenarios have been conducted in a PT bus system as well as in the DART system [RAU, TIAN et al., 2018], which is an autonomous PT system with a modular setup. This is further described in Section 5.2.3. Through the modular setup, it is more suited for capacity reallocation than buses as a platoon of modules can be split and parts of its capacity can be dispatched while maintaining the service's headway on the original line. In contrast, a bus cannot split and its reallocation means a reduced level of service on its original line.

This is also confirmed by the results of BACHMANN, RAU et al. [2021]. The redirection of passengers already significantly reduces the delay whereas the additional reallocation of capacities lowers them further. The DART system caters for better results than the bus system in this matter.

The conceptional framework of BACHMANN, RAU et al. [2021] is picked up and developed further by BACHMANN, DANDL et al. [2023] and implemented in Python [PYTHON SOFT-WARE FOUNDATION, 2022] to conduct a simulation study. It is then combined with the simulation tool Simulation of Urban MObility (SUMO) [LOPEZ et al., 2018] to conduct a more sophisticated simulation study as the previously introduced work.

The tested scenarios take place in the Mandl network [MANDL, 1979] with a PT network design by UL ABEDIN [2019]. Two different incident durations (30 and 60 minutes) and two different disposition measures are tested, namely: rerouting, in which the PT services are rerouted, and line-splitting, in which the disrupted PT lines serve in loops on both sides of the incident location in the network (Section 2.2.1). Passengers are provided with path advice which is computed by a heuristic and by an optimisation approach. The results are then compared with each other. As VAN DER HURK et al. [2018], the authors also consider different compliance rates to take passengers who do not follow the path advice into account. Values of 57%, 100%, and logarithmic calculated values are used for the passengers' compliance. A PT user survey by the German Federal Ministry of Transportation and digital Infrastructure [Translated from German: Bundesministerium für Vekehr und digital Infrastruktur] (BMVI) revealed that 57% of the participants would be willing to follow an operator's path advice [BMVI, 2019]. The logarithmic compliance rates are adopted from VAN DER HURK et al. [2018].

BACHMANN, DANDL et al. [2023] conclude that path advice has the potential to significantly reduce passenger delays during incidents. Further findings are that the optimisation approach achieves better results than the heuristic and that their method is more effective in the combination with rerouting PT lines than line-splitting. Moreover, their results reveal that the incident duration has a considerable influence on delay reduction. The longer the incident the lower the positive effect of their PCIM method. One case in which logarithmic calculated compliance rates are used shows an increase in delay. The authors explain this by overcrowding effects such as also stated by CATS and JENELIUS [2014]. It could also show that the logarithmic function might not be suitable to reflect a complex situation of an affected passenger who has to decide on taking path advice or not. As described in Section 2.1.4, many factors influence passengers' reactions to incidents and to connection suggestions. In addition, BACHMANN, RAU et al. [2021] derive from their results that the additional reallocating of capacities during incidents can support passenger redirection and further reduce delays.

Lastly, the work by MO, KOUTSOPOULOS, SHEN et al. [2023] is introduced in this section. They mathematically formulate passenger flow distribution to run several scenarios in an eventbased public transit simulator. PT users with the same OD relation receive the same path advice. A historical incident case in the metro network of Chicago, United States of America (USA), is used. Like most of the aforementioned studies, this work also considers capacity constraints. In addition, uncertain passenger demands are considered to take PT users into account who leave the system due to the incident.

The result of the path advice optimisation is compared to three benchmark scenarios. In the first, the passenger flows are distributed evenly, it is therefore referred to as uniform path share. In the second, paths' capacities are considered during the assignment process, which is therefore named capacity-based path shares, and the last is called status-quo path share as this case reflects the actual decisions of the passengers on the day of the incident, which are inferred from smart card data. All results show an improvement in the passengers' situation through the given path advice.

Comparison: All these presented works have in common that their focus lies rather on the passenger than on the supply side. By informing the passengers either about the incident and the changes to the PT service or by giving them concrete path advice, they are directly involved in the passenger delay reduction. Table 2.5 reveals that the goal of decreasing passenger delay during incidents is approached and accomplished in various ways by the investigations.

The focus of CATS and JENELIUS [2014] lies on the influence of PI on a PT system's vulnerability; VAN DER HURK et al. [2018] focus on path advice and capacity adjustment; MÜLLER-HANNEMANN et al. [2019] concentrate on the algorithmic and computational implementation of path advice; ZHU and GOVERDE [2019a] bring the reassignment of passengers into focus; LENG and CORMAN [2020] concentrate on passenger behaviour during incidents; LENG, LIAO et al. [2020] and M. RAHIMI et al. [2021] focus on the mutual influence between *supply-centric* measures and PI availability. RAHIMI SIEGRIST and CORMAN [2021] centre their work on the positive and negative indirect effects of incidents; BACHMANN, DANDL et al. [2023] focus on path advice whereas BACHMANN, RAU et al. [2021] additionally focus on capacity readjustment. The focus of the work by MO, KOUTSOPOULOS, SHEN et al. [2023] lies on the distribution of passenger flows. Therefore, although the studies presented have the same objective, namely to reduce passenger delays during incidents, they are focused on different issues.

One of the crucial aspects which need to be examined in PCIM is the remaining capacity on alternative paths, to avoid secondary disturbances [CATS and JENELIUS, 2014; BACHMANN, TSAKARESTOS et al., 2023]. Hence, it is not surprising that each of the introduced academic investigations in PCIM considers the PT supply's capacity. However, as suggested by CATS and JENELIUS [2014], it needs effective PI schemes and proactive fleet management to avoid such negative effects. Effective PI schemes could be interpreted here as concrete path advice, redirecting passengers onto paths with sufficient remaining capacity. Therefore, it is also interesting to compare the various considerations of the works introduced here in this matter. PI is subdivided into its content, schemes, and channels:

1) PI content: CATS and JENELIUS [2014], ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] inform the passengers about expected arrival times during the incident, and the latter two also inform about train occupancies. The same is true for the *Corman group*¹ who instead of mentioning estimated arrival times mention the estimated duration of the incident as well as the affected PT lines to the travellers. MÜLLER-HANNEMANN et al. [2019] do not inform about occupancy in general but suggest connections with low occupancy, furthermore, the authors assume that the passengers inform themselves and choose their path selfishly. VAN DER HURK et al. [2018], BACHMANN, RAU et al. [2021], BACHMANN, DANDL et al. [2023] and MO, KOUTSOPOULOS, SHEN et al. [2023] give concrete path advice to affected passengers.

2) PI schemes: MÜLLER-HANNEMANN et al. [2019], BACHMANN, RAU et al. [2021], BACHMANN, DANDL et al. [2023] and MO, KOUTSOPOULOS, SHEN et al. [2023] do not mention any schemes; ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] disseminate PI at stops and/or on trains; CATS and JENELIUS [2014] test four different PI schemes, namely: no information, information at stops only, clustered information at transfer hubs, and network-wide information; the *Corman group*¹ simulates five PI schemes that differ in the time of dissemination: no information, timely information (at the start of the incident), advanced information (before the incident), "equilibrium with disruption" which is a special case of the advanced information and at the start of the trip. VAN DER HURK et al. [2018] assume perfect information availability.

3) PI channels: as mentioned in Section 2.1.5, there are a variety of available communication channels in PT systems between the PT operator and the PT users. CATS and JENELIUS [2014], VAN DER HURK et al. [2018], MÜLLER-HANNEMANN et al. [2019] and ZHU and GOVERDE [2019a] do not clarify the PI channels they take into account; BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023] mention the use of all available channels, whereas the *Cormann group*¹ assumes appropriate available channels. MO, KOUTSOPOULOS, SHEN et al. [2023] consider smartphones, websites, and PI displays at stops. To conclude the works' considerations on PI, all of them consider PI as an important part of PCIM. However, some of them clearly identify PI channels to be used and also clarify when the information is to be given, whereas other works simply mention the necessity of information dissemination.

To take the reaction of passengers on the given PI into account, the travellers are dealt with in two different ways in the approaches presented. CATS and JENELIUS [2014] and ZHU

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

and GOVERDE [2019a] and the *Corman group*¹ handle the passengers individually, whereas the rest considers indivisible passenger groups. Even though it is easier to find sufficient room in the system for individuals, MÜLLER-HANNEMANN et al. [2019] argue that it is impossible to know who travels with whom. Furthermore, BACHMANN, DANDL et al. [2023] emphasize the availability of collective PI channels (i.e. speakers and dynamic PI displays) through which it is impossible to inform PT users individually. According to the PT user survey presented in Chapter 3 the collective PI channels are preferred by PT users during incidents. In addition, BACHMANN, TSAKARESTOS et al. [2023], who formulate requirements for an ideal PCIM method, point out that it could lead to confusion and mistrust if people travelling together would receive different path advice and that PI needs to be consistent across all channels along the whole journey. In summary, most works handle the passengers as indivisible groups, as it is who of the affected travels together and some PI channels do not provide for the possibility of informing PT users individually.

In any case, just because PI or even path advice is given, it does not necessarily mean that people follow given suggestions. Most of the academic investigations consider such behaviour. In the case of CATS and JENELIUS [2014] and the *Corman group*¹, this matter is handled by a path choice model (Section 2.1.2) which is influenced by the used PI scheme. In ZHU and GOVERDE [2019a], the path choice depends on the set maximum for the accepted delay. VAN DER HURK et al. [2018] and BACHMANN, DANDL et al. [2023] consider different rates of compliance. Deterministic values as well as a logarithmic function to calculate compliance rates are used. BACHMANN, RAU et al. [2021] and MO, KOUTSOPOULOS, SHEN et al. [2023] assume complete compliance and MÜLLER-HANNEMANN et al. [2019] do not mention such considerations. Mainly there are two approaches concerning the consideration of travellers' reaction to given PI. Studies which take path advice into account, consider different compliance rates or simply full compliance, whereas studies which provide affected passengers with the arrival times of services and crowding information use path choice models.

To implement the approaches introduced here, different tools and algorithms have been used in the studies. CATS and JENELIUS [2014] use the simulation tool BusMezzo [CATS, 2022]; VAN DER HURK et al. [2018] mention the developed algorithm advice and rolling stock rescheduling problem with uncertainty (ARSRU); MÜLLER-HANNEMANN et al. [2019] use the round-based public transit optimized router (RAPTOR) [DELLING et al., 2015] to find alternative paths and the Gurobi Optimizer [GUROBI OPTIMIZATION LLC, 2022] to solve their optimisation problem; ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] mention the event-activity network as well as the adapted fix and optimise (AFaO) algorithm they developed; the *Corman group*¹ uses the simulation tool MATSim [HORNI et al., 2016] to conduct their case studies; BACHMANN, RAU et al. [2021] present a numerical investigation whereas BACHMANN, DANDL et al. [2023] use SUMO and the Gurobi Optimizer [LOPEZ et al., 2018; GUROBI OPTIMIZATION LLC, 2022], both mention the developed PCIM method. MO, KOUTSOPOULOS, SHEN et al. [2023] mention a event-based public PT simulator. All of the works use different tools to implement the developed methods. The only

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

tool which is named more than once is the Gurobi Optimizer for solving optimisation problems.

In most works optimisation plays a role. CATS and JENELIUS [2014] and BACHMANN, RAU et al. [2021] just mention the possibility of optimising in the outlook; VAN DER HURK et al. [2018], MÜLLER-HANNEMANN et al. [2019] and BACHMANN, DANDL et al. [2023] and the *Corman group*¹ minimise the passenger delay; ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] aim on optimising the passenger-operator balance, and MO, KOUT-SOPOULOS, SHEN et al. [2023] optimise the passenger flow distribution.

Besides the used tools and optimisation goals, the presented investigations also differ in the inputs and assumptions considered in their case studies. VAN DER HURK et al. [2018], ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] use a part of the Dutch railway network; The *Corman group*¹ use the PT network of Zurich, Switzerland; CATS and JENELIUS [2014] test their approach in the PT network of Stockholm, Sweden; MÜLLER-HANNEMANN et al. [2019] use a part of the German long-distance railway network; BACHMANN, RAU et al. [2021] do not conduct a simulative but numerical investigation and use a simple artificial network to do so; BACHMANN, DANDL et al. [2023] apply a PT network design by UL ABEDIN [2019] of the Mandl network [MANDL, 1979]. The metro network of Chicago, USA, is utilised by MO, KOUTSOPOULOS, SHEN et al. [2023]. Concluding, all works use different environments for their case studies, besides the exception of the Dutch railway network which is used by VAN DER HURK et al. [2018], ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020]. Nevertheless, the rest of the input considered by them differs.

Furthermore, the publications vary in the assumed duration of the incident it ranges from half an hour in BACHMANN, RAU et al. [2021], CATS and JENELIUS [2014], ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] to four hours in VAN DER HURK et al. [2018].

Moreover, the demand input is assumed by ZHU and GOVERDE [2019a], ZHU and GOVERDE [2020] and BACHMANN, RAU et al. [2021], whereas the rest considers historical travel data.

The relative gained reduction in passenger delays of the presented academic investigations, shows a wide range with most results showing a positive effect of PCIM on the passenger delay reduction. CATS and JENELIUS [2014] accomplish a relative reduction of passenger delays of up to 4%, whereas one case shows an increase of 2.5% which is explained by overcrowding effects. The results of the case study by VAN DER HURK et al. [2018] show a relative passenger delay reduction of up to 30% while reducing the number of affected passengers by 20%. Nevertheless, in one case the relative delay is increased by 2%. MÜLLER-HANNEMANN et al. [2019] present results between 10.5% and 11.5% of relative delay reduction. The outcome of the work by ZHU and GOVERDE [2019a] ranges between 13.8% and 46.7% of relative delay can be further reduced due to adjusting the rolling stock rescheduling scheme to the outcome of their dynamic passenger assignment model. The *Corman group*¹ shows significant improve-

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

ments in the passengers' situation with relative delay reductions between 93.2% and 99.8%. The PCIM method by BACHMANN, RAU et al. [2021] achieves a relative delay reduction by 63.3% given path advice. This is further improved by the reallocation of capacity to 73.2% in a bus system and to 73.8% in an autonomous PT system. The highest accomplished reduction of passenger delays by BACHMANN, DANDL et al. [2023] is at 96.0%, however, in one case the delay increases by 15%. This is explained by overcrowded bus stops as well as that the used logarithmic function by VAN DER HURK et al. [2018] has not been fully adjusted to their passenger redirection approach. Furthermore, as elaborated on in Section 2.1.4, there are many factors influencing travellers' reactions to incidents as well as their willingness to follow path advice. Hence, the logarithmic function by VAN DER HURK et al. [2018] which merely considers the aspect of the travel time difference between the suggested path and the fastest alternative path does probably not live up to the complexity of passengers' willingness to comply to path advice. MO, KOUTSOPOULOS, SHEN et al. [2023] accomplish a relative delay reduction for the affected passengers of up to 20.6%. Nonetheless, also in this work, there is a case in which the delay increases by 6.4%. This is explained through the rationality with which passengers choose their path during normal PT operations. In summary, also the results of the various academic investigations introduced here concerning the relative delay reduction of affected passengers show a wide range of outcomes between an increase in delay of 15% and a reduction of up to 99.8%. However, as implied in this section there is a large variety among the approaches, input and environments introduced here through which it is inadequate and inappropriate to directly compare the results of the presented investigations. Nevertheless, the outcomes clearly indicated the potential of PCIM methods in general.

Another aspect of the academic investigations is the consideration of different kinds of PT modes. VAN DER HURK et al. [2018], MÜLLER-HANNEMANN et al. [2019], ZHU and GOVERDE [2019a], ZHU and GOVERDE [2020] and MO, KOUTSOPOULOS, SHEN et al. [2023] solely consider trains or metro, whereas CATS and JENELIUS [2014] and the *Corman group*¹ take several PT modes into account, however, in the last mentioned works the incident also occurs in the rail-bound part of the network. BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023] focus on bus operations.

As elaborated on in Section 2.2.1, *supply-centric* disposition measures are well investigated and part of the daily business of OCCs. In addition to the incident which disrupts the PT service, these disposition measures also cause alterations to the PT service, and thus the available alternative paths for the PT users. Therefore, it is not surprising that all investigations consider disposition timetables that reflect not only the service changes through the incident but also the taken disposition measures. While doing so, some works focus on specific measures. In the paper by CATS and JENELIUS [2014] it is holding; VAN DER HURK et al. [2018] and ZHU and GOVERDE [2020] and the *Corman group*¹ mention rescheduling, while ZHU and GOVERDE [2020] also use short-turning and flexible stopping and the *Corman group*¹ focus on rerouting and increase in capacity. Rerouting is also named by BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023], as well as line-splitting. In summary,

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

all academic investigation seem to share the understanding that it is necessary to consider *supply-centric* disposition measures taken when developing PCIM methods. Consequentially, all of them consider corresponding disposition timetables, whereas some of them even integrate such measures as part of their approach.

Even though disposition timetables, which are a result of the dispatchers' decisions, are taken into account, the OCC is mentioned, at most. The same goes for an integration of PCIM into OCCs as suggested by BACHMANN, TSAKARESTOS et al. [2023]. The MÜLLER-HANNEMANN et al. [2019], ZHU and GOVERDE [2019a] and ZHU and GOVERDE [2020] and the *Corman group*¹ do not mention it. This is one of the major differences in comparison to the approaches in practice, which are introduced in the next section.

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

Reference	Cats and Jenelius [2014]	van der Hurk et al. [2018]	Müller- Hannemann et al. [2019]	Zhu and Goverde [2019; 2020]	Corman group ¹	Bachmann et al. [2021; 2023]	Mo, Kout- sopoulos, Shen et al. [2023]	
Focus	Passenger in- formation	Path advice, capacity adjustment	Computa- tional imple- mentation of path advice	Passenger re- assignment	Passenger be- haviour	Path ad- vice, capacity adjustment	Passenger flow distribu- tion	
Main goal	Analysis of PT system vulnerability	Reduction of passenger delay	Reduction of passenger delay	Reduction of passenger delay	Reduction of passenger delay	Reduction of passenger delay	Reduction of passenger delay	
PT system's capacity	Considered	Considered, adjusted	Considered	Considered, adjusted	Considered	Considered, adjusted	Considered	
PI content	Expected ar- rival times	Path advice, capacity shortages	Self-informing passengers assumed	Expected arrival times, train occupan- cies	Affected lines, disrup- tion duration, vehicle capa- cities	Path advice	Path advice	
PI schemes	At stops, clustered, network wide	Perfect in- formation is assumed	Not men- tioned	At stations and/or on trains	Timely, advanced information	Not men- tioned	Not men- tioned	
PI channels	Mentioned	Not con- sidered	Mentioned	Mentioned	Assuming appropriate channels	On all avail- able channels	Mobile phones, websites, displays at stops	
	Continues							

Table 2.5: Overview of academic investigations [BACHMANN, TSAKARESTOS et al., 2023]

ЗG

Table 2.5: Overview of academic investigations [BACHMANN, TSAKARESTOS et al., 2023] (continued)							
Reference	Cats and Jenelius [2014]	van der Hurk et al. [2018]	Müller- Hannemann et al. [2019]	Zhu and Goverde [2019; 2020]	Corman group ¹	Bachmann et al. [2021; 2023]	Mo, Kout- sopoulos, Shen et al. [2023]
Passenger handling	Individually	Indivisible passenger groups	Indivisible pas- senger groups	Individually / indivisible pas- senger groups	Individually	Indivisible pas- senger groups	Indivisible passenger groups
Compliance rate / path choice	Path choice dependent on PI scheme	Deterministic, logarithmic compliance assumed	Not men- tioned	Path choice dependent on accepted delay	Path choice dependent on PI scheme	Deterministic, logarithmic [VAN DER HURK et al., 2018] compliance assumed	Full compli- ance assumed
Used tools / algorithms	BusMezzo	ARSRU	RAPTOR, Gurobi	Event-activity network, AFaO	MATSim	SUMO, Gur- obi / PCIM method	Event-based PT simulator
Optimisation	Mentioned	Minimise pas- senger delay	Minimise pas- senger delay	Passenger- operator balance	Minimise pas- senger delay	Mentioned, minimise passenger delay	Passenger flow distribu- tion
Case study	Stockholm PT network	Part of the Dutch railway network	Part of the German rail- way network	Part of the Dutch railway network	Zurich PT network	Simple arti- ficial, Mandl network	Metro net- work of Chicago
Incident	0.5 hours, five different dis- rupted lines	3 – 4 hours, unsure period, link closure	Train cancella- tions based on historical data	1 – 2 hours, link closure	3 hours, link closure	0.5, 1 hour(s), link closure	1 hour, line suspension
Continues							

Table 2.5. Overview of academic investigations [DACHMANN, 1 SARARES105 et al., 2025] (Continued))		
Reference	Cats and Jenelius [2014]	van der Hurk et al. [2018]	Müller- Hannemann et al. [2019]	Zhu and Goverde [2019; 2020]	Corman group ¹	Bachmann et al. [2021; 2023]	Mo, Kout- sopoulos, Shen et al. [2023]
Demand in- put data	Historical travel data	Historical travel data	Historical travel data	Assumed demand	Historical travel data	Assumed demand / [MANDL, 1979]	Historical travel data
Modes of transport	Urban PT modes	Railway	Railway	Railway	Urban PT modes	Buses	Metro
Disposition measures	Disposition timetable, holding	Disposition timetable, rescheduling	Disposition timetable	Disposition timetable, short turn- ing, flexible stopping, rescheduling	Disposition timetable, rescheduling, rerouting, capacity increases	Disposition timetable, rerouting, line-splitting	Disposition timetable
Integration in OCC	Mentioned	Mentioned	Not men- tioned	Not men- tioned	Not men- tioned	Mentioned	Mentioned

Table 2.5: Overview of academic investigations [BACHMANN, TSAKARESTOS et al., 2023] (con
--

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

37

Approaches in Practice

In addition to the previously discussed academic investigations, approaches in practice are reviewed as well. In contrast to the academic investigations, which focus on ideal and optimised solutions to reduce passengers' delay in simulation environments, the focus of the approaches in practice lies on the feasibility of PCIM methods.

Literature Review: It is common practice in well-developed PT systems to provide passengers with the type and location of incidents. However, information about suggested alternative paths is rarely to be found in practice, especially in terms of capacity-aware passenger redirection. The dissemination of alternative paths, if applicable, is done in different ways in practice. Either in forms of collective information via dynamic PI displays at stops and in vehicles or through speaker announcements and staff. If path advice is given, it is usually based on pre-planned alternative paths for recurring disruptions and vulnerable parts of the network.

In the United Kingdom (UK) such pre-planned, static alternative paths are published by the National Rail of UK on their website [NATIONAL RAIL (UK), 2023]. The necessity of providing PT users not only with the information about an incident but also with alternative paths is also expressed by the customer information strategy of the Transport for London (TfL). The provision is realised through staff at stations that receive real-time information on the PT system's status via social media such as Twitter. The crowding at stations is also calculated based on electronic ticket data and provided in real-time [TFL, 2016; TFL, 2017]. In addition, the real-time PI is made available for third parties via an open source data platform [TFL, 2023].

Nevertheless, some agencies do not publish any or very little information about their customer information strategies. Transport for New South Wales (TfNSW) for instance, mention in their incident management framework that informing the PT user is important and that it is the responsibility of the communications coordinators to ensure that corresponding PI is given [$T_{\rm FNSW}$, 2023].

Approaches that go beyond the aforementioned state of practice are mostly part of smartphone apps and referred to as Mobility-as-a-Service (MaaS). One example is the French smartphone app "monRER A" by the the autonomous administration of the Paris transport services [Translated from French: Régie Autonome des Transports Parisiens] (RATP). With this app, user can monitor their trip and receive information about delays and alternative paths in accordance with their origin and destination, in case an incident occurs. An alternative path is taken into account as such by the app when it fulfils the criterion of reducing the user's travel time [RATP, 2023].

A customisable MaaS app is offered by the German company Mobimeo GmbH. One relatively famous example of this app is the DB track agent [Translated from German: DB Streckenagent] from the German railway operator DB. This app tracks the journey of its user and suggests alternative paths in case of disruption. Micro mobility services such as bike and e-scooter sharing are also included [MOBIMEO, 2023]. Comparable to the previous approach is the service provided by the Montréal transport company [Translated from French: Soviété de Transport de Montréal] (STM). Twitter notifications are used to provide PT users with information about an incident such as location and an estimation of its duration. In addition, a short redirection with alternative metro lines is sent with a suggestion to search for an alternative path by bus via the STM app [INIT, 2023; STM, 2023].

These examples give an adequate representation of the state of practice of PCIM in welldeveloped PT systems. Even though path advice is provided in some cases, it is done without the consideration of capacity constraints, which can lead to secondary disturbances [CATS and JENELIUS, 2014; BACHMANN, TSAKARESTOS et al., 2023]. Nevertheless, approaches in practice are introduced below that take capacity constraints into account when redirecting passengers.

In Germany, a project about the targeted redirection of passengers was conducted over several phases and years. A pre-phase, financed by the German Federal Ministry of Transport, Building, and Housing [Translated from German: Bundesministerium für Vekehr, Bauen und Wohnen] (BMVBW), had the main goal to define passenger redirection plans in case of often reoccurring incidents, using the experience of every day OCC practice [BMVBW, 2001]. Two standardised interfaces were used which have been developed by the association of German transport companies [Translated from German: Verband Deutscher Verkehrsunternehmen] (VDV), the VDV453 which links ITCS of different operators, and the API VDV454, which connects ITCS with online PT trip planners [VDV, 2001, pp. 164–165; VDV, 2018a; VDV, 2018b]. The latter enables travellers to look for paths through the PT network based on real-time data. BMVBW [2001] suggests that PCIM need to be integrated in the system architecture of an OCC.

Building upon this pre-phase, in the first phase of the project, which has been financed by the German Federal Ministry of Transport, Building, and Urban Development [Translated from German: Bundesministerium für Vekehr, Bauen und Stadtentwicklung] (BMVBS), the theoretical framework for the redirection of passenger flows has been developed and adopted to concrete procedures and calculations [BMVBS, 2009]. One result is the identification of three distinct passenger groups, divided by their geographical relation towards the incident site, which need different treatment during an incident:

(a) passengers who are already in the system and in the close vicinity of the incident without any alternative paths, who are in need of the deployment of additional PT vehicles;

(b) travellers who are also already in the system and on their way towards the incident, however, who have still the possibility of altering their journey in accordance with the incident situation, who are in need of corresponding PI at stops and in PT vehicles to avoid the incident and unnecessary delays;

(c) PT users who have not started their trip yet and have thereby the possibility of replanning their journey through the PT network and also to post-pone their trip, if possible, and are, therefore, in need of corresponding internet-based PI via PT operators' websites and online trip planners. This concept was tested in an offline trial run, using the PT trip planner called electronic timetable information [Translated from German: Elektronische Fahrplan Auskunft] (EFA). The output of EFA is path advice for the passenger groups (b) and (c) considering qualitatively the capacities of the PT lines, meaning that PT lines with similar capacities as the disrupted lines are considered for the path advice. The concept and its results were discussed by a panel of German operators, who concluded that passenger redirection during incidents is generally reasonable. The suitability of existing operation software was explored and further research gaps were identified. Moreover, shortcomings of existing interfaces have been found [BM-VBS, 2009].

BMVBS [2011] used these findings in the second phase of the project to develop two corresponding prototypes for the cities of Berlin and Stuttgart in Germany to verify the potential of this PCIM system. Offline data from the respective OCCs was used as inputs. One finding was that the number of possible alternative paths needs to be reduced to keep the passenger redirection applicable for both, the passengers and the dispatchers in the OCCs. Another major outcome was the awareness of the crucial role of the capacities and the passenger volume of the PT system for the redirection process. Due to the lack of APC systems or comparable sources (Section 2.1.1) in the PT systems of Berlin and Stuttgart, realistic online data was not available. Furthermore, it is concluded that the minimum incident duration for a collective passenger redirection is 30 minutes, considering the processing times of the OCC in such situations as well as the extension of travel times due to the longer alternative paths.

The consortium of PT operators of this project phase has agreed on the statement that for an incident duration of less than 30 minutes, the simple information about the existence of an incident would be sufficient for PT users to re-plan their trips. Since this consortium was not accompanied by an academic institution, details about the developed prototype, the conducted tests, and the results have not been published. The main conclusion has been that further refinement of interfaces is necessary and that one of the biggest challenges is the prediction of passenger demand. The most important achievement of phase two was the development of the two prototypes mentioned before.

Based on these, online tests were conducted in the third and last phase of the German passenger redirection project which was financed by the BMVI. The aforementioned lack of real-time demand data was taken into account during this project phase by limiting the number of potential paths even further. Furthermore, the targeted groups for redirection were reduced to group (a), which is at the incident site, and part of group (b) which is already in the close vicinity of the incident. This was mainly since at the time of the project phase, PT trip planner apps on smartphones were widely used. This lead to a simplification of the original idea of wide-area redirections to individualised PI.

Nevertheless, this phase has catered for a communication standard for incident occurrences from OCCs to operators' trip planner apps. Moreover, a field test with the further developed prototype from phase two were carried out in the operation area of the transport association Berlin/Brandenburg [Translated from German: Verkehrsverbund Berlin/Brandenburg] (VBB), Germany. To ensure consistency between the individual and collective information channels, the PI was transmitted to the mobile trip planner Hanover consulting timetable information system [Translated from German: Hannover Consulting Fahrplan Auskunftssystem] (HAFAS)

as well.

Because real-time disposition timetables, reflecting the altered PT supply during the incident was not available during the field trial, path calculations were done based on static timetables which led to some miscalculations. Apart from that, the system performed well from the operators' perspective. However, the suggestion of not ideal alternative paths from the passengers' point of view, especially of experienced PT users led to the realisation that transparency about the incident and its consequences is the key of gaining higher compliance rates during the redirection process.

The results of a customer trial reveal that about 57% of the customers are willing to follow path advice by the operator. This is the same share of survey participants which would follow connection suggestions in case of a disruption in long-distance railway services according to WILKE [2023]. Further main conclusions are that a collective redirection is technically feasible and reasonable, however, to be successful such a PCIM system depends on real-time data on the demand of the residual capacity of the available paths and the disposition timetable [BMVI, 2019].

A passenger redirection project from Japan is presented by TSUCHIYA et al. [2006]. A prototype for the Japanese railway network is introduced that provides passengers at stations with a map of alternative paths during an incident to help them find a suitable path. For the prototype's input, a database with historical incidents and their durations was built to estimate an incident duration which is considered in the path advice. The system calculates the travel time of all possible paths for each station to each destination station, taking into account the estimated incident duration as well as a recovery time, which reflects the time from the dissolution of the incident's source to the return to normal operations.

Furthermore, the prototype is capable of considering more than one incident. PT users who want to receive path advice in correspondence to their travel plan can use an online app. $T_{\rm SUCHIYA}$ et al. [2006] demonstrate that in a less dense network such as a long-distance railway network, pragmatic solutions to the passenger redirection problem are feasible. It is indicated that an operational prototype has been developed which is yet to be tested in the field.

A more individualised approach comes from Italy by BRUGLIERI et al. [2015]. The authors developed a prototype algorithm as an extension for PT trip planners to allow their users to look for paths during an incident. The algorithm considers the closure of lines and stations as incident scenarios and was integrated into the PT trip planner of Milan, Italy.

Residual capacity in the PT system has not been considered and incident information is assumed to be provided by the OCC. Since this work focuses on the redirection of individuals and not on the system's optimal redirection of the collective of affected passengers it is not considered in the following summary and comparison of the approaches in practice introduced here.

Comparison: Table 2.6 which is also adopted from BACHMANN, TSAKARESTOS et al. [2023] shows the main characteristics of these publications. Even though the focus of all approaches in practice mainly lies on the feasibility of PCIM methods, they still have distinctions in their approaches, goals, additional foci, and conducted field trials or case studies,

respectively.

In addition to the general aforementioned focus, the approaches in practice have a common main goal which is determining feasible path advice [BMVBW, 2001; TSUCHIYA et al., 2006; BMVBS, 2009; BMVBS, 2011; BMVI, 2019].

Furthermore, the work by BMVBS [2009] focuses on the development of procedure concepts for a PCIM in an OCC. Another focal point for BMVBS [2011] is the development of a prototype as well as a lab test. In addition, BMVI [2019] focuses on technical aspects and a customer trial with the prototype developed by BMVBS [2009]. TSUCHIYA et al. [2006] also focus on the development of a prototype. The approaches in practice have the common goal of validating the feasibility of an implementation of PCIM in practice. In addition, some of them developed corresponding prototypes in the process.

As the academic investigations, all approaches in practice also consider the capacity of the supply. However, most of them do so in a qualitatively manner.

As mentioned in Section 2.1.5, PI plays a crucial role in PCIM. In all approaches in practice, its content is path advice, BMVBS [2011] and BMVI [2019] also broadcast the location and kind of incident. As mentioned in Section 2.1.5, knowing the reason for an incident increases the understanding of PT users, which is also pointed out by BMVI [2019].

Regarding PI schemes, BMVBS [2009] considers informing clustered passenger groups, which is adopted by BMVBS [2011] and BMVI [2019]. TSUCHIYA et al. [2006] present PI at all stations. BMVBW [2001] merely mentions PI schemes in general as well as PI channels, the latter is also true for BMVBS [2011]. BMVBS [2009] considers speakers, displays, and PT trip planners as possible channels; BMVI [2019] smartphones and PT trip planners and TSUCHIYA et al. [2006] smartphones and displays. As in the academic investigations, also in the approaches in practice, the role of PI in PCIM is stressed. However, here, the focus lies more on the considered PI channels than on the information schemes as it is the case in the academic investigations. Furthermore, the content is path advice in all works, which is not the case for the academic investigations.

All works handle the affected passengers as a collective and none of them considers a compliance rate. However, BMVI [2019] determined a possible compliance rate of 57% through a customer trial.

Since no simulations have been conducted, the used tools are rather of practical nature, such as PT trip planners [BMVBS, 2009], own prototypes [TSUCHIYA et al., 2006; BMVBS, 2011], OCC infrastructure [BMVI, 2019], or a database with preselected scenarios [TSUCHIYA et al., 2006]. BMVBW [2001] presents purely conceptional work. As investigations in practice have mostly also economic intentions, publishing findings are often not intended or has a side role, at most. Therefore, it is harder to find publications, which is one of the reasons that most of the mentioned approaches in practice are from Germany because these are the works, to which the author has access. However, even in such reports, developed algorithms are not described. This might be one of the explanations for the still-existing gap between research and practice which is further discussed in the next section. BMVBS [2011] and BMVI

[2019] did also not share their algorithms, the others did not even mention any.

None of the approaches in practice mentions any kind of optimisation. Nevertheless, all of them conducted some kind of field trial. BMVBW [2001] solely focuses on the visualisation of path advice. BMVBS [2009] presents calculated examples, whereas TSUCHIYA et al. [2006] conducted an offline prototype test. BMVBS [2011] tested multiple scenarios in two German cities, Berlin and Stuttgart, and BMVI [2019] did a field trial in real operation in Berlin/Brandenburg, Germany.

The approaches in practice merely mention PI channels and OCC infrastructure as used tools. Algorithms and prototypes are mentioned at most, which is partly due to the fact that commercial interests are involved in these works and that it is less common to make findings publicly available in practice. Nonetheless, some of them conducted field trials.

When it comes to considered incidents, the practical works show also different considerations. BMVBW [2001] defines incident classes. BMVBS [2009] considers an incident duration of 30 minutes and various scenarios of disrupted PT services. TSUCHIYA et al. [2006] and BMVBS [2011] solely mention incidents but do not specify any features. BMVI [2019] takes historical information of 31 incidents into account. However, all approaches in practice solely mention the necessity of a demand input but do not specify it.

The four publications about the German project consider various urban PT modes [BM-VBW, 2001; BMVBS, 2009; BMVBS, 2011; BMVI, 2019], whereas TSUCHIYA et al. [2006] apply their developed prototype in the railway system of Japan.

BMVI [2019] is the only practical study that does not consider *supply-centric* disposition measures. Since the main focus of these works has been the feasibility in practice, all of them considered the integration in an OCC, the heart of incident management in PT systems [CARREL, MISHALANI et al., 2010; BRIEM, BUCK, MAGDOLEN et al., 2020; BACHMANN, BRIEM et al., 2022]. BMVBW [2001] considers it in the system's architecture, BMVBS [2009] go a step further and developed a concept for the PCIM to be an integrated part of the OCC. Based on that, BMVBS [2011] presents the prototypes as part of the OCC and BMVI [2019] links them to the OCC's existing system architecture via standardised interfaces.

Table 2.6: Overview of approaches in practice [BACHMANN, TSAKARESTOS et al., 2023]								
Reference	BMVBW [2001]	BMVBS [2009]	BMVBS [2011]	BMVI [2019]	Tsuchiya et al. [2006]			
Focus	Feasibility	Concept for pro- cedures in an OCC	Prototype devel- opment and lab test	Customer trial of the prototypes from BMVBS [2009], technical aspects	Prototype devel- opment			
Main goal	Feasible path sug- gestions	Feasible path sug- gestions	Feasible path sug- gestions	Feasible path sug- gestions	Feasible path sug- gestions			
PT system's ca- pacity	Considered	Considered	Qualitatively con- sidered Path advice, loca-	Qualitatively con- sidered Path advice, loca-	Qualitatively con- sidered			
PI content	Path advice	Path advice	tion, and kind of the incident	tion, and kind of the incident	Path advice			
PI schemes	Mentioned	Informing clustered groups of passengers	Based on the concept of BM- VBS [2009]	Based on the concept of BM- VBS [2009]	PI presented at all stations			
PI channels	Mentioned	Speakers, displays and PT trip plan- ners	Mentioned	Mobile phones, PT trip planners	Displays and mo- bile phones			
Passenger hand- ling	Passenger collect- ive	Passenger collect- ive	Passenger collect- ive	Passenger collect- ive	Passenger collect- ive			
Compliance rate / Path choice	Not mentioned	Not mentioned	Not mentioned	Evaluated compli- ance rate via ques- tionnaire	Not mentioned			
Continues								

Table 2.0. Overview of approaches in practice [DACHMANN, TSAKARESTOS et al., 2025] (continued)						
Reference	BMVBW [2001]	BMVBS [2009]	BMVBS [2011]	BMVI [2019]	Tsuchiya et al. [2006]	
Used tools / Al- gorithms	Purely conceptual work	PT trip planner / Pseudocodes	Own prototypes / Algorithms not disclosed	OCC infrastruc- ture / Algorithms not disclosed	Own prototype using a database with pre-selected scenarios	
Optimisation	Not mentioned	Not mentioned	Not mentioned	Not mentioned	Not mentioned	
Case study	Exemplary visual- isation	Calculated ex- amples	Multiple scenarios in Berlin and Stut- tgart	Test in real opera- tion in Berlin	Offline prototype test	
Incident	Definition of incid- ent classes	0.5 hours, different service disruptions	Mentioned, not specified	31 historical incid- ents	Mentioned, not specified	
Demand input data	Mentioned	Mentioned	Mentioned	Mentioned	Mentioned	
Modes of trans- port	Urban PT modes	Urban PT modes	Urban PT modes	Urban PT modes	Railway	
Disposition measures	Considered	Considered	Considered	Not considered	Considered	
Integration in OCC	Considered in sys- tem architecture	Concept planned as an integral part of an OCC	Prototypes planned as an integral part of an OCC	Prototypes linked to the OCC via standardised inter- faces	Considered	

Table 2.6: Overview of approaches in practice	[BACHMANN, TSAKARESTOS et al., 2023]	(continued)
---	--------------------------------------	-------------

2.3 Summary and Research Gaps

The sheer diversity of incidents and their characteristics undoubtedly implies that the management of incidents is very complex, as has also been pointed out in this chapter. Most investigations about incident management focused on *supply-centric* disposition measures, describing the readjustment of PT supply to a present incident situation and its return to operations as scheduled.

Typical measures are holding, stop-skipping, rescheduling, rerouting, short-turning, linesplitting, and various measures to provide for additional capacity. Less has been done in the field of PCIM. Furthermore, most of the scientific literature presented here focuses on railbound transport either by merely considering railway services or in case of investigations taking several PT modes into account, the incident takes place in the railway part of the respective network. The only papers about capacity-aware passenger redirection or PCIM, respectively, which are focussing on bus operations are by BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023] which describe earlier stages of the methodology presented here. Due to this reason, even though this thesis considers multi-modal urban PT systems, there is an emphasis on bus operations.

Furthermore, the academic investigations show a weak connection to practice meaning that aspects which need to be considered for an implementation of PCIM in practice are ignored, such as the infrastructure of OCCs, as also pointed out by BACHMANN, TSAKARESTOS et al. [2023]. However, the approaches in practice introduced here, show a weak connection to the state of the art in research, meaning that the developed solutions do not live up to possibilities which have already been explored in research, such as optimisation of path advice algorithms.

Therefore, to contribute to the closure of the gap between research and practice, OCCs were visited [BACHMANN, BRIEM et al., 2022], and a PT user survey was conducted. The survey results are presented in the next chapter. Those help to understand the two sides which need to be connected during incidents through PCIM, the PT users and PT operators or the passenger and supply side, respectively. This shapes the methodology presented in this thesis as indicated in Figure 1.1.

Moreover, the findings of the OCC visits and the PT user survey has supported the development of the model presented in Chapter 4. This model is a novel contribution to the research field of PCIM.

This literature review clearly points out the potential of PCIM in PT for its users and operators due to the reduction of delays and pressure on the PT system in terms of demand on alternative paths and accelerated recovery from incidents. Further important findings are:

- factors which influence the reaction of passengers to incidents,
- the necessity of considering passengers not following path advice (compliance),
- available PI channels in a PT system,
- retrievable data concerning the PT supply,
- and available information about an incident as well as disposition measures taken by OCC staff.

Chapter 3

Passenger Perceptions of Incidents and Incident Management

One of the most important aspects of the method presented here is moving the passengers' view during incidents into focus (Chapter 1). Therefore, it is crucial to understand what actually happens to passengers during incidents, how they perceive them, how they feel about incidents, and how such inconvenient situations can be improved for them. A basis for this understanding has been laid out in Section 2.1.4. To complement this basis and to improve the understanding of the passengers' perception of incidents and incident management a PT user survey was conducted. Its setup and results are presented in the following.

3.1 Passenger Survey Setup

This web-based PT user survey was conducted in Berlin, Germany and Singapore. Both cities are very big, with a population of 3.7 and 5.6 million inhabitants, respectively. In both cities a large and complex PT network is in operation. In Berlin, the registered cars to inhabitants ratio is 33.3% (in 2020) and in Singapore, 11.5% (in 2020), which indicates the importance of the PT system in both cities [SINGAPORE DEPARTMENT OF STATISTICS, 2022; STATISTICS OFFICE BERLIN-BRANDENBURG, 2022]. The level of PT service is comparable as well as the available resources and objectives of the PT systems.

Table 3.1 gives some information about the survey. In Berlin, the survey took place at the end of 2020, ran for 55 days, and the answers of 658 participants were analysed. The survey platform was provided by the Berlin transport company [Translated from German: Berliner Verkehrsbetriebe] (BVG) which is the main PT provider in Berlin. BVG also promoted the survey on their website [BVG, 2023] and on the BVG app. In Singapore, the survey ran at the beginning of the year 2021 for 43 days. The answers of 189 participants were analysed. The survey was advertised via the websites of the PT operator Tower Transit and the research company TUMCREATE, as well as via the Facebook profile of the German Embassy in Singapore, a Facebook group of PT users and a Telegram group for surveys conducted in Singapore [TOWER TRANSIT, 2020; TUMCREATE, 2021; DUROV, 2023; META, 2023].

Unfortunately, there is an obvious gap between the number of participants in the two cities. This is despite the fact that more efforts have been made in Singapore to reach more people, as previously mentioned. For the survey in Singapore a confidence level of about 80% is achieved with a margin of error of 5%, whereas for the survey in Berlin a confidence level of 99% is achieved with a margin of error of 5% [HERZ et al., 1992; MOMENTIVE, 2023].

Therefore, a higher level of confidence would have been desirable for the survey in Singapore. Nevertheless, the results still give an insight in PT users' perception of incidents and incident management. Additionally, the results of the two cities are somewhat comparable as shown in the next section.

City	Start date	End date	Number of days	Number of participants
Berlin	28.10.2020	21.12.2020	55	658
Singapore	04.01.2021	15.02.2021	43	189

Table 3.1: Information about the PT user survey

The survey contains a revealed preference part that casts light on incidents which the participants experienced themselves, as well as on the background of the participants. A stated preference part is about incident situations in an autonomous PT system.

Through the questions about the experienced incident, the participants have been asked details about the incident they encountered, how the PT operator and they reacted to the incident, how they received information and which, and about their preferences regarding PI during incidents. Furthermore, they also were asked how they felt about this event. One important question in this section was about their willingness to follow a path suggestion, given by the operator during an incident. In the stated preference part, the participants were asked to imagine that their experienced incident would happen in an autonomous PT vehicle. Again, they were asked how they felt about this incident in the new imagined environment and what information they would like to perceive and how. Since there is no PT driver in autonomous vehicles to turn to, incident situations might be more stressful for passengers.

The survey part about the background of the participants contained questions to put the given answers in perspective of the participants' age, profession, education, transport habits, etc.

3.2 Survey Results

The presentation of the results is structured as follows. First, the sample description is presented to get an overview of the background of the participants and to better understand the given answers. Second, the answers concerning the incidents which the participants experienced are presented. Third, their preferences in PI provision during incidents are introduced. The results presented last show answers to questions regarding the perceived emotions of the PT users during the experienced incident as well as their attitude towards the question how they feel about an incident taken place in an autonomous PT system.

Pie charts show the results of single choice questions, bar charts the results of multiple choice questions, and columns charts the results of scaling questions. Thereby, the type of the corresponding question is indicated by the kind of chart. The survey comprised more than 50 questions of which only those which are most relevant to the topic of this thesis are presented.

3.2.1 Sample Description

The target group of this survey were PT users who knew the respective PT network reasonably well and experienced at least one incident in it. Therefore, the questionnaire started with two screening questions about whether the participant had ever experienced an incident in PT and if they lived in the respective city and for how long. Participants who had never experienced an incident in PT or did not reside in the respective city were excluded. The rest of the participants have almost completely lived at least for two years in the respective city (96.2% in Berlin and 98.4% in Singapore, Figure B.1). This is also important, as participants were asked to answer the questions *without* the consideration of the COVID-19 pandemic, which hit the global society at the beginning of 2020 [WOLRD HEALTH ORGANIZATION, 2021].

Furthermore, as the target group were PT users, they were asked about their primary means of transport as well as the frequency with which PT has been used.

Please note that "N" below each chart expresses the number of participants who answered the respective question. As can be seen from Figure 3.1, for most participants PT was by far the most important means of transport. Compared to the previously mentioned car to inhabitants ratios, also for the participants of the survey in Berlin, the car plays a bigger role (6.8%) than for the participants in Singapore (3.2%). Section 2.1.4 also discusses the influence of car ownership, not only on the travellers' reactions to incidents, but also on mode choice in general. In Berlin, "Cycling" was named by 7.6% of the participants, and 2.6% by the participants in Singapore. In the questionnaire of Singapore, "personal mobile device (PMD)" was added to the category of "Cycling", which is the umbrella term for e-scooters and comparable devices. Walking and riding motorcycles or taxis play a minor role for the participants of both cities.

To understand the participants' transport habits even better, Figure 3.2 presents the participants' usage frequency of PT. Again, the answer options differ slightly between the two cities, in the questionnaire for Berlin an "(Almost)" was added to the answers "Daily" and "Never" as well as the answer option "Don't remember". The second answer option in Singapore was "Most days", whereas in Berlin it was "Several times [a week]", meaning the same. Most participants use PT daily, with 68.5% in Berlin and 52.9% in Singapore. Most of the rest use PT either "Several times [a week]"/"Most days" with 19.9% in Berlin and 39.7% in Singapore or once a week with 6.4% in Berlin and 5.3% in Singapore. The remaining answers appear in an insignificant number with only 1.5% in Berlin and 0.5% in Singapore stating that they never use PT.

Looking at the three presented questions, it can be assumed that the target group of PT users who had an understanding of the respective PT network is well represented by the sample.

Within this group of people, the goal was to have a good distribution of participants in matters of age, gender, education level, and labour status. In this way, different demographic groups within the respective societies are represented fairly evenly.

In Berlin, the under-20-year-olds (5.5%) and the 70+ olds (4.0%) are rather underrepresented, whereas the rest is fairly distributed with shares of 11.9% for 60 to 69 year-olds to 25.2% for 30 to 39 year-olds. Nevertheless, the age distribution is comparable to the results from the census of 2011 [STATISTICS OFFICE BERLIN-BRANDENBURG, 2022]. However,



Figure 3.1: In the last 2 years, before the COVID-19 pandemic, which has been your primary means of transport?



Figure 3.2: In the last 2 years, BEFORE the COVID-19 pandemic, how often do you take PT?

the age distribution in the census refers to all inhabitants. A shift in the age distribution among PT users might be possible as especially young and old people rely on PT as some of

them are not allowed or not capable of driving a car or bicycle. In Singapore, the 20 to 29 year-olds are strongly overrepresented with 55.6% resulting in a comparative underrepresentation of the rest. Especially the age groups of 50 to 59 year-olds (2.1%), 60 to 69 year-olds (0.5%), and 70+ year-olds (0.5%) are strongly underrepresented, which needs to be kept in mind when interpreting the results of the survey. The remaining 40 to 49 year-olds (8.6%), under 20 year-olds (13.4%) and 30 to 39 year-olds (19.3%) are fairly represented compared to Singapore's census of 2010 [SINGAPORE DEPARTMENT OF STATISTICS, 2022]. Please note that in Singapore two participants have been excluded from the results of this question as they answered this question with a four-digit number. It is therefore assumed that they did not wish to reveal their actual age (Figure B.2).

Regarding the gender of the participants, we have a slight shift towards male with 58.2% of male, 40.7% of female and 1.7% of diverse in Berlin and 58.7% of male, 40.1% of female and 0.5% of other in Singapore (Figure B.3). However, in the censuses of both cities, there is a slight switch towards females with 48.6% of males and 51.4% of females in Berlin and 49.3% of males and 50.7% of females in Singapore [SINGAPORE DEPARTMENT OF STATISTICS, 2022; STATISTICS OFFICE BERLIN-BRANDENBURG, 2022]. Both censuses do not consider other gender identifications.

The last two distributions to describe the survey sample are about the highest accomplished education and current labour status of the sample. In Berlin, about 30% finished school but did not yet complete vocational training or gain a job degree of any kind. Over 40% have obtained an academic degree, which shows a bias towards the higher educated part of the society, compared to the census. In Singapore, compared to the census, there is also an obvious bias towards the higher educated part of the society, with almost half having a high school or job degree, which is summarised in this questionnaire, and more than 40% having an academic degree, which fits well to the bias age distribution (Figure B.2).

Concerning the labour status, there is a bias towards the working part of the society in Berlin with over 70% working either full-time (56.5%), part-time (11.6%) or self-employed (5.6%) compared to 48.9% in the census. In Singapore, according to the census, 67.6% of the population is working, whereas among the survey participants about half are working. There is a clear bias towards students with 38.6%, which again fits well with the bias regarding the age in the sample from Singapore (Annexe B.1).

Therefore, it can be concluded that the samples of both cities have a slight bias towards males and the samples' education level is higher than the education level of the respective society. Regarding the age, the sample of Berlin is somewhat fitting, whereas the Singaporean sample is too young with a focus on the age of 20 to 29 years. The Berlin sample has a bias towards the working part of the society and the Singaporean sample towards students. These biases need to be kept in mind, when interpreting the following survey results [SINGAPORE DE-PARTMENT OF STATISTICS, 2022; STATISTICS OFFICE BERLIN-BRANDENBURG, 2022].



Berlin (N=658) Singapore (N=189)

Figure 3.3: Which of the following happened during the incident?

3.2.2 Experienced Incidents

This section includes questions which focus on the incidents experienced by the participants. First, the question about what happened is shown in Figure 3.3. In addition to understanding what kind of incident the participants encountered, this question was also meant to help the participants refresh their memory about the incident. As Figure 3.3 shows, most of the incidents involved the breakdown of a PT vehicle, especially rail-bound vehicles. It is to be mentioned that a few days before the survey was launched in Singapore a mass rapid transit (MRT) breakdown occurred in the PT system, which on the one hand might be an explanation for this imbalance in terms of different kinds of experienced an incident in PT. However, also in Berlin, most participants experienced a breakdown in a railway service, the suburban train (SUBT). Please note that the questions given to participants in Berlin are slightly different to the ones given in Singapore. In Berlin, traffic congestion, ambulance deployment, and traffic accidents are standing-alone answers, whereas in Singapore the latter two were given as examples for traffic congestion. This question helps to understand the given answers to the following questions.

The next question is about at what stage of their journey the participants were when they noticed the incident. This gives an idea about the available information and about which information channels passengers were likely to had access to, such as speaker announcements



Figure 3.4: Which stage of your trip were you at when you found out about the incident?

or PI displays at stops (Figure 3.4, Section 2.1.5). As one can see, the differences between Berlin and Singapore are quite big here, whereas in Berlin most people noticed the incident while standing at the stop or being at the origin, in Singapore, most people were in a PT vehicle, on a MRT or a light rail transit (LRT). The location of PT users at the point of time at which they notice an incident can influence the way they are informed about it. In both cities most of the participants, 32.5% in Berlin and 31.7% in Singapore, have been informed by the respective PT operator (Figure B.8).

In the results from Singapore, it is noticeable that a lot of people noticed the incident due to the affected travel speed of the respective PT vehicle (22.8%). Furthermore, the most given answer which fell under the category "Other" were "the vehicle stopped unexpectedly or broke down" with 6% of all given answers. This can be explained by the kind of incident most participants experienced which involved some kind of PT vehicle breakdown. The third most given answer is that "the MRT did not arrive or depart" (14.3%). Passengers on board a breaking down vehicle usually notice the incident very quickly through the unexpected slowing down or stopping of the PT vehicle, whereas PT users at a station often realise it through the non-arriving vehicle. This fits well with the answers in Figure 3.4 about the place where people found out about the incident since most survey participants from Singapore were on a PT vehicle when they noticed the incident. In Berlin, apart from the most frequent answer which is also "informed by the operator", the second, third, and fourth most frequent answers are about not arriving or departing PT vehicles (46.0% in total). This again fits well with the results shown in Figure 3.4, since most people in Berlin noticed the incident while being at a stop. Please note that there are different rail-bound PT modes available in Berlin and Singapore, therefore, the answers differ, as you can see in the legends below the charts. Whereas there are Tram, SUBT and metro in Berlin, there are MRT and LRT in Singapore (Annexe B.2).

As mentioned before, the location of the participants also influences the availability of information channels. At stops and on PT vehicles there can be PI displays and speakers for announcements. Moreover, at stops, one will also find station personnel or ticket inspectors, and on board you are more likely to find a driver. Additionally, at the origin or destination people most likely gain information over smartphone apps or websites. The participants were asked if they got information from the operator and if yes, through which channel and what kind of information they were provided with (Figures 3.5 and B.9).

Multiple choices were possible for these questions. As about 30% of all participants in both cities received official information (Figure B.8), "N" is smaller in these questions. Among the people who answered the question, speaker announcements are the primary information channel in both cities. Whereas in Berlin, only the PI displays at stops and in PT vehicles represents another important channel, in Singapore, also the operators' staff and social media play an important role. The three most chosen channels, speakers, displays, and operators' staff, are all to be found at stops as well as on PT vehicles. Passengers were not only asked which channel was used to inform them, but also what information they were provided with.



Berlin (N=355) Singapore (N=121)

Figure 3.5: Which of these channels did you obtain official information from?

In both cities, the cause of the incident, affected PT services, and possible delays were often transmitted information with 69.6% in Berlin and 32.3% in Singapore (Figure B.9). One differ-

ence between the two cities in the context of these kinds of information is that the information about affected PT services and possible delays was more often given in Berlin with 27.9% and 42.8%, respectively, than it was given in Singapore with 23.3% and 33.9%, respectively. However, more information was provided in Singapore regarding the delay duration (26.5% in Singapore and 15.2% in Berlin) and advice about switching to another means of transport or PT line (27.5% in Singapore and 21.2% in Berlin). In Singapore, at least 9% got an advice for another PT line, but, how to go from there was not given (Annexe B.2).

The last question of the section about experienced incidents is about the reaction of the participants to the incident. It becomes obvious that the number one reaction to incidents in both cities was to wait for the next vehicle on the desired PT line. This is possible if the incident merely affects the PT vehicle but not the stop or the whole line as it would be the case for a blocked railway track for example (Section 2.1.3).

Please note that the participants in Berlin had one more answer option than the participants in Singapore in the question shown in Figure 3.6. It is also to be mentioned that the majority of the participants of Singapore, whose answers fall under "Other" (7.4%) pointed out that they were stuck in the PT vehicle and therefore did not have an option other than waiting in the PT vehicle. As Figure 3.3 indicates, most participants experienced some kind of vehicle breakdown. Relatively many people also chose to take another, unaffected path, in Berlin, it is the second strongest answer to this question. People who know the PT system well or are even experienced in reacting to incidents are usually doing well in finding an alternative path quickly. Also, a lot of people switched to other means of transport such as private cars, taxis, cycling, or walking. Especially severe incidents with long delays motivate people to find another possibility of getting to their destination if they are in a hurry, which was true for most of the participants.

For 78.3% or respectively 58.5% of the participants in Singapore and Berlin, the trip was not postponable according to the question about the urgency of the trip. Section 2.1.4 points out that there are many factors which influence passengers' reactions to incidents, car ownership and the purpose of the trip are for example two of them. Another answer option was about switching to shuttle services, such as rail replacement services. However, these are only established by the dispatchers if the incident is very severe and takes a lot of time to be dissolved (Section 2.2). As aforementioned, shortly before the survey was launched in Singapore, a MRT breakdown occurred and a shuttle service was probably established, which explains the comparably high number of people who switched to a shuttle service in Singapore.

For more information about the incidents, experienced by the survey participants it is referred to Annexe B.2. These charts presented show that incidents are very diverse in their kind and severity as well as their impact on and perception of PT users and the possible reactions to them.



Figure 3.6: How did you react to the incident?

3.2.3 Passengers' Preferences

The previous section presents how survey participants received information and what its content was. However, this does not necessarily mean that this has been their preferred information channel during incidents. As incidents are usually unexpected and confusing events, people might want to be informed differently. Therefore, the participants were asked about their preferred communication channel during incidents and which information they would like to receive in such situations (Figures 3.7 and 3.8).

The participants could tick up to two options. In this way, it was ensured that the participants did not tick all answers which helped to reveal actual preferences. In both cities, the speaker announcements and dynamic PI displays at stations and in vehicles play the most important role, however, in Berlin, the PI displays are the most preferred information channel with 67.3% and speaker announcements as second most preferred information channel with 45.3%. In Singapore, it is the other way round, speaker announcements are the most preferred channel with 63.0% and PI displays with 44.4% as the second most preferred channel. Also, the PT staff, such as PT drivers or station personnel, play an important role in both cities when it comes to PI dissemination. In Singapore, it is the third most chosen channel with 34.9% and in Berlin with 18.8% the fourth most. Whereas in Berlin the third most preferred information channel is the PT operators app (26.3%), this information channel plays only a minor role in Singapore (7.9%). This can be explained by the different ride fare systems in Berlin and Singapore. Singapore's PT system uses smart cards and credit cards and the fare is calculated for each trip individually by distance travelled [MINISTRY OF TRANSPORT, GOV-ERNMENT OF SINGAPORE, 2022]. In Berlin, there is a zone system and the fare depends on the number of zones passed on a trip. Moreover, it is necessary to purchase a ticket in Berlin,



■ Berlin (N=658) ■ Singapore (N=189)

Figure 3.7: What are your preferred communication channels during incidents?

which can be also done via the BVG app [STATE OF BERLIN, 2022]. Therefore, the app has more functions than PT operators' apps in Singapore, which serve mainly the purpose to inform.

A significant difference between the cities appears also in the usage of social media (e.g. Facebook, Twitter, Telegram, etc.). Whereas in Berlin only 7.6% of the participants chose this answer, in Singapore it was 28.0%. This is partly explainable by the fact that in Singapore, the survey was also published in a Facebook group and a Telegram channel, whilst the Berlin survey was solely shared on BVG's website and smartphone app. PT users who noticed the survey on a certain information channel were likely familiar with this channel and probably preferred it for incident information as well. Furthermore, the location where the participants were, when the incident occurred, might also influence the answers as it had probably also an influence on the question about through which channel they received information (Figure 3.5). However, since the establishment of the smartphone with all its functions and capabilities, social media, websites and PT apps can easily be used from anywhere.

Another topic regarding the preferences of information during incidents is the content of the information the passengers receive. In order to be able to react properly to an incident, travellers need adequate information [BACHMANN, TSAKARESTOS et al., 2023]. As it is shown in Figure 3.6 the three most often used reactions to an incident were: waiting for the next PT vehicle of the same line, using another line or a shuttle service, or switching to another means



Figure 3.8: What would have been the most important information for you during this incident?

of transport. To choose the most convenient option, travellers need certain information such as the to-be-expected delays. In Figure 3.8, it is presented that only very few people ticked the answer option about not needing any information in incident situations (Berlin: 2.9%; Singapore: 0.5%). In Berlin, the to-be-expected duration of the incident and the expected delays due to the incident is the most important information for the survey participants with 55.0% or 44.5%, respectively. In Singapore, the preference of the to-be-expected delays is clear with 61.9%, followed by an alternative path within the mass PT network. Mass PT network includes all means of PT which transport most of the passenger demand (e.g. bus, MRT, metro, etc.), but do not include modes that transport individual persons and small groups, such as taxis, which are also part of PT. In Berlin, people were more interested in the cause of the incident (25.2%), than alternative paths (23.1%), followed by affected PT lines (17.5%), and suggestions for alternative means of transport (12.8%). Please note that the last-mentioned answer was not an option in the survey conducted in Singapore. There, the affected PT lines are almost as interesting to the participants (28.0%) as the duration of the incident (27.0%). The cause of the incident plays the smallest role in Singapore, with 22.8%. The rather small but existing differences between Berlin and Singapore in this question could be explained by the information the participants were used to. This would mean that in Berlin the cause of an incident was more often communicated than in Singapore, for example. In Singapore, also more people answered that they would like a path suggestion, almost 20.0% more than in Berlin. This might show that PT users in Berlin were more used to finding their

alternative path, which also fits with their desire for information about the expected delay and incident duration. Part of the 20% might also come through the additional answer option in Berlin about suggestions for alternative means of transport.



Figure 3.9: If the operator provides you with an alternative path within the PT network, would you take it?

Since path advice is a core aspect of the method presented in Chapter 4, participants were asked if they would follow a path suggested by the PT operator and what are the most important characteristics of an alternative path. As to be seen in Figure 3.9, the results of Singapore and Berlin are very comparable concerning alternative paths. A quarter in Berlin to almost a third in Singapore answered that they *would* follow a path suggested by the operator. However, in both cities, over 60%, in Berlin almost 70%, stated that they *may* follow a suggestion, depending on the suggested alternative path. Only 7.4% in Singapore and 6.5% in Berlin *would not* follow a path suggestion, either because they would find an alternative path themselves or leave the PT system to switch to another means of transport.

Also, when it comes to the most important characteristics of an alternative path, the answers of the participants of the two different cities are comparable (Figure B.10). In both cities, the fastest path is the preferred path with 36.6% in Berlin and 48.7% in Singapore, followed by the path with the most reliable travel time with 17.5% in both cities. The third most common answer in Berlin was about the path with the same destination stop with 17.2% (9.0% in Singapore). In Singapore, the third most common answer is the one with the highest chance of not being crowded and thus also the one with the highest chance of being seated with 10.6% (2.9% in Berlin). The path with the fewest transfers played only a role in Berlin with

15.2% (3.2% in Singapore). In both cities, amenities and shops basically played no role with 0.5% in Berlin and 0.0% in Singapore. The fact that less walking and fewer transfers play a minor role in Singapore can be explained by the distribution of age among the participants in Singapore, which is a bias towards young people who are generally less averse to walking than older generations (Figure B.2).



3.2.4 Perceived Emotions

Figure 3.10: How confident are you that an autonomous, i.e. driverless, vehicle (e.g. an autonomous bus) will handle a similar incident well? (0 Not at all confident, 5 Neutral, 10 Extremely confident)

This thesis was part of the research project "Towards the Ultimate Public Transport System" which investigated the aforementioned DART system, an autonomous PT system [RAU, TIAN et al., 2018; TUMCREATE, 2021]. Therefore, in this section, the presented questions were asked about two different scenarios. The first scenario was about the incident the participants experienced. For the second scenario, the participants were asked to imagine that a similar incident takes place in an autonomous PT system.

The first question in this section aimed at gaining a better understanding of the differences between the responses to the two scenarios above. Therefore, participants were asked how confident they were about an autonomous vehicle handling an incident well, which is similar to the one they experienced (Figure 3.10). Depending on the kind of incident a participant

experienced, the PT vehicle itself played a stronger or weaker role in the situation. Therefore, it can be assumed that according to the kind of incident, people also expected an autonomous vehicle to handle a certain situation well or not.

In addition, PT vehicles are not yet well established in today's traffic. At the time of the survey, only two experiments were conducted in the two cities. In Berlin, there was an autonomous shuttle operating on the campus of the Charité hospital and in Singapore there was an autonomous bus, serving a PT line on the campus of the Nanyang Technological University (NTU) [BVG AND CHARITÉ, 2017; VOLVO AB, 2019] (Section 2.1.3). Therefore, experiences with autonomous vehicles and especially with autonomous vehicles were rather limited, if existent at all. This explains the high share of neutral answers. In general, the answers are somewhat balanced with a tendency towards less confident, which is particularly strong in Berlin, due to more than 20% of the people who answered with "not at all confident". As mentioned before, PT operators helped to broadcast the survey. In Singapore, the survey was advertised on the website of the PT operator Tower Transit, as well as, in a Facebook group of PT users, a Telegram channel for surveys conducted in Singapore and on websites of TUMCREATE, and the Facebook profile of the German embassy in Singapore. In Berlin, the survey was solely published on BVG's website and smartphone app. Websites of PT operators are not only visited by PT users but most likely also by PT drivers, who naturally have a dismissive attitude towards autonomous PT systems, as they are reducing the number of available driver jobs. Therefore, there might be a bias in these answers for some of the participants in Berlin.

The next two figures support the statement in the last paragraph, of people not being too sure about autonomous vehicles yet. Figure 3.11 shows the participants' answers to the question about their perceived level of pleasantness when thinking about the risk that a similar incident might occur. In both cities, the most given answer is "5 Neutral" and the rest strongly leans towards different levels of unpleasant. However, whereas in Berlin 32.1% answered with "5 Neutral" and 18.8% with "0 Unpleasant", in Singapore it's 21.7% for "5 Neutral" and 10.1% for "0 Unpleasant". In Singapore, more people used the answer options between "10 Pleasant" and "5 Neutral" than the participants from Berlin.

In general, people from Singapore seem also a bit more optimistic towards their next incident experience in PT than people from Berlin, as the answer options between 6 and 9, which are the different levels of pleasant are higher than in Berlin. Nevertheless, "10 Pleasant" was given by 2.4% in Berlin and merely by 0.5% in Singapore. How come some people have a pleasant feeling when thinking about the risk of experiencing an incident in PT again, which is usually or even mostly an unpleasant event? Two possible explanations might be that the incident did not disrupt the participants' travel plans as expected or that the PT operator reacted very well to the incident and resolved it rather quickly.

Figure 3.12 shows the same question however, the participants were asked to imagine that they experience the incident while being in an autonomous PT vehicle: "Can you indicate how unpleasant or pleasant you feel now, at this moment, when you think about the risk of a similar incident occurring with an autonomous, i.e. driverless, vehicle?". Compared to the emotions towards the actually experienced incident, the participants reacted a lot more neut-



Figure 3.11: Can you indicate how unpleasant or pleasant you feel now, at this moment, when you think about the risk of a similar incident occurring again? (0 Unpleasant, 5 Neutral, 10 Pleasant)

rally, with 47.3% in Berlin and 33.3% in Singapore, when it came to imagining the incident in an autonomous PT system. This is understandable as it is a stated preference question and as mentioned before the experiences with autonomous PT vehicles were rare if existed at all. More interestingly, the number of people who felt unpleasant towards the risk of experiencing an incident in an autonomous PT vehicle dropped in both cities compared to the experienced incident. In Berlin, there is a more significant drop than in Singapore. In general, participants seem to expect autonomous vehicles to handle the same situation better than the experienced incident has been handled.


Figure 3.12: Can you indicate how unpleasant or pleasant you feel now, at this moment, when you think about the risk of a similar incident occurring with an autonomous, i.e. driverless, vehicle? (0 Unpleasant, 5 Neutral, 10 Pleasant)

In the previous section, we saw that 24.9% to 29.6% of the people *would* follow path suggestions by the PT operator if given during an incident and 63.0% to 68.5% *may* follow it (Figure 3.9). In Figures 3.13 and 3.14, it is shown how willing people are to follow path instructions during incidents on a scale between "0 Not at all willing" and "10 Extremely willing". Moreover, the same question is presented for the scenario in an autonomous PT system, to see if this influenced their willingness to follow such instructions.

Even though "5 Neutral" is the most given answer in Berlin, with 24.5%, the majority lies between 7 and 10 with 10 as the second most given answer with 18.7%. In Singapore, it looks very similar, with most answers lying between 7 and 10, however, here, the two answers given most often are 7 with 20.6% and 8 with 18.5%. "5 Neutral" is significantly less represented (16.9%) than in Berlin, nevertheless, so is 10 (10.6%). The answers from 0 to 4 are considerably less represented in both cities. This fits very well with the answers given in Figure 3.9 and also shows that most of the people who answered with "maybe" tend differently strongly towards "yes".

When comparing the answers for the experienced incident with the one given for the scenario with an autonomous PT vehicle, it becomes obvious that the number of positive answers (6 to 10) declines strongly towards neutral, which increases by 5.9% in Berlin and by 9.6% in Singapore. Especially in Berlin, the negative answers (0 to 4) show a clear increase with "0



Figure 3.13: During such incidents, how willing are you to follow path instructions from the transport operator? (0 Not at all willing, 5 Neutral, 10 Extremely willing)

Not at all willing" growing by 7.4%. In Singapore, next to 5, 4 shows the most significant increase with 8.5%.

Hence, the participants from Singapore were more optimistic towards autonomous PT vehicles than the participants from Berlin, when it came to following given path suggestions. This can be explained by the fact that also in the question if participants would follow path suggestions, more people from Singapore answered "yes" than people from Berlin (Figure 3.9). The answers given in Figures 3.9, 3.13, and 3.14 are of particular interest for the PCIM method presented here, as providing travellers affected by an incident with path advice is a crucial part of it, as explained in Chapter 4.



Figure 3.14: During such incidents, how willing are you to follow path instructions from the autonomous, i.e. driverless, vehicle? (0 Not at all willing, 5 Neutral, 10 Ex-

3.2.5 Correlations

tremely willing)

Figure 3.15 shows the correlation matrices of some of the questions which have been asked in the PT user survey. Such matrices can give an idea about relations between the answers given in the asked questions. In such colour-formatted correlation matrices, as they are used here, the colour indicates whether a correlation is positive (blue) or negative (red). If two answers – A and B – have a positive correlation (blue), it means that if answer A is ticked, it is likely that answer B is ticked as well. In case of a negative correlation (red), if answer A is ticked, it is likely that answer B is not ticked.

Another important aspect is the strength of the correlation between two answers, indicated by the intensity of the colour of a field. The more intense a field's colour, the stronger the correlation, as it is also indicated by the scales next to the matrices. The weaker a correlation between two answers, the more fades the colour. Hence, white fields indicate no correlation between the respective answers. Logically, each answer has a positive correlation of 1.0 with itself, which is why the fields in the diagonal of the matrix from the top left to the bottom right are filled with an intense blue colour [FRIENDLY, 2002].

In Figure 3.15, "Willingness" refers to the participants' willingness to follow path instructions in a today's PT system (Figure 3.13). The next six or respectively seven answers refer to the

trip stage of the participants, at the origin, at the first stop, at a transfer stop, at the last stop, on a PT vehicle, at the destination, or elsewhere (Figure 3.4). Please note that the last answer option was only given in the questionnaire for Berlin.

The following five answers refer to the channels, the participants usually use to find their path through the PT network, which are referred to as path search channel (PSC) (Figure B.6). The participants could choose between the options of using an app (of an operator or third party), the operator's website, network maps displayed at stations and on PT vehicles, asking PT staff, family and friends or other travellers, or another channel.

The next seven answers refer to the participants' preferred PI channel during incidents, which is named incident channel (IC) here (Figure 3.7). The given options were PI displays at stops or on vehicles, speakers, through PT staff, via an app (of an operator or third party), through an operator's website, via social media, or another channel.

The last column and row refer to the participants' age. The values of the willingness and age is normalised, meaning their range goes from 0 to 1 instead of their actual values with corresponding values in between to make them better comparable to the other answers which values are either 0 or 1, depending on whether a participant ticked (1) the respective answer or not (0) [ROTHE et al., 1996]. Please note that for Berlin the participants chose among the age groups shown in Figure B.2, whereas in Singapore, the participants put in their exact age in years. Through the normalisation those two are also comparable, while the steps from 0 to 1 are finer in Singapore. As all questions had more than one option, the correlation is mostly not very strong. The more options the less obvious can be the correlation. The fact that vice versa is mostly not given for question with multiple answer options might be an explanation for why the correlation gets less obvious. Nevertheless, the correlation matrices in Figure 3.15 still give an idea about the correlation between some of the answers.

Looking at the results from Berlin, among the PSC, there is an indication that participants. who find their path through the PT network via an app, also prefer to receive incident information via an app, the blue colour in the corresponding field is relatively intense. This is an understandable correlation, if someone is familiar with a certain information channel, chances are high that this person also turns to that channel when it comes to incidents. This might also be the explanation for the positive correlation between PSC- and IC-other in the matrix for Singapore. Furthermore, their is an evidently negative correlation between PSC-app and age in Berlin, as the corresponding field shows a relatively intense red colour. This means that the older participants are the more likely it is that they did not tick PSC- or IC-app, whereas the correlation is stronger for PSC, also due to the fewer answer options. A stronger affinity of younger generations towards technology such as smartphones and apps is a reasonable assumption here. In Singapore, a noticeable negative correlation regarding the age is to be seen in the relation to PSC-map, while there is a positive relation towards IC-other. As there is a bias towards people between 20 and 29 years in the data from Singapore (Section 3.2.1), this correlation needs to be interpreted with caution. It might mean that younger people rather use maps to find their paths and that elderly rather prefer other channels than the listed during incidents, for example asking another traveller.

Regarding the stage of the trip when encountering the incident, in Berlin, there is a weak

correlation (colour with low intensity) between people who were at a transfer stop, when they experienced the stated incident and have a preference towards PI displays when it comes to incident information. If a stop is equipped with such a display, it probably also provides waiting people with incident information. The people who ticked both answers regarding their location and IC probably received information through the displays and might have already built some trust in this channel. A comparable correlation seems to exist between participants who encountered the stated incident while they were on a PT vehicle and IC-speaker and IC-staff, whereas there is a negative correlation of comparable intensity towards IC-PI display and with a smaller intensity towards apps. A possible explanation is that in contrast to the people at stops, many passengers, who experienced an incident while being on a PT vehicle received incident information through speaker announcements and PT staff, most likely the driver, and therefore had built some trust in these channels. As only two options could be given for this question, fewer people ticked PI display. In Singapore, there is a relatively strong correlation between the trip stage "Vehicle" and the other stages which is probably due to the fact that many participants in Singapore have been on a vehicle (75.7%) while noticing the incident (Figure 3.4).

Concerning the willingness to follow path instructions, there are no strong correlations. One noticeable correlation in Berlin is that people who have been on a vehicle when encountering the incident are more likely to follow path instructions. This might be related to the two IC channels which show a positive correlation with the location PT vehicle, staff and speaker announcements. This probably indicates that passengers rather trust path suggestions when PI is provided by one of these channels. It is also perceivable that people who were at their destination are less willing. This is understandable, once a passenger reaches the destination, there is no need for any information regarding the trip, unless the next trip starts shortly after.

Other noticeable correlations are mainly among different answers of the same question. In Singapore, there is a detectable negative correlation between the participants' willingness to follow path instructions and the trip stage "transfer stop". This is somewhat unfavourable, because a transfer stop is a practical location for path suggestions due to the multiple options of PT lines. This is especially true for transfer hubs with a large variety of lines. Moreover, as in Berlin, there is a positive correlation with IC-PT staff which might be an indication of people trusting path instructions more when given by staff.

Among the other questions, correlations are even weaker and therefore are not presented here. This correlation analysis shows that there are no strong correlations found here, especially regarding the preferences towards specific IC and the willingness to follow path instructions, which are relevant for the model presented in Chapter 4. Furthermore, the correlations to be found in the results from Berlin differ from the ones from Singapore. However, in both cities some relations between the age and certain PI channels as well as trip stages and PI channels are pointed out here. These indicate comparable patterns, even though Berlin and Singapore are two very different cities in terms of geographic location, climate, culture, and population density. The correlation matrices also show that the group of people who is willing to be redirected is very heterogeneous, which indicates that path advice should be receivable by all groups of society via all available PI channels as it is also suggested by WILKE [2023]. As already pointed out in Section 2.1.4, there are many factors influencing a passenger's reaction to an incident and to follow a given suggestion for alternative connections.



Figure 3.15: Correlation matrices

Chapter 4

Model Formulation

REdirecting Passengers and reAllocating Capacities Knowingly (REPACK) is the name of the algorithmic PCIM method which solves the problems and achieves the goals mentioned in the introduction (Chapter 1). Two earlier versions of the PCIM method presented here have been published by BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023]. Therefore, this chapter contains overlapping elements with the mentioned publications since the general idea and the conceptual framework have not changed significantly in the course of its development. Nonetheless, compared to the two previous versions, many aspects have been refined and completely new elements have been integrated. Corresponding comments can be found in the following sections.

4.1 Real Life Story

This section explains the general idea behind REPACK, which is to redirect passengers by giving them concrete path advice and simultaneously reallocating capacities during incidents, as its name already indicates. Figure 4.1 provides a simple toy network (STN) to explain the method. It has been adopted from UL ABEDIN [2019], modified, and already used by BACHMANN, RAU et al. [2021]. It includes seven nodes with stops and four PT lines. The numbers next to the links are the link travel times in minutes and the colour of a link refers to the line which operates on it. In this PT system, all lines are served by buses. The black curved link represents an adjacent parallel street which is not used during normal operations, meaning when no incident occurred.

Imagine you are travelling through the STN. Your origin is Node 3 and your destination is Node 6. You choose to take the fastest path, which means taking the red line to your destination node. While you are still at Node 3, an incident suddenly occurs between Nodes 4 and 5. This completely blocks the PT services between those two nodes for a certain duration. The dispatchers in the OCC react to the incident by rerouting the red and yellow lines along adjacent parallel streets, which is represented by the black curved link in Figure 4.1. This causes an increase in travel time of 14 minutes. Through dynamic PI displays at your stop and the PT app on your smartphone, you receive the information that your original trip is disrupted by this incident for 30 minutes and the corresponding delay is provided. If you know the PT network well, you can quickly conclude that your fastest alternative path is to take the cyan line until Node 4 and then transfer to the rerouted red or yellow line to finally reach your destination without much delay. However, this is probably what most travellers who are in the



Figure 4.1: Simple Toy Network (STN) [BACHMANN, RAU et al., 2021]

same situation as you would conclude. As pointed out by CATS and JENELIUS [2014] and in Section 2.2.2, PI about incidents can also lead to secondary incidents due to overcrowding effects, leading to additional delays. This is especially true if the disrupted PT mode has a higher capacity than the modes of the fastest alternative path. Therefore, one central point of REPACK is to provide incident-affected travellers with concrete path advice to avoid too many passengers transferring to the same alternative path, hence to avoid overcrowding on alternative paths.

Assume that the cyan line would be overloaded by the additional demand if all affected passengers who come from Node 1 or 3 and want to go to Node 4 or 6 would choose the cyan line during the incident. In this case, it would be useful to split them in accordance with the available remaining capacities of the cyan and violet lines. However, if the path advice comes from collective information channels such as dynamic PI displays or speaker announcements, it is impossible to give path advice to individuals. It is only possible to give the same path advice to all passengers travelling in the same direction. Furthermore, imagine that you are not travelling alone, but with your favourite travel companion. If the two of you would receive different path advice, for example through a PT app on your smartphones, it could lead to confusion and dissatisfaction among the informed passengers like you and your travel companion. Hence, it is reasonable to group the affected passengers in correspondence

to the node at which they are receiving path advice and their destination. Let us assume that the passenger group coming from Node 1 is the group which is supposed to stay on the cyan line until Node 4 as they are already on this line. Consequently, you, standing with your travel companion at Node 3, receive the information to take the violet line and then change at Node 4 onto the red or yellow line to reach your destination at Node 6. At this point you probably ask yourself, why not take the faster cyan line? Therefore, it might be helpful for you and the other PT users affected by the incident to be informed about the reason for this redirection, which in this case, is the capacity shortages on the cyan line. The better the affected passengers understand the situation, the more willing they will be to follow given instructions (Section 2.1.5). As the results from the PT user survey presented in Chapter 3 show, the cause of the incident is relevant to the affected passengers. Furthermore, the survey revealed that collective PI channels such as speaker announcements and dynamic PI displays play the most important roles during incidents from the passengers' point of view (Figure 3.7).

Imagine you and your travel companion follow the suggestion and take the violet line. In this case, you reach your destination later than planned, however, you reach it without any additional delays caused by waiting for the incident to be resolved or denial of boarding due to overcrowding. If not enough travellers were to follow the suggestion, or even if you were not to follow it but instead were to take the cyan line, you might experience further delays because of the exceeding of the cyan line's remaining capacity. Exceeding capacities leads to delayed departure due to prolonged boarding and alighting times as well as boarding denial and additional waiting time for subsequent PT vehicles. If you take the advised path but too many of the other affected passengers do not follow the path advice, they might cause delays on the red line and you could face additional delays at Node 4. However, due to the additional availability of the yellow line, the delays are probably much less than the delays the passengers on the cyan line have to face. In any case, a sufficient number of affected passengers must follow the path advice, to avoid secondary disturbances. Furthermore, imagine the passenger redirection is additionally supported by buses which have been withdrawn from low demand PT lines and reallocated to the cyan line to avoid the exceedance of its capacity. In this way, it might even be possible to redirect all affected passengers via the cyan line, including you and your travel companion, which would further reduce passenger delays.

This example gives reason to assume that path advice can reduce passenger delays and that reallocation of capacities decreases them even further. This is shown by BACHMANN, RAU et al. [2021] in a numerical example. REPACK redirects affected passengers during incidents on alternative paths with sufficient remaining capacity and reallocates capacities to support the passenger redirection while avoiding severe delays on the PT lines from which the capacities are withdrawn.

4.2 Problem Statement

In the previous section, a picture was painted showing how a PCIM method, and specifically REPACK, could be perceived by passengers in practice. This section describes the theory behind this picture, stating the theoretical problem which is to be solved by REPACK. The problem to be solved is that there is an incident that causes passenger delays. It is therefore

this method's main goal to reduce the delays of affected passengers as much as possible. One important assumption is that the incident's location, the disrupted PT services, as well as an estimation of the incident's duration can be provided by the OCC staff. Through interviews of OCC directors and trainers and observations of dispatchers in OCCs during their work, it has been concluded that dispatchers have quite a good intuition about the duration of an incident [BRIEM, BUCK, MAGDOLEN et al., 2020; BACHMANN, BRIEM et al., 2022]. Furthermore, in Section 2.2.1 it has been elaborated upon how dispatchers take disposition measures to resolve the incident and reduce its negative effects on the PT system. The timetable which depicts the changes in the service during the incident and thus also the corresponding disturbances and the disposition measures taken (e.g. rerouting or splitting PT lines) is called a disposition timetable. When an incident occurs in a PT network, it usually does not affect the whole network, but rather specific PT lines and thus not all PT users $u \in U$ in the network, with u being one user and U being the set of all PT users. Hence, the following kinds of passengers can be distinguished:

- Affected PT users $U_a \subset U$, are the passengers whose trips are directly affected by the incident since they planned to take one of the disrupted PT lines. The focus of the presented methodology lies on them, U_a being the set of all affected users.
- Indirectly affected PT users U_{ida} ⊂ U. These travellers take PT lines which are not disrupted by the incident, but which are however under increased pressure due to the additional demand of the u ∈ U_a transferring on to these PT lines. Additional demand can lead to secondary incidents due to overcrowding effects (Section 2.2.2), U_{ida} being the set of all indirectly affected users.
- Unaffected PT users $U_{una} \subset U$ are the passengers who are not at all affected by the incident because their trip takes place long before or after the incident when there are no consequences of the incident, or because they take PT lines which are neither disrupted nor occupied by transferred affected passengers, U_{una} being the set of all unaffected users.

Even though the PT network is disturbed by an incident which causes delays for the PT users, it is assumed that no one leaves the PT system. Another goal is therefore to provide the passengers with path advice in such a way that everybody is accommodated in the PT system and an adjusted equilibrium state is reached. Furthermore, it is assumed that an OD matrix, as an estimation of the passenger demand, is available. As elaborated by GKIOT-SALITIS and CATS [2021], there are various sources of information about the demand which can be used to derive a corresponding OD matrix either based on historical data or even in real-time, based on how well a PT system is equipped in this matter. Typical sources for demand data are automatic passenger counts and fare collection, ticket sales, and sample census data (Section 2.1.1). Although it is assumed that no traveller leaves the PT system, it would be unreasonable to assume full compliance. As one can see in Figure 3.9 (Chapter 3), some passengers would rather find a path on their own or keep their original trip plans. To consider these passengers within REPACK, two factors are taken into account. First, a self-redirection rate which expresses the share of passengers who prefer to redirect themselves is considered.

Second, a compliance rate is taken into account, as mentioned in the previous section. The self-redirection rate is one of the various aspects which is not part of the earlier versions of the method described here, presented by BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023]. The compliance rate refers to the share of passengers who do follow the given path advice. The rest, the passengers who do not redirect themselves or follow the given path advice, are assumed to stay on their original path.

It goes without saying that for passengers to be able to follow given path advice, they need to receive it. For this purpose, several PI channels are available in PT systems, as described in Section 2.1.5. However, depending on the PT system, only some of these channels might be available to the travellers. Furthermore, in Figure 3.7 (Chapter 3), the survey participants stated different preferences regarding the PI channel through which they would want to receive information about the incident and its consequences. Hence, an information rate is used to consider passengers who do not receive path advice and to be able to take different cases of availability of PI channels into account.

Another restriction to the solution to this problem is that it is reasonable to not redirect passengers individually but to group them. First, it is not known who is travelling with whom, and giving a group of people who are travelling together different path advice can lead to confusion. Second, in general, if people get redirected at the same stop and have a common destination, especially on the collective PI channels (e.g. speakers, displays) the information should be consistent, otherwise, it might lead to confusion and distrust in such a PCIM method (Section 2.1.5). As stated in the literature review in Section 2.2.2, passengers are usually handled as indivisible groups in PCIM methods.

Here, the affected passengers are grouped into OD pairs $od \in OD_a$, in accordance with their origin of redirection o and destination d, od being one OD pair, ODa being the set of all affected OD pairs, subset of the set of all OD pairs OD ($OD_a \subset OD$). An origin of redirection is the stop from which onward the passengers need an alternative path through the PT network. For travellers who did not start their trip before the beginning of the incident, this stop is the same as the starting stop of their original trip. For passengers who are already on their way, it is the next stop of their current trip.

When passengers are redirected, it should also be ensured that secondary incidents through overcrowding are avoided. Therefore, the remaining capacity on the alternative paths needs to be taken into account. Consequently, another assumption is that corresponding information about the PT system's capacity and occupancy is accessible.

The solution to the stated problem should thereby be that path advice given by the operator should be disseminated to all affected passengers in a manner such that: 1) travellers with the same origin of redirection and destination receive the same path advice, 2) the path advice is to be broadcast consistently across all PI channels, and 3) the given path advice considers the remaining capacity of the alternative paths. If an approach solves this problem accordingly, it fulfils the requirements for CAPR as stated by BACHMANN, TSAKARESTOS et al. [2023].

The bullet list below summarises the mentioned assumptions:

- dispatchers in OCCs can estimate the incident duration,
- nobody leaves the PT system,
- the passenger demand on the PT system is given as an OD matrix,
- the capacity and occupancy of the PT vehicles is known.

To reach the main goal of minimising the overall delay of $U_a \subset U$, which is expressed mathematically in Equation 4.1, REPACK has been developed. The delay is hereby defined by the difference between the originally planned arrival time of a passenger to their actual arrival time at the respective destination.

$$\min\sum_{u\in U_a}\Delta tt_u\tag{4.1}$$

Where:

 $u \in U_a = \mathsf{PT}$ user in set of all affected PT users, $\Delta t t_u = \mathsf{delay} \text{ of } \mathsf{PT} \text{ user } u \in U_a.$

The procedure of the PCIM method REPACK consists of four algorithmic processes that are presented in Figure 4.2. In the following flow charts, in this chapter, eclipses refer to the terminals (i.e. its start and end), rectangles represent processing steps, diamonds indicate decisions at what processes split, and parallelograms refer to information or data.



Figure 4.2: Procedure of REPACK

The first process – situation analysis presented in Section 4.4 – retrieves the needed information about the incident, the PT supply and the passengers' trips. The second process – passenger redirection presented in Section 4.5 – exploits the gathered information to redirect the passengers affected by the incident in the PT network. Note that during the second process, the remaining capacity on the alternative paths and thus all $u \in U_{ida}$ who occupy part of the capacity on the alternative paths are considered. Additionally, the third process – capacity reallocation presented in Section 4.6 – checks the PT system for dispensable capacities which could be reallocated to support the passenger redirection. Once a redirection strategy is set, the last process – information dissemination presented in Section 4.7 – then disseminates corresponding path advice across all available PI channels and informs the PT drivers about the reallocation of their vehicles. Before diving into the first process of the REPACK procedure, the next section describes how the PT network, including the timetable and especially the aforementioned disposition timetable reflecting the PT service during the incident, is represented during the procedure.

4.3 Network Graph

To search the PT system for available paths as well as check their travel times and remaining capacity, the PT supply needs to be represented properly. As in BACHMANN, DANDL et al. [2023], this is done with a directed network graph G = (N, L) with nodes N and links L. Figure 4.3 shows a part of the PT network presented in Figure 5.2 in Section 5.2.2 as represented in the graph introduced here. Stop areas (e.g. s1) consist of two types of nodes: 1) stop nodes $N_P \subset N$ which reflect the physical PT stops and 2) line nodes $N_L \subset N$ which represent the PT lines serving a particular stop node. A line node is connected to its corresponding stop node via a link (dashed lines), which is for transferring between lines as well as getting off and on the PT vehicles. Each stop area includes at least two stop nodes, reflecting the PT stops for operations in opposite directions, such as two stops across a street. Walking links (dashed arrows) connect the different stop nodes (e.g. s1_1, s1_2) of a stop area. Larger transport hubs with several stops, such as stop area s2 in Figure 4.3 can have even more than two stop nodes. If a stop node is served by more than one PT line, it is connected to a corresponding number of line nodes. These represent the PT lines which service the corresponding stop node. The colour of a line node indicates the colour of the PT line in Figure 5.2 (Chapter 5). The PT links (arrows) connect the line node of a particular PT line with the line node of the same PT line of the next stop, reflecting the line route. The PT vehicles serving a line node are stored as attributes at each node with there departure time and remaining capacity in the graph.

When an incident occurs, it is not the regular timetable which needs to be represented by G, but the disposition timetable, which takes the incident itself into account as well as the disposition measures taken by the OCC (see Section 2.2.1). It thereby represents the PT supply in line with the condition of the PT system during the incident. The incident can then be expressed by its start (t_i^s) and end time (t_i^e) as well as by the links it disrupts $L_i \subset L$. From these links, the disrupted PT lines and affected PT vehicles can be derived as well. Since the capacity of the system is a crucial aspect in PCIM, it must also be represented in G in order to be able to check the available paths found for their remaining capacity. As mentioned in the previous section, it is assumed that the capacity and occupancy of all PT vehicles are known. Therefore, the remaining link capacity c_l of $l \in L$ can be calculated by summing up the remaining free places on all PT vehicles crossing l. A path $p \in P$ is defined as a specific order of $l \in L$, P being the set of all paths. The remaining path capacity c_p is thereby defined by the smallest c_l among all links which are part of p ($L_p \subset L$).



Figure 4.3: Example for the used representation of the PT supply [BACHMANN, DANDL et al., 2023]

4.4 Situation Analysis

This process is the first in the procedure of REPACK and is shown in Figure 4.4. After the incident occurred, Step 1 is to gather the incident data. Important incident data is its location and with that the disrupted PT lines as well as an estimation of its duration. Even though each incident has its own characteristics and its duration is hard to predict, through interviews with OCC directors and trainers, as well as through observations of dispatchers during their work, it is known that especially experienced dispatchers have quite a good intuition about the duration of an incident [BRIEM, BUCK, MAGDOLEN et al., 2020; BACHMANN, BRIEM et al., 2022] (Section 2.2.1).

Next (Step 2), the travel data is retrieved which includes the supply side, meaning the planned PT schedule as well as the passenger side i.e., information about the passengers' trips. If this method is implemented in practice, it is conceivable that the timetable can be retrieved via an interface such as GTFS or together with the incident information from an ITCS via an interface such as the VDV454 (Sections 2.1.5 and 2.2.2). Depending on how well a PT system is equipped, the passenger trip information can be catered for through the PT operators' model of the PT network and corresponding estimates of the passenger demand on the day the incident occurred. For example, it is imaginable that day categories (e.g. weekdays, holidays, special event days) are formed which show typical passenger demands. In accordance with the day of the incident a category is picked to predict the passenger demand.



Figure 4.4: Situation analysis process

In PT systems with APC or AFC, it might also be possible that real-time demand data is available (Section 2.1.1).

Based on the timetable and the incident information, it can then be determined in Step 3 which PT vehicles would be affected by the incident. If this information is merged with the passenger trip data, the affected passengers and their OD relations can be identified (Step 4). In Step 5 the origins of affected passengers' redirections are specified. An origin of redirection describes the stop from which passengers need to be redirected. For affected travellers whose trip has not started at the time the incident occurs (t_i^s) , the origin stop of their trip is the same as the origin stop of their redirection, whereas for affected travellers who already started their trip it is the next stop they are passing or the stop at which they are transferring at the time the incident occurs. Through this step and based on the previously received passengers rip data, it is known from which stop to which stop in the network the affected passengers need alternatives to their original path.

As pointed out in Section 4.1, it is reasonable to group the affected travellers according to the origin of their redirection and their destination (Step 6). OD is the set of all OD pairs in the network. $OD_a \subset OD$ is the set of all od which are affected by the incident. This grouping method avoids passengers travelling together receiving different path advice as suggested by MÜLLER-HANNEMANN et al. [2019]. Moreover, it helps to prevent confusion among the affected travellers, especially when the path advice is disseminated via collective PI channels such as speaker announcements and dynamic PI displays at stops and in PT vehicles, which are the preferred channels of travellers during incidents as shown in Figure 3.7. Before coming to the redirection process, the actual PT service being operated during the incident needs to be examined. For this purpose, the disposition timetable, which includes the disposition measures (Section 2.2.1) taken by the dispatchers in the OCC, is collected (Step 7). Additionally, the corresponding available paths in this altered PT network are fetched.

The collection of available paths in accordance with the incident and the disposition measures taken is a process that could be preprocessed in practice. It is conceivable that a corresponding database could be built in which the available paths are saved and associated with different possible incident situations, which would speed up Step 7. A filtering process could then further adjust such a comprehensive list of available paths to the detailed characteristics of the present situation. This filter can be part of Step 7 or 8, in which each $od \in OD_a$ is associated with a set of potential alternative paths for $od (P_{od} \subset P)$ which connect the respective od's origin of redirection with its destination.

Here, a set of all available simple paths is found using the method by SRYHENI [2022]. The characteristic "simple" refers to the absence of loops in a path. The algorithm validates every opportunity to go from one vector or node to another within the corresponding graph (G = (N, L)), representing the disposition timetable as described in Section 4.3. By saving the already visited nodes while going through the network, it is ensured that each node which is part of a path is only visited once. However, this method might be too inefficient for larger networks with a high number of nodes, as the number of possible connections grows by a factor of |n|! with |n| being the number of nodes. In such cases, a k-shortest path algorithm as described in Section 2.1.2 is probably more appropriate.

This set of paths is sorted by the estimated travel time of paths, from the fastest to the slowest alternative path. The actual instead of the perceived travel time (Section 2.1.2) is used here, as it is assumed that in special situations such as incidents, people are more willing to take a more inconvenient path, if necessary, in the form of crowded PT services or a longer perceived travel time, in order to reach their respective destination with a minimum of delay [CARREL, HALVORSEN et al., 2013]. The fastest alternative path of a set of paths is depicted as $\hat{p} \in P_{od}$ in the following. It is also the path which is assumed to be taken by passengers who rather redirect themselves in an incident situation than follow the operator's path advice (Step 9). As known from the survey results in Figure 3.9 (Section 3.2.3), some passengers prefer to redirect themselves, meaning choosing an alternative path on their own. Commuters, for example, who know the PT network well, probably prefer to take the fastest alternative path. Therefore, the methodology presented here considers a self-redirected share of affected passengers for each $od \in OD_a$. The self-redirection rate ξ is multiplied by the number of passengers associated with an od. Afterwards, a random sample of the corresponding size is taken to build the respective group of self-redirected passengers, which is referred to as

 sg_{od} . This is done for each $od \in OD_a$ before all sg_{od} are then assigned to their corresponding $\hat{p} \in P_{od}$ in Step 10. These two steps differ significantly from the earlier versions of this method which do not consider the passengers who redirect themselves [BACHMANN, RAU et al., 2021; BACHMANN, DANDL et al., 2023].

Ultimately, this process analysed a present incident situation. That is, all $p \in P_{od}$ of each $od \in OD_a$ are identified and the sg_{od} are assigned, so, the redirection process can begin.

4.5 Passenger Redirection

In the following, two approaches are presented to solve the passenger redirection problem. Therefore, at the 11th step in Figure 4.5, the algorithm splits into two approaches. The blue area contains the rule-based heuristic approach and the red area contains the optimisation approach.

4.5.1 Heuristic Approach

To reduce passenger delay, it makes sense to redirect passengers only as long as it is faster than to wait for the incident to be resolved and stay on the original path $0 \in P_{od}$. For this purpose, Equation 4.2 is formulated to calculate the redirection duration T_{ad}^{p} , which expresses the period during which path advice is disseminated to a certain $od \in OD_a$ on a certain path $p \in P_{od}$. It has been adopted from BACHMANN, DANDL et al. [2023] and slightly modified. A time constant t_{con} which is used by BACHMANN, DANDL et al. [2023] to consider delays due to missed PT vehicles and interference with other passengers has been replaced by the product of the longest headway h_p among the PT lines which are part of path p and the factor ψ , by which \check{h}_p is multiplied. Through \check{h}_p , differences in the headway among the various PT lines which are part of path p are taken into account. Therefore, product of \check{h}_p and factor ψ can be used to adjust Equation 4.2 to a certain PT system to achieve better results. Besides the consideration of missed PT vehicles and interference with other passengers, this product can also used to prolong the redirection duration if a negative value is set for ψ . If for example an incident is severe and caters for a recovery time, describing the period between the dissolution of the incident and the end of delays induced by the incident (Section 2.2.2), it might be reasonable to extent the redirection duration. This is further discussed in Section 5.2.1. Once T_{od}^{p} elapses, the point of time is reached at which it makes more sense for the passengers of od to wait for the remaining time until the dissolution of the incident and stick with $0 \in P_{od}$.

$$T_{od}^{p} = (t_{i}^{e} - t_{i}^{s}) + (tt_{0}^{od} - tt_{p}) - \breve{h}_{p} \cdot \psi$$
(4.2)

Where:

 tt_p = travel time needed for path p,

- $tt_0^{od} =$ travel time needed for the original path 0 of an od,
- t^e_i = end time of the incident,
- t_i^s = start time of the incident,
- \check{h}_p = longest headway among the PT lines which serve a section of p,
- ψ = calibration parameter to adjust T_{od}^p .

Based on T_{od}^p , the redirection demand of od on a particular path p can be examined. It expresses how many people are redirected if OD pair od is assigned to path p. After the passengers of sg_{od} are deducted from all passengers associated with an $od \in OD_a$, the rest is divided into a redirection group rg_{od}^p and a waiting group wg_{od}^p . The redirection group rg_{od}^p are actually redirected whereas the waiting group wg_{od}^p refers to the passengers who are waiting for the dissolution of the incident and staying on $0 \in P_{od}$. The decision of who belongs to which group is done based on T_{od}^p .

For each passenger of an od who is not in sg_{od} , it is therefore checked if the respective arrival time at the origin of redirection (t_{arr}^o) lies before the end of the corresponding T_{od}^p , measured from the start time of the incident (t_i^s) , as shown in Equation 4.3. If so, the redirection is reasonable for them and they are assigned to rg_{od}^p ; if not, they are assigned to wg_{od}^p . Since the redirection of passengers is only reasonable as long as it reduces their delay: the smaller tt_p , the longer T_{od}^p and therefore the more passengers in group rg_{od}^p .

$$t_{arr}^o < t_i^s + T_{od}^p \tag{4.3}$$

Where:

 $t^o_{arr} = arrival$ time at the origin of redirection,

 t^s_i = start time of the incident,

 T_{od}^p = redirection duration of passengers associated with od for path p.

However, passengers who receive path advice do not necessarily comply with it. As shown in Figure 3.9, a few people would redirect themselves, some people *would* follow the path advice, and most just *might* follow path advice. The people who redirect themselves are already taken care of through group sg_{od} , however, some people might also not follow path advice because they would prefer to stick with $0 \in P_{od}$. To consider these travellers, a compliance rate η is taken into account. This rate corresponds to the size of the share of affected and compliant passengers. In addition, the information rate ζ , which expresses the share of affected passengers who actually receive the path advice, might also reduce the number of people who would follow it. Depending on which information channels are available to the passengers, the information rate can be less than 100%. Furthermore, in accordance with Figure 3.7, the most important information channels are the collective ones (i.e. speakers, dynamic PI displays at stops). Since the space at displays is limited and very long announcements are confusing, it is not possible to inform all rg_{od}^p via the collective channels. Therefore, for these channels, the information rate is smaller. The rest $(1 - \eta \cdot \zeta)$ is additionally associated with wg_{od}^p .

The heuristic approach (blue area in Figure 4.5) builds on the idea that the larger rg_{od}^p , the larger the reduction of passenger delays once a fitting path $p \in P_{od}$ is found. Therefore, the heuristic approach starts with the $od \in OD_a$ with the largest rg_{od}^p for its fastest path $(\hat{p} \in P_{od})$. The smaller tt_p is, the longer is T_{od}^p and the larger is rg_{od}^p . To bring $OD_a \subset OD$ in corresponding order, in Step H12 of the heuristic approach, T_{od}^p is calculated for $\hat{p} \in P_{od}$ for each $od \in OD_a$, which enables the examination of the sizes of $rg_{od}^{\hat{p}}$ and $wg_{od}^{\hat{p}}$ for each $od \in OD_a$ on their respective $\hat{p} \in P_{od}$ (Step H13). In Step H14 of the heuristic approach, $OD_a \subset OD$ can then be sorted accordingly by the size of $rg_{od}^{\hat{p}}$.

Then, for each OD pair $od \in OD_a$ taken in Step H15, the corresponding set of available paths $(P_{od} \subset P)$ is checked in Step H16. P_{od} is sorted by the paths' travel time $(tt_p, \text{Step 8}$ in Figure 4.4), with the fastest path $\hat{p} \in P_{od}$ on top. As long as no fitting path $p \in P_{od}$ is found, the next p is inspected. In Step H17, T_{od}^p of the currently examined p is calculated to then examine in Step H18 the corresponding size of rg_{od}^p and wg_{od}^p as per the set parameters (i.e. η , ζ , ξ , ψ). Through the size of rg_{od}^p , it is known how much room is needed if OD pair od is assigned to path p. To avoid secondary incidents on path p due to overcrowding, as mentioned before [CATS and JENELIUS, 2014], the remaining capacity of path p (c_p) is examined in Step H19.

In the heuristic approach, this is done segment-wise. A path p is divided into path segments $(ps \in PS_p)$ in accordance with the PT lines which are serving on path p. Due to the segment-wise calculation of c_p , the travel time of rg_{od}^p along path p can be considered. When passengers travel along a path, they occupy space on the corresponding PT lines for the corresponding path segment $ps \in PS_p$. For each path segment they need a certain travel time. Correspondingly, on the subsequent path segments they occupy space later. Therefore, for a redirection group rg_{od}^p , it is reasonable to check the remaining capacity of a segment of path p in the period where this segment would actually be occupied by rg_{od}^p . At this point, it can be checked whether the current path p possesses sufficient remaining capacity for the size of rg_{od}^p and if the additional passenger demand can consequently be accommodated (Step H20). If it does, in Step H22a, the passengers in rg_{od}^p are redirected onto path p, whereas the passengers in wg_{od}^p are informed to stay on $0 \in P_{od}$. The path to which rg_{od}^p is redirected is referred to as the assigned path $\tilde{p} \in P_{od}$. To consider these groups in the next steps, the remaining capacities of the paths are updated accordingly in the PT network graph (Figure 4.3).

If Step H20 is answered with "NO", it is checked if there are still paths $p \in P_{od}$ left in Step H21. If there are, the process returns to Step H16 and picks the next path $p \in P_{od}$ and repeats Steps H17 to H20. If not, rg_{od}^p and wg_{od}^p are assigned to $0 \in P_{od}$ in Step H22b and it is checked in Step H23 if all $od \in OD_a$ have been processed. If not, the process goes back to Step H15, takes the next $od \in OD_a$ and repeats Steps H16 to H23. This step-by-step assignment of passenger groups is comparable to the greedy approach presented by MÜLLER-HANNEMANN et al. [2019], however, it works with a random order of passenger groups. If the question in Step H23 can be answered with "YES", the heuristic approach of the passenger redirection process reaches its end and leads to the next process of REPACK, which is the capacity reallocation process. However, before explaining this process, the optimisation approach of the passenger redirection is described in the following section.



Figure 4.5: Passenger redirection process

4.5.2 Optimisation Approach

In addition to the previously explained heuristic approach, the passenger redirection is also formulated as an optimisation problem. Instead of handling $OD_a \subset OD$ in a specific order (largest to smallest $rg_{od}^{\hat{p}}$) as the heuristic approach does, the goal of the optimisation approach is to find the ideal match between the available paths $p \in P_{od}$ and the $od \in OD_a$ to be redirected to minimise the overall delay of all $u \in U_a$ (Equation 4.1). To do so, in Step 12 of the optimisation approach, T_{od}^p (Equation 4.2) is calculated for each OD pair $od \in OD_a$ and each path $p \in P_{od}$. Based on this, the sizes of rg_{od}^p and wg_{od}^p can be examined for each OD pair $od \in OD_a$ and each path $p \in P_{od}$ in Step O13, as per the set parameters (i.e. η , ζ , ξ , ψ), by checking whether passengers arrive at their respective origin of redirection, before T_{od}^p is elapsed (Equation 4.3). Note that for this step, the self-redirected group sg_{od} is subtracted from all PT users associated with $od \in OD_a$ ($u \in U_{od}$), before the remaining passengers of $U_{od} \subset U$ are added to the redirection or waiting group (rg_{od}^p and wg_{od}^p).

At this point of the optimisation approach, the number of passengers in rg_{od}^p and wg_{od}^p is known for each OD pair $od \in OD_a$ and each path $p \in P_{od}$. Thereby, the corresponding total travel time TT_{od}^p for each OD pair $od \in OD_a$ on each path $p \in P_{od}$ can be computed in Step O14 (Equation 4.4). Note that this is also done for the respective original paths $0 \in P_{od}$ to include the cases in which no alternative path could be found (Step O15).

$$TT_{od}^{p} = |rg_{od}^{p}| \cdot tt_{p} + \left(|wg_{od}^{p}| \cdot tt_{0}^{od} + \sum_{u \in wg_{od}^{p}} t_{u}^{i}\right)$$
(4.4)

Where:

 TT_{od}^p = total travel time of OD pair *od*, if assigned to path *p*,

- tt_p = travel time needed for path p,
- tt_0^{od} = travel time needed for the original path 0 of OD pair od,
- t_u^i = waiting time of PT user $u \in U_{od}$ until the end of the incident (t_i^e) .

The total travel time of OD pair od on path $p(TT_{od}^p)$ expresses the incurred loss of time of an OD pair $od \in OD_a$ if it is assigned to a specific path $p \in P_{od}$. Thus, the overall incurred loss of time can be calculated for a specific solution to the passenger redirection problem which is used by a optimisation solver to find the ideal match between available paths and all rg_{od}^p .

As already emphasised in the explanation of the heuristic approach, the consideration of the remaining capacity on the alternative paths is a crucial point in the PCIM method presented to avoid secondary incidents through overcrowding effects. Therefore, in Step O16, the remaining capacity c_l of each link $l \in L$ is determined by checking the remaining capacity of each PT vehicle crossing a link l in a specific period. Here, the period of the incident from t_i^s to t_i^e is used. Moreover, staying within the available remaining capacity is set as an optimisation constraint in Step O17 (Equation 4.7).

The path-link matrix is represented by $M \in 0, 1^{P_{od}xL}$, in which the element m_{pl} is 1 for every link $l \in L$ that is part of path p ($l \in L_p$). Like the heuristic approach, the optimisation approach thereby also considers all indirectly affected PT users $U_{ida} \subset U$ who are assigned to their respective path and thereby occupying parts of the PT network's capacity. The constraint (Equation 4.6) ensures that each redirection group rg_{od}^p is solely assigned to one specific path $p \in P_{od}$ through x_{od}^p (Step O18). Equation 4.5 shows the mathematical formulation of the optimisation problem to which the objective of minimising the overall delay of all affected passengers is set in Step O19.

$$\min_{x_{od}^{p}} \sum_{od \in OD_{a}} \sum_{p \in P_{od}} x_{od}^{p} \cdot \Delta t t_{od}^{p}$$
(4.5)

subject to:

$$\sum_{od \in OD_a} \sum_{p \in P_{od}} x_{od}^p \cdot |rg_{od}^p| \cdot m_{pl} \le c_l \qquad \forall \quad l \in L$$
(4.6)

$$\sum_{p \in P_{od}} x_{od}^p = 1 \qquad \forall \quad od \in OD_a$$
(4.7)

Where:

$$\begin{array}{ll} m_{pl} &= \text{binary element of } M \in 0, 1^{P_{od}xL} \text{ which is 1 if link } l \in L_p, \\ M \in 0, 1^{P_{od}xL} &= \text{path-link matrix,} \\ c_l &= \text{remaining link capacity of } l \in L, \\ x_{od}^p &= \text{binary decision variable which is 1 if } od \text{ is assigned to } p. \end{array}$$

In Step O20, the optimisation function (Equation 4.5) as well as the prepared inputs are then entered in the Gurobi Optimizer. This is a solver of mathematical optimisation problems such as the one presented here. For further information about Gurobi, it is referred to GUROBI OPTIMIZATION LLC [2022]. This optimiser solves the optimisation problem formulated in Equation 4.5 and hence, finds the ideal match between each OD pair $od \in OD_a$ and a specific path $p \in P_{od}$ so its objective of minimising the overall travel time of $OD_a \subset OD$ is fulfilled (Step O21). In Step O22, this association of each OD pair $od \in OD_a$ with a path $p \in P_{od}$ is retrieved and can afterwards be used in Step O23 to assign all rg_{od}^p and wg_{od}^p accordingly. At this step, the optimisation approach is at its end and also leads to the capacity reallocation process, as the heuristic approach does.

4.6 Capacity Reallocation

As pointed out in Section 2.2.1, it is becoming increasingly difficult for operators to find enough drivers operating the vehicles to conduct the PT service as planned [BACHMANN, BRIEM et al., 2022]. Hence, chances that operators can afford to keep standby vehicles and drivers available are very low, if existent at all. Therefore, if additional capacity is needed, it is often taken from other lines which are less in demand. Keeping this in mind, the introduced process examines if buses can be withdrawn from some of the PT lines with lower demand to be reallocated to the PT lines which are needed to support the passenger redirection. In the following, the first type of PT lines is called support PT lines ($PL_{sup} \subset PL$), and the latter type is called insufficient PT lines ($PL_{ins} \subset PL$). Both types are subsets of all PT lines PL. In the previously described process of the passenger redirection, an OD pair $od \in OD_a$ could not be assigned to its fastest alternative path $\hat{p} \in P_{od}$ if this path did not possess sufficient remaining capacity to accommodate its corresponding redirection group rg_{od}^p . It is therefore the aim of this process to reallocate dispensable buses to the insufficient lines which are required to assign an OD pair $od \in OD_a$ to a faster alternative path $(tt_p < tt_{\tilde{p}}^{od})$, to improve the redirection strategy for the corresponding OD pair od if possible. If it is, the delays of the corresponding OD pair and thus the total delay of the affected passengers can be further reduced compared to the redirection strategy which has been found during the passenger redirection process.

Figure 4.6 shows the step-by-step procedure of the capacity reallocation process. First, in Step 24, the buses are identified that are actually operating during the incident. Since rail-bound services, such as tram and metro trains, are very restricted regarding their routing possibilities due to the boundaries of railway infrastructure, only buses are considered. Buses operate on the denser street network; therefore, they are more flexible than rail-bound vehicles and it is more likely for them to be able to support another line. When a bus is withdrawn from its originally scheduled line, it causes additional delays on that line. In order to keep these delays low and avoid that the reallocation of capacities causes more delays than it reduces, it is checked for each considered bus if the following bus can accommodate the additional demand on the original line of the considered bus (Step 25). In this case, the corresponding travellers would additionally wait for the period of the corresponding line's headway, at most. Therefore, the support lines from which the buses are withdrawn should ideally have a relatively small headway, depending on the PT system and situation.

In Step 26, based on the number of buses that can be withdrawn from the support lines $(PL_{sup} \subset PL)$, the dispensable capacity is computed. Especially during the peak demand of a day, it is possible that no bus can be spared because all buses B are needed at their original assignment. Hence, in Step 27 it is checked if dispensable capacity is actually available. If not, the redirection strategy set in the passenger redirection process remains unchanged, the capacity reallocation process ends and the subsequent information dissemination process starts. If there are dispensable capacities, the process continues with Step 28.

In this step, the travel times between different line routes are retrieved. These can either be preprocessed based on historical data or, if the PT system is adequately equipped, real-time travel times can be used. Thereby, it can be determined in this step how long each available bus would need from the beginning of its originally scheduled line route to the beginning of any other line route they might be reallocated to. At this point in the procedure, it can be examined which of the affected OD pairs $od \in OD_a$ can be redirected onto a faster alternative path $p \in P_{od}$. This process is built heuristically, based on the same ideas as the heuristic redirection approach introduced in the previous section. Thereby, the list of affected OD pairs $OD_a \subset OD$ is sorted by the size of the redirection group rg_{od}^p in association with the fastest alternative path $\hat{p} \in P_{od}$. The idea is, the faster the path on which an affected OD pair $od \in OD_a$ is assigned to $(\tilde{p} \in P_{od})$, the longer the corresponding redirection duration T_{od}^p . the larger rg_{od}^p , and hence, the bigger the reduction of the overall delay (Step 29). In the next step, Step 30, it is then checked whether the currently investigated $od \in OD_a$ is already assigned to its corresponding fastest path or not $(\tilde{p} \in P_{od} = \hat{p} \in P_{od})$. If yes, the situation for this od cannot be improved and the next OD pair $od \in OD_a$ is taken. If not, in Step 30b one alternative path $p \in P_{od}$ after another is taken to check whether od can be redirected onto a faster alternative path than it is currently redirected onto, starting with the fastest path $\hat{p} \in P_{od}$.



Figure 4.6: Capacity reallocation process

The following steps are the same as in the heuristic approach in Section 4.5.1. Therefore, in Step 31, the redirection duration is calculated for the currently investigated path $p \in P_{od}$ to afterwards examine the sizes of the corresponding redirection and waiting groups (rg_{od}^p) and wg_{od}^p). Since the capacity reallocation process in the overall procedure (Figure 4.2) is performed after the redirection process, the self-redirected group sg_{od} of each OD pair od is already considered in this examination of group sizes.

In the following, the remaining capacity of path $p(c_p)$ is determined (Step 33). As in the heuristic redirection process, this is done segment-wise. A path is divided into path segments $(ps \in PS_p)$ in accordance with the PT lines which are serving path p, PS_p being the set of all path segments of path p. In this way, the remaining capacity of each segment of p can be determined for the period for which rg_{od}^p would actually occupy a certain path segment $ps \in PS_p$. In case the OD pair od has not been redirected onto path p during the passenger redirection approach, path p has not sufficient remaining capacity. In Step 34, it then can be identified which path segments provide insufficient remaining capacity ($ps \in PS_p$), PS being the set of path segments with insufficient capacity, and hence which PT lines would need additional buses so that redirection group rg_{od}^p of OD pair od can be redirected onto path $p \in P_{od}$.

By reallocating buses, the number of free places of certain insufficient path segments $ps \in$ PS_p and thus their remaining capacity can be increased. Furthermore, it can be determined from when to when a path segment would be occupied by the passengers of rg_{od}^{p} . Each rg_{od}^{p} has a certain travel period during which the group is travelling along its path p through the PT network, starting with the departure of its earliest member and ending with the arrival of its latest member. This travel period can be divided in correspondence to the path segments. Then, starting at Step 35, each insufficient path segment $ps \in PS_p$ is investigated regarding the travel period of rg_{od}^p on this particular ps as well as its number of missing free places to fit rg_{od}^p . This number of places is calculated in Step 36. The next PT line is then taken from the set of support lines $PL_{sup} \subset PL$ to which previously identified potential supporting buses (Step 25) are associated with. The set of support lines $PL_{sup} \subset PL$ is sorted by the travel time a bus $b \in B$ takes to get from the start of its original line route at the support line to the start of the line route of the insufficient PT line of $ps \in PS_p$ (tt_{pl}^b) . In this way, the process first checks the buses which are fastest at the insufficient line $pl \in PL_{ins}^{ps}$, PL_{ins}^{ps} being the set of PT lines with insufficient capacity, associated with path segment ps. The faster the buses $b \in B_{sup}$ are reallocated, the more buses can support the insufficient line during the incident.

Furthermore, PT drivers are legally obliged to observe driving times. Therefore, the drivers' duty roster is planned accordingly and it is reasonable that through the capacity reallocation the drivers do not exceed their driving times. This also shows how complex the process of capacity reallocation can get. Here, legally prescribed driving times and duty rosters are not considered. Then, in Step 38, the next bus of the currently examined supporting PT line $b \in B_{sup}$ is checked to see whether it would arrive at the path segment $ps \in P\bar{S}_p$ in time or not. Here, "in time" means that bus b would arrive at the corresponding start of the insufficient line's route early enough to arrive at the start of the corresponding path segment $ps \in P\bar{S}_p$ within the period in which this segment would actually be occupied by the redirection group rg_{od}^p of the currently investigated OD pair od. Equation 4.8 expresses the period in which a bus needs to arrive at the corresponding path segment $ps \in Pod$ to be

considered as a supporting bus.

$$t_{s,ps}^{p} <= t_{dep}^{b} + tt_{pl}^{b} + tt_{ps}^{b} < t_{e,ps}^{p}$$
(4.8)

Where:

- $t^p_{s,ps}$ = start time of passenger redirection at path segment $ps \in \bar{PS}_p$ of path $p \in P_{od}$,
- $t^p_{e,ps} =$ end time of passenger redirection at path segment $ps \in PS_p$ of path $p \in P_{od}$,
- t_{dep}^{b} = original departure time of bus $b \in B_{sup}$ at $PL_{sup} \subset PL$,
- tt^{b}_{pl} = travel time of bus $b \in B$ from its original line route to the route of line $pl \in PL^{ps}_{ins}$, tt^{b}_{ps} = travel time of bus $b \in B$ on line $pl \in PL^{ps}_{ins}$ to the start of path segment $ps \in PS_{p}$.

In Step 39, it is then checked if the number of missing places, calculated in Step 36 can already be compensated by the supporting buses $b \in B_{sup}$ found so far. If yes, it is checked whether all insufficient path segments $ps \in P\bar{S}_p$ have been checked. If not, the next segment $ps \in P\bar{S}_p$ is taken in Step 35 and the previously described steps are repeated. If all insufficient path segments $ps \in P\bar{S}_p$ have been checked in Step 44a, the redirection group rg_{od}^p is assigned to the corresponding path $p \in P_{od}$, and the waiting group wg_{od}^p is assigned to the original path of the OD pair $0 \in P_{od}$. Afterwards, the found support buses $b \in B_{sup}$ are reallocated to the corresponding insufficient PT lines to ensure that path p has sufficient remaining capacity to fit the redirection group rg_{od}^p (Step 45). Additionally, the corresponding paths' capacities are updated in the network graph G.

As the last step in this capacity reallocation process, it is ensured that all affected OD pairs $od \in OD_a$ have been checked (Step 46). If not, the next OD pair od is examined starting at Step 29; if so, the process ends and the following information dissemination process begins.

If in Step 39 the number of additional places is insufficient, the process continues from there further to Step 41, determining the availability of additional supporting buses at the currently examined support line $PL_{sup} \subset PL$. If it does, it goes to Step 38 and verifies the sufficiency of the added places through the found supporting buses $b \in B^{pl}_{sup}$. If there is still not enough room, the process examines if there is a support line $pl \in PL_{sup}$ left for the insufficient path segment $ps \in PS_p$ (Step 42). If there is at least one line available, the process takes the next support line in Step 37 and continues accordingly. If not, the process determines whether there are alternative paths $p \in P_{od}$ left in Step 43. Note that Step 43 is only executed until the process reaches the path to which the OD pair od has been assigned during the redirection process $(\tilde{p} \in P_{od})$. Otherwise the corresponding rg_{od}^p stays on the assigned path $\tilde{p} \in P_{od}$ because all remaining paths $p \in P_{od}$ have a longer travel time, as the set of paths is sorted by travel time. Taking a path p listed below the assigned path $\tilde{p} \in P_{od}$ would thereby worsen the situation for the passengers of OD pair od. If there is a corresponding path $p \in P_{od}$ left, the process investigates the next path p, starting from Step 30b. If not, the redirection group rg_{od}^{p} is again assigned to $\tilde{p} \in P_{od}$ and the waiting group wg_{od}^p to the original path $0 \in P_{od}$. This can be the same path, in the case that no faster alternative path with sufficient remaining capacity has been found (Step 44b).

Again, in Step 46, it is ensured that all affected OD pair $od \in OD_a$ have been checked, if not, the next OD pair od is taken and the process continuos at Step 29; if so, the capacity reallocation process ends and the next process about the dissemination of information is started. A new redirection strategy is set.

4.7 Information Dissemination

The last process of the REPACK algorithm is about the dissemination of the computed redirection strategy. This includes the PI as well as the information which is needed by the PT staff.

Therefore, in Step 47, it is checked whether capacity has been reallocated in the previous process. If so, the corresponding drivers of the supporting buses ($b \in B_{sup}$) are informed about their altered assignment (Step 48). In case no support bus $b \in B_{sup}$ could be found, this step is consequently skipped.



Figure 4.7: Information dissemination process

The following steps are about informing the affected passengers in accordance with the set redirection strategy. To do so, the relevant information channels need to be collected from all available PI channels the PT system has (Step 49). Especially for collective information channels such as speakers and dynamic PI displays at stops and in PT vehicles it is necessary to identify the relevant ones. Afterwards, each affected OD pair $od \in OD_a$ needs to be associated with a certain set of PI channels. To avoid confusion each OD pair od should receive the same path advice across all channels. Furthermore, it is not appropriate but rather confusing if path advice is disseminated at stops the corresponding rg_{od}^p are not passing and in vehicles which they are not riding. As different OD pair od have different origins of redirection and some of them also use different PT lines along their distinct assigned paths $\tilde{p} \in P_{od}$, each OD pair $od \in OD_a$ needs to be associated with certain stops and vehicles its members are using to ensure that the corresponding speakers and PI displays disseminate right path advice (Step 50).

In Step 51 path advice and, if applicable, additional information (e.g. kind of incident, special comments regarding the redirection) is formulated for each OD pair $od \in OD_a$, which

is then disseminated in Step 52 across all previously collected and associated PI channels. For PI channels such as PT trip planners on operators' websites and PT smartphone apps the given path advice can be picked in accordance with the entered origin and destination of the PT users.

Then, in Step 53, it is checked whether the corresponding redirection duration T_{od}^p , calculated during the passenger redirection and capacity reallocation process, elapsed. If not, the information dissemination continues, otherwise, REPACK ends. At this point it is more convenient for all passengers of each OD pair $od \in OD_a$ to wait for the dissolution of the incident which ideally does not take much longer. Then the PT service returns to its scheduled operation and all passengers can take their usual paths from their respective origin to their respective destination.

Chapter 5

Case Studies and Model Evaluation

In the following, a simulation study is presented, showing the results of various scenarios and cases in which the previously introduced model REPACK is tested in PT systems. This study assesses the effectiveness of REPACK in terms of reducing passenger delays in incident situations.

5.1 Model Implementation

Among the publicly available tools the open-source software SUMO has been preferred to others for several reasons: SUMO is a microscopic traffic simulator that is under continuous development at the German Aerospace Center [Translated from German: Deutsches Zentrum für Luft- und Raumfahrt] (DLR) [LOPEZ et al., 2018; DLR, 2022]. Before finally the decision has been made on SUMO, several other simulation environments were considered. At first, the commercial software PTV Visum by PTV Planning Transportation Traffic [Translated from German: Planen Transport und Verkehr GmbH] (PTV) was considered, a macroscopic traffic simulator which has been rejected here as it does not offer the level of detail needed to deal with the processes of REPACK. Through the literature described in Section 2.2.2, the attention turned to the publicly available mesoscopic simulation tool BusMezzo that is under development at the Smart Public Transport Lab at the Delft University of Technology [CATS, 2022]. This tool's functions are very suitable for the processes of REPACK and it has already been used for investigations on incident management and PI in PT systems [CATS, LARIJANI, KOUTSOPOULOS et al., 2011; CATS, KOUTSOPOULOS et al., 2011]. Unfortunately, the available documentation is limited [CATS, 2022]. The publicly available simulation software SUMO is well documented and provides helpful functions, such as feeding passenger routes as input to a simulation run, which is an important characteristic to simulate certain reactions of passengers to given path advice [DLR, 2022]. In addition, it comes along with helpful Python scripts to process simulation outputs and prepare simulation inputs. Therefore, the decision was made to use SUMO as a simulation environment.

The REPACK algorithm itself has been implemented in the object-oriented programming language Python which is employed by a large user community and provides for a variety of helpful packages [PYTHON SOFTWARE FOUNDATION, 2022]. Several Python scripts have been written to implement REPACK as described in Chapter 4.

Figure 5.1 shows the step-by-step procedure of each simulation study conducted. In Step 1, a scenario without any incident is simulated to synthetically generate the "historical" travel data.



Figure 5.1: Simulation study procedure

The simulation output contains, among other things, the taken paths and associated travel times of the passengers, as well as the travel times and occupancies of the PT vehicles. The results of this scenario show the travel times under normal undisturbed conditions. Therefore, this serves as the lower benchmark scenario and is used to calculate passenger delays in the scenarios with an incident. The closer the results of REPACK are to the lower benchmark scenario, the better (Equation 4.1).

In Step 2, a scenario with an incident is conducted, however, without executing REPACK. Nevertheless, *supply-centric* disposition measures (e.g. PT line-splitting, PT line-rerouting), as described in Section 2.2.1, are performed. This serves as higher benchmark scenario. Therefore, for every scenario in which an incident occurs, a disposition timetable is defined which reflects the condition of the PT service as it is provided during the incident. This timetable includes the reduction of the service quality due to the incident as well as the changes to the service through the considered disposition measures.

Based on this timetable, a network graph, as explained in Section 4.3, is built. It is implemented using the Python package NetworkX [HAGBERG et al., 2022]. This graph is used to conduct path searches and to check the available remaining capacities of paths identified (Step 3).

Step 4 refers to the first process of REPACK, "Situation Analysis" (Section 4.4). As its

name indicates, this process analyses the present incident situation. The necessary incident information, such as its duration and location as well as disrupted PT services, is assumed. As mentioned in Section 4.4, in real use cases, this kind of information can be provided by the dispatchers in OCCs or even directly through an ITCS [BACHMANN, BRIEM et al., 2022]. In this process, the affected passengers are identified and associated with OD pairs in accordance with their origin of redirection and destination. Furthermore, a set of available paths in the PT network is assigned to each of these OD pairs. Another output of this process is the assignment of the passengers who redirect themselves, meaning finding and taking an alternative path on their own. It is assumed that these passengers take the fastest path, as this path characteristic has been stated as the most important one by the participants of the PT user survey presented in Chapter 3 (Figure B.10).

The next process, "Passenger Redirection" (Section 4.5) finds suitable alternative paths for each OD pair. Since two approaches have been developed to solve the passenger redirection problem - heuristic and optimised redirection - the process branches out in Step 5 depending on the decision of which approach to use in a specific case study. The decision which approach is taken is done manually, nevertheless, the results presented below also indicate in which case, which approach should be preferred. As a result of this process each of the previously identified affected OD pairs $od \in OD_a$ is assigned to a specific alternative path $\tilde{p} \in P_{od}$ for the period of the redirection duration T_{od}^p or is assigned to the respective original path $0 \in P_{od}$ if no alternative path holding sufficient remaining capacity can be found (Section 4.5).

To be able to evaluate the results of the passenger redirection process, in Step 6, a simulation run is performed in which the incident occurs, the *supply-centric* disposition measures are taken, and the passengers are redirected in accordance with the outcome of the passenger redirection process. This is followed by the process "Capacity Reallocation" (Step 7). In this step, the process described in Section 4.6 checks whether dispensable capacities are available in the network and if their reallocation can support the passenger redirection. Its outcome is the reallocation of the corresponding buses from their original PT line to a line with insufficient capacity at which they are needed to assist the passenger redirection. Since there is the possibility that no sufficient dispensable capacities are available or that none of the dispensable buses would reach the insufficient PT lines in time, it is checked in Step 8 if in Step 7 buses are reallocated. If not, the procedure continues directly with the last step to evaluate the simulation results. If buses are reallocated, a simulation run is conducted in which the incident occurs, the *supply-centric* disposition measures are taken and the passenger redirection, supported by the capacity reallocation, is done.

In SUMO, the reallocation of buses is realised through SUMO's API, called traffic contol interface (TraCl) [DLR, 2022]. Whenever one of the reallocated buses enters the network, TraCl is used to change the line, route, and stops of the corresponding bus in accordance with its reallocation.

As the last step, the results of all simulation runs of the case study are evaluated. By subtracting the individual travel times of each passenger in the lower benchmark scenario from each of the other scenarios (no redirection, heuristic or optimised passenger redirection, and capacity reallocation), the respective delays as well as other indicators can be calculated to evaluate the performance of REPACK and its components (Step 10).

To evaluate REPACK, the relative delay of passengers compared to the lower benchmark

scenario without an incident is used as key performance indicator. In addition, the total delay, the delay distribution, and the delay per time interval are used as indicators, to better understand the characteristics of the delay. The delay of a passenger includes the different parts of travel time, in-vehicle, waiting, and walking time. As waiting time is the part which is perceived as most inconvenient (Section 2.1.2), it is also investigated. Lastly, crowding at stops and in-vehicles is a known side effect of incidents (Section 2.2.2). Therefore, it is examined if REPACK has an influence on the traveller crowding at stops and the occupancy of PT vehicles.

The process "Information Dissemination" (Section 4.7) is not part of the case study procedure described here, as it explains mainly steps that are relevant for the implementation of REPACK in practice, such as broadcasting path advice via the available PI channels (Section 2.1.5) and informing the corresponding PT drivers about the capacity reallocation. The implementation of REPACK into practice is further discussed in Section 6.3.

5.2 Implementation in the Mandl Network

As first environment to test the performance of REPACK, the Mandl network is used (Figure 5.2). It has been published by MANDL [1979] with link travel times and an OD demand matrix. The original network and demand matrix can be found in Annexe C. The network contains of 15 nodes and 21 links with link travel times from two to ten minutes. An important advantage of using such a benchmark network is the publicly available information on the travel times and demand, so future works in the field of PCIM can also use it and the results are comparable to the results of this work due to the same inputs. Due to the same reason it has been used by several investigations examining the PT network design and frequency setting problem, some of which are presented and compared by UL ABEDIN et al. [2018]. Here, a PT network design by UL ABEDIN [2019] is used as an environment for the following case studies.

Criteria for the choice of this PT network setup for the Mandl benchmark network have been that ideally all links of the network are covered by at least one PT line to have a relatively complex PT network for such a comparatively small road network. As REPACK relies on the availability of alternative paths in the network a minimum complexity is necessary to be able to demonstrate REPACK's functionality.

Furthermore, the link between Nodes 8 and 10 has been chosen as the incident location as it is centrally located in the network. As shown in Figure 5.2b, the centrality of the link caters for a relatively high number of affected passengers. Furthermore, it ensures the aforementioned availability of alternative paths, which would have not been given if the incident would occur between Nodes 1 and 2, for example. Since in the presented PT network design of the Mandl network, two lines are operating on this link and are therefore disrupted by the assumed incident, the consequences are correspondingly severe. These are features that are necessary to demonstrate the effectiveness of REPACK, since the more central the incident, the more PT users are affected and the more users can be potentially redirected.

Besides, the line route course, the timetable is an important input to represent the PT service in the simulation. For the sake of simplicity, the timetable has been modified from its



Figure 5.2: Mandl PT network [UL ABEDIN, 2019]

original setting by UL ABEDIN [2019]. First, all lines are served by buses with a capacity of 100 passengers. Second, there are only two headway settings, five minutes for the red, grey, and yellow lines and ten minutes for the cyan and violet lines.

Another important input is demand. There are different kinds of demand, namely: pedestrian, bicyclist, car, truck, and other motorized individual traffic demand, as well as PT user demand. Here, only PT user demand is considered. PT user demand is derived from the original OD matrix by MANDL [1979] (Annexe C). However, it has been modified to better suit the testing of REPACK. First, it is assumed that the whole demand, represented by the aforementioned matrix, is PT user demand as it also has been by the aforementioned investigations about PT network design [UL ABEDIN et al., 2018]. Second, the OD relations which are affected by the assumed incident are increased to also increase the severity of the consequences of the assumed incident as more passengers are affected by it and need an alternative path. This has been done to ensure the visibility of REPACK's effect on the passenger delays. Which relations have been changed and to which extent is shown in Annexe C. Third, the overall demand has been increased to adjust it to the simplification of the service and the corresponding increased service frequency which also leads to an increased severity of the incident's disturbances and helps to investigate the effect of REPACK. BACHMANN, DANDL et al. [2023] use double the original demand over a period of four hours. Their results show severe delays also for unaffected passengers who the authors explain through crowding at the bus stops. In order to avoid this effect or at least weaken it, here, the OD matrix is multiplied by 1.5 instead of 2.0. As in the aforementioned publication, the demand is spread over four hours. The OD matrix used here is shown in Table C.1 in Annexe C. Based on this matrix, the origins, destinations, and initial paths for the passenger demand are determined. For this, the dynamic user assignment tool "duarouter", which is part of SUMO, has been used [DLR, 2022]. This tool uses the Dijkstra algorithm (Section 2.1.2) to assign a particular path to each passenger, including specific PT lines to take by a certain passenger.

To consider the common line problem that describes situations in which more than one line could be used for a connection, here, the passengers' paths are checked for such situations and the list of available lines are complemented accordingly for each connection of each passenger [KURAUCHI et al., 2003]. In addition to the initial paths, this is also done during later steps in REPACK, when passengers are redirected onto alternative paths.

To minimise the possibility that the results of the case studies merely occur coincidentally, five different random seeds¹ are considered. These, for example, also cater for a variety of departure times for each passenger, which changes the simulation procedure for each seed. Moreover, they have also been used to create the initial path assignments for the lower benchmark scenario without an incident from the same OD matrix (Table C.1). These initial path assignments have been used for the lower as well as the higher benchmark scenarios. Ideally, the results of the cases in which REPACK is executed show lower overall passenger delays than the cases without. The closer the results are to the lower benchmark, the better.

Concerning the relative delays of affected passengers, the results show a maximum variance of 7.1%. With a margin of error of 2.5%, a confidence interval of at least 95%, depending on the considered case, can be achieved. Therefore, the number of seeds seems to be sufficient to ensure the legitimacy of the results [FGSV, 2006].

5.2.1 Scenarios and Cases

Incident Features and Disposition Measures

For all scenarios and cases in the Mandl network, the incident is located between Nodes 8 and 10. Furthermore, three different incident durations are tested, namely: 30, 60, and 90 minutes. For each incident duration and random seed, in addition to the lower benchmark scenario, two different scenarios with several different tested cases have been conducted. A scenario is defined by the disposition measure that is taken as a *supply-centric* response to the incident. This measure impacts the PT supply during the incident and shapes the disposition timetable.

In the first scenario, the applied measure is PT line-splitting (LS). It divides the disrupted line into two parts, one on each side of the incident's location. On both sides, the PT service is short-turned and the two line parts operate in loops on both sides of the incident. For the case study presented here, this means that the upper part of the red line operates between Nodes 1 and 8 and the lower part of it between Nodes 10 and 12. The upper part of the yellow line operates between Nodes 8 and 9, and the lower part between Nodes 10 and 11 (Figure 5.3a).

In the second scenario, the applied measure is PT line-rerouting (LR). In this scenario, the disrupted PT lines are rerouted via alternative routes (Section 2.2.1). For the cases conducted in Scenario LR, the red line is rerouted between Nodes 8 and 10 via Nodes 7 and 15, not

¹Namely: 281218, 210619, 230619, 271019, and 120320
stopping at the latter two nodes. The yellow line is rerouted via Node 7, not stopping at this node. Furthermore, for the yellow line, the stop at Node 8 is cancelled during the period in which the rerouting is in progress (Figure 5.3b). The two scenarios represent the higher benchmark scenarios for the respective cases. The disposition measures are conducted for the whole period of the incident duration.



Figure 5.3: Mandl PT network with disposition measures

Parameters

Besides the scenario-specific disposition measures, the REPACK model described in Chapter 4 includes several parameters which need to be set. The parametrisation for the case study is shortly described in the following.

Self-redirection Rate ξ : The self-redirection rate ξ expresses the share of affected passengers who prefer to find an alternative path on their own instead of following given path advice. It is assumed that these people would choose the fastest alternative path irrespective of the PT system's state. In accordance with the survey results presented in Section 3.2.3, about 5% of the asked participants would rather redirect themselves than follow given path advice (Figure 3.9). As pointed out in Section 2.1.4, the reaction of passengers to incidents is influenced by a large variety of factors and therefore difficult to predict. Hence, it is reasonable to assume that this is also true for the passengers' willingness to follow given path advice. There are many factors that may influence the decision, be it the distance to the destination, car or bike ownership, weather, and many more. Therefore, to better understand the effect of people redirecting themselves on the performance of REPACK, a wider range of values for ξ have been tested: 0%, 10%, and 30%. Since 30% is already 25% more than the share of

people who stated in the presented PT user survey that they would redirect themselves, higher values have not been considered.

Compliance Rate η and Information Rate ζ : The number of affected passengers of each affected OD pair $od \in OD_a$ is multiplied by the compliance rate η to calculate the number of people who would follow the path advice. Additionally, this number is multiplied by the information rate ζ , which is the share of passengers who receive the path advice. As shown in Section 3.2.3, most of the participants of the survey preferred to be informed via a collective PI channel (i.e. speaker announcement, dynamic PI displays) during incidents (Figure 3.7). Depending on how well equipped a PT system is in terms of the distribution of dynamic PI displays and speakers at stops and in PT vehicles, passengers may or may not receive path advice. Since the number of affected passengers of each OD pair $od \in OD_a$ is multiplied by both η and ζ , it is reasonable to set them as one parameter for the case studies to avoid redundancies. For example, the results would be the same for a case in which η is 50% and ζ is 70%, compared to a case in which η is 70% and ζ is 50%.

According to the results from the survey in Section 3.2.3, about one-quarter of the participants *will* follow given path advice, whereas about 60% *might* follow it. This implies a wide range of possible outcomes. As previously mentioned, it is likely that a lot of factors (e.g. weather, bike or car ownership, etc.) influence a passenger's reaction to an incident as well as the passenger's willingness to follow given path advice (Section 2.1.4). Therefore, it is difficult to predict how many people would actually follow path advice in a given incident situation. Furthermore, the reachability of passengers via collective PI channels can vary also depending on the location of the incident and of the passengers in the network. Whereas larger transfer hubs are often well-equipped with PI channels, this might not be the fact for small bus stops at the outskirts of a PT network. If no collective PI channels are available, the passengers rely on information via websites, social media, and PT smartphone apps. It is therefore reasonable to also test a wide range for the product of the compliance and information rates ($\eta \cdot \zeta$). Here, 0%, 35%, 70%, and 100% are tested. Furthermore, a special case is tested in which only a certain number of OD pairs is completely reachable and has thereby an information rate of 100%, whereas it is lower for the remaining OD pairs.

Once the number of passengers receiving the path advice and the number of those following it are calculated, a random sample of the corresponding size is taken of each OD pair $od \in OD_a$. As described in Section 4.5, it is then checked for these passengers whether they arrive at their respective origin of redirection on time before the redirection duration T_{od}^p elapsed or not (Equation 4.2). If they do, they are assigned to the redirection group rg_{od}^p , which is redirected onto the assigned alternative path $\tilde{p} \in P_{od}$. If not, they are assigned to the waiting group wg_{od}^p which stays on the original path $0 \in P_{od}$.

These are different cases compared to the test cases presented by BACHMANN, DANDL et al. [2023], in which an information rate of 100% and compliance rates of 100%, 57%, and a logarithmic calculated compliance rate are considered. The 57% is the share for people who will follow given path advice, according to the results of a PT user trial by BMVI [2019]. The logarithmic compliance rate is introduced by VAN DER HURK et al. [2018] and takes the travel time difference between the assigned alternative path ($\tilde{p} \in P_{od}$) and the fastest alternative path ($\hat{p} \in P_{od}$) into account. If the travel time of the assigned path $\tilde{p} \in P_{od}$ is more than ten minutes longer than the travel time of the fastest alternative path $\hat{p} \in P_{od}$, the compliance rate drops rapidly. However, as pointed out in Section 2.1.4, various factors can influence the passengers' willingness to follow given path advice. Therefore, the mentioned difference in the paths' travel times is probably just one of several factors which also need to be considered in such an equation. Furthermore, WILKE [2023] points out that there are also influencing factors such as vouchers for free drinks and reduction in fare which could motivate people to follow a given connection suggestion. Consequently, a wide range of values for the compliance rate is tested here to cover a variety of different outcomes.

Factor ψ : In BACHMANN, DANDL et al. [2023], the last part of the equation to calculate the redirection duration T_{od}^p is a subtraction of five minutes. This is done to avoid missed buses leading to a disadvantage for redirected passengers. Five minutes has been chosen, as all lines of the case study conducted by BACHMANN, DANDL et al. [2023] have a headway of five minutes.

In more complex PT systems there is a variety of headways. Therefore, as explained in Section 4.5.1, the subtraction of a time constant in Equation 4.2 to calculate T_{od}^p is replaced by a product of two parameters. The first factor of the above product is the headway of the PT line that has the longest headway among the lines serving a given path $p(\check{h}_p)$, and the second factor, which is deterministically defined, is the factor ψ . This change caters for the consideration of different headways. As mentioned in Section 4.5, the product of factor ψ and \check{h}_p , however, can also be used to extend the redirection duration in case of a recovery time, when a negative value is used for ψ . Through a rough screening of possible values for ψ , the following values have been chosen for testing: 1, 0, -1, -2, -3.

Most of the values are negative, leading to the redirection duration T_{od}^p being extended by the product of ψ and \check{h}_p . However, the rough screening revealed that in some cases an extension of the redirection leads to better results concerning the average delays of all U and the affected passengers $U_a \subset U$. As explained, this indicates a recovery time.

The recovery time describes the period between the end of the incident and the point of time at which no more incident-related delays occur, for example, due to crowding at stops or in PT vehicles. Equation 4.2, however, only considers the incident duration, but not the recovery time, as it is very difficult to predict it, as pointed out by TSUCHIYA et al. [2006]. The authors suggest a linear function to predict the development of the recovery time, which is strongly related to timetable rescheduling in railway operations. In an urban PT system, many factors can influence the course of the recovery time: the demand, the occupancy of the PT services after the end of the incident, and traffic conditions, for example. Therefore, here, the adjustment towards the recovery time is done through $\psi \cdot \check{h}_p$.

Minimum Redirection Duration Lastly, the minimum redirection duration is mentioned here. The minimum redirection duration is the threshold which needs to be met by the calculated redirection duration to trigger the passenger redirection for an affected OD pair $od \in OD_a$. If the calculated redirection duration is shorter than 5 minutes, the corresponding redirection group rg_{od}^p is not redirected. Shorter redirection durations would probably not be appropriate and would lead to confusion in practice, as PT users also need time to adapt to an incident situation.

5.2.2 Results of the Case Study in the Mandl Network

With two disposition measures – Scenarios LS and LR – three incident durations (t_i) , three self-redirection rates (ξ) , four values for the compliance and information rates product $(\eta \cdot \zeta)$, and five values for factor ψ , in total, 360 study cases have been conducted. The relative delay of the affected passengers of all these cases is presented in Annexe C. In this section, some of these results are shown and discussed in detail.

To give a first brief insight into the results of the conducted case study, Tables 5.1 and 5.2 present the best outcome, in terms of lowest relative overall delay of the affected passengers in percent of each conducted case in Scenarios LS and LR with a self-redirection rate of $\xi = 0\%$. The lower the overall passenger delays are, the better. Moreover, of the four approaches used – heuristic redirection (HR), optimised redirection (OR), capacity reallocation in combination with heuristic redirection (CR+HR), and capacity reallocation in combination with optimised redirection (CR+OR) – the one that achieved the best outcome is named. If more than one approach achieved the best result, all of them are listed. The smallest relative delay for each incident duration t_i and ψ is presented in bold print. Further result tables are to be found in Annexe C.

ψ	t_i	$\eta\cdot\zeta$				
		0%	35%	70%	100%	
1	30 minutes	HR, OR: 100.2	OR: 71.5	OR: 49.9	CR+OR: 36.3	
	60 minutes	HR, OR: 100.0	OR: 63.3	HR: 34.7	OR: 17.9	
	90 minutes	HR, OR: 100.1	OR: 61.1	OR: 30.8	OR: 12.4	
0	30 minutes	HR, OR: 100.2	OR: 66.8	OR: 42.4	CR+HR: 29.5	
	60 minutes	HR, OR: 100.0	HR, OR: 61.7	HR: 32.2	OR: 15.3	
	90 minutes	HR, OR: 100.1	OR: 59.7	HR, OR: 29.1	OR: 10.7	
-1	30 minutes	HR, OR: 100.2	OR: 64.9	HR: 39.0	OR: 27.7	
	60 minutes	HR, OR: 100.0	OR: 59.8	HR: 30.8	OR: 14.6	
	90 minutes	HR, OR: 100.1	HR, OR: 59.0	HR: 28.1	OR: 10.1	
-2	30 minutes	HR, OR: 100.2	HR: 64.2	HR: 37.8	OR: 28.6	
	60 minutes	HR, OR: 100.0	HR: 59.0	HR: 30.2	OR: 14.4	
	90 minutes	HR, OR: 100.1	OR: 58.5	HR, OR: 27.8	OR: 10.0	
-3	30 minutes	HR, OR: 100.2	HR: 63.7	HR: 37.1	OR: 29.5	
	60 minutes	HR, OR: 100.0	OR: 58.2	HR: 29.9	OR: 14.6	
	90 minutes	HR, OR: 100.1	OR: 58.1	HR: 27.5	OR: 10.1	

Table 5.1: Lowest relative delays of each case for $U_a \subset U$ with $\xi = 0\%$ in Scenario LS (HR: Heuristic Redirection; OR: Optimised Redirection; CR+HR: Capacity Reallocation with Heuristic Redirection; CR+OR: Capacity Reallocation with Optimised Redirection)

Generally, it can be observed that the longer an incident lasts, the more the delay is reduced through REPACK. This can be explained by the fact that the longer an incident lasts, the greater the delays. Since more passengers are affected by it, and because waiting for its

dissolution takes correspondingly longer as well. Furthermore, it probably worsens the crowding at stops, which prolongs the recovery time after the dissolution of the incident.

The characteristics of ψ seem to be complex. It is a factor by which the headway of the PT line with the longest headway (\check{h}_p), serving path p, is multiplied to subtract the product from the redirection duration (Equation 4.2). In this way the calculation of the redirection duration is adjusted to the service's longest headway of a path p and gives the opportunity to consider the recovery time (Section 5.2.1).

The best result of 10.0% among the outcomes of the cases with a self-redirection rate of 0% has been accomplished by the optimised redirection (OR) and a ψ of -2 for an incident duration $t_i = 90$ minutes. It should be noted that similar results of 10.1% are to be found for the same cases with values for $\psi = -1$ and -3, again, accomplished by Approach OR.

As mentioned before, the recovery time after the dissolution of an incident is very hard to estimate, as factors such as demand and capacity availability influence how quickly a PT system recovers from an incident. These two factors again depend on the time of day — if it's during on-peak hours, off-peak hours, or at the beginning or end of a peak — as well as on staff management in terms of availability of staff (Section 2.2).

As different PT lines operate with different headways, the longest headways on the found alternative paths also play a role. In this simulation study only two headways are considered: 5 minutes for the grey, red, and yellow lines and 10 minutes for the cyan and violet lines (Figure 5.2). In more complex, real-life PT systems, there are wider ranges of headways, which makes factor ψ a more important model element with a high influence on the redirection duration and outcome. Multiplying a deterministic value with the longest headway might therefore be too simple an approach, leaving room for further research.

In general, across all conducted cases, Scenario LR gives results with a less effective delay reduction (Table 5.2) compared to the results of Scenario LS. Here, the best result is a delay reduction of 39.5%, compared to the best result in Scenario LS of 90.0%. This is only accomplished with $\eta \cdot \zeta = 100\%$. With $\eta \cdot \zeta = 35\%$ the delay reduction ranges from 6.5% to 15.5% depending on the incident duration. An obvious explanation is that rerouted PT lines, which still serve most or even all of their stops mean a less severe disruption of the PT service than line-splitting. The delay is not dependent on the incident duration, but on the additional travel time through the altered route of the corresponding buses. This delay is significantly smaller than having to wait for the dissolution of an incident before being able to continue a trip on the planned PT line, as is the case for the passengers in Scenario LS. Therefore, the gap between the two levels of severity of the incident's disturbances of the two scenarios grows with the increase in the incident duration t_i .

The travel time on the red line, for example, increases by about 3.5 minutes. This is also an explanation for the insignificant effect of changing the value of ψ in this scenario. For most of the affected passengers staying on the rerouted line might still be the best alternative. This raises the question of the reasonableness of passenger redirection in such cases. However, as the results of some of the cases show, it still can lead to significant delay reductions under certain circumstances.

Interestingly, in Table 5.2 all of the results in the case of $\eta \cdot \zeta = 100\%$ are 100.0% compared to the no-redirection case. Again, this can be explained by the lower severity of Scenario LR

ψ	t_i	$\eta\cdot\zeta$				
		0%	35%	70%	100%	
1	30 minutes	HR, OR: 100.0	HR: 93.5	OR: 86.9	HR: 80.8	
	60 minutes	HR, OR: 100.0	OR: 89.3	HR: 78.4	HR: 70.1	
	90 minutes	HR, OR: 100.0	HR: 85.5	HR: 71.4	HR: 60.8	
0	30 minutes	HR, OR: 100.0	HR: 93.1	HR: 85.7	CR+HR: 78.0	
	60 minutes	HR, OR: 100.0	OR: 89.1	OR: 78.0	HR: 69.5	
	90 minutes	HR, OR: 100.0	HR: 84.5	OR: 71.6	HR: 60.5	
-1	30 minutes	HR, OR: 100.0	HR: 93.4	HR: 86.1	CR+HR: 77.9	
	60 minutes	HR, OR: 100.0	OR: 88.9	HR: 77.9	HR: 69.4	
	90 minutes	HR, OR: 100.0	OR: 84.5	HR: 71.6	HR: 60.5	
-2	30 minutes	HR, OR: 100.0	HR: 93.4	HR: 86.1	CR+HR: 77.9	
	60 minutes	HR, OR: 100.0	OR: 88.9	HR: 77.9	HR: 69.4	
	90 minutes	HR, OR: 100.0	OR: 84.5	HR: 71.6	HR: 60.5	
-3	30 minutes	HR, OR: 100.0	HR: 93.4	HR: 86.1	CR+HR: 77.9	
	60 minutes	HR, OR: 100.0	OR: 88.9	HR: 77.9	HR: 69.4	
	90 minutes	HR, OR: 100.0	OR: 84.5	HR: 71.6	HR: 60.5	

Table 5.2:	Lowest relative delays of each case for $U_a \subset U$ with $\xi = 0\%$ in Scenario LR
	(HR: Heuristic Redirection; OR: Optimised Redirection;
	CR+HR: Capacity Reallocation with Heuristic Redirection;

CR+OR: Capacity Reallocation with Optimised Redirection)

through which less travellers are queuing at the stops at Nodes 8 and 10, as the red and yellow lines still serve their stops. Due to this fact, there is less interaction among the travellers, generally less disturbances and thus less room for different outcomes in basically the same situation compared to the no-redirection case.

As can be derived from Tables 5.1 and 5.2, the more severe the disturbances of an incident on a PT system, the higher the likelihood for longer recovery times and thus the better the results with a negative value for factor ψ . Therefore, for further analysis and discussion on the effect of the different approaches and the impact on their performance by various products of compliance and information rates, self-redirection rates, and incident durations, in the following, for the more disruptive Scenario LS the results with $\psi = -1$ and for Scenario LR the results with $\psi = 0$ are used. In this way the calculation of the passenger redirection is slightly adjusted to the disruptiveness of a scenario and the likelihood for a perceptible recovery time.

Relative and Total Delay

Figures 5.4a to 5.4d present the results of the simulation cases with an incident duration t_i of 30 minutes and a self-redirection rate ξ of 0% in Scenario LS. The upper two diagrams show the relative delay and the bottom the total delay, with the case of "No Redirection" as the higher benchmark case. The two diagrams on the left present the results for the affected PT users $U_a \subset U$ whereas those on the right side show the results for all PT users U.



Figure 5.4: Overall relative and total delays of $U_a \subset U$ and U with $t_i = 30$ minutes and $\xi = 0\%$ in Scenario LS ($\psi = -1$)

In Figure 5.4a, which shows the relative delays of the affected passengers, it is immediately noticeable that the delay continuously falls with an increase in the compliance and information rates product $(\eta \cdot \zeta)$. As the self-redirection rate ξ is 0%, here, the cases with a product of compliance and information rates of 0% show also a relative delay of about 100%, as no passengers have been redirected and the outcome of the simulation is, therefore, the same as in the case with no redirection. In addition, the approaches including the capacity reallocation are merely present in the case with $\eta \cdot \zeta = 100\%$. Reasons for this are described in the following.

Kindly note that the capacity reallocation has not been executed for some of the conducted cases and seeds. The capacity reallocation process is only done if the criteria for its execution have been met. These are the availability of dispensable capacity and that there is insufficient capacity on the fastest alternative path or alternative paths that are at least faster than the assigned one $\tilde{p} \in P_{od}$ (Section 4.6). The results of the approaches including capacity reallocation are therefore only considered in the evaluation process if the capacity reallocation process has been executed for all five random seeds of a case. Due to the different seeds, the travellers' departure times vary which can sometimes lead to the occupation of a vehicle that would be dispensable with the departure times caused by another seed, and the same applies to the need for capacity reallocation. Therefore, for the cases of $\eta \cdot \zeta < 100\%$, it might be the case that sufficient remaining capacity is available on the alternative paths.

If this is the case, it might also be an answer to the question why the optimised redirection performs slightly worse than or equal to the heuristic besides the case with $\eta \cdot \zeta = 100\%$, in which it performs better than the heuristic. The validation of sufficient capacity on the alternative paths is done differently in the two approaches. Whereas in the optimised redirection approach, the capacity on all alternative paths is calculated for a fixed period which is equal to the incident duration, in the heuristic redirection approach, the remaining capacity is determined segment-wise (Section 4.5). The segment-wise determination of the capacity considers the travel time of a redirection group rg_{od}^p along an alternative path $p \in P_{od}$. It is therefore more precise in calculating the remaining capacity and might assign more redirection groups on faster paths, as long as there is sufficient remaining capacity on the respective fastest alternative paths. It can also lead to the assignment of fewer passengers to avoid overcrowding effects. This advantage of the heuristic over the optimised redirection seems to last until the fastest paths do not possess sufficient remaining capacity anymore, as seems to be the case with $\eta \cdot \zeta = 100\%$. Here, the optimised redirection performs better than the heuristic. An explanation for this is that the optimisation does not assign the redirection group of each affected OD pair $od \in OD_a$ step by step, going from the biggest to the smallest redirection group rg_{od}^{p} , as is done in the heuristic, but all redirection groups are assigned in a way that the overall delay is minimised.

The two approaches including the capacity reallocation – CR+HR and CR+OR – perform similarly to the optimised redirection (OR). The fact that the capacity reallocation does not improve the outcome of the redirection approaches can be explained by the disadvantages which come along with it. Even though the relevant paths for passenger redirection are supported by extra capacity and additional runs, the reallocation of buses reduces the levels of service on the lines they have been withdrawn from. This quality reduction is due to the lower capacity as well as the longer headway. Furthermore, when capacities are reallocated, REPACK checks if the line from which the buses are withdrawn still possesses sufficient capacity for its demand. Additionally, it is validated that the reallocated bus arrives at the line with insufficient capacity in time, meaning before the corresponding redirection duration T_{od}^p elapses. However, it does not check how many passengers actually profit from a reallocated bus and it does not check the temporal offset to the regularly scheduled buses of the supported line. If a reallocated bus arrives shortly before or after a regularly scheduled bus at a stop of the line which is supported by it, there might be no passengers left at the stop to be picked up by the second vehicle. The

capacity reallocation would be in vain in such a case and might even aggravate the situation as it reduces the service quality on the original line of a reallocated bus. At the same time it does not improve the service at the line with insufficient capacity which can also lead to more people being redirected than the corresponding alternative path can take as the additional capacity does not arrive at the alternative path when it is needed by the redirected passengers.

Considering all passengers (Figure 5.4b) the relative delays are smaller than the ones only taking affected passengers into account (Figure 5.4a). This is logical, as the number of the affected passengers of about 1,100 passengers, depending on the random seed value used (Section 5.2.1), is smaller than that of all 25,486 passengers. The delays of the affected passengers, however, are more severe. Other passengers who do not use the disrupted PT lines, but are affected by the crowding at the stops and in vehicles, are referred to as indirectly affected passengers ($U_{ida} \subset U$), whose delays also play a role in the consideration of the delay of all passengers (U). The relation between the outcomes of the relative delays is comparable between U and $U_a \subset U$. Looking at the total delays in Figures 5.4c and 5.4d, it becomes immediately evident that the delays of the indirectly affected passengers are not to be underestimated, with about half of the delays being associated with them. Besides, the indirectly affected passengers $U_{ida} \subset U$ who are affected through crowding, passengers who travel after the dissolution of the incident on the previously disrupted PT lines are especially likely to face delays as the system has not yet returned to its normal state, as affected passengers queue at stops. This raises the question of the definition of the affected passengers. So far it is limited to the passengers who would travel on the disrupted link during the incident. This definition might need to be redesigned, as it also needs to consider passengers of the same OD relations travelling after the dissolution of the incident on the previously disrupted links. Furthermore, it again stresses the necessity to investigate the recovery time of the PT system.

Figure 5.5 shows the outcome of the same cases but with an incident duration t_i of 60 minutes. The results show a comparable picture to that in the cases with $t_i = 30$ minutes. Nevertheless, two observations are easy to make. First, the overall relative delay is smaller than in the cases with a shorter incident duration. As a longer incident duration also causes an increase in the number of affected passengers, consequently the overall delay also rises, through which the effect of REPACK grows as well. Second, the capacity reallocation is again executed only in the case with a product of compliance and information rates of 100%. However, it performs noticeably worse than the optimised redirection, which reaches the best outcome in this example with an overall relative delay of 10.0% for all passengers U and 14.6% for the affected passengers $U_a \subset U$.

When dispensable buses are reallocated, the service quality on the PT line from which they have been withdrawn is reduced in terms of extending the service headway. This leads to longer waiting times at the first stop and transfer stops of travellers. Therefore, the capacity reallocation might be continuously more disruptive with an increasing incident duration. Moreover, it might be reasonable to limit the number of reallocated buses that have been withdrawn from the same line to limit the negative effects of capacity reallocation. Comparing the total number of delays of all and affected passengers, it can also be seen here that the delays of affected passengers – about 2,270 people – account for more than half of the delays of all passengers – 25,486 people. Hence, the relation between the affected $U_a \subset U$ and



Figure 5.5: Overall relative and total delays of $U_a \subset U$ and U with $t_i = 60$ minutes and $\xi = 0\%$ in Scenario LS ($\psi = -1$)

indirectly affected passengers $U_{ida} \subset U$ seems to change with an increase in t_i . This indicates that the delays of the indirectly affected travellers $U_{ida} \subset U$ rise slower with an increase in the incident duration than do the delays of the affected passengers. Nonetheless, in all cases with $\eta \cdot \zeta > 0\%$, all approaches reduce the passenger delay significantly compared to the higher benchmark case with no redirection.

In Figure 5.6, the results with an incident duration of 90 minutes are presented. The trend implied by the differences between the previous two figures, namely that the relative delay



Figure 5.6: Overall relative and total delays of $U_a \subset U$ and U with $t_i = 90$ minutes and $\xi = 0\%$ in Scenario LS ($\psi = -1$)

further decreases with increasing incident duration, continues, with the optimisation approach performing best, with 7.0% for U and 10.1% for $U_a \subset U$, again, with a product of compliance and information rates of 100%. As before, this can be explained by the increasing total delay due to the longer incident duration through which the effectiveness of REPACK seems to further rise as well. The more people are affected – here, about 3,410 people – the more can be redirected, and the bigger the positive effect of passenger redirection. Another trend that continues is the relation between the total delay of all and the affected passengers. With an incident duration of 90 minutes the total delay of the affected passengers is about 57% of

the total delay of all passengers whereas for $t_i = 30$ minutes they are 50% and for $t_i = 60$ minutes they are 55%. So, with a growing incident duration the influence of the delays of the indirectly affected passengers $U_{ida} \subset U$ seems to shrink. The fact that this effect is stronger between an incident duration of 30 and 60 minutes than between 60 and 90 minutes might be an indication that this effect does not grow linearly but weakens with an increase in the incident duration.

The relation between the two capacity reallocation approaches and the heuristic and optimised redirection is also nearly the same as in these cases with $t_i = 60$ minutes. Hence, the previously described disadvantage of the capacity reallocation of reducing the service quality on the PT lines from which the dispensable vehicles are withdrawn does not further worsen with a longer incident duration. Nonetheless, it raises the question of how this effect further develops with incident durations longer than 90 minutes. The ratio between the demand on the support line, from which the buses are withdrawn, and the demand on the line with insufficient capacity, to which the buses have been dispatched, is probably also a factor that influences the success of the capacity reallocation. Furthermore, so is the headway of the support line. The longer the headway of the support line, the longer the additional waiting time of the travellers on the support line, induced by the capacity reallocation. Moreover, as previously mentioned, another explanation for the performance of the reallocation approaches is the timing of the reallocated buses.

The following Figures 5.7a to 5.7f present the relative delay of the affected passengers $U_a \subset U$ with a self-redirection rate ξ of 10% on the left side and a rate of 30% on the right side with an increasing incident duration from 30 to 90 minutes from the top to the bottom of this set of figures. Since in these cases passengers are considered who redirect themselves and do not follow given path advice or who simply wait for the dissolution of the incident, even in the case with $\eta \cdot \zeta = 0\%$, the relative delay is lower than in the case with no redirection due to the reduced delays of the self-redirected travellers. Logically, the relative delay is lower with a higher value for ξ .

In cases with $t_i = 30$ minutes the approaches with a capacity reallocation – CR+HR and CR+OR – are not executed, at least not with all five random seeds used (Section 5.2.1). This might be due to the following reasons: Firstly, an increase of ξ and t_i does also increase the number of self-redirected passengers who occupy places on the fastest alternative paths as these are the paths they are assumed to take. With $t_i = 30$ minutes this effect seems to be too low to activate the capacity reallocation. Secondly, as the passengers who redirect themselves are chosen by chance, a situation might occur in which there are no fitting dispensable buses as in the cases with $\xi = 0\%$. This is due to self-redirected travellers occupy them. With an increase in the incident duration, the chance for dispensable capacities grows, as do the period during which they are looked for and the period during which they need to reach the PT line with insufficient capacity. Therefore, the longer the incident duration, the more severe the delays it causes, however, the higher the chance for dispensable capacity (Section 4.6). Hence, for the longer incident durations of 60 and 90 minutes, the capacity reallocation is executed. In the case of $\xi = 30\%$, CR+HR performs best with a relative delay for the affected passengers of 18.7% with $t_i = 60$ minutes. Interestingly, in this case only

CR+HR, but not CR+OR is executed. This might be due to the differences between the two redirection approaches through which the passengers are differently redirected, which seems to cater for no fitting dispensable capacities in the case of the optimised redirection.

In the cases with $t_i = 90$ minutes CR+HR and CR+OR perform best, reducing the relative delay of the affected passengers by 84.2%. In these cases of $\xi = 30\%$, the advantages of CR+HR and CR+OR beat their previously discussed disadvantages. This leads to the capacity reallocation approaches performing best. Furthermore, it is also possible that some of the self-redirected passengers profit from the reallocated capacity as it is dispatched to faster alternative paths, ideally to the respective fastest alternative paths of the affected OD pairs $od \in OD_a$. The capacity reallocation increases not only the capacity on the line it is dispatched to, but also raises the frequency due to the extra vehicles. This reduces the waiting time for some of the redirected passengers and might also lower the waiting time for some of the self-redirected passengers. Nevertheless, the fact that the overall passenger delay reduction is smaller in the cases with a higher self-redirection rate implies that the overall redistribution of passengers is not ideal.

Generally speaking, the results with a self-redirection rate of higher than 0% show that even with people redirecting themselves, for example experienced commuters, REPACK still reduces the overall delay significantly. According to the previously presented results of the PT user survey, a self-redirection rate between 0% and 10% can be expected (Section 3.2.3).

In addition to the previously shown results for the products of compliance and information rates of 0%, 35%, 70%, and 100%, in the following, results with finer steps for $\eta \cdot \zeta$ are presented for an incident duration of 60 minutes. This is done to investigate possible key values for these parameters for REPACK to be reasonable or to perform well. For example, to identify a necessary minimum value for the product of compliance and information rates.

Figure 5.8 shows the relative delay in 10% steps of $\eta \cdot \zeta$ of the affected passengers ($U_a \subset U$) on the left side and all passengers (U) on the right side. The results underline the reasonability of REPACK, as already with $\eta \cdot \zeta = 10\%$, there is a reduction of the passenger delay of about 10%. The reduction of the delay develops mainly linearly with respect to the increase in the product of the compliance and information rates ($\eta \cdot \zeta$). The approaches with capacity reallocation are only executed at $\eta \cdot \zeta = 100\%$; however, the optimised redirection performs best in this case, with a relative delay of 14.6% for the affected and 10.0% for all passengers.

A product of compliance and information rates of 80% seems to be the tipping point where the advantages of the optimised redirection outweigh the advantages of the heuristic redirection. With a smaller product of compliance and information rates the heuristic performs better in most cases, whereas with a value of over 80%, the optimisation performs better. The to be expected product of the compliance and information rates can thereby be an indicator for the decision on which of the two passenger redirection approaches to be used.



No Redirection Heuristic Redirection (HR) Optimised Redirection (OR) 75 Relative Delay [%] 50 49.8 38.7 33.1 33.0 32.1 32.2 25 0 $\begin{array}{cccc} - & 0 & 35 & 70 & 100 \\ Compliance Rate (\eta) \cdot & Information Rate (\zeta) [\%] \end{array}$ (b) $U_a \subset U$; $t_i = 30$ minutes; $\xi = 30\%$ 100 No Redirection Heuristic Redirection (HR) Optimised Redirection (OR Capacity Reallocation + HR ΗŔ 75 Relative Delay [%] 50 43.5 2 31.2 25 0 $\begin{array}{cccc} - & 0 & 35 & 70 & 100 \\ \text{Compliance Rate }(\eta) & \text{Information Rate }(\zeta) \, [\%] \end{array}$ (d) $U_a \subset U$; $t_i = 60$ minutes; $\xi = 30\%$ No Redirection Heuristic Redirection (HR) Optimised Redirection (OR) Capacity Reallocation + HR Capacity Reallocation + OR 100 75 Relative Delay [%] 50 42.2 25 20.2 n - 0 35 70 100 Compliance Rate (η) · Information Rate (ζ) [%] (f) $U_a \subset U$; $t_i = 90$ minutes; $\xi = 30\%$

100

Figure 5.7: Overall relative delays of $U_a \subset U$ with $\xi > 0\%$ in Scenario LS ($\psi = -1$) 112



Figure 5.8: Overall relative delays of $U_a \subset U$ and U with $\eta \cdot \zeta$ in 10% steps with $t_i = 60$ minutes in Scenario LS ($\psi = -1$)



Figure 5.9: Overall relative and total delays of $U_a \subset U$ and U with $t_i = 30$ minutes and $\xi = 0\%$ in Scenario LR ($\psi = 0$)

As previously pointed out, the Scenario LR is a less disruptive disposition measure than LS. This is especially true if the additional travel time on the rerouted PT lines is not considerably prolonged, as is the case here with an additional travel time of about 3.5 minutes for the red line between Nodes 8 and 10. The yellow line skips Stop 8 in the scenario of the rerouting which means that passengers cannot reach this node by using the yellow line. The delay of the buses of this line is 3 minutes between Nodes 10 and 15. Therefore, a shorter recovery time is to be expected which is why for Scenario LR results with $\psi = 0$ are presented.

The total delays in Figure 5.9 support the statement that line-rerouting is less disruptive



Figure 5.10: Overall relative and total delays of $U_a \subset U$ and U with $t_i = 60$ minutes and $\xi = 0\%$ in Scenario LR ($\psi = 0$)

than line-splitting. Whereas in Scenario LS, an incident duration of 30 minutes causes a total delay of all passengers of 1097.1 hours (Figure 5.4d), in Scenario LR, however, it causes a total delay of 125.3 hours (Figure 5.9c). Furthermore, the difference between the total delays of all passengers (U: 25,486, Figure 5.9d) and the delays of the affected passengers ($U_a \subset U$: approximately 1,100, Figure 5.9c) is noticeably lower than in the Scenario LS (Figures 5.4d and 5.4c). Since the rerouting is less disruptive for the affected passengers $U_a \subset U$, it is also less disruptive for the indirectly affected passengers $U_{ida} \subset U$. Looking at the results in Scenario LR with incident durations of 60 and 90 minutes (Figures 5.10 and 5.11), it is

evident that the effectiveness of REPACK rises with an increase of the incident duration, as in Scenario LS. Therefore, even in situations with less disruptive disposition measures such as line-rerouting, REPACK becomes more reasonable with longer incident durations.

The relation between the total delays of all passengers U and of the affected passengers $U_a \subset U$ shows a similar development as in Scenario LS, with an increase in the incident duration, the influence of the delays of the indirectly affected passengers ($U_{ida} \subset U$) shrinks. In case of $t_i = 30$ minutes the delays of the affected passengers amounts to 64% of all passengers, with $t_i = 60$ minutes to 78% and with $t_i = 90$ minutes to 82%. Again this effect shrinks with an increase of the incident duration, here, more obvious than in Scenario LS.

Another interesting observation is that only CR+HR, but not CR+OR, is executed in the cases with $t_i = 30$ minutes and that neither CR+HR nor CR+OR is executed in any other of the cases with a self-redirection rate ξ of 0%. This is probably due to the same reasons which have been discussed before concerning the execution criteria of the capacity reallocation process. Furthermore, even though CR+HR performs best from the perspective of the affected passengers, it worsens the situation of all passengers. Hence, the disadvantages for the line from which the reallocated buses have been withdrawn seem to be higher than the advantages for the supported line.

The results with a self-redirection rate of $\xi > 0\%$ show that the increase in the self-redirection rate does not have a significant effect on the relative passenger delay in Scenario LR which again can be explained by the less disruptive nature of this scenario, as well as with by the fact that for most affected passengers, the rerouted lines might still be the best alternative path which possesses the same capacity as in the undisrupted scenario. This fact might lead to most of the affected passengers being redirected onto them. Hence, the difference between self-redirected travellers and travellers redirected by given path advice is lower than in Scenario LS.

Among these cases, there is only one in which an approach with capacity reallocation is executed. With $\xi = 10\%$ and $t_i = 30$ minutes, CR+HR performs best and accomplishes a delay reduction of 22.3% in the case with $\eta \cdot \zeta = 100\%$. Through the self-redirected travellers who occupy room on the fastest alternative paths, the positive effects of CR+HR seem to outweigh its disadvantages better than with $\xi = 0\%$, which means no self-redirected PT users. At least for the affected passengers, whereas for all passengers, Approach HR performs best. This indicates that also the level of severity and taken disposition measures might be a support to decide on which redirection approach to use. However, the differences between the results of Approach HR compared to OR are insignificant in most cases, in Scenario LR.

The presentation of the relative delays in 10% steps shows a relatively linear development with respect to an increase of $\eta \cdot \zeta$, as in Scenario LS. However, here, the trend line is flatter, which fits well with the previously presented results. As argued before, this makes the execution of REPACK less reasonable in case of a less disruptive disposition measure, especially with $\eta \cdot \zeta < 40\%$, for which cases the delay reduction is less than 10%.

In summary, REPACK demonstrates a significant positive effect on the overall relative and

total delay of affected passengers with a relative delay reduction of up to 90% under ideal circumstances. However, the results also show certain factors that influence REPACK's performance perceivably such as the severity and duration of an incident, as well as the level of disruptiveness of the disposition measures. The product of compliance and information rates also shows significant influence on the results with a somewhat linear relation. Moreover, a rising number of self-redirected passengers decreases the effectiveness of REPACK.



Figure 5.11: Overall relative and total delays of $U_a \subset U$ and U with $t_i = 90$ minutes and $\xi = 0\%$ in Scenario LR ($\psi = 0$)





Figure 5.12: Overall relative delays of $U_a \subset U$ with $\xi > 0\%$ in Scenario LR ($\psi = 0$) 119



Figure 5.13: Overall relative delays of $U_a \subset U$ and U with $\eta \cdot \zeta$ in 10% steps with $t_i = 60$ minutes in Scenario LR ($\psi = 0$)

Relative Waiting Time

In the following the waiting time in dependence of the product of compliance and information rates is presented in order to evaluate the effect of REPACK on the waiting time of passengers (Figure 5.14). The waiting time is a part of the travel time of each PT user. As explained in Section 2.1.2 the waiting time is less convenient than the riding or in-vehicle time of a passenger. It is the accumulated time during which travellers are waiting at stops for the PT service they intend to take. For the affected passengers $U_a \subset U$ (left side), the waiting time similarly decreases compared to the relative delay. For all passengers U (right side), the relative delay decreases more rapidly than the relative waiting time. Therefore, the waiting time decreases more quickly for the affected passengers. This indicates that REPACK also caters for an advantage in waiting time for the redirected passengers. This reduces their perceived travel time even quicker than the actual travel time (Section 2.1.2).



Figure 5.14: Overall relative waiting time and delay of $U_a \subset U$ and U with $\eta \cdot \zeta$ in 10% steps in Scenario LS with $t_i = 60$ minutes ($\psi = -1$)

As for Scenario LR, the relative waiting time drops quicker for the affected $U_a \subset U$ (left side) than for all passengers U (right side). In general, the development of the waiting time in this scenario fits well to the development of the relative delay and the fact that the line-rerouting is less disruptive than the line-splitting (Figure 5.15a).



Figure 5.15: Overall relative waiting time and delay of $U_a \subset U$ and U with $\eta \cdot \zeta$ in 10% steps in Scenario LR with $t_i = 60$ minutes ($\psi = 0$)

Delay Distribution

In addition to the analysis of the relative and total delay summed up over all and the affected passengers, this section presents the delay distribution of the affected $U_a \subset U$ and redirected passengers $U_{red} \subset U$ to give insight into how REPACK influences this aspect of passenger delays. Figure 5.16 shows the delay distribution of the affected passengers, again with $\psi = -1$. The figures on the left side are the cases with a self-redirection rate ξ of 0% and on the right side with $\xi = 30\%$. The two figures at the top have an incident duration of 30 minutes and at the bottom of 90 minutes.



Figure 5.16: Delay distribution of $U_a \subset U$ in Scenario LS ($\psi = -1$)

For these results box plots are used. The bottom border of the box represents the 25th percentile (first quartile) of the passengers, and the top border delineates the 75th percentile (third quartile). The horizontal line within each box is the median (i.e. the 50th percentile, second quartile). The bottom whisker describes the 0th percentile (zeroth quartile) and the top whisker the 100th percentile (fourth quartile). The data points outside the whiskers represent outliers. To define outliers, here, the 1.5IQR rule is used. The inter quartile range (IQR) is the range from the first to the third quartile. If a value lies beyond the product of 1.5 times IQR plus the third quartile (75th percentile) or minus the first quartile (25th percentile), the value is defined as an outlier [DAWSON, 2011; WIKIPEDIA, 2023].

Generally, it is apparent that a reduction in the number of waiting travellers also reduces the distribution of passenger delays. This goes along with an increase in the number of outliers that are above the top whisker. The more redirected travellers, the fewer waiting travellers due to the absence of available alternative paths or due to their late arrival at the origin of the redirection after the redirection duration elapsed (Section 4.5). However, if there is no available alternative path for a certain OD pair, the corresponding passenger delay is rather severe, compared to the delay of the redirected passengers.

For the cases with $\xi = 0\%$ and $\eta \cdot \zeta = 35\%$, the IQR is the largest (Figures 5.16a and 5.16c). This can be explained by the fact that in these cases the number of waiting people is higher than that of the redirected passengers. Furthermore, the waiting travellers include those who arrive at the incident site not just shortly before the dissolution of the incident, but also during its entire duration. With a rising product of compliance and information rates, the number of these long-waiting passengers decreases and consequently does the IQR. However, it also raises the number of outliers who are most likely these few still present long waiting passengers for whom no alternative path is available. The case in which $t_i = 90$ minutes shows that the optimised redirection performs best in this matter. Since the more severe delay of long waiting travellers can perceivably increase the average overall delay of all affected passengers, the optimisation rather redirects these passengers than a bigger group that faces a shorter delay.

The results of the cases with $\xi = 30\%$ show a similar picture. Nonetheless, in these cases, the distribution starts to decline even with $\eta \cdot \zeta = 0\%$, which shows the largest IQR in these cases, even with a product of compliance and information rates of 0%. This can be explained through the 30% of the affected passengers who redirect themselves and thereby stretch the deviation of passenger delays. Interestingly, in these cases, the optimised redirection does not show a smaller range of outliers. The reason for this might be the fact that those passengers who redirect themselves occupy relevant space that outliers might take when redirected.

With an increase of the incident duration, the development of the results from the cases with $\eta \cdot \zeta = 0\%$ to 100% becomes slightly more obvious, but its course is similar.

Another striking observation that can be made is the fact that some travellers show a negative delay which means that their travel time is shorter than in the lower benchmark scenario without an incident. The zero line runs between first and second quartile for $\eta \cdot \zeta >=$ 35% with $\xi = 0\%$ and for $\eta \cdot \zeta >= 0\%$ with $\xi = 30\%$. For $\eta \cdot \zeta = 100\%$ this effect shrinks, which means that for some of the affected passengers it is best if not all $u \in U_a$ follow path advice or are informed. This indicates that through passenger redirection that causes a



Figure 5.17: Delay distribution of $U_{red} \subset U$ and $U_a \subset U$ in Scenario LS ($\psi = -1$)

redistribution of passengers within the PT network, for some of the affected passengers the situation is even better than without an incident in terms of their travel time.

The Braess paradox could also be an explanation for this phenomenon. This paradox expresses that a disturbed system sometimes performs better than the undisturbed [AMELI et al., 2020]. In contrast to this, however, is the fact that the disrupted PT system without passenger redirection does not show this phenomenon, or at most for individual passengers.

To compare the delay distribution of the affected passengers $U_a \subset U$ to the delay distribution of the redirected passengers $U_{red} \subset U$, in Figure 5.17 the redirected passengers' delay distribution with $\xi = 0\%$ and incident durations of 30 and 90 minutes is presented in comparison to the results of the affected PT users $U_a \subset U$. The figures on the left show the results for $U_{red} \subset U$ and the ones on the right for $U_a \subset U$. These figures help to explain the right shift of the median of the delay distribution of the affected passengers, as for the redirected passengers the IQR is lower and with an increasing product of compliance and information rates from left to right, the delay distribution of the redirected passengers moves closer and closer to the distribution of the affected passengers. This is logical, as with an increase in the number of redirected passengers, the number of waiting passengers decreases and with $\eta \cdot \zeta = 100\%$, most of the affected passengers are redirected.

For Scenario LR, the delay distribution of the affected passengers looks noticeably different compared to Scenario LS (Figure 5.18a). The negative consequences of the line-rerouting are significantly lower in terms of delays for most of the affected passengers compared to the line-splitting. This can also be seen in the delay distributions. Therefore, the outliers extend the range of the occurred delays greatly through which the boxes shrink to an almost unreadable size. Therefore, the outliers are removed here. Some of these diagrams with outliers are to be found in Annexe C.

Due to the reduced overall delays even without passenger redirection, the differences between the various cases with a range of used products of compliance and information rates are also smaller than in Scenario LS. The median only drops minimally in the cases with $\eta \cdot \zeta > 35\%$. However, this can only be seen through the position of the median line in the diagram, since the change occurs exclusively in the decimal range without any change to the rounded value depicted in the diagram. It is noticeable that the optimised redirection has a higher IQR in the cases with $\eta \cdot \zeta = 100\%$. This is especially true for the cases with $\xi = 0\%$. Again, it might be the case that through the less precise calculation of the alternative paths' remaining capacities, the redirection worsens the situation for a few passengers. This also fits the results with $\xi = 30\%$, for which this occurs less often as more passengers take their fastest alternative path and do not follow the path advice. Generally speaking, the delay distribution in Scenario LR is insignificant, which is not surprising, as the difference in additional travel time of the reouted lines is 3.5 minutes.

For the same reason, there is almost no difference in the delay distribution of the redirected passengers $U_{red} \subset U$, which is presented in Figure 5.19 in comparison to the results of the affected passengers $U_a \subset U$. The results for the redirected PT users are on the left side, whereas the results for the affected PT users are on the right side.



Figure 5.18: Delay distribution of $U_a \subset U$ in Scenario LR ($\psi = 0$)

5.2 Implementation in the Mandl Network



Figure 5.19: Delay distribution of $U_{red} \subset U$ and $U_a \subset U$ in Scenario LR ($\psi = 0$)

Delay per Departure Time

The total and relative delay as well as the delay distribution grants insight into the delays over the whole simulation period of each case study. However, those results do not cast any light on the development of the passenger delay over time. Therefore, Figures 5.20a to 5.20d show the average delay of all passengers per time interval. The delays are averaged in 10-minutes steps of the passengers' departure times. The red bar at the bottom of the diagram shows the incident duration and the violet bar shows the redirection duration. Since the redirection duration varies for the redirection groups of the different affected OD pairs depending on the alternative path they are redirected onto (Section 4.5), the end of the violet bar is determined by the mean of the departure times of the respective last redirected passenger in each case over all five random seeds.

The figures demonstrate that the course of the development of the passenger delays is similar between all the conducted approaches. However, an obvious gap exists between the case with no redirection and the ones in which passengers have been redirected. Logically, this gap is bigger in the cases with a product of compliance and information rates of 100% (Figures 5.20b and 5.20d) than in the cases with a rate of 35% (Figures 5.20a and 5.20c). As in these cases the delay reduction is more substantial due to the larger number of passengers that are redirected. Furthermore, the more travellers follow the path advice, the quicker the delay decrease at the end and shortly after the incident. For the cases with no redirection the delays fall slowly and go far beyond the end of the incident which makes a negative value for ψ , which prolongs the redirection duration, reasonable.

Through a negative value of ψ the redirection duration T_{od}^p is prolonged by $|\psi| \cdot \check{h}_p$. With $\psi = -1$ this still caters for a redirection duration shorter than the incident duration. Due to the fact that the rest of Equation 4.2, to calculate T_{od}^p , mainly depends on the travel time difference between the fastest alternative path $\hat{p} \in P_{od}$ and the assigned alternative path $\tilde{p} \in P_{od}$ of an affected OD pair $od \in OD_a$. This also explains the fact that the redirection durations in the two Scenarios LS and LR are nearly the same at about 57 minutes. Even though the value for factor ψ is set differently (-1 versus 0). This is because the differences in travel time between $\hat{p} \in P_{od}$ and $\tilde{p} \in P_{od}$ are likely to be greater in the more disruptive Scenario LS than in Scenario LR.

The approaches with the capacity reallocation are only executed in the cases with $\eta \cdot \zeta = 100\%$. As in the results for the relative delays the optimised redirection performs best.

Figure 5.21 fits well with the previously presented results of the relative and total delays as well as the delay distributions in Scenario LR. The gaps between the case with no redirection and the cases with a passenger redirection are perceptible but insignificant. The delays end with the end of the incident, even in the case of $t_i = 90$ minutes. This makes a negative value for ψ unreasonable in Scenario LR as there is no or an insignificant recovery time. This supports the decision for a value of $\psi = 0$.







Figure 5.21: Average delays by departure time of $u \in U$ with $\xi = 0\%$ in Scenario LR ($\psi = 0$)

PT User Crowding at Stops

As explained in Section 5.2.2, REPACK also reduces the waiting time of affected passengers. This can also lead to a decrease in passenger crowding at stops, as affected travellers wait for a shorter duration at a stop before they can board a PT vehicle. For this, the six stops which are directly before and after the incident site at Nodes 8 and 10 in the Mandl network have been picked for this investigation, as the crowding is likely to be most severe at these stops.



Figure 5.22: Stop setup at Nodes 8 and 10 in Scenario LS

Figures 5.22 and 5.23 show the location of the investigated stops concerning their position in the network. As described in Section 4.3, each node in the network represents a stop area that contains at least two stops (i.e. one for each line direction at the opposite sides of a street or respective link). At junctions with several links where different PT lines meet, this number can be even higher. As Figures 5.22 and 5.23 show, Node 8 includes two stops, whereas Node 10 includes four. The coloured arrows show how the PT lines that serve these nodes approach and leave a stop area and which stop is served by which line and by which direction of a line. Figure 5.22 shows the setup in Scenario LS and Figure 5.23 in Scenario LR.

Passengers are able to walk between the stops of a stop area which is not depicted here to avoid confusion through overlapping lines. However, this is shown in Figure 4.3. In Scenario LS, the split lines serve two stops of each stop area before the vehicles turn around to drive along the line route in the opposite direction. In Scenario LR, only one stop is served. Which stop depends on the direction of the line route.



Figure 5.23: Stop setup at Nodes 8 and 10 in Scenario LR

Figures 5.24a to 5.24d show the number of passengers at the chosen stops over time. The left side shows Scenario LS and the right side Scenario LR. The upper two figures show the corresponding cases with no redirection, the middle two with the heuristic redirection approach, and the bottom two with the optimised redirection approach. The product of compliance and information rates is 100% in these results, the self-redirection rate 0%, $\psi = -1$ for Scenario LS, and $\psi = 0$ for Scenario LR. The incident duration is 60 minutes. These results are taken from the simulation runs are performed for a single random seed.

Each data point in the diagrams indicates a bus stopping at the respective stop, its departure time determines the x-value of the data point, whereas the number of passengers who have been waiting at that time at the stop defines the y-value. The orange and brown lines describe the crowding at the stops of Node 8, and the blue and green lines show the crowding at the stops of Node 10. The relatively high number of people waiting at the stops at the beginning of the scale can be explained by the fact that the passengers start to enter the network right at the beginning of the simulations, whereas the PT service starts later with different departure times for each PT line and direction. Once the PT service starts, the number of passengers at stops drops rapidly.

The crowding effects of the incidents underline the previous statement of the stronger severity of Scenario LS compared to Scenario LR. As in the latter, the service is not turning around, but keeps serving the same stops, just with an additional travel time of 3.5 minutes, the crowding is limited, even without passenger redirection. In Scenario LS the crowding increases quickly at Stops s10_3 and s8_2. These are the last two stops before the PT service
would enter the edge which is disrupted due to the assumed incident. Even after the incident, it takes time until the crowding decreases as travellers continuously enter the network and the number of waiting PT users at the stops exceeds the capacity of one bus of 100 passengers. These observations also fit with the results from the previous section about the delays per time interval, as in the case with no redirection, the delays decay slowly and far beyond the end of the incident.

Since the split lines serve the rest of their routes, the Stops s8 1 and s10 4, as well as the stops served by the not disrupted grey line, do not show a significant rise in the number of waiting passengers. After the dissolution of the incident, the crowding at Stop s8_1 increases. It is the stop behind the incident site. After the dissolution of the incident, passengers travel again along the edge between Nodes 8 and 10. The red and yellow lines start to pick up the strongly accumulated traveller demand at Stop s10_3. Some of them get off at Stop s8_1 to transfer from the red line to the grey or yellow line, or from the yellow to the grey or red line. As it can be seen in Figure 5.2b a lot of passengers are travelling from Node 8 to Node 6. If however, passengers travel on the yellow line between Nodes 8 and 10, they transfer to the grey or red line at Stop s8_1. This rising number of passengers at this stop will likely to continue until the accumulated traveller demand at Stops s10_3 is completely dissipated. As the accumulated demand at Node 8 is lower, so is the crowding effect at Stop s10_4. Furthermore, this crowding development after the dissolution depends on the OD relations. As most of the passengers who travel from Node 8 to 10 have Node 10 or 11 as their destination (Figure 5.2b), they do not transfer at Node 10, hence, the low crowding at Stop s10_4 compared to Stop s8 1.

Comparing the passenger crowding in the case of HR to the case of OR in Scenario LS, it is noticeable that the crowding a Stop s8_1 is more severe during the incident in the case of HR. Hence, more passengers transfer at this stop during the heuristic redirection than in the optimised redirection which might be an indication that the optimised redirection redirects more passengers to the cyan line (Figure 5.3a) than Approach HR.

In Scenario LR, the effects of the incident on the crowding levels of the depicted stops are significantly lower. Only Stop s8_1 shows a small rise after the end of the incident. As previously implied by the results from Scenario LS, this stop is used by some passengers to transfer to another line (Figure 5.24b). This peak is not to be found in the results from the case with passenger redirection, shown in Figure 5.24d. So, even though the crowding levels are already very low during the case with no redirection they are even lower in the case with passenger redirection in Scenario LR. These observations again fit well with the results of the delays per time interval in the previous section, in which Scenario LR also shows insignificant delays during and no delays right after the incident. Comparing the case of HR to OR in this scenario, there are no noticeable differences as the crowding in general in Scenario LR is insignificant.



Figure 5.24: Passenger crowding at Nodes 8 and 10 with $t_i = 60$ minutes in both scenarios 136



Figure 5.25: Passenger crowding at Nodes 8 and 10 with $t_i = 60$ minutes in Scenarios LS and LR

Regarding the Approaches CR+HR and CR+OR, Figure 5.25 shows the crowding for these approaches at the investigated stops. However, only for Scenario LS as the capacity reallocation has not been executed in Scenario LR for the set of parameters used here. There are no noticeable differences in the stop-crowding levels between the two approaches with capacity reallocation. This indicates that in these cases, the noticeable differences in the results between the heuristic and optimised redirection might be eliminated through the capacity reallocation process, which uses a similar heuristic procedure as the heuristic redirection (Section 4.6).

Compared to the approaches without capacity reallocation, it is noticeable that the levels of some of the crowding lines change. This is explainable by the reallocated buses. The orange line which shows the passenger crowding at Stop s8_1 is higher with Approach HR than with Approaches CR+HR and CR+OR. In Scenario LS, buses have been reallocated onto the grey line. These cater for extra pick-ups at Stop s8_1, which lowers the crowding line accordingly. The comparison of Approaches CR+HR and CR+OR with the optimised redirection (OR) is more complex. Whereas the levels of the light green and light blue crowding lines for Stops s10_1 and s10_3 decrease, the orange crowding line for Stop s8_1 increases. As the capacity reallocation process redirects the passengers similarly to the heuristic redirection, more passengers transfer at Stop s8_1 as in the heuristic redirection, hence, the increase of the level of the orange line. Due to the same reason, the light blue line also develops in the same way as in the heuristic redirection. However, the lower level of the light green line might be caused by reallocated buses serving the grey line. Therefore, these results indicate a positive effect of the Approaches CR+HR and CR+OR on the crowding levels at stops which is logical due to the extra runs.

Nevertheless, what these diagrams do not show is how the reallocation of capacities might

increase the crowding at the stops of the lines from which the reallocated buses are withdrawn. As shown in previous results, the approaches with a capacity reallocation only lead in a few cases to stronger reductions of passenger delays than the Approaches HR and OR without capacity reallocation, which indicates an increase in passenger crowding at those stops.

On the whole, it can be said that REPACK has a noticeable positive effect on the crowding levels at stops. This effect becomes more significant with an increase in the severity of an incident. With an increase in the self-redirection rate, a decrease in crowding at stops is likely, whereas, with an increase in the incident duration, a rising in the crowding effect is to be expected as more travellers are affected for a longer period. This leads to more people waiting at the stops for longer durations.

PT Line Occupancy

This section presents the effects of REPACK on the occupancy of the different operating PT lines. Again, $\psi = -1$ for the results of Scenario LS and $\psi = 0$ for the results in Scenario LR. A self-redirection rate of $\xi = 0\%$ is set. Furthermore, a product of compliance and information rates of 100% is used. The incident durations of 30 and 90 minutes are considered to also see an effect of the increase in the incident duration. Figures 5.26a and 5.26c on the left show the results for Scenario LS and Figures 5.26b and 5.26d on the right show the results for Scenario LR. The upper two show the cases with an incident duration of 30 minutes, and the lower two with $t_i = 90$ minutes. The first letter of the line identifications refers to the line colour (e.g. Y: yellow line), and the second letter refers to the direction of the corresponding line route (N: northward; S: southward; E: eastward; W: westward). The lines which have been split or rerouted are marked with a star (*). For the evaluation of the line occupancy, the period of the incident has been chosen in correspondence to the arrival of the PT vehicle. Hence, the results of the red and yellow line without a disposition measure depict buses that already passed the incident site, but have not yet arrived at their destination at the time when the incident occurs.

In Scenario LS the different approaches show a similar picture regarding the occupancy of the different PT lines. The most noticeable differences are to be seen in a shift towards the split red and yellow lines as well as the grey line when comparing the no redirection case with the redirection and reallocation approaches. This shift is more obvious in the case with an incident duration of 90 minutes. Here, there is also a shift towards the cyan line. The cyan and the grey line are the two obvious choices if one wants to travel around the incident site in Scenario LS. The line occupancies indicate that for shorter incident durations, most redirected passengers use the grey line. For a longer incident duration, the capacity of the grey line might not fit all redirection groups for which the cyan line is also increasingly used.

In Scenario LR, more people travel with the rerouted red and yellow line. As argued before, in this scenario, this is likely to still be the fastest path for some or even most of the affected passengers. Interestingly, in the case with an incident duration of 30 minutes, when comparing the no-redirection case with the passenger-redirection and capacity-reallocation approaches, the occupancy of the rerouted southward yellow line is reduced whereas the occupancy of the



Figure 5.26: PT line occupancies in Scenarios LS and LR

rerouted northward yellow line increases. Logically, this cannot be due to the same passengers. Therefore, it might be the case that affected passengers travelling southward are redirected onto other lines than the yellow line, whereas affected passengers travelling northward are redirected onto the yellow line. This phenomenon can also be observed in the case with an incident duration of 90 minutes, however, as in Scenario LS, here, there is also a shift towards the grey line.

By and large, the changes in the line occupancy are most significant between the noredirection cases and the other cases with passenger redirection. The size of the network and the number of lines, which limits the number of alternative travel options, are possible reasons. Nevertheless, the results show a noticeable shift and point out some lines which are of importance to the passenger redirection, such as the grey line, for instance. The capacity reallocation approaches do not show a significantly different picture of the line occupancy than the redirection approaches. However, there are some peaks in the Scenario LR in the case with an incident duration of 30 minutes of Approach CR+OR which might be an indication of vehicles being withdrawn. In Scenario LS most reallocated vehicles have been withdrawn from the violet line and dispatched to the grey line, whereas for Scenario LR, most reallocated vehicles have been withdrawn from the cyan and violet lines to support the red line.

Special Cases with Limited Information Availability

In addition to the cases which have been previously presented in this section, special cases for each scenario – LS and LR – are conducted in which the information rate ζ is dependent on the size of the redirection group on the fastest alternative path of each affected OD pair $od \in OD_a (rg_{od}^{\hat{p}})$ (Section 4.5). Only a limited number of OD pairs has an information rate of 100%.

Since dynamic PI displays do not have unlimited space to disseminate information and speaker announcements cannot continuously broadcast information, as it would rather confuse than help, only a limited number of OD pairs can be informed via the collective PI channels. This is unfortunate, as these channels are preferred by the survey participants for receiving information during incidents (Figure 3.7). It is reasonable that the OD pairs with the biggest redirection groups $rg_{od}^{\hat{p}}$ are chosen, as in this way more people profit from the path advice. It is assumed that these OD pairs have an information rate of 100%. For the rest of the OD pairs, an information rate of 25% is considered, as this is about the share of survey participants who prefer social media, websites, and PT smartphone apps as PI channels during incidents. This is tested for the first five OD pairs with the largest $rg_{od}^{\hat{p}}$. The compliance rate η is set to 100%.

Figures 5.27a and 5.27c show the case in which only the aforementioned five OD pairs are completely reachable; Figures 5.27b and 5.27d show the case in which for all OD pairs $\zeta = 100\%$. The two figures on the top show the cases with $t_i = 30$ minutes and the two on the bottom with $t_i = 90$ minutes. The self-redirection rate ξ increases from left to right from 0% to 30%. The results express that for a disruptive event such as in Scenario LS, REPACK is also useful, even if only the five OD pairs with the largest redirection groups completely receive the path advice and of the redirection groups of the remaining OD pairs, only a quarter receives the path advice. However, this is for $\eta = 100\%$. For lower compliance rates, which are likely according to the results of the PT user survey presented in Section 3.2.3, the relative delay increases.

For Scenario LR this is not the case (Figure 5.28a). With a rising incident duration, the positive effect of REPACK on the relative passenger delay increases. In addition to the fact that a longer incident duration leads to more severe delays and therefore to stronger reductions in delays for the informed passengers. For example, in Scenario LS and Approach HR



Figure 5.27: Relative delays of $U_a \subset U$ comparing merely five $od \in OD_a$ to all $od \in OD_a$ informed with $\zeta = 100\%$ in Scenarios LS ($\psi = -1$)

with $t_i = 30$ minutes, the number of redirected passengers is 666, which is about 60.5% of the 1,100 affected passengers. With $t_i = 90$ minutes, the number of redirected passengers is about 1,971, which is 58% of the 3,410 affected passengers. Therefore, there is a higher chance that the remaining capacity on the fastest alternative paths is raised due to a reduced number of informed passengers. This improves the situation for the informed travellers. For comparison, in Scenario LS and heuristic passenger redirection with a product of compliance

and information rates of 100%, the number of redirected passengers is about 1,100 for $t_i = 30$ minutes and 3,357 for $t_i = 90$ minutes. Hence, in the former case, all affected passengers (1,100) are redirected and in the latter, almost all affected passengers (3,410) are redirected. Thus, the results are better for the cases with no limitations on the information availability.



(a) $U_a \subset U$; Scenario LS; $t_i = 30$ minutes; 5 OD



(c) $U_a \subset U$; Scenario LR; $t_i = 90$ minutes; 5 OD



(b) $U_a \subset U$; Scenario LS; $t_i = 30$ minutes; all OD



(d) $U_a \subset U$; Scenario LR; $t_i = 90$ minutes; all OD



5.2.3 Implementation in an Autonomous PT System

In the previous case study, all PT lines have been served by buses. In addition, in this section, the setup and results of a case study are described in which the PT lines are served by the aforementioned DART system. RAU, TIAN et al. [2018] and RAU, JAIN et al. [2019] introduce the setup and properties of the DART system. It is an autonomous and connected PT system with a modular setup. Instead of buses, PT lines are served by autonomous bus modules which offer a capacity of 30 passengers. The general idea behind the system is that the timetable is designed in a way that single modules serve different feeder lines and meet at the beginning of a common section, a trunk line, to couple and serve this common section as a platoon of modules.

Platooning of vehicles describes the driving of several vehicles behind each other with small intra-platoon gaps between the vehicles within a platoon. Through vehicle-to-vehicle (V2V) communication, vehicles are capable of sharing certain information such as location, speed, and acceleration with each other. This enables them to keep a smaller gap between each other, compared to mandatory safety gaps which need to be kept by human drivers.

The platooning of trucks on highways is an especially well-researched area in literature. Two investigated advantages of truck platooning are energy and space savings through the reduced gap size between the vehicles and the resulting improved slipstream effect [BACHMANN, 2017; BACHMANN and SETHURAMAN, 2019; LARSEN et al., 2019].

In addition, in recent years more and more investigations regarding platooning in urban traffic have been conducted. One finding, for instance, is the advantages concerning the increased throughput of vehicles as well as advantages for traffic light signalling as an electronically coupled platoon needs less time to cross an intersection than the members of such a platoon would need as individually driving vehicles [LIORIS et al., 2017; Y. WU et al., 2022].

Moreover, there also have been investigations regarding platooning in PT systems in recent years. CAO et al. [2022], for instance, introduce a coupling and decoupling process of PT vehicles and its integration in the PT schedule planning process. This is also a concept used in the DART system. SETHURAMAN et al. [2019] conclude that platoons of DART modules cater for energy savings as well as an increase in the effectiveness of PT prioritisation at signalised intersections. This is argued through the fact that coupled modules arrive together at intersections which reduces the number of arrivals of PT vehicles at intersections.

In addition to the aforementioned advantages of platooning concerning energy and space savings as well as the improvement of traffic flow, BACHMANN, RAU et al. [2021] imply that platooning of DART modules can also improve capacity reallocation during incidents, which has been tested in a numerical example. The conceptional framework of the capacity reallocation process by BACHMANN, RAU et al. [2021] has been further developed into an applicable model and is presented in Section 4.6. It is used here to implement the DART system into a simulation study in the Mandl network.

Case Study Setup in the DART System

Instead of 30 passengers per module, a capacity of 33 passengers is assumed here and the lines are served by platoons of three DART modules. In this way, a platoon has about the same capacity as one bus in the previous case study (99 versus 100 passengers). This improves the comparability between both PT systems. Due to the modular setup of the DART system, a capacity reallocation can be undertaken here by just reallocating one or two modules to the PT lines with insufficient capacity, instead of the whole platoon. Therefore, in contrast to the process in a PT bus system as described in Section 4.6 when searching the PT system for dispensable capacities, it is not checked whether the following platoon can take the passengers of the previous one to then reallocate the complete platoon of three modules. Instead, it is checked if the occupancy of a platoon is below one-third or two-thirds of its capacity of 99 passengers. In the former case, two of the three modules, which form a platoon, can be reallocated. In the latter case, one module can be reallocated. In this way, the decay of service quality through the reallocation on the line, from which the modules are withdrawn, is smaller compared to the previous studies, in which whole buses have been withdrawn. This is due to the fact that even though the capacity is reduced, the headway is not extended as still one module or a platoon of two modules serves their original line.

Please note that the dynamics of platooning have not been implemented in SUMO. Instead, modules and platoons are represented by buses of corresponding size and capacity. Whole platoons of three modules have the same size as the buses in the previous study, but with a capacity of 99 passengers. If dispensable capacity is identified and matched with an affected OD pair $od \in OD_a$, such a bus, representing a whole platoon of three modules, is reduced to one-third or two-thirds of its size as well as its capacity (i.e. 33 or 66 passengers), depending on whether a platoon can spare one-third or two-thirds of its capacity. In addition, a second vehicle is inserted in the simulation with a corresponding size and capacity, representing the decoupled module or modules which are reallocated. As in the previous case study, for this procedure, SUMO's API TraCI has been used [DLR, 2022]. It is assumed that the dynamics of platooning as the coupling and decoupling process do not have a significant influence on the passengers' travel time and that it can therefore be represented in this way without significant effects on the passenger delays.

Unfortunately, for $\psi = -1$ in Scenario LS and $\psi = 0$ in Scenario LR, the capacity reallocation is not executed in the case study with the DART system. Therefore, to be able to compare the two PT systems concerning the capacity reallocation process, factor ψ has been set here to -2 in both scenarios – LS and LR – through which the capacity reallocation is executed. The redirection duration T_{od}^p is thereby longer in the following case study than in the previous one (Equation 4.2). For Scenario LS, the study is done with an incident duration of 60 and 90 minutes and for Scenario LR with $t_i = 30$ minutes.

For these cases, a self-redirection rate ξ of 10% as well as products of compliance and information rates of 0%, 35%, 70%, and 100% are used. The timetable, demand, and disposition measures are as set as in the previous case study (Figure 5.29).



Figure 5.29: Mandl PT network served by the DART system with disposition measures

Results of the Case Study in the DART System

Due to the similar vehicle capacity and the same timetable and disposition measures, it can be expected that the results for the passenger redirection Approaches HR and OR are somewhat similar to those in the previous case study, in which the PT lines have been served by buses. Additionally, minimal or no differences can be expected regarding the total delay in the noredirection case. For the capacity reallocation approaches – CR+HR and CR+OR – however, an improvement in the results might be expected due to the improved flexibility in reallocating capacities due to the modular setup of the DART system. Still, these advantages are more likely to be seen in the results for all passengers than for the affected passengers, as the main advantage of the modular setup of the DART system is that the headway on the original lines of reallocated modules can be maintained even during the reallocation process.

As expected, the total delays in the no-redirection case show merely insignificant differences. For Scenario LS with $t_i = 60$ minutes and $\xi = 10\%$, the total delays in the bus system are 2075.1 hours, whereas they are 2093.6 hours in the DART system. For Scenario LR with $t_i = 30$ minutes and $\xi = 10\%$, the total delay of the affected passengers is 80.7 hours in the bus system and 81.2 hours in the DART system.

Figure 5.30 presents the relative delays in Scenario LS with $t_i = 60$ minutes and $\xi = 10\%$. The upper two figures show the relative delays of the affected passengers $U_a \subset U$, whereas the lower two pictures present the relative delays of all passengers U. On the left side there are the results of the case study in the DART system and on the right side there are the results of the case study in the bus system. Generally, the results develop similarly from $\eta \cdot \zeta = 0\%$ to $\eta \cdot \zeta = 100\%$ for both PT systems. With an increase in the product of compliance and information rates, the delay reduction improves. However, in contrast to the expectations, there are already visible differences between the results of the two systems concerning the



Figure 5.30: Overall relative delays in Scenario LS in a bus and the DART system with $t_i = 60$ minutes and $\xi = 10\%$ ($\psi = -2$)

passenger redirection approaches – HR and OR – without the capacity reallocation. The bus system performs better in all cases with $\eta\cdot\zeta<100\%$. It is rather doubtful that this difference can be explained by the reduced capacity of the DART modules by one place, but is rather due to simulation dynamics of random interactions between the travellers and the PT vehicles.

Another obvious observation is that for the DART system, both capacity reallocation approaches – CR+HR and CR+OR – are executed, whereas for the bus system, merely Approach CR+HR is done. However, in contrast to the expectations, the bus system performs

better in the cases with a capacity reallocation than the DART system. This is true for the affected as well as all passengers. Reasons for this can be that this kind of micromanagement of capacities is over-shooting the intentions of providing better service during incidents or that sending smaller vehicles with less capacity to the line with insufficient capacity does not makes sense if not all affected passengers at a stop can board the additional vehicle due to insufficient capacity. Another possible explanation for this might be the differences in the process of the identification of dispensable capacity as explained in the previous section. Through this, the timings of the DART modules might be different and less fitting than the timings of the buses. As discussed before in Section 5.2.2, when reallocated vehicles arrive shortly before or after a regularly scheduled bus of the supported PT line, the reallocation process might worsen the situation for the passengers rather than improve it. This effect might be stronger for the DART than for the PT bus system in these cases.

Interestingly, Approach OR performs best in DART system in the compared cases with a delay reduction of 89.4%, which again, can only be explained by the variances through simulation dynamics. Nevertheless, the differences between the results of both systems are rather insignificant and motivate further investigations to find out the reasons for the differences.

In Figure 5.31 the results for Scenario LR with $t_i = 30$ minutes and $\xi = 10\%$ are presented for both PT systems. The results for the DART system are again on the left side and for the bus system on the right side, with the relative delays of $U_a \subset U$ on top and of U on the bottom of this set of figures.

Here, the differences in the passenger redirection approaches – HR and OR – are less noticeable between the two PT systems compared to the results from Scenario LS. Furthermore, the DART system performs here slightly better than the bus system with $\eta \cdot \zeta < 100\%$. With full compliance and information availability, the bus system performs insignificantly better.

As in Scenario LS, Approach CR+HR is executed in both systems whereas CR+OR is only executed in the DART system. Here, the DART system accomplishes the best result among all cases with a relative delay of 77.0% for the affected passengers $U_a \subset U$ and 84.0% for all passengers U. This means that the circumstances for the advantage of the modular setup of the DART system have been right or at least better in this case. This invites further research to find out which circumstances need to be met for the DART system to be able to unfold its full potential during capacity reallocation, especially because the difference to the second best result in the case of the bus system with Approach HR, with a relative delay of 79.2% for the affected passengers and 86.2% for all passengers, is just insignificantly worse.

In summary, it can be said that the DART system performs similarly to the bus system during the capacity reallocation, in Scenario LS slightly worse, in Scenario LR slightly better. As the previous case study in Section 5.2.2 point out, the capacity reallocation process needs further research and adjustment to better fit its task of supporting passenger redirection during incidents. This is further discussed in Chapter 6.



Figure 5.31: Overall relative delays in Scenario LR in a bus and the DART system with $t_i = 30$ minutes and $\xi = 10\%$ ($\psi = -2$)

5.3 Implementation in the Network of the City of Kassel

In addition to the Mandl network, REPACK has also been implemented for the network of the city of Kassel, Germany, to test it in a more complex and more realistic network. Kassel is a big city with 207,622 inhabitants [CITY OF KASSEL, 2023]. The Kassel region special purpose association [Translated from German: Zweckverband Raum Kassel] (ZRK), of which the city administration of Kassel, the North Hessian transport association [Translated from German: Nordhessischer Verkehrsverbund] (NVV), and the Kassel transport company [Translated from

German: Kasseler Verkehrsgesellschaft] (KVG) are members, have provided a transport model of the operating area of the NVV. This area includes the northern half of the federal state of Hesse. As the focus of this thesis lies on urban PT systems, the city centre of Kassel has been extracted from the transport model. Since this model is implemented in PTV Visum, the used part of it was extracted using the "Subnetworkgenerator". This is a tool to cut out a part of a PTV Visum transport model [PTV GROUP, 2023].

However, the transport model was not directly convertible into a SUMO network, therefore, the "OSMWebWizard", which is a tool to extract networks from Open Street Map (OSM) and to save them as SUMO network, has been used to build the basis of the simulation model used here [DLR, 2022; OSM FOUNDATION, 2023]. Nevertheless, the transport model from Kassel contains the OD zones from which traffic originates and to which traffic flows. The geographical definition of these zones has been transferred to the created SUMO network. The transport model also contains OD matrices for pedestrian, cyclist, car, and truck demand, as well as PT user demand of 24 hours of a usual weekday.

If the demand has been assigned onto corresponding paths before the subnetwork has been extracted, the corresponding demand for the subnetwork can be exported from the transport model in PTV Visum. For this process, PTV Visum creates so-called cordon zones. These represent the origin and/or destination of OD relations which lie outside the created subnetwork to be able to also include this demand. Unfortunately, these do not contain geographical area information and have therefore not been transferred to the SUMO network. Nonetheless, since in the model used here, only urban PT services have been simulated, and demand beyond the borders of the subnetwork also uses regional and long-distance PT services, it would not have been possible to assign the external demand appropriately. The PT users who travel within the city is about 70% of the overall demand (including the cordon zones). Therefore, this demand has been increased accordingly, to compensate for the missing trips from outside the city centre.

Furthermore, the period of the day during which the demand is simulated has been chosen to be the four hours from 6 am to 10 am. This period is about the period of the morning peak during which the traffic demand is highest throughout the day [HEIMBÖCKL, 2005; KLEIN, 2021]. So, even though 4 hours are about 16.7% of the day, it has been assumed that about 35% of the daily PT users are travelling in this period. This leads to a demand of about 31,390 people. Since the focus of this work lies on PT, people who only use walking as a transport mode have been filtered out, which leaves a demand of about 25,300 PT users, depending on the random seed. The same number of random seeds as in the Mandl network have been used. To limit the complexity of this case study, cyclists, motorised individual traffic, and freight transport have not been considered, as it would have also caused the need for calibrating the traffic lights. These have been set to actuated signalling which means that they react to the traffic demand, which in this case are the PT vehicles.

The transport model of Kassel also contains PT line routes and timetables which have further been complemented by schedule information from KVG's website [KVG, 2023]. The city centre of Kassel that has been used here comprises a complex PT network which is served by 20 bus lines (10, 11, 12, 13, 14, 16, 17, 21, 25, 26, 27, 30, 32, 37, 38, 51, 52, 100, 110, 500), 7 tram lines (1, 3, 4, 5, 6, 7, 8) and 3 "RegioTram" lines (RT1, RT4, RT5) [KVG,

2023]. The "RegioTram" is a regional tram service that also serves suburban areas. However, in the transport model used here, only the parts of the lines which are in the simulated area are considered. The level of complexity is therefore a lot higher than the level of complexity of the Mandl network (Figure 5.2). Figure 5.32 shows a map of the simulated part of the city of Kassel with the PT services that have been taken into account here. The bus lines are depicted by cyan lines, the tram lines by violet lines, and the regional tram lines by yellow lines. A capacity of 100 passengers is assumed for the buses, a capacity of 180 passengers for the tram trains and a capacity of 220 passengers for the regional tram trains.



Figure 5.32: PT network of the city of Kassel

5.3.1 Case Study Setup in the Kassel Network

As has been discussed in Section 5.2.2, factor ψ , which influences the redirection duration T_{od}^p (Equation 4.2), is difficult to set probably and needs further research. Moreover, in the Kassel network, the headways of the lines vary between 10 minutes and a full hour, which increases the influence of ψ on T_{od}^p immensely. Therefore, to ensure that T_{od}^p is not set illogically, as

it could lead to PT users being redirected for a too-short or too-long period, ψ has been set here to 0. The incident duration t_i is 60 minutes.

The results from the case study in the Mandl network show that REPACK is more reasonable in situations with more disruptive disposition measures such as PT line-splitting than less disruptive such as PT line-rerouting. Therefore, for the case study in the Kassel network, a scenario with line-splitting is set. Since this disposition measure is more common in railway operations, as the rail network offers fewer diversion options than the more closely meshed road network (Section 2.2.1), the regional tram lines RT1 and RT4 have been split in this Scenario LS. A track-switch failure is assumed between the central station and the stop "Kirchditmold" as shown by the red cross on the map in Figure 5.32. The two lines can short-turn on both ends of the incident site but not serve the stops as scheduled during the incident. The two split lines RT1 and RT4 run with a headway of 60 minutes.

As said in Section 4.4, the implementation of the all-simple-paths algorithm, which has been used to create a set of all simple paths in the Mandl network [SRYHENI, 2022], is too complex for a network as large as the network of Kassel. The complexity of the algorithm is O(|n|!), where |n| is the number of nodes in the network. Therefore, to collect the necessary set of alternative paths for Kassel, a k-shortest path algorithm (Section 2.1.2) has been used [MEOOOW, 2022]. K indicates the number of paths that are supposed to be found for one OD relation, if available. This value has been set to 100, to ensure that enough alternative paths are found for the affected OD pairs. However, in a network as complex and as large as this network, even with k = 100, there might be paths that are not considered due to this limit on the number of alternative paths. Furthermore, instead of searching the network for all OD relations, only those have been searched which are needed for the affected OD pairs $od \in OD_a$. In this case, just about 200 relations have been searched, instead of over 200,000. Due to the five random seeds, some of the origins of redirection are different in the various simulation runs.

5.3.2 Results of the Case Study in the Kassel Network

The results of the case study conducted in the Kassel network show a smaller reduction of the delay of the affected and all passengers than the results from the case studies in the Mandl network. However, in case of a product of the information and compliance rates of 100%, the delays of the affected passenger can still be reduced by 38.5% for the affected and 50.9% for all passengers. As in the case studies in the Mandl network, the delay reduction grows with an increase in the product of compliance and information rates. One reason for the lower effect of REPACK might be the less central location of the incident than in the previous case studies. This caters for a lower level of severity of the incident's disturbances for example in terms of fewer affected passengers of about 180 passengers with $t_i = 60$ minutes. The previous case studies show that the more severe the incident, the better the effectiveness of REPACK. Furthermore, even though the incident is not severe for the system, it is severe for the affected passengers through the line-splitting of the PT lines RT1 and RT4. Hence, a negative value for factor ψ might lead to better results than $\psi = 0$ as it is set here. Nevertheless, this case also shows that REPACK has a positive effect on passenger delays if an incident is less

centrally located.

Another striking observation is that for all passengers U, the heuristic redirection approach (HR) performs significantly better than the optimised redirection (OR). This becomes continuously more obvious with a rising product of compliance and information rates. Moreover, it is a noticeable difference to the results of the case studies in the Mandl network in which the OR performed better than the HR with $\eta \cdot \zeta > 80\%$ (Section 5.2.2).

Another significant difference is that this observation is more true for all passengers rather than for the affected ones, meaning that HR shows a smaller negative effect on the indirectly affected passengers $U_{ida} \subset U$ than OR. This can be explained by the different procedures of calculating the remaining capacities when comparing Approach HR with OR (Section 4.5). Through a segment-wise calculation of the remaining capacity in Approach HR, which takes the travel time of the redirection groups (rg_{od}^p) on the alternative paths into account, the estimation of it is more precise than in Approach OR. In this approach, the incident duration is used as the period to calculate the remaining capacity. Due to the longer headways, this might be a bigger disadvantage in more complex scenarios with a larger range of headways. In this case, this probably leads to passengers being redirected to services with insufficient remaining capacity. This causes an increase in the delays of the indirectly affected passengers. Another possibility might be that the complexity of a network might be another decision support factor to decide whether Approach HR or OR is used for passenger redirection. This is further discussed in Section 6.1.

Moreover, the difference between the total delay of the affected passengers compared to all passengers is striking for the case with $\eta \cdot \zeta = 100\%$ in which the total delay of all passengers U is lower than for the affected passengers $U_a \subset U$ in the case of the heuristic redirection approach. This means that the redirection improves the situation for the unaffected $U_{una} \subset U$ and/or indirectly affected passengers $U_{ida} \subset U$ more than for $U_a \subset U$. The fact that this only happens in the case of the heuristic redirection (HR) but not in Approach OR might be again due to the differences in the calculation of the remaining capacity of alternative paths among these two approaches, which is more precise in the latter than in the former (Section 4.5).

Another explanation can be that REPACK needs further adjustments and development to be more suitable for an application in more complex and more realistic networks.

The representation of the PT service by the network graph as explained in Section 4.3 might be too simple for such a complex case. For example, traffic lights could lead to delays of the buses and a later arrival at stops which is not considered in the graph. A representation that takes the actual arrival times of PT services into account would be more precise. This becomes even more important when the demand for individual traffic modes such as cars and freight transport is considered. Additionally, some aspects such as calibrating the traffic lights and implementing PT prioritisation at traffic lights have not been taken into account here, which also probably influences the outcome. Moreover, implementing the PT service exactly as it is in practice likely influence the results. In this case study, only a simplified version of the actual timetable which considers the headways but not additional services during peak hours might be a reason for these results.



Figure 5.33: Overall relative and total passenger delays with an incident duration (t_i) of 60 minutes and $\xi = 0\%$ in Scenario LS ($\psi = 0$)

Furthermore, factor ψ and the consideration of the recovery time need more investigation and might play a bigger role in more complex networks than they do in more simple networks such as the Mandl network. In addition, the fact that not all possible alternative paths are considered to reduce the computational effort for calculating them might cause that better alternative paths are not taken into consideration.

These explanations might also be elements which influence the Approaches CR+HR and CR+OR, which have not been executed in the Kassel network. Furthermore, this can be due to the following reasons:

first, Kassel is served by bus, tram, and regional tram lines, however, for the reallocation of capacities, merely buses have been considered in the search for dispensable capacities. This is due to the fact that only buses have the necessary flexibility to be reallocated in the way proposed in Chapter 4. Furthermore, buses are not capable to run on tram tracks, at least if these are not integrated in the road. Therefore, only path segments with insufficient capacity which are served by bus lines have been considered for the deployment of dispensable buses. This reduces the possibilities to find matching dispensable capacities.

Second, the bigger network also means that the start and end of lines are more distant from each other, which means it takes buses longer to get from their original line to the lines with insufficient capacity. Moreover, it is less likely that they get there in time before the respective redirection duration elapses. If the reallocation of capacities during their run would be considered, it could lead to a higher chance for matches, especially if they also would just serve a certain segment of a line with insufficient capacity, which might be more applicable in practice. However, it also lifts the level of complexity of the capacity reallocation problem.

Third, in the Kassel network, headways vary between 10 minutes and a full hour. For relatively long headways of for example more than 10 minutes, depending on the network and situation, a capacity reallocation is probably not expedient, as the negative effects on passenger delays through the withdrawal of vehicles from their original line. Further, it makes it harder to find buses in the incident period. However, such situations might be the ones in which autonomous modular system such as the aforementioned DART system is able to make full use of its advantages. Through the modular setup a platoon of modules can be split and a line with insufficient capacity can be supported while the rest of the platoon can still serve the originally assigned line. In this case, the headway stays the same on the original line and also longer headways are not a problem. Hence, a bigger, denser, and more complex PT network might lead to better results for the DART system compared to the results presented in Section 5.2.3.

Nevertheless, REPACK caters for a significant reduction in the delays of the affected as well as all passengers and shows thereby a potential for an application in practice. Furthermore, the applicability of REPACK in a multimodal PT network is demonstrated.

5.4 Key Findings from the Case Studies

Firstly, the results imply that the application of REPACK is especially reasonable if the incident and the consequential disposition measures are sufficiently disruptive. This is because the results of the cases in Scenario LS in which the disrupted PT lines are split show a significantly stronger delay reduction than the results of the cases in Scenario LR in which the lines are rerouted.

Secondly, with an increase in the incident duration the effectiveness of REPACK grows significantly. Therefore, the incident duration is an important aspect of setting a minimum threshold for the execution of passenger redirection as well. BMVI [2019] presents the outcome of an approach in practice concerning passenger redirection. The consortium which conducted this investigation is mainly composed of practitioners. One finding of this work is the definition of a minimum incident duration of 30 minutes. This fits well with the results

of the previously presented case studies. However, shorter incident durations are not tested here or by any of the works presented in Section 2.2.2. Nevertheless, as already mentioned, an incident must have a certain minimum level of disturbance for the execution of REPACK to be meaningful. Even though each incident has its characteristics according to the outcome of the aforementioned observations in OCCs [BACHMANN, BRIEM et al., 2022], the definition of a generally valid minimum threshold for the execution of REPACK is appropriate and makes it more applicable in practice. In terms of the incident duration, 30 minutes seems to be a good choice as the minimum, however, the severity of the incident needs to be considered in this decision.

Thirdly, another factor that influences the effectiveness of REPACK is the availability of alternative paths and the travel time differences between these paths and the disrupted original path of affected passengers. If no alternative paths are available, passengers cannot be redirected. Therefore, the closer to the city centre an incident happens, where the PT network is close-meshed, the more likely the availability of alternative paths. The fact that the passenger delays are more severe in the case study in Kassel than in the case study in the Mandl network might also be caused by the less centrally located incident and fewer available alternative paths.

Furthermore, as demonstrated by the comparison of the two Scenarios LS and LR in the Mandl network (Section 5.2.2), the more severe the disturbances in terms of length of delays on the originally planned path of the affected passengers and the bigger the difference in travel time between the disturbed original path and the available alternative paths, the longer the redirection duration T_{od}^p (Equation 4.2). With a longer redirection duration, more passengers are redirected and profit from REPACK, hence, increasing its effectiveness. In Scenario LR, for instance, the PT lines are rerouted and the additional travel time due to the detour is just 3.5 minutes. Therefore, for less disruptive scenarios, passenger redirection might not be appropriate for shorter incident durations of less than 60 minutes. This indicates that the previously mentioned definition of a minimum incident duration should be set in accordance with the level of severity of an incident.

Fourthly, as previously mentioned, the difference in travel times between the disturbed original path and the alternative paths of the affected passengers also influences the redirection duration T_{od}^p ; the bigger the difference, the shorter the redirection duration. However, it is also influenced by the factor ψ . A positive value for it shortens T_{od}^p ; a negative value prolongs it (Equation 4.2). The previously presented results show that a negative value for ψ leads to better results with more severe incident situations such as in Scenario LS whereas for Scenario LR $\psi = 0$ is a good setting. Still, the presented results indicate that a deterministic value for ψ might be a too simple approach, as it needs to be set according to the recovery time. This is further discussed in the next chapter.

Lastly, the capacity reallocation process has merely been executed in cases with a product of compliance and information rates of 100%, which is a rather unlikely rate according to the outcome of the PT user survey presented in Chapter 3 (Figure 3.9). It might be that the reallocation process is only applicable in cases in which the capacity on the alternative paths is strongly exceeded or in which rail replacement services must be supported (Section 2.2). As pointed out by BACHMANN, BRIEM et al. [2022], finding drivers becomes continually more difficult for PT operators, which leads to the unavailability of standby and extra vehicles. If extra capacity is needed as presented here or in the form of rail replacement services to bridge disrupted railway services, it might be necessary to withdraw it from other lines. For such a situation, the introduced capacity reallocation process might be very suitable. However, as the results show, this process also gives room for improvement. This is discussed in the following chapter.

Chapter 6

Discussion

The previous chapter presents and explains the results of case studies conducted in the Mandl network as well as the network of the city of Kassel. The first section of this chapter elaborates on the limitations of the REPACK model introduced here, followed by a section underlining the contribution of this work and comparing it to the literature presented in Section 2.2.2. The final section discusses steps needed to implement REPACK in practice.

6.1 Model Limitations

"All models are wrong, but some are useful" [Box, 1979]. It is the author's conviction that the REPACK model is useful. However, as valid for all models, it has some limitations which are elaborated on and discussed in the following.

First, REPACK is not yet capable of adjusting to changes in the estimated incident duration. Whenever an incident occurs and is reported to the OCC, dispatchers estimate its duration. The return of the PT service to its schedule depends on various factors such as the severity of the incident, its location and closeness to a depot or standby services, the availability of staff and extra vehicles, and the traffic condition, just to name a few. Therefore, it is difficult to estimate an incident duration, which is why this estimation strongly relies on the experience of the dispatchers in OCCs, as pointed out by BACHMANN, BRIEM et al. [2022]. In any case, an incident can evolve differently than expected. Therefore, the capability of adjusting the estimated incident duration is useful. When the estimated incident duration is adjusted, the set redirection strategy needs to be taken into account. Sudden changes in the path advice might lead to confusion and mistrust. Therefore, the advised paths need to stay the same and just the redirection duration is adjusted if the alternative paths possess sufficient remaining capacity for the changed redirection duration. However, this might be an issue for cases in which the redirection duration is prolonged and the remaining capacities on the alternative paths cannot accommodate the consequential additional demand.

Second, related to the issue of the estimated duration is the prediction of the recovery time, which expresses the period between the end of the incident and the point of time at which no further delays induced by the incident are caused. TSUCHIYA et al. [2006] introduce a linear function to predict the development of the recovery time, however, their approach heavily relates to rescheduling of railway services. Furthermore, here, this is rather understood as part of the incident duration, as rescheduling is a disposition measure in railway operations (Section 2.2). Nevertheless, including a prediction for the recovery time is expedient to accomplish a more precise redirection duration. Factors that influence the recovery time

are the demand and the occupancy of services, which influence how quickly the accumulated passenger demand at stops can be picked up.

Concerning the calculation of the redirection duration T^p_{od} , such a prediction would also help to improve Equation 4.2. This can be achieved due to a further development of factor ψ from a deterministic value to a recovery-time-dependent sub-model. Such a sub-model probably needs to be adjusted to the characteristics of a PT system, as well as the location and duration of an incident, and the demand and occupancy levels at the point of time the incident occurs.

Third, REPACK merely considers one incident, however, it is possible that more than one incident occurs at the same time. Furthermore, an initial incident can also trigger secondary disturbances (Section 2.1.3). Similarly to the previously described case of adjusting the estimated incident duration, the set redirection strategy needs also to be taken into account if a second incident or disturbance occurs. Changing the redirection strategy, meaning the already provided path advice for the OD pairs, is inappropriate as mentioned before, at least if it happens within a short period. The minimum incident duration for a switch in the redirection strategy is yet to be determined. Otherwise, the passenger redirection for the second disturbance is more restricted due to the already disseminated path advice.

Fourth, Approach HR gives room for improvement. The heuristic sorts the affected OD pairs $od \in OD_a$ by the corresponding size of its respective redirection group $rg_{od}^{\hat{p}}$ (Section 4.5). Another reasonable approach would be to organise the $od \in OD_a$ by the potential savings of their overall delays, which are not only dependent on the size of $rg_{od}^{\hat{p}}$, but also on the difference in travel time between the alternative paths. Especially with smaller products of compliance and information rates of $\eta \cdot \zeta < 80\%$, which are more likely according to the results from the PT user survey (Figure 3.9, Section 3.2.3), the heuristic approach performs better than Approach OR, which indicates the expediency of redirecting the affected OD pairs in a certain order. Nevertheless, as pointed out here, delays could be further reduced with a different definition of this order of OD pairs.

Fifth, the aforementioned statement about Approach HR is therefore also an argument for the advancement of the optimised approach of the passenger redirection process (Approach OR). The consideration of a certain order of the affected OD pairs $od \in OD_a$ during the redirection or a stronger inclusion of the product of compliance and information rates into the optimisation, especially into the optimisation function, might be useful. Further, in the optimisation approach, the travel time of the redirection groups rg_{od}^p along a path $p \in P_{od}$ when calculating the remaining capacity is not considered as in Approach HR, in which the capacities are calculated segment-wise (Section 4.5).

When a redirection group travels on an alternative path, the start and end time of the journey on a certain path segment are determined by the departure of the first passenger of the corresponding redirection group at the beginning of the segment and the arrival of the last group member at the end of a segment. This is true for each path segment, defined by the used PT lines, which are used along the path. The period in which a certain rg_{od}^p needs free room on the first path segment is earlier than the period for the second and third segments. This can lead to the remaining capacity being over- or underestimated. Consequently, the achievable reductions can decrease in the overall passenger delays, which is also an explanation for the Approach HR performing better in some cases than Approach OR, which was especially evident in the case study conducted in the network of Kassel (Section 5.3). Therefore, the optimisation

can be improved by including the travel time of the redirection groups when calculating the remaining capacity on alternative paths.

The fact that Approach HR performs in some cases better than Approach OR also invites investigations for a hybrid model in which both approaches are integrated into one. In the previous chapter, factors which influence which approach performs better are already mentioned – e.g. product of compliance and information rates, network complexity, and incident duration.

Sixth, this thesis merely presents a heuristic approach to the reallocation process. As the heuristic passenger redirection process, the heuristic reallocation process might also be improved by the previously described order of the redirection groups, sorted by the potential reduction in passenger delay rather than by the size of the respective redirection group $rg_{od}^{\hat{p}}$ of the affected OD pairs $od \in OD_a$. Furthermore, just like the redirection process, the reallocation process is mainly a matching problem, however, the matching of affected passengers with available alternative paths is more complex when including the possibility of reallocating dispensable capacities. Nevertheless, it is an optimisable problem that invites the development of an optimisation approach for the capacity reallocation process.

Moreover, the passenger redirection and capacity reallocation process have been presented separately to investigate which element of REPACK has which effect. However, a merger of the approaches would be expedient as it might lead to an acceleration of the process and better results through the consideration of dispensable capacities during the passenger redirection procedure.

Seventh, in the case studies conducted, the Approaches CR+HR and CR+OR, which include the reallocation of capacities, are only executed if all missing places could be provided by dispensable capacities. Another approach could also be that if this is not achievable, the available dispensable capacities are used. In this way, at least some of the corresponding redirection groups could be redirected for a correspondingly shorter redirection duration. As some of the results indicate, capacity reallocation not only increases the remaining capacity on alternative paths, but also lowers the waiting time for some passengers through the extra departures at the corresponding stops at the supported PT line. Moreover, capacity reallocation is only considered for the whole stretch of the line with insufficient capacity ($pl \in PL_{ins}^{ps}$) and only if a bus is dispensable from the beginning of its original assignment. To enhance the level of flexibility and increase the chance for the execution of the capacity reallocation process, the possibility of reallocating a bus during its run on its original line and just for the part of the insufficient line $pl \in PL_{ins}^{ps}$ which serves the path segment $ps \in PS_p$ with insufficient remaining capacity might improve the process. Another advantage of increased flexibility during this process can be the timing of the reallocation. As discussed in the previous chapter, the fact that the capacity reallocation approaches perform best merely in a few cases might also derive form reallocated buses arriving right before or after regularly scheduled buses at stops. This limits the positive effect of the process while still causing additional delays on the lines from which the reallocated buses are withdrawn.

Eighth, REPACK is developed for mass PT services and does neither consider other modes and services such as walking, cycling as well as bike, e-scooter, or car sharing, nor on-demand services. Especially when the remaining distance to a traveller's destination is short, walking is probably the best alternative. Additionally, walking to a nearby alternate stop might lead to bigger delay reductions for some passengers. In particular, for small redirection groups of one or a few affected passengers, mode-sharing and on-demand services probably offer a suitable alternative and reduce the demand on the mass PT system during the incident. However, these services have a significantly smaller capacity than mass PT services and are therefore not suitable for the redirection of bigger groups. That being said, including several on-demand vehicles such as taxis in the capacity reallocation process could probably help to reduce the overall delay of affected passengers more efficiently. According to the aforementioned interviews and observations in OCCs, this is also done in practice [BACHMANN, BRIEM et al., 2022].

Ninth, a path choice model is not included in REPACK. Works such as CATS and JENELIUS [2014] and LENG and CORMAN [2020], which are introduced in Section 2.2.2, use a path choice model to predict the reaction of passengers to incident information. However, these works do not consider concrete path advice. Extending a path choice model accordingly might help to better predict the compliance rate η of affected passengers. This is further discussed in the following.

Tenth, even though an information rate is considered, differences in the information rate concerning different locations in the network are just mentioned, but not implemented in the methodology. This would also be a useful extension for the special case in which only five OD pairs have an information rate of 100% and the rest of 25% (Section 5.2.2).

The model limitations clearly show a wide range of possibilities of improvement. Even though REPACK already shows mainly a strong positive effect on the passenger delays during incidents, it is likely that even better results can be accomplished when the listed limitations are removed or at least weakened. The following section discusses this work's contribution as well as the comparison to other works in the field of PCIM.

6.2 Contribution and Comparison

Section 2.2.2 introduces the to-date academic investigations and approaches in practice in the field of *passenger-centric* incident management (PCIM) in PT systems. Among the academic investigations presented are four that propose the provision of path advice and are thereby closest to the methodology presented here. However, two of them, BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023] present earlier stages of the PCIM method described here. BACHMANN, RAU et al. [2021] show the conceptional framework of a PCIM method including heuristic passenger redirection and capacity reallocation, and test the method in a numerical example in a simple toy network (STN) which is used here to tell the real-life story in Section 4.1. BACHMANN, DANDL et al. [2023] present a further developed version of the passenger redirection heuristic and introduce an optimised approach to it. These two approaches of the passenger redirection process from BACHMANN, DANDL et al. [2023] as well as the framework of the capacity reallocation process from BACHMANN, RAU et al. [2021] are merged and further developed into the REPACK model presented here. The most significant advancements are the consideration of self-redirected passengers through ξ , the advanced version of Equation 4.2 to calculate the redirection duration T_{od}^p through which factor ψ is introduced as well as the development of the conceptional framework of the capacity reallocation process into an applicable algorithm. The latter upgrade makes the reallocation process applicable for simulation studies and brings it closer to implementation in practice. Furthermore, here, REPACK is tested in a multimodal PT network model which is based on an actually existing network.

The other two works which present methods for the provision of path advice are by VAN DER HURK et al. [2018] and MO, KOUTSOPOULOS, SHEN et al. [2023]. The former publication demonstrates a PCIM method to provide customers of railway services with path advice in case of severe disruption. The introduced algorithm, called ARSRU, includes also a rolling-stock rescheduling model, based on the work of KROON et al. [2015], which are connected through an iterative loop to accomplish an ideal outcome.

The most noticeable difference to REPACK is the environment for which the methods have been developed. Whereas VAN DER HURK et al. [2018] focus on regional and long-distance railway operations, REPACK is meant for multimodal urban PT systems. Railway networks, especially of regional and long-distance services, are more coarse-meshed than urban PT networks. Moreover, the headways are often longer, and incidents that include the blockage of a track are often more severe, as trains cannot overtake each other like buses can in a close-meshed street network. The situations in the two systems are therefore different but comparable. Also, the work by VAN DER HURK et al. [2018] as well as most of the other works introduced in Section 2.2.2 consider indivisible passenger groups in accordance with the origin and destination of passengers or by their origin of redirection and destination.

The authors also consider a compliance rate of several deterministic values as well as a logarithmic calculated value. The latter takes the travel time difference between the suggested and the fastest path into account. As argued before, the difference between the fastest and the suggested path is certainly one factor that influences the willingness of affected passengers to follow given path advice, but other influencing factors are likely too. As elaborated in Section 2.1.4, weather, bike and car ownership, closeness to the destination, and further aspects influence the reaction of passengers to incident situations and therefore can also have positive or negative effects on the compliance rate of affected passengers.

Furthermore, the aforementioned difference in path travel times rather influences the willingness of experienced travellers, such as daily commuters, who are aware of the fastest alternative path, than that of the rest. REPACK takes PT users who prefer to redirect themselves separately into account through the self-redirection rate ξ . Moreover, VAN DER HURK et al. [2018] do not consider PI channels or an information rate to take into account the reachability of affected passengers. As the survey results in Section 3.2.3 show, most of the participants prefer a collective PI channel (i.e. dynamic PI displays or speaker announcements) as the source of information during incidents. Therefore, in Section 5.2.2 a special case is presented where only five of the affected OD pairs are fully reachable, while of the rest only 25% of the passengers of each affected OD pair $od \in OD_a$ are reachable. Nevertheless, ARSRU considers uncertainties in the estimation of the incident duration which is yet to be implemented in REPACK.

The general idea behind the rolling stock rescheduling model is similar to the capacity reallocation process presented here, which is the provision of additional capacity to support passenger redirection. However, as rolling stock rescheduling faces more restrictions due to

the boundaries of the deployment of trains and the usually coarse-meshed nature of railway networks, the reallocation of buses is more flexible, which however induces a certain further complexity. Through REPACK, buses are withdrawn from their original assignment and deployed to other lines to support passenger redirection. Among all the works done in the field of PCIM, VAN DER HURK et al. [2018] present the method which is the closest to REPACK, however, as discussed in this section, the method shows several differences to REPACK.

Mo, KOUTSOPOULOS, SHEN et al. [2023] present a mathematical formulation of passenger flow distribution in an urban PT network. The authors consider full compliance, and even though PI channels such as mobile phones and websites are mentioned, no information rate or information schemes are taken into account. As with most of the works presented in Section 2.2.2, MO, KOUTSOPOULOS, SHEN et al. [2023] do consider indivisible passenger groups as well as capacity constraints. Other than that, through the mathematical approach to flow distributions, the work by MO, KOUTSOPOULOS, SHEN et al. [2023] is significantly different from the other publications presented in Section 2.2.2.

The rest of the academic investigations introduced in Section 2.2.2 do not consider path advice, but either assume that affected travellers inform themselves about alternative connections [MÜLLER-HANNEMANN et al., 2019] or inform passengers about an incident and changes to the arrival times of PT services or taken disposition measures.

CATS and JENELIUS [2014] and the *Corman group*¹ implement the availability of this kind of information in path choice models through which the affected PT users are then assigned to paths. Therefore, there is no room for a compliance rate, as on the one hand, no concrete path advice is given to which the travellers could comply, and on the other hand, how the passengers react to the given incident information is decided by the path choice model.

Path choice models are sophisticated and helpful tools to predict travellers' path choices. This reveals the possibility of combining parts of the presented methodologies by implementing path advice into path choice models (Section 2.1.2). However, for this, the factors which influence the compliance to given path advice need further research to complement the work by WILKE [2023], who investigates incentives to influence passengers' willingness to follow path advice, in the context of urban PT services (Section 2.1.4).

Furthermore, CATS and JENELIUS [2014], ZHU and GOVERDE [2019a], as well as the *Corman group*¹ consider different locations and schemes of information. CATS and JENELIUS [2014], for instance, compare information dissemination at stops and at transfer hubs as well as network-wide dissemination. ZHU and GOVERDE [2019a] take into account information being available at stations and/or trains. The *Corman group*¹ tests different information schemes concerning the time at which the information is disseminated. As mentioned in the previous section, location-dependent information rates are an especially expedient advancement for REPACK.

The results of the case studies presented in the various investigations concerning relative passenger delay reduction show a wide range of outcomes from negative values [CATS and JENELIUS, 2014; VAN DER HURK et al., 2018; BACHMANN, DANDL et al., 2023; MO,

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

KOUTSOPOULOS, SHEN et al., 2023], to reduction of almost 100% [LENG and CORMAN, 2020] (Section 2.2.2, [BACHMANN, TSAKARESTOS et al., 2023]). With a range of relative delay reductions from less than 10% with $\eta \cdot \zeta = 35\%$ in Scenario LR to a reduction of 90% with $\eta \cdot \zeta = 100\%$ in Scenario LS, the results of REPACK fit well with the published works in the field of PCIM. However, all these works use different simulation environments, networks, and inputs, which makes a direct comparison inadequate and unreasonable. The usage of the same publicly available benchmark networks, such as the Mandl or the Mumford networks, as used for the comparison of algorithms to solve the PT network design problem [UL ABEDIN et al., 2018], would be desirable.

In the approaches in practice which are introduced in Section 2.2.2, the focus is different from that of the previously mentioned academic investigations, even though the general idea and aim of reducing passenger delays through adequate PI during incidents are the same. As pointed out by BACHMANN, TSAKARESTOS et al. [2023] as well as in Section 2.2.2, most well-equipped PT systems to date provide their users with information about the occurrence of an incident as well as the consequential disturbances. The provision of concrete path advice is not state-of-the-art in practice yet. TSUCHIYA et al. [2006] present a prototype of a passenger guidance system for the railway system in Japan. Affected passengers receive the information on whether they should wait or take another path to their destination in case of an incident. Considered PI channels are dynamic displays at stations and mobile phones. However, the system's capacity is only qualitatively taken into account, and the developed algorithms are not disclosed, optimisation not mentioned.

BMVI [2019] presents the outcome of a PCIM project with several phases in Germany. Over the development of the project, the consortium, mainly composed of practitioners, moved continuously away from the use of collective PI channels towards individual information channels such as smartphone apps [BMVBW, 2001; BMVBS, 2009; BMVBS, 2011; BMVI, 2019]. This contradicts the findings of the PT user survey presented in Section 3.2.3, which imply the importance of collective PI channels, especially during incidents. The system's capacity is also only qualitatively considered and as in the previously mentioned work, algorithms are not disclosed either and optimisation is not mentioned. All approaches in practice, which are introduced in Section 2.2.2 mention the necessity of an integration of a PCIM method into the OCC of a PT system, as also suggested by BACHMANN, TSAKARESTOS et al. [2023].

Besides the aforementioned publications by BACHMANN, RAU et al. [2021] and BACH-MANN, DANDL et al. [2023], this work presents the first approach in the field of PCIM focussing on bus operations and includes a capacity reallocation process which withdraws dispensable buses from PT lines to dispatch them to other lines to support passenger redirection. This has been done, since the interviews and observations in OCCs give evidence of the fact that finding drivers who cater for the availability of standby and extra vehicles has constantly become harder [BACHMANN, BRIEM et al., 2022]. This could also be a problem when an incident in the railway part of an urban PT network occurs and a rail replacement service needs to be established to bridge the disrupted segment of the railway service (Section 2.2). If not enough extra buses are available, such a bridging service could be supported by the capacity reallocation process to withdraw buses from other lines to dispatch them to the bridging

service. It goes without saying that the passenger redirection process should also be executed in such a situation to give the affected passengers adequate path advice.

One of the most decisive contributions of this thesis is that it brings academic research and approaches in practice in the field of PCIM one step closer together. As pointed out by BACHMANN, TSAKARESTOS et al. [2023] and also explained in Section 2.2.2, there is a gap between these two perspectives. Academic research presents optimised procedures with which optimal paths are found for affected passengers, but it lacks the connection to practice in terms of available data and information dissemination equipment. Approaches in practice strongly focus on the available infrastructure, but fail the implementation of state-of-the-art research.

Through interviews and observations in OCCs (Section 2.2.1), the base of understanding of PT operations – the supply side – is built. Additionally, through the PT user survey, presented in Chapter 3, insight into the passengers' perception of incidents and incident management is given that supports the understanding of the passenger side. The supply and passenger sides are to be connected through PCIM methods such as REPACK, as also implied in Figure 1.1. This connection is elaborated in the following.

The redirection duration T_{od}^p which is introduced here is important information for the dispatchers in OCCs. During the information dissemination process, described in Section 4.7, available PI channels for the affected OD pairs are identified. The redirection duration T_{od}^p then defines for each of the affected OD pairs $od \in OD_a$ for how long the path advice is disseminated. After the redirection duration is elapsed, the advice to wait can be broadcast until the incident is resolved or alternatively until the recovery time is elapsed. In addition, the presented PT user survey gives insight into the users' preferences regarding PI channels during incidents. As these are mostly collective channels – i.e. dynamic PI displays and speaker announcements (Section 3.2.3) – the reachability of all affected OD pairs might not be certain. Therefore, the information rate ζ enables the inclusion of this limitation. Furthermore, the option that for only a limited number of the affected OD pairs, the information rate is higher since they can be informed via the collective channels, whereas for the rest, the information rate is lower, as they are dependent on other PI channels, is implemented. In Section 5.2.2, such a case has been tested with five affected OD pairs that have an information rate of 100% and the rest a rate of 25% following the results from the aforementioned survey (Section 3.2.3).

Nevertheless, works by CATS and JENELIUS [2014], ZHU and GOVERDE [2019a], as well as the *Corman group*¹ couple the location of information dissemination with the reachability of PT users. Hence, connecting the information rate in REPACK with the location where affected passengers receive the path advice would be expedient. As argued before, the availability of collective PI channels is different within a PT network depending on the kind, size, and location of a stop. Whereas large transfer hubs in the city centre are often well-equipped with PI channels, small bus stops at the outskirts of a network might not be equipped with any

¹LENG and CORMAN [2020], LENG, LIAO et al. [2020], M. RAHIMI et al. [2021] and RAHIMI SIEGRIST and CORMAN [2021]

PI channels. Depending on how well a PT network is generally equipped in this matter, even stops in the city centre might not be equipped adequately.

However, even though REPACK already presents a method that is relatively close to implementation in practice compared to other works in the field of PCIM, further steps are necessary and need to be considered before an application of REPACK in practice is thinkable. These steps are discussed in the following section.

6.3 Implementation in Practice

As pointed out by BACHMANN, TSAKARESTOS et al. [2023] as well as by the investigations of practitioners presented in Section 2.2.2, when implementing a PCIM method into practice, it is useful to integrate it in the system architecture of an ITCS. The ITCS is the overall system connecting the OCC with the vehicles and the PI systems [VDV, 2001, pp. 182-183; PANGILINAN et al., 2008]. The OCC is the heart of a PT system where important information is gathered, especially during and about incidents (Section 2.2.1), which is important input for REPACK. Hence, a logical first step is the consideration of the sources for necessary inputs. Incident information such as location and an estimation of duration as well as the disrupted PT lines are important inputs. It is conceivable that this information is manually provided by the dispatchers when they receive the information over the radio from the PT drivers. However, this makes an already stressful situation even more stressful for the dispatchers. Another opportunity to provide for the necessary information is to establish an interface to the ITCS as proposed by BACHMANN, TSAKARESTOS et al. [2023]. Through a connection to the ITCS also information regarding the disposition measures taken by the dispatchers as supply-centric reaction to the incident can be passed and thus the disposition timetable, which is necessary to find alternative paths for affected passengers, can be made available. Therefore, a coupling of a PCIM method like REPACK with ITCS is expedient.

Figure 6.1 by BACHMANN, TSAKARESTOS et al. [2023] shows a high level system architecture of an ITCS in which CAPR is integrated. The incident reporting and instructing the disposition measures are done verbally via radio. The PT vehicles' positions and capacities are transmitted automatically. The OCC feeds back the vehicles delays, based on the delay calculation, which compares vehicles' positions with the timetable [BACHMANN, BRIEM et al., 2022]. The disposition measures are decided by the dispatcher. The consequential disposition timetable, the incident information (i.e. location, duration), the vehicles' capacities, and the passenger demand can then be entered in the ITCS. Afterwards, PCIM methods like REPACK can compute path advice based on the given information as described in Chapter 4. Together with other information such as departure times and free text written by a dispatcher, path advice can be sent to all available PI channels (e.g speakers, dynamic PI displays, PT trip planners).

Therefore, in addition to connecting REPACK with the OCC and integrating it in the system architecture of the ITCS, the PI channels need to be capable of disseminating path advice. This also concerns the aforementioned interfaces, the GTFS and the VDV454 (Sections 2.1.5 and 2.2.2) [VDV, 2018b; MOBILITY DATA, 2022].

Moreover, the reallocation process of an PCIM method such as REPACK would profit from such an interface through which important information about the fleet and its occupancy can



Figure 6.1: CAPR integrated in the ITCS architecture [BACHMANN, TSAKARESTOS et al., 2023]

be transmitted. Hence, not only CAPR but all elements of REPACK should be integrated.

A related second step concerns the incident duration. As described by BACHMANN, BRIEM et al. [2022] and in Section 2.2.1, a great extent of incident management relies on the experience of dispatchers, as does the estimation of the incident duration, which is an important input for REPACK. This might lead to wrong estimations and readjustments of the estimated incident duration. As pointed out in Section 6.1, the ability of REPACK to adjust to changes in the incident duration is yet to be implemented and is a sensible advancement. However, as far as a categorisation of incidents regarding their location in the network as well as the type of incident, time of day, and other aspects are concerned, it would be expedient if an OCC can provide a corresponding database, which contains information about common and rare incidents. In accordance with factors such as the time of the day, weekday, and to be expected demand level, the database could then provide an estimation for the incident duration.

The demand is subject of the third step. To be able to check alternative paths for sufficient remaining capacity, at least an estimation of the demand is needed. Most PT operators have at least some historical travel data through ticket sales, random samples, or censuses, as this data is also needed for planning purposes [UL ABEDIN, 2019]. It is thinkable that a database of such historical travel data is prepared and categorised by the day of the week and the time of day to get a demand estimation for the time when an incident occurs. Such a database could also include the unusual demand levels which are caused due to special events such as festivals or sport competitions. Furthermore, in well-developed PT systems, there is the

possibility of connecting REPACK with real-time demand data that can be collected through AFC and APC. With the help of such real-time data the precision of REPACK could be further improved.

Speaking of redirecting demand onto alternative paths, the provision of a set of available paths is another necessary input for REPACK and defines the fourth step. As pointed out in Sections 4.4 and 5.3 as well as by MÜLLER-HANNEMANN et al. [2019], the calculation of paths can be computationally expensive and take a long time, which is not convenient for a system that needs to react quickly. Therefore, also in the case of paths' provision, a database could be set up, preparing possible sets of paths regarding certain incident situations. In a construction kit manner, such a database could be built with certain incident locations, incident types, and estimations of durations to then select a possible set of paths. The building procedure of such a database could be guided by link vulnerability analysis such as the one by CATS and JENELIUS [2014] to start with the incidents with the most severe consequences for the PT system and the links at which an incident is most likely.

Regarding the paths, an advancement of REPACK is also necessary concerning the destination of redirection groups rg_{od}^p of the affected OD pairs $od \in OD_a$. So far, REPACK considers the final destination stops of affected passengers. In a large real-life PT network such as the one used in Section 5.3, this is not reasonable as it leads to more OD pairs and thereby smaller redirection groups, which can limit the number of reachable travellers as pointed out before. Hence, it would be more considerate to use the last transfer hub or relatively centrally located stop of affected passengers as the destination of redirection. In this way, more passengers could be associated with the same OD pair, even though their final stop is not the same.

A thinkable fifth step concerns computational time. As previously pointed out, looking for paths is computationally expensive. However, also in this regard, REPACK needs improvement. With an Intel Xeon W-2133 CPU with 3.60 Gigahertz and 32.0 Gigabytes of RAM, the assignment process from Step 6 in the situation analysis process (Figure 4.4) to Step 23 in the passenger redirection process (Figure 4.5) needs about 20 seconds in the Mandl network, as also mentioned by BACHMANN, DANDL et al. [2023]. However, most of the time needed, about 18 seconds, is used for the retrieval and processing of the simulation output from the higher benchmark scenario, which can probably be accelerated. If the capacity reallocation process is executed, an additional 30 seconds are needed from Step 24 to Step 46 (Figure 4.6). So, in total REPACK would need about one minute in practice, not including the accessing and processing of the data coming from the ITCS and aforementioned databases concerning the available alternative paths and the demand, which will likely also add processing time. Furthermore, in the network of Kassel (Section 5.3), Steps 6 to 23 need about 5 minutes. As REPACK is supposed to react quickly on suddenly occurring incident situations, this process needs to be accelerated.

Speaking of timing, Figure 6.2 combines Figure 2.1 which presents the incident management procedure of an OCC (left track) with Figure 4.2 presenting the procedure of REPACK (right track). It is supposed to give an idea of the timing of REPACK's execution in relation to the processes in an OCC. At the point of time at which the dispatcher decided on the disposition measures (Step 3, left track) on which basis the disposition timetable is created, REPACK can be executed to start analysing the situation (Step A). As indicated by the not existing gaps between the processes of REPACK, it continues with the subsequent Steps B, C, and D

without any delay. This is in contrast to the steps conducted in the OCC during which the dispatchers have to wait until certain events occur, such as the arrival of the transport warden at the incident site.

In Step 7, the OCC updates the instructions to the PT drivers. If an incident develops differently than expected or the dispatchers receive additional information from the transport warden, disposition measures might need to be adjusted and the estimation of the incident duration might also change. As discussed before, it is therefore expedient if REPACK is also capable of adjusting to changes in the inputs such as the incident duration, which happens in Step E and update the redirection strategy accordingly. At the same time, also the instruction to the drivers is updated. This step is not yet implemented in REPACK. Once the situation is cleared, REPACK ends as well, until the next incident occurs.

A sixth step is the consideration of staff management. As pointed out by CARREL, MIS-HALANI et al. [2010] and BACHMANN, BRIEM et al. [2022], shortages in staff can worsen incidents and play a vital role in incident management. The capacity reallocation process does not yet consider the legal constraints concerning the driving hours of the PT drivers. Certain reallocations might therefore not be possible if they extend the driving times significantly.

A possible seventh step concerns a REPACK smartphone app or app add-on by which the passengers could be redirected. Figure B.7 shows that the survey participants are willing to share some information about their trips and travel habits for the service of getting path advice during incidents. Furthermore, such an app could consider travellers' preferences regarding redirection as also suggested by CEDER and JIANG [2019] for path searches under undisturbed conditions. In any case, as argued in Section 3.2.5, path advice should be disseminated through all available PI channels.

The eighth and last necessary step for REPACK's implementation into practice is a rather minor one but still needs consideration. When such a PCIM method is rolled out in practice, the PT users need to first build up trust in the system before most of them would follow given path advice. As shown in Figure 3.9, most of the passengers answered that they *might* follow given path advice. Additionally, the presented results in Chapter 5 reveal that the effectiveness of REPACK grows with an increase in the product of compliance and information rate. Being "REPACKed" might therefore not be a promising term to gain trust. On the users' end of REPACK, a more appealing name, such as Path Advice for Passengers during Incidents (PAPI) or Passenger ROUting during Distruption (PROUD), should be established. Another thinkable way to motivate people to follow path advice is by monetary incentives for travellers who follow path advice by reducing the fare of the corresponding rides as also the survey results from WILKE [2023] suggest. In urban PT systems that are equipped with AFC such an approach is technically feasible.

In general, further investigations could help to better understand travellers' reactions to incidents and path advice. For example if these would examine factors which influence the passengers' willingness to follow given path advice and develop a model to predict such behaviour, such as the work by WILKE [2023] for long-distance railways services. Furthermore, such knowledge could be implemented in path choice models and improve the effectiveness of PCIM methods such as REPACK as pointed out before.



Figure 6.2: REPACK integrated in the incident management procedure of OCCs, extended from BACHMANN, BRIEM et al. [2022]
Chapter 7

Conclusions

7.1 Summary

This dissertation presents the background as well as the state-of-the-art of incident management in public transport (PT). Here, incident management is divided into two parts: *supply-centric* incident management that describes the supply side during incidents in PT, and *passenger-centric* incident management (PCIM) that connects the supply side with the passenger side as Figure 1.1 implies. To better understand the passenger side, especially during incidents, a PT user survey has been conducted, the setup and results of which are described in this thesis as well.

The knowledge which has been gained through the aforementioned elements of this thesis is the basis for the development of the model for REdirecting Passengers and reAllocating Capacities Knowingly (REPACK) during incidents in urban PT systems. This model is a new contribution to the research area of PCIM. The results of several case studies in the Mandl network as well as one case study in the network of the city of Kassel, Germany, are presented and discussed. Furthermore, the reallocation of capacity is tested in an autonomous PT system.

This has been done to answer the following research questions, which are also introduced in Chapter 1:

1. To what extent does path advice lower the overall passenger delay during incidents?

This question can be answered through the literature review on PCIM in PT. From the works presented in Section 2.2.2, four consider path advice, also including the two investigations presenting earlier stages of the REPACK model [VAN DER HURK et al., 2018; BACH-MANN, RAU et al., 2021; BACHMANN, DANDL et al., 2023; MO, KOUTSOPOULOS, SHEN et al., 2023]. The outcomes of the studies by VAN DER HURK et al. [2018] and MO, KOUT-SOPOULOS, SHEN et al. [2023] show a relatively wide range from -6.4% to 30% relative delay reduction for affected passengers, which indicates that path advice has the potential to conceivably reduce passenger delays during incidents. However, the outcomes also indicate limitations and cases in which it is not reasonable to redirect passengers.

Also, REPACK shows great potential to reduce passenger delays induced by incidents in urban PT systems. Under ideal circumstances, the relative delay of affected passengers can be reduced by 90%, which is significantly more than the aforementioned results. But, a direct

comparison of the results of different works is inappropriate and inadequate as the models have been tested with different inputs, in diverse networks, and simulation environments. Nevertheless, it is evident that path advice can reduce overall passenger delays during incidents to a great extent.

2. Under which circumstances is the redirection of passengers reasonable?

The fact that path advice can also lead to an extension of passenger delays as implied by the previously mentioned negative results. This means that the circumstances for a passenger redirection to be reasonable need to be met. The results presented in Chapter 5 express that the location of an incident plays a decisive role as it influences the number of available alternative paths. Passengers can only be redirected if alternative paths are available.

Moreover, the type of incident and the taken disposition measures play an important role as well. The comparison of the two tested scenarios – LS, in which the disrupted PT lines are split; LR, in which the disrupted PT lines are rerouted – show that the more disruptive the incident and the taken disposition measures, the more effective is REPACK. Furthermore, the longer the incident duration, the more people are affected by it, and hence, the more affected passengers can profit from REPACK, which also increases its effectiveness. In some cases in the less disruptive Scenario LR, the effect of REPACK was very low and therefore it is probably not expedient to employ it in these cases. As discussed in Chapter 6, the definition of an incident's minimum level of disruptiveness as the threshold for the execution of REPACK is useful.

3. What are factors that influence the effectiveness of passenger redirection?

In addition to the incident-related circumstances such as location, duration, and taken disposition measures, factors are also identified that influence the effectiveness of REPACK.

Firstly, the product of the two parameters compliance and information rates shows a significant influence on the outcome of passenger redirection. The more people who are reachable and follow provided path advice, the higher the effectiveness of REPACK.

Secondly, a growing share of affected passengers who redirect themselves lowers the effectiveness of REPACK. About these matters, the presented results of the PT user survey also give a useful insight into the passengers' willingness to follow given path advice as well as into preferences regarding the passenger information (PI) channels through which they retrieve information during an incident.

Thirdly, the quality of the calculation of the redirection duration introduced here influences how much REPACK can lower passenger delays. In this regard, factor ψ plays a role. For the more disruptive Scenario LS -1 is a good set, and for the less disruptive Scenario LR 0 is a good set, as the more disruptive a scenario is, the longer is the respective recovery time. The recovery time is the time between the dissolution of the incident and the end of the delays induced by an incident. A negative value for factor ψ prolongs the redirection duration through which it can be adjusted to the recovery time. However, as discussed before, a deterministic value for factor ψ might be a too simple approach. Therefore, more research is needed concerning this matter.

4. Does capacity reallocation support passenger redirection?

In addition to the two passenger redirection approaches – heuristic redirection (HR) and optimised redirection (OR) – introduced here, these are tested in combination with the capacity reallocation process described in Section 4.6. BACHMANN, RAU et al. [2021] imply that through reallocating capacities, passenger redirection can be supported and passenger delays further reduced. However, this work which presents the conceptional framework of REPACK is merely tested in a numerical example, in contrast to the simulation case studies in this thesis, in which REPACK is tested at its present stage.

The results show that only in rare cases the aforementioned criteria for the execution of the capacity reallocation process are met. It only happens with a product of compliance and information rates of 100%, which indicates that in cases in which fewer people are redirected, there is no need for capacity reallocation. Therefore, it might be more interesting for cases in which the remaining PT system's capacity is smaller than in the cases conducted here or the PT system is stronger demanded, respectively.

Furthermore, among the cases in which the capacity reallocation is done, there are only a few cases in which it performs better than the passenger redirection process. Consequently, in most cases, the disadvantages of the capacity reallocation of reducing the service quality on the lines from which reallocated buses are withdrawn outweigh the advantages on the lines to which they are dispatched to support the passenger redirection. The fact that this is also true for the cases in an autonomous PT system with a modular setup, called DART, in which these disadvantages are weakened through the flexibility of the modular setup, indicates that other factors also influence the performance of the capacity reallocation. For example, the timing of the deployment at the supported line as discussed before. Nevertheless, in some cases, the capacity reallocation reduces passenger delays further as suggested by BACHMANN, RAU et al. [2021], but gives also room for further research.

In summary, REPACK shows great potential for reducing passenger delays during incidents. Performance-influencing circumstances such as the incident duration and factors such as the compliance rate have been identified. Due to the closeness of this work to practice, as discussed in Chapter 6, this thesis presents a contribution to PCIM in PT which is another step closer to implementation in practice. The presented work supports the statement that PCIM methods such as REPACK can improve the reliability of PT systems and by that their attractiveness. This can lead to a rise in ridership, which consequently lead to a reduction in greenhouse gas emissions.

7.2 Outlook

First of all, REPACK needs further evaluation and testing in different cases, especially regarding the complexity of the network and variety of available mass PT modes (i.e. metro, trams, bus) as well as the characteristics of the incident concerning its location in the network and its duration. Furthermore, only two disposition measures are considered here in combination with REPACK. As Section 2.2.1 presents, there are a variety of *supply-centric* measures taken by dispatchers in OCCs to mitigate the negative effects of incidents. Chapter 6 reveals that

factor ψ needs further development from a deterministic value to a more sophisticated submodel considering the recovery time and severity of an incident. To do so, it needs to be investigated which factors influence the recovery time and to which extent.

Additionally, the presented limitations of REPACK need to be removed or at least weakened. These include the possibility of adjusting to an altered estimation of the incident duration, being able to handle more than one incident, further improving the heuristic passenger redirection approach by sorting the affected origin-destination (OD) pairs in another manner, improving the optimised passenger redirection approach by considering the travel time of redirection groups when calculating the remaining paths' capacities, and further develop the calculation of the redirection duration as previously mentioned.

Moreover, the capacity reallocation needs further research to improve its performance through better timing of the deployment and redefining its criteria for execution. Furthermore, an optimisation approach for the capacity reallocation approach is yet to be developed as well as the integration of the passenger redirection and capacity reallocation processes into one. Additionally, the capacity reallocation process provides the possibility of also including on-demand services, such as taxis to support passenger redirection. Speaking of other modes, the passenger redirection process only considers mass PT services and not other modes like walking, cycling, or shared mobility, such as bike, car, or e-scooter sharing services. Those can lead to further path alternatives, especially for smaller redirection groups and, thereby, to even stronger delay reductions. In addition, this can exploit new alternative paths for coarse-meshed areas of the PT network.

The results from the case study in the Kassel network indicate that further adjustments are needed for REPACK to function well in more complex and realistic networks. The representation of the disposition timetable through the network graph described in Section 4.3 might be a good point to start in this matter. A representation that takes into account the actual arrival times of PT vehicles is necessary when REPACK is to be implemented in practice. This would also consider the influences through mixed traffic which additionally includes private and freight vehicles. Once REPACK is implemented into practice, the experience can be used to further improve and validate it.

Finally, factors influencing the compliance rate of affected passengers need further research. If these are better understood, such knowledge can be used to develop path choice models which can take path advice during incidents into account and to better predict travellers' reactions to incidents and path advice.

As stated in this section, the research area of PCIM has just been opened up and gives plenty of room for further improvement to accomplish even stronger delay reductions.

7.3 Own Publications

This last section of this dissertation presents the studies that have been conducted or supported by the author of this thesis. As indicated in Table 7.1, most of them are already published and further are planned. In the following, it is described how these works contributed to this thesis or, in case of the planned papers, on basis of which section of this thesis they will be developed. Most importantly the works by BACHMANN, RAU et al. [2021] and BACHMANN, DANDL et al. [2023] presenting earlier stages of the REPACK model are to be pointed out. Therefore, they are cited accordingly in the Chapters 4 and 5.

Furthermore, the publication by BACHMANN, BRIEM et al. [2022] gives profound insight into the daily practice of dispatchers in OCCs of urban PT systems. Thereby, it contributes to Section 2.2.1.

In addition, a literature review about CAPR is conducted by BACHMANN, TSAKARESTOS et al. [2023] which is the basis of Section 2.2.2.

One case study of this thesis is conducted in the autonomous modular PT system, called DART, which, among other things, uses the advantages of vehicle platooning. The explanation of platooning in general and its advantages in the context of PT systems is supported by the works of BACHMANN [2017], BACHMANN and SETHURAMAN [2019] and SETHURAMAN et al. [2019] in Section 5.2.3.

Moreover, WEINSZIEHR et al. [2023] study the detection of the phenomenon of bus bunching, which is mentioned in Section 2.1.3.

Generally speaking, these works as well as the peer review they went through, helped to improve the quality of this thesis.

Regarding the planned publications, the first listed planned work will present an updated version of REPACK shown in Chapter 4. It will include an improvement of the heuristic as well as optimisation approach as discussed in Section 6.1 and an optimised approach for the capacity reallocation process. These upgrades probably further reduce the passenger delays.

The second planned paper will couple REPACK with the model FleetPy which is introduced by ENGELHARDT et al. [2022] to manage fleets of on-demand services. It will investigate how passenger redirection can be supported by reallocating on-demand services during incidents. As mentioned in Section 2.2.1 this is already done in practice but not yet supported by algorithms or optimisation. As on-demand services are more flexible than buses due to their smaller size and the fact that they are not assigned to a scheduled PT lines, this might lead to better results compared to the reallocation of buses. Even better results might be gained by a combination of buses and on-demand services. Therefore, it will be an extension of the capacity reallocation process explained in Section 4.6.

In addition to the potential of REPACK, also room for further research and improvements is shown. If the aforementioned limitations can be removed or at least weakened and the aforementioned extensions are implemented, REPACK can lead to an even greater reduction in passenger delays than it already does. Furthermore, the steps listed in Section 6.3 can help to build the bridge between practice and research in the field of PCIM. The time is right to use the existing realisations to implement PCIM methods such as REPACK into practice. As this dissertation and aforementioned studies in the field of PCIM (Section 2.2.2) show, this can improve PT systems' reliability and by that their attractiveness. An increase in the attractiveness of more environmentally friendly means of transport, such as PT, motivates people to use it instead of motorized individual means such as cars. This reduces greenhouse gas emissions and thus contributes to the achievement of humanity's greatest challenge of adjusting our way of life to a sustainable one so that planet Earth remains habitable for our kind and for other living beings.

Reference	Title	Place of publication	
Published			
	Simulative assessment of the	Master's Thesis	
BACHMANN [2017]	impact of a platoon assistant on	at Technical University	
	the capacity of motorways	of Munich	
BACHMANN and SETTINDAMAN [2010]	Simulative Assessments of Energy and	mobil.TUM Conference	
DACHMANN and SETHORAMAN [2019]	Space Savings due to Platooning	2019	
SETULDAMAN at al [2010]	Effects of Bus Platooning in an	IEEE Intelligent Transportation	
SETHORAMAN et al. [2019]	Urban Environment	Systems Conference	
	Redirecting Passengers and	100th Annual Meeting	
BACHMANN, RAU et al. [2021]	Reallocating Capacities during	of the Transportation	
BACHMANN BRIEM et al [2022]	Incidents in Public Transport	Research Board	
	Dynamics and Processes in Operations	IFFE Transactions on Intelligent	
BACHMANN, BRIEM et al. [2022]	Control Centers in Urban Public	Transportation Systems	
	Transport: Potentials for Improvement	Transportation Systems	
	Optimized Passenger Redirection	102nd Annual Meeting	
BACHMANN, DANDL et al. [2023]	During Incidents in Urban Public	of the Transportation	
BACHMANN, DANDL et al. [2023]	Transportation Systems	Research Board	
	State of the Art of Passenger Redirection		
BACHMANN, TSAKARESTOS et al. [2023]	during Incidents in Public Transport Systems,	Public Transport	
	Considering Capacity Constraints		
	Detection of Bus Bunching	11th Symposium	
WEINSZIEHR et al. [2023]	through the Analysis of	of the European Association	
	Prevalent Public Transport Control Data	for Research in Transportation	
Planned	Planned		
	REPACK: REdirecting Passengers and		
Planned	ReAllocating Capacities Knowingly during	Transportation	
	incidents in public transport systems		
	Integration of On-Demand Services	Transportation Possarch	
Planned	in the Reallocation of Capacities During	Down D. Mathadalaria	
	Incidents in Public Transport Systems	Fart D: Methodological	

Conclusions

Table 7.1: Published and planned studies

Bibliography

- ABKOWITZ, MARK; HOWARD SLAVIN; ROBERT WAKSMAN; LARRY S. ENGLISHER; NIGEL H.M. WILSON (1978). Transit service reliability. URL: https://rosap.ntl.bts.gov/ view/dot/9212/dot_9212_DS1.pdf (visited on 11/12/2022).
- ADELÉ, SONIA; SABINE TRÉFOND-ALEXANDRE; CORINNE DIONISIO; PIERRE-ALAIN HOYAU (2019). 'Exploring the behavior of suburban train users in the event of disruptions'.
 In: Transportation Research Part F: Traffic Psychology and Behaviour 65, pp. 344–362. ISSN: 13698478. DOI: 10.1016/j.trf.2019.08.009.
- ALLEN, JAIME; JUAN CARLOS MUÑOZ; JUAN DE DIOS ORTÚZAR (2019). 'Understanding Public Transport Satisfaction: Using Maslow's Hierarchy of (Transit) Needs'. In: Transport Policy 81, pp. 75–94. ISSN: 0967070X. DOI: 10.1016/j.tranpol.2019.06.005.
- AMELI, MOSTAFA; JEAN-PATRICK LEBACQUE; LUDOVIC LECLERCQ (2020). 'Improving traffic network performance with road banning strategy: A simulation approach comparing user equilibrium and system optimum'. In: Simulation Modelling Practice and Theory 99, p. 101995. ISSN: 1569-190X. DOI: 10.1016/j.simpat.2019.101995.
- ANDOR, MARK A.; LUKAS FINK; MANUEL FRONDEL; ANDREAS GERSTER; MARCO HORVATH (2021). 'Free public transport: acceptance among the population and possible effects on mobility behavior [Translated from German: Kostenloser ÖPNV: Akzeptanz in der Bevölkerung und mögliche Auswirkungen auf das Mobilitätsverhalten]'. In: List Forum für Wirtschafts- und Finanzpolitik 46.3, pp. 299–325. ISSN: 0937-0862. DOI: 10.1007/s41025-020-00207-y.
- ARRHENIUS, SVANTE (1896). 'XXXI. On the influence of carbonic acid in the air upon the temperature of the ground'. In: *The London, Edinburgh, and Dublin Philosophical Magazine and Journal of Science* 41.251, pp. 237–276. ISSN: 1941-5982. DOI: 10.1080/ 14786449608620846.
- ASSOCIATION OF GERMAN TRANSPORT COMPANIES [TRANSLATED FROM GERMAN: VEREHRSVERBAND DEUTSCHER VERKEHRSUNTERNEHMEN] (VDV) (2001). Telematik im ÖPNV in Deutschland - Telematics in public transport in Germany. Vol. 7. Blaue Buchreihe des VDV. (bilingual source). Duesseldorf: Alba Verlag. ISBN: 3-87094-648-2.
- Association of GERMAN TRANSPORT COMPANIES [TRANSLATED FROM GERMAN: VEREHRSVERBAND DEUTSCHER VERKEHRSUNTERNEHMEN] (VDV) (2018a). VDVscript 453: Actual-data-interface: Connection protection, dynamic passenger information, visualisation, general message service [Translated from German: VDV-Schrift 453: Ist-Daten-Schnittstelle: Anschlusssicherung, Dynamische Fahrgastinformation, Visualisierung, Allgemeiner Nachrichtendienst]. Cologne. URL: https://www.vdv.de/453v24-sds.pdfx? forced=false (visited on 05/09/2023).
- Association of German Transport Companies [Translated from German: Verehrsverband deutscher Verkehrsunternehmen] (VDV) (2018b). VDV-

script 454: Actual-data-interface – Timetable information: Reference data timetable information, timetable information [Translated from German: VDV-Schrift 454: Ist-Daten-Schnittstelle – Fahrplanauskunft: Referenzdaten Fahrplanauskunft, Fahrplanauskunft]. Co-logne. URL: https://www.vdv.de/454v1.2.2-sds.pdf.pdfx (visited on 05/09/2023).

- AUTONOMOUS ADMINISTRATION OF THE PARIS TRANSPORT SERVICES [TRANSLATED FROM FRENCH: RÉGIE AUTONOME DES TRANSPORTS PARISIENS] (RATP) (2023). Multimodal, multimedia information in real-time. URL: https://www.ratp.fr/en/gro upe-ratp/passenger-information-operating-system/multimodal-multimediainformation-real-time (visited on 13/02/2023).
- BACHIR, DANYA; GHAZALEH KHODABANDELOU; VINCENT GAUTHIER; MOUNIM EL YACOUBI; JAKOB PUCHINGER (2019). 'Inferring dynamic origin-destination flows by transport mode using mobile phone data'. In: *Transportation Research Part C: Emerging Technologies* 101, pp. 254–275. ISSN: 0968-090X. DOI: 10.1016/j.trc.2019.02.013.
- BACHMANN, FREDERIK R. (2017). 'Simulative assessment of the impact of a platoon assistant on the capacity of motorways [Translated from German: Simulative Wirkungsuntersuchung eines Kolonnenassistenten auf die Kapazität der Fernstraßeninfrastruktur]'. Master's Thesis. Munich: Technical University of Munich. URL: https://mediatum.ub.tum. de/1403474?show_id=1415808 (visited on 19/12/2018).
- BACHMANN, FREDERIK R.; LARS BRIEM; FRITZ BUSCH; PETER VORTISCH (2022). 'Dynamics and Processes in Operations Control Centers in Urban Public Transport: Potentials for Improvement'. In: *IEEE Transactions on Intelligent Transportation Systems* 23.10, pp. 17819–17834. ISSN: 1558-0016. DOI: 10.1109/TITS.2022.3158821.
- BACHMANN, FREDERIK R.; FLORIAN DANDL; ARSLAN ALI SYED; ROMAN ENGEL-HARDT; KLAUS BOGENBERGER (2023). 'Optimized Passenger Redirection During Incidents in Urban Public Transportation Systems'. In: 102nd Annual Meeting of the Transportation Research Board. URL: https://trid.trb.org/View/2117725 (visited on 17/04/2023).
- BACHMANN, FREDERIK R.; ANDREAS RAU; FRITZ BUSCH (2021). 'Redirecting Passengers and Reallocating Capacities during Incidents in Public Transport'. In: 100th Annual Meeting of the Transportation Research Board. URL: https://trid.trb.org/view/ 1759725 (visited on 17/02/2021).
- BACHMANN, FREDERIK R.; GANESH SETHURAMAN (2019). 'Simulative Assessments of Energy and Space Savings due to Platooning'. In: mobil. TUM 2019: Transportation Systems of the Future: International Scientific Conference on Mobility and Transport. Munich. URL: https://webarchiv.typo3.tum.de/BGU/mobil-vt/fileadmin/w00bqi/www/ mobilTUM2019/Sessions/Manuscripts/6044.pdf (visited on 25/03/2022).
- BACHMANN, FREDERIK R.; ANONTIOS TSAKARESTOS; FRITZ BUSCH; KLAUS BOGEN-BERGER (2023). 'State of the Art of Passenger Redirection during Incidents in Public Transport Systems, Considering Capacity Constraints'. In: *Public Transport*. ISSN: 1613-7159. DOI: 10.1007/s12469-023-00344-5.
- BAI, YUAN; LINA KATTAN (2014). 'Modeling Riders' Behavioral Responses to Real-Time Information at Light Rail Transit Stations'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2412.1, pp. 82–92. ISSN: 0361-1981. DOI: 10.3141/ 2412-10.

- BAST, HANNAH; DANIEL DELLING; ANDREW GOLDBERG; MATTHIAS MÜLLER-HANNEMANN; THOMAS PAJOR; PETER SANDERS; DOROTHEA WAGNER; RENATO F. WERNECK (2016). 'Route Planning in Transportation Networks'. In: *Algorithm Engineering*. Ed. by LASSE KLIEMANN; PETER SANDERS. Vol. 9220. Lecture Notes in Computer Science. Cham: Springer International Publishing, pp. 19–80. ISBN: 978-3-319-49486-9. DOI: 10. 1007/978-3-319-49487-6_2.
- BAUM, HERBERT J. (1973). 'Free Public Transport'. In: Journal of Transport Economics and Policy 7.1, pp. 3–19. ISSN: 00225258. URL: http://www.jstor.org/stable/20052304 (visited on 18/12/2022).
- BELLMAN, RICHARD (1958). 'On a routing problem'. In: *Quarterly of Applied Mathematics* 16.1, pp. 87–90. ISSN: 0033-569X. DOI: 10.1090/qam/102435.
- BERLIN TRANSPORT COMPANY [TRANSLATED FROM GERMAN: BERLINER VERKEHRS-BETRIEBE] (BVG) (2023). URL: https://www.bvg.de/en (visited on 09/03/2023).
- BERLIN TRANSPORT COMPANY [TRANSLATED FROM GERMAN: BERLINER VERKEHRS-BETRIEBE] (BVG) AND CHARITÉ (2017). BVG and Charité testing autonomouses minibuses [Translated from German: BVG und Charité testen autonome Kleinbusse]. URL: https:// www.charite.de/service/pressemitteilung/artikel/detail/bvg_und_charite_ testen_autonome_kleinbusse/ (visited on 01/12/2020).
- BERREBI, SIMON J.; SEAN ÓG CRUDDEN; KARI E. WATKINS (2018). 'Translating research to practice: Implementing real-time control on high-frequency transit routes'. In: *Transportation Research Part A: Policy and Practice* 111, pp. 213–226. ISSN: 0965-8564. DOI: 10.1016/j.tra.2018.03.008.
- BOJIC, IVA; CHUNLING LUO; XINRONG LI; DANIEL ZEHE; DAVID ECKHOFF; CARLO RATTI (2021). 'Optimal Railway Disruption Bridging Using Heterogeneous Bus Fleets'. In: *IEEE Access* 9, pp. 90656–90668. DOI: 10.1109/ACCESS.2021.3091576.
- BOX, GEORGE E.P. (1979). 'Robustness in the Strategy of Scientific Model Building'. In: *Robustness in Statistics Workshop*. Elsevier, pp. 201–236. ISBN: 9780124381506. DOI: 10. 1016/B978-0-12-438150-6.50018-2.
- BRAKEWOOD, CANDACE; KARI WATKINS (2019). 'A literature review of the passenger benefits of real-time transit information'. In: *Transport Reviews* 39.3, pp. 327–356. ISSN: 0144-1647. DOI: 10.1080/01441647.2018.1472147.
- BRIEM, LARS (2023). 'Analysis of disruptions and dispositions in public transport [Translated from German: Analyse von Störungen und Dispositionen im ÖPNV]'. Dissertation. Karlsruhe: Karlsruher Institute of Technology. URL: https://publikationen.bibliothek.kit. edu/1000162270 (visited on 27/09/2023).
- BRIEM, LARS; SEBASTIAN H. BUCK; HOLGER EBHART; NICOLAI MALLIG; BEN STRASSER; PETER VORTISCH; DOROTHEA WAGNER; TOBIAS ZÜNDORF (2017). 'Efficient Traffic Assignment for Public Transit Networks'. In: 16th International Symposium on Experimental Algorithms (SEA 2017). Ed. by COSTAS S. ILIOPOULOS; SOLON P. PISSIS; SIMON J. PUGLISI; RAJEEV RAMAN. Vol. 75. Leibniz International Proceedings in Informatics (LIPIcs). Dagstuhl, Germany: Schloss Dagstuhl-Leibniz-Zentrum fuer Informatik, 20:1– 20:14. ISBN: 978-3-95977-036-1. DOI: 10.4230/LIPIcs.SEA.2017.20.
- BRIEM, LARS; SEBASTIAN H. BUCK; MIRIAM MAGDOLEN; LARUA LANGE; PETER VORTISCH (2020). 'Incident Management in Public Transport Surveying Dispatchers'

Actions'. In: *European Transport Conference*. URL: https://for2083.mathematik.uni-kl.de/documents/publications/tpv_etc_2020.pdf (visited on 17/04/2023).

- BRUGLIERI, MAURIZIO; FRANCESCO BRUSCHI; ALBERTO COLORNI; ALESSANDRO LUÈ; ROBERTO NOCERINO; VINCENZO RANA (2015). 'A Real-time Information System for Public Transport in Case of Delays and Service Disruptions'. In: Transportation Research Procedia 10, pp. 493–502. ISSN: 23521465. DOI: 10.1016/j.trpro.2015.09.003.
- BUEHLER, RALPH; JOHN PUCHER; REGINE GERIKE; THOMAS GÖTSCHI (2017). 'Reducing car dependence in the heart of Europe: lessons from Germany, Austria, and Switzerland'.
 In: *Transport Reviews* 37.1, pp. 4–28. ISSN: 0144-1647. DOI: 10.1080/01441647.2016. 1177799.
- CALLENDAR, G. S. (1938). 'The artificial production of carbon dioxide and its influence on temperature'. In: *Quarterly Journal of the Royal Meteorological Society* 64.275, pp. 223–240. ISSN: 00359009. DOI: 10.1002/qj.49706427503.
- CANCA, DAVID; EVA BARRENA; GILBERT LAPORTE; FRANCISCO A. ORTEGA (2016). 'A short-turning policy for the management of demand disruptions in rapid transit systems'. In: Annals of Operations Research 246.1-2, pp. 145–166. ISSN: 1572-9338. DOI: 10.1007/ s10479-014-1663-x.
- CAO, ZHICHAO; SILIN ZHANG; AVISHAI CEDER (2022). 'Novel Coupling-Decoupling Strategy for Scheduling Autonomous Public Transport Vehicles in Over-crowded Corridors'. In: *Applied Mathematical Modelling*. ISSN: 0307904X. DOI: 10.1016/j.apm.2022.01.020.
- CAREY, MALACHY (1999). 'Ex ante heuristic measures of schedule reliability'. In: *Transportation Research Part B: Methodological* 33.7, pp. 473–494. ISSN: 01912615. DOI: 10.1016/ S0191-2615(99)00002-8.
- CARREL, ANDRÉ (2009). 'Diagnosis and assessment of operations control interventions : framework and applications to a high frequency metro line'. Master's Thesis. Massachusetts Institute of Technology. URL: https://dspace.mit.edu/bitstream/1721.1/55152/2/ 607531524-MIT.pdf (visited on 17/04/2023).
- CARREL, ANDRÉ; ANNE HALVORSEN; JOAN L. WALKER (2013). 'Passengers' Perception of and Behavioral Adaptation to Unreliability in Public Transportation'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2351.1, pp. 153–162. ISSN: 0361-1981. DOI: 10.3141/2351–17.
- CARREL, ANDRÉ; RABI G. MISHALANI; NIGEL H. M. WILSON; JOHN P. ATTANUCCI; ADAM B. RAHBEE (2010). 'Decision Factors in Service Control on High-Frequency Metro Line: Importance in Service Delivery'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2146.1, pp. 52–59. ISSN: 0361-1981. DOI: 10.3141/2146-07.
- CASEY, BRADLEY; ASHISH BHASKAR; HAO GUO; EDWARD CHUNG (2014). 'Critical Review of Time-Dependent Shortest Path Algorithms: A Multimodal Trip Planner Perspective'. In: *Transport Reviews* 34.4, pp. 522–539. ISSN: 0144-1647. DOI: 10.1080/01441647. 2014.921797.
- CATS, ODED (2022). *BusMezzo*. URL: https://odedcats.weblog.tudelft.nl/busmezz o/ (visited on 01/12/2022).

- CATS, ODED; ERIK JENELIUS (2014). 'Dynamic Vulnerability Analysis of Public Transport Networks: Mitigation Effects of Real-Time Information'. In: *Networks and Spatial Economics* 14.3, pp. 435–463. ISSN: 1572-9427. DOI: 10.1007/s11067-014-9237-7.
- CATS, ODED; ERIK JENELIUS (2015). 'Planning for the unexpected: The value of reserve capacity for public transport network robustness'. In: *Transportation Research Part A: Policy and Practice* 81, pp. 47–61. ISSN: 0965-8564. DOI: 10.1016/j.tra.2015.02.013.
- CATS, ODED; GERT-JAAP KOPPENOL; MARTIJN WARNIER (2017). 'Robustness assessment of link capacity reduction for complex networks: Application for public transport systems'. In: *Reliability Engineering & System Safety* 167, pp. 544–553. ISSN: 09518320. DOI: 10.1016/j.ress.2017.07.009.
- CATS, ODED; HARIS N. KOUTSOPOULOS; WILCO BURGHOUT; TOMER TOLEDO (2011). 'Effect of Real-Time Transit Information on Dynamic Path Choice of Passengers'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2217.1, pp. 46–54. ISSN: 0361-1981. DOI: 10.3141/2217-06.
- CATS, ODED; ANAHID NABAVI LARIJANI; HARIS N. KOUTSOPOULOS; WILCO BURG-HOUT (2011). 'Impacts of Holding Control Strategies on Transit Performance'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2216.1, pp. 51– 58. ISSN: 0361-1981. DOI: 10.3141/2216-06.
- CATS, ODED; ANAHID NABAVI LARIJANI; ÁSDÍS ÓLAFSDÓTTIR; WILCO BURGHOUT; INGMAR J. ANDRÉASSON; HARIS N. KOUTSOPOULOS (2012). 'Bus-Holding Control Strategies'. In: Transportation Research Record: Journal of the Transportation Research Board 2274.1, pp. 100–108. ISSN: 0361-1981. DOI: 10.3141/2274-11.
- CATS, ODED; YUSAK O. SUSILO; TRIIN REIMAL (2017). 'The prospects of fare-free public transport: evidence from Tallinn'. In: *Transportation* 44.5, pp. 1083–1104. ISSN: 0049-4488. DOI: 10.1007/s11116-016-9695-5.
- CEBALLOS, GERARDO; PAUL R. EHRLICH; ANTHONY D. BARNOSKY; ANDRÉS GARCÍA; ROBERT M. PRINGLE; TODD M. PALMER (2015). 'Accelerated modern human-induced species losses: Entering the sixth mass extinction'. In: Science advances 1.5, e1400253. ISSN: 2375-2548. DOI: 10.1126/sciadv.1400253.
- CEBECAUER, MATEJ; WILCO BURGHOUT; ERIK JENELIUS; TATIANA BABICHEVA; DAVID LEFFLER (2021). 'Integrating Demand Responsive Services Into Public Transport Disruption Management'. In: *IEEE Open Journal of Intelligent Transportation Systems* 2, pp. 24– 36. DOI: 10.1109/0JITS.2021.3057221.
- CEDER, AVISHAI (1989). 'Optimal design of transit short-turn trips'. In: *Transportation Research Record: Journal of the Transportation Research Board* 1221.557, pp. 8–22. ISSN: 0361-1981.
- CEDER, AVISHAI (2016). Public Transit Planning and Operation: Modeling, Practice and Behavior. 2nd. Boca Raton: Chapman and Hall/CRC. ISBN: 9781466563919.
- CEDER, AVISHAI; YUVAL HADAS; MICHAEL MCIVOR; ALBERT ANG (2013). 'Transfer Synchronization of Public Transport Networks'. In: *Transportation Research Record: Journal* of the Transportation Research Board 2350.1, pp. 9–16. ISSN: 0361-1981. DOI: 10.3141/ 2350–02.

- CEDER, AVISHAI; YU JIANG (2019). 'Personalized public transport mobility service: a journey ranking approach for route guidance'. In: *Transportation Research Procedia* 38, pp. 935–955. ISSN: 23521465. DOI: 10.1016/j.trpro.2019.05.048.
- CHAN, RAYMOND; JOSEPH L. SCHOFER (2014). 'Role of Social Media in Communicating Transit Disruptions'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2415.1, pp. 145–151. ISSN: 0361-1981. DOI: 10.3141/2415-16.
- CHANG, JAMES; JOHN COLLURA; FRANÇOIS DION; HESHAM RAKHA (2003). 'Evaluation of Service Reliability Impacts of Traffic Signal Priority Strategies for Bus Transit'. In: *Transportation Research Record: Journal of the Transportation Research Board* 1841.1, pp. 23–31. ISSN: 0361-1981. DOI: 10.3141/1841-03.
- CHAPMAN, LEE (2007). 'Transport and climate change: a review'. In: Journal of Transport Geography 15.5, pp. 354–367. ISSN: 09666923. DOI: 10.1016/j.jtrangeo.2006.11.008.
- CHOW, VINCENT T.F.; KA WING SUNG; HELEN M. MENG; KA HO WONG; GARY K. S. LEUNG; YONG-HONG KUO; KELVIN K.F. TSOI (2016). 'Utilizing Real-Time Travel Information, Mobile Applications and Wearable Devices for Smart Public Transportation'. In: 2016 7th International Conference on Cloud Computing and Big Data (CCBD). Ed. by IEEE STAFF. Piscataway: IEEE, pp. 138–144. ISBN: 978-1-5090-3555-7. DOI: 10.1109/CCBD.2016.036.
- CITY OF KASSEL [TRANSLATED FROM GERMAN: STADT KASSEL (2023). Figures at a glance [Translated from German: Zahlen im Überblick]. URL: https://www.kassel.de/daten-und-karten/statistik-kassel.php (visited on 24/02/2023).
- COOK, JOHN; DANA NUCCITELLI; SARAH A. GREEN; MARK RICHARDSON; BÄRBEL WINKLER; ROB PAINTING; ROBERT WAY; PETER JACOBS; ANDREW SKUCE (2013). 'Quantifying the consensus on anthropogenic global warming in the scientific literature'. In: *Environmental Research Letters* 8.2, p. 024024. ISSN: 1748-9326. DOI: 10.1088/1748-9326/8/2/024024.
- DAGANZO, CARLOS F.; JOSH PILACHOWSKI (2011). 'Reducing bunching with bus-to-bus cooperation'. In: *Transportation Research Part B: Methodological* 45.1, pp. 267–277. ISSN: 01912615. DOI: 10.1016/j.trb.2010.06.005.
- DAWSON, ROBERT (2011). 'How Significant is a Boxplot Outlier?' In: Journal of Statistics Education 19.2. DOI: 10.1080/10691898.2011.11889610.
- DE CEA, JOAQUÍN; ENRIQUE FERNÁNDEZ (1993). 'Transit Assignment for Congested Public Transport Systems: An Equilibrium Model'. In: *Transportation Science* 27.2, pp. 133– 147. ISSN: 0041-1655. DOI: 10.1287/trsc.27.2.133.
- DELGADO, FELIPE; JUAN CARLOS MUNOZ; RICARDO GIESEN (2012). 'How much can holding and/or limiting boarding improve transit performance?' In: Transportation Research Part B: Methodological 46.9, pp. 1202–1217. ISSN: 01912615. DOI: 10.1016/j.trb. 2012.04.005.
- DELLING, DANIEL; THOMAS PAJOR; RENATO F. WERNECK (2015). 'Round-Based Public Transit Routing'. In: Transportation Science 49.3, pp. 591–604. ISSN: 0041-1655. DOI: 10.1287/trsc.2014.0534.
- DIJKSTRA, E. W. (1959). 'A note on two problems in connexion with graphs'. In: *Numerische Mathematik* 1.1, pp. 269–271. ISSN: 0945-3245. DOI: 10.1007/BF01386390.

- DOLLEVOET, TWAN; DENNIS HUISMAN; MARIE SCHMIDT; ANITA SCHÖBEL (2012). 'Delay Management with Rerouting of Passengers'. In: *Transportation Science* 46.1, pp. 74– 89. ISSN: 0041-1655. DOI: 10.1287/trsc.1110.0375.
- DRABICKI, ARKADIUSZ; ODED CATS; RAFAŁ KUCHARSKI; ACHILLE FONZONE; AN-DRZEJ SZARATA (2023). 'Should I stay or should I board? Willingness to wait with real-time crowding information in urban public transport'. In: *Research in Transportation Business & Management*, p. 100963. ISSN: 2210-5395. DOI: 10.1016/j.rtbm.2023.100963.
- DUROV, PAVEL (2023). Telegram: a new era of messaging. URL: https://telegram.org (visited on 04/12/2023).
- DZIEKAN, KATRIN; KARL KOTTENHOFF (2007). 'Dynamic at-stop real-time information displays for public transport: effects on customers'. In: *Transportation Research Part A: Policy and Practice* 41.6, pp. 489–501. ISSN: 0965-8564. DOI: 10.1016/j.tra.2006.11. 006.
- EBERLEIN, XU JUN (1995). 'Real Time Control Strategies in Transit Operations Models and Analysis'. Dissertation. Boston, MA: Massachusetts Institute of Technology. URL: https:// dspace.mit.edu/bitstream/handle/1721.1/11387/33226950-MIT.pdf?sequence= 2 (visited on 17/04/2023).
- EBERLEIN, XU JUN; NIGEL H. M. WILSON; DAVID BERNSTEIN (1999). 'Modeling Real-Time Control Strategies in Public Transit Operations'. In: *Computer aided transit scheduling*.
 Ed. by NIGEL H. M. WILSON. Vol. 471. Lecture Notes in Economics and Mathematical Systems. Berlin: Springer, pp. 325–346. ISBN: 978-3-540-65775-0. DOI: 10.1007/978-3-642-85970-0 16.
- EBERLEIN, XU JUN; NIGEL H. M. WILSON; DAVID BERNSTEIN (2001). 'The Holding Problem with Real-Time Information Available'. In: *Transportation Science* 35.1, pp. 1–18. ISSN: 0041-1655. DOI: 10.1287/trsc.35.1.1.10143.
- ENGELHARDT, ROMAN; FLORIAN DANDL; ARSLAN ALI SYED; YUNFEI ZHANG; FABIAN FEHN; FYNN WOLF; KLAUS BOGENBERGER (2022). FleetPy: A Modular Open-Source Simulation Tool for Mobility On-Demand Services. DOI: 10.48550/ARXIV.2207.14246.
- EUROPEEAN ENVIRONMENT AGENCY (2022). Air quality in Europe 2021: Sources and emissions of air pollutants in Europe. URL: https://www.eea.europa.eu/publicati ons/air-quality-in-europe-2021/sources-and-emissions-of-air (visited on 18/12/2022).
- FERRIS, BRIAN; KARI WATKINS; ALAN BORNING (2010). 'OneBusAway: Results from Providing Real-Time Arrival Information for Public Transit'. In: CHI '10: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. Ed. by ELIZABETH MYN-ATT; DON SCHONER; GERALDINE FITZPATRICK; SCOTT HUDSON; KEITH EDWARDS; TOM RODDEN. New York: Association for Computing Machinery, pp. 1807–1816. ISBN: 9781605589299. DOI: 10.1145/1753326.1753597.
- FRIEDRICH, MARKUS; MATTHIAS MÜLLER-HANNEMANN; RALF RÜCKERT; ALEXAN-DER SCHIEWE; ANITA SCHÖBEL (2018). 'Robustness as a Third Dimension for Evaluating Public Transport Plans'. In: 18th Workshop on Algorithmic Approaches for Transportation Modelling, Optimization, and Systems (ATMOS 2018). Ed. by RALF BORNDÖR-FER; SABINE STORANDT. Vol. 65. OpenAccess Series in Informatics (OASIcs). Dagstuhl,

Germany: Schloss Dagstuhl–Leibniz-Zentrum fuer Informatik, 4:1–4:17. ISBN: 978-3-95977-096-5. DOI: 10.4230/OASIcs.ATMOS.2018.4.

- FRIENDLY, MICHAEL (2002). 'Corrgrams: Exploratory displays for correlation matrices'. In: *The American Statistician* 56.4, pp. 316–324. ISSN: 0003-1305. DOI: 10.1198/000313002533.
- FU, QIAN; RONGHUI LIU; STEPHANE HESS (2012). 'A Review on Transit Assignment Modelling Approaches to Congested Networks: A New Perspective'. In: *Procedia - Social* and Behavioral Sciences 54, pp. 1145–1155. ISSN: 18770428. DOI: 10.1016/j.sbspro. 2012.09.829.
- GE, LIPING; MALEK SARHANI; STEFAN VOSS; LIN XIE (2021). 'Review of Transit Data Sources: Potentials, Challenges and Complementarity'. In: *Sustainability* 13.20, p. 11450. DOI: 10.3390/su132011450.
- GE, LIPING; STEFAN VOSS; LIN XIE (2022). 'Robustness and disturbances in public transport'. In: *Public Transport* 14.1, pp. 191–261. ISSN: 1613-7159. DOI: 10.1007/s12469-022-00301-8.
- GENTILE, GUIDO; KLAUS NÖKEL (2016). Modelling Public Transport Passenger Flows in the Era of Intelligent Transport Systems: COST Action TU1004 (TransITS). Springer International Publishing. ISBN: 9783319250823. URL: https://link.springer.com/book/ 10.1007/978-3-319-25082-3 (visited on 15/12/2022).
- GERMAN AEROSPACE CENTER [TRANSLATED FROM GERMAN: DEUTSCHES ZENTRUM FÜR LUFT- UND RAUMFAHRT] (DLR) (2022). SUMO User Documentation. URL: http s://sumo.dlr.de/docs/ (visited on 01/12/2022).
- GERMAN FEDERAL MINISTRY OF TRANSPORT AND DIGITAL INFRASTRUCTURE [TRANS-LATED FROM GERMAN: BUNDESMINISTERIUM FÜR VERKEHR UND DIGITALE IN-FRASTRUKTUR] (BMVI) (2019). Collective dynamic passenger guidance - Phase 3: final report [Translated from German: Kollektive dynamische Fahrgastlenkung - Phase 3: Endbericht]. Bonn. URL: https://fops.de/wp-content/uploads/2020/07/70.0917_ Schlussbericht.pdf (visited on 04/09/2023).
- GERMAN FEDERAL MINISTRY OF TRANSPORT, BUILDING AND HOUSING [TRANS-LATED FROM GERMAN: BUNDESMINISTERIUM FÜR VERKEHR, BAUEN UND WOHNEN] (BMVBW) (2001). Strategy for cross-modal passenger guidance in public transport with the help of networked dynamic passenger information in conurbations [Translated from German: Strategie für eine verkehrsmittelübergreifende Fahrgastlenkung im ÖPNV mit Hilfe vernetzter dynamischer Fahrgastinformation in Ballungsräumen]. Bonn. URL: https:// trid.trb.org/view/947527 (visited on 04/09/2023).
- GERMAN FEDERAL MINISTRY OF TRANSPORT, BUILDING AND URBAN DEVELOPMENT [TRANSLATED FROM GERMAN: BUNDESMINISTERIUM FÜR VERKEHR, BAUEN UND STADTENTWICKLUNG] (BMVBS) (2009). Development, strategy and tools for dynamic public transport routing: final report [Translated from German: Entwicklung, Strategie und Instrumentarium für ein dynamisches ÖV-Routing: Endbericht]. Bonn. URL: https:// trid.trb.org/view/930804 (visited on 04/09/2023).
- GERMAN FEDERAL MINISTRY OF TRANSPORT, BUILDING AND URBAN DEVELOPMENT [TRANSLATED FROM GERMAN: BUNDESMINISTERIUM FÜR VERKEHR, BAUEN UND STADTENTWICKLUNG] (BMVBS) (2011). Dynamic passenger routing on collective media

[Translated from German: Dynamisches Fahrgast-Routing auf kollektiven Medien]. Bonn. URL: https://trid.trb.org/view/1118753 (visited on 04/09/2023).

- GERSHENSON, CARLOS; LUIS A. PINEDA (2009). 'Why does public transport not arrive on time? The pervasiveness of equal headway instability'. In: *PLOS ONE* 4.10, e7292. DOI: 10.1371/journal.pone.0007292.
- GKIOTSALITIS, KONSTANTINOS; ODED CATS (2021). 'At-stop control measures in public transport: Literature review and research agenda'. In: *Transportation Research Part E: Logistics and Transportation Review* 145, p. 102176. ISSN: 13665545. DOI: 10.1016/j.tre. 2020.102176.
- GKIOTSALITIS, KONSTANTINOS; Z. WU; ODED CATS (2019). 'A cost-minimization model for bus fleet allocation featuring the tactical generation of short-turning and interlining options'. In: *Transportation Research Part C: Emerging Technologies* 98, pp. 14–36. ISSN: 0968-090X. DOI: 10.1016/j.trc.2018.11.007.
- GLOBAL FOOTPRINT NETWORK (2022). This year, Earth Overshoot Day fell on July 28 Earth Overshoot Day marks the date when humanity has used all the biological resources that Earth regenerates during the entire year. URL: https://www.overshootday.org/ (visited on 18/12/2022).
- GUROBI OPTIMIZATION LLC (2022). Gurobi Optimizer Reference Manual. URL: https: //www.gurobi.com/ (visited on 28/04/2022).
- HAGBERG, ARIC; DAN SCHULT; PIETER SWART (2022). NetworkX Reference. URL: http s://networkx.org/documentation/stable/_downloads/networkx_reference.pdf (visited on 08/12/2022).
- HAMBURGER HOCHBAHN AG (2021). The future is self-driving. URL: https://www. hochbahn.de/hochbahn/hamburg/en/home/projects/expansion_and_projects/ project_heat (visited on 08/12/2022).
- HART, PETER; NILS NILSSON; BERTRAM RAPHAEL (1968). 'A Formal Basis for the Heuristic Determination of Minimum Cost Paths'. In: *IEEE Transactions on Systems Science and Cybernetics* 4.2, pp. 100–107. ISSN: 0536-1567. DOI: 10.1109/TSSC.1968.300136.
- HASSAN, L. K.; B. F. SANTOS; J. VINK (2021). 'Airline disruption management: A literature review and practical challenges'. In: *Computers & Operations Research* 127, p. 105137. ISSN: 03050548. DOI: 10.1016/j.cor.2020.105137.
- HEIMBÖCKL, U. (2005). 'Traffic Surveys in Public Transport: Methods of Optimisation of Glaetting Procedures in the Preparation of Hydrographs/Passenger Counts in Public Transport [Translated from German: Verkehrserhebungen im ÖPNV: Methoden der Optimierung von Glaettungsverfahren bei der Aufbereitung von Ganglinien'. In: *Der Nahverkehr* 23.7/8.
- HELFERS, ANNA; MARISSA REISERER; NATALIE SCHNEIDER; MIRJAM EBERSBACH; CARSTEN SOMMER (2022). 'Should I Stay or Should I Go? Risk Perception and Use of Local Public Transport During the COVID-19 Pandemic'. In: *Frontiers in Psychology* 13, p. 926539. ISSN: 1664-1078. DOI: 10.3389/fpsyg.2022.926539.
- HERSHBERGER, JOHN; MATTHEW MAXEL; SUBHASH SURI (2007). 'Finding the k shortest simple paths'. In: ACM Transactions on Algorithms 3.4, p. 45. ISSN: 1549-6325. DOI: 10. 1145/1290672.1290682.
- HERZ, RAIMUND; HANS G. SCHLICHTER; WILFRIED SIEGNER (1992). Applied Statistics for Transport and Regional Planners [Translated from German: Angewandte Statistik für

Verkehrs- und Regionalplaner]. 2nd ed. Vol. 42. Werner-Ingenieur-Texte. Düsseldorf: Werner. ISBN: 3804119719.

- HOFMANN, D. J.; J. H. BUTLER; E. J. DLUGOKENCKY; J. W. ELKINS; K. MASARIE; S. A. MONTZKA; P. TANS (2006). 'The role of carbon dioxide in climate forcing from 1979 to 2004: Introduction of the Annual Greenhouse Gas Index'. In: *Tellus B: Chemical and Physical Meteorology* 58.5, pp. 614–619. ISSN: 0280-6509. DOI: 10.1111/j.1600– 0889.2006.00201.x.
- HORNI, ANDREAS; KAI NAGEL; KAY W. AXHAUSEN, eds. (2016). The Multi-Agent Transport Simulation MATSim. London: Ubiquity Press. ISBN: 9781909188754. DOI: 10.5334/baw. (Visited on 26/07/2021).
- HRELJA, ROBERT; TOM RYE (2022). 'Decreasing the share of travel by car. Strategies for implementing 'push' or 'pull' measures in a traditionally car-centric transport and land use planning'. In: *International Journal of Sustainable Transportation*, pp. 1–13. ISSN: 1556-8318. DOI: 10.1080/15568318.2022.2051098.
- HU, HUA; YUNFENG GAO; JIE YU; ZHIGANG LIU; XIN LI (2016). 'Planning Bus Bridging Evacuation during Rail Transit Operation Disruption'. In: Journal of Urban Planning and Development 142.4, p. 04016015. ISSN: 0733-9488. DOI: 10.1061/(ASCE)UP.1943-5444.0000335.
- HU, YUZHEN; BAOGUANG XU; JONATHAN F. BARD; HONG CHI; MIN'GANG GAO (2015). 'Optimization of multi-fleet aircraft routing considering passenger transiting under airline disruption'. In: *Computers & Industrial Engineering* 80, pp. 132–144. ISSN: 03608352. DOI: 10.1016/j.cie.2014.11.026.
- HUGHES, TERRY P.; MICHELE L. BARNES; DAVID R. BELLWOOD; JOSHUA E. CINNER; GRAEME S. CUMMING; JEREMY B. C. JACKSON; JOANIE KLEYPAS; INGRID A. VAN DE LEEMPUT; JANICE M. LOUGH; TIFFANY H. MORRISON; STEPHEN R. PALUMBI; EGBERT H. VAN NES; MARTEN SCHEFFER (2017). 'Coral reefs in the Anthropocene'. In: Nature 546.7656, pp. 82–90. DOI: 10.1038/nature22901.
- HUSSAIN, ETIKAF; ASHISH BHASKAR; EDWARD CHUNG (2021). 'Transit OD matrix estimation using smartcard data: Recent developments and future research challenges'. In: *Transportation Research Part C: Emerging Technologies* 125, p. 103044. ISSN: 0968-090X. DOI: 10.1016/j.trc.2021.103044.
- INIT AG (2023). Showcase Montr'eal. URL: https://www.initse.com/ende/projects/ projects-north-america/montreal-showcase/ (visited on 13/02/2023).
- INTERGOVERNMENTAL PANEL ON CLIMATE CHANGE (IPCC) (2018). Global Warming of 1.5°C: IPCC Special Report on Impacts of Global Warming of 1.5°C above Pre-industrial Levels in Context of Strengthening Response to Climate Change, Sustainable Development, and Efforts to Eradicate Poverty. Campridge, United Kingdom: Cambridge University Press. DOI: 10.1017/9781009157940.
- ITANI, ALAA; AMER SHALABY (2021). 'Assessing the Bus Bridging Effectiveness on the Operational Resilience of the Subway Service in Toronto'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2675.9, pp. 1410–1422. ISSN: 0361-1981. DOI: 10.1177/03611981211007836. URL: https://journals.sagepub.com/doi/pdf/10.1177/03611981211007836 (visited on 08/12/2022).

- KASSEL TRANSPORT COMPANY [TRANSLATED FROM GERMAN: KASSLER VERKEHRSGESELL-SCHAFT AG] (KVG) (2023). KVG. URL: https://www.kvg.de/en/ (visited on 24/02/2023).
- KIM, KYUNG MIN; SUNG-PIL HONG; SUK-JOON KO; DOWON KIM (2015). 'Does crowding affect the path choice of metro passengers?' In: *Transportation Research Part A: Policy and Practice* 77, pp. 292–304. ISSN: 0965-8564. DOI: 10.1016/j.tra.2015.04.023.
- KLEIN, TIMOTHEUS (2021). 'New daily hydrographs of source and destination traffic evaluation of the MiD data on the use-specific daily variation of traffic demand [Translated from German: Neue Tagesganglinien des Quell-und Zielverkehrs-Auswertung der MiD-Daten zum nutzungsspezifischen Tagesgang der Verkehrsnachfrage]'. In: *Straßenverkehrstechnik* 65.3, pp. 183–189. URL: https://www.argus-hh.de/wp-content/uploads/2021/04/SVT_ 03-2021_Seiten_183-189.pdf (visited on 25/03/2023).
- KNOWLES, J.; D. CORNE (1999). 'The Pareto archived evolution strategy: a new baseline algorithm for Pareto multiobjective optimisation'. In: *Proceedings of the 1999 congress on evolutionary computation*. IEEE, pp. 98–105. ISBN: 0-7803-5536-9. DOI: 10.1109/CEC. 1999.781913.
- KROON, LEO; GÁBOR MARÓTI; LARS NIELSEN (2015). 'Rescheduling of Railway Rolling Stock with Dynamic Passenger Flows'. In: *Transportation Science* 49.2, pp. 165–184. ISSN: 0041-1655. DOI: 10.1287/trsc.2013.0502.
- KURAUCHI, FUMITAKA; MICHAEL G. H. BELL; JAN-DIRK SCHMÖCKER (2003). 'Capacity Constrained Transit Assignment with Common Lines'. In: *Journal of Mathematical Modelling and Algorithms* 2.4, pp. 309–327. ISSN: 1572-9214. DOI: 10.1023/B: JMMA.0000020426.22501.c1.
- LAM, WILLIAM H. K.; MICHAEL G. H. BELL, eds. (2007). Advanced modeling for transit operations and service planning. Bingley, UK: Emerald. ISBN: 9780080442068. DOI: 10. 1108/9780585475226.
- LARSEN, RUNE; JEPPE RICH; THOMAS KJÆR RASMUSSEN (2019). 'Hub-based truck platooning: Potentials and profitability'. In: *Transportation Research Part E: Logistics and Transportation Review* 127, pp. 249–264. ISSN: 13665545. DOI: 10.1016/j.tre.2019. 05.005.
- LEE, JINWOO; AMER SHALABY; JOHN GREENOUGH; MIKE BOWIE; STANLEY HUNG (2005). 'Advanced Transit Signal Priority Control with Online Microsimulation-Based Transit Prediction Model'. In: *Transportation Research Record: Journal of the Transportation Research Board* 1925.1, pp. 185–194. ISSN: 0361-1981. DOI: 10.1177/0361198105192500119.
- LENG, NUANNUAN; FRANCESCO CORMAN (2020). 'The role of information availability to passengers in public transport disruptions: An agent-based simulation approach'. In: *Transportation Research Part A: Policy and Practice* 133, pp. 214–236. ISSN: 0965-8564. DOI: 10.1016/j.tra.2020.01.007.
- LENG, NUANNUAN; ZHENGWEN LIAO; FRANCESCO CORMAN (2020). 'Role of Timetable, Rolling Stock Rescheduling, and Information Strategies to Passengers in Public Transport Disruptions'. In: Transportation Research Record: Journal of the Transportation Research Board 2674.9, pp. 135–147. ISSN: 0361-1981. DOI: 10.1177/0361198120927000.
- LIN, TEDDY; AMER SHALABY; ERIC MILLER (2016). 'Transit User Behaviour in Response to Service Disruption: State of Knowledge'. In: *Canadian Transportation Research Forum*

51st Annual Conference - North American Transport Challenges in an Era of Change//Les défis des transports en Amérique du Nord à une aire de changement Toronto, Ontario, May 1-4, 2016. URL: file:///C:/Users/gu92fol/Downloads/agecon-trf-1381.pdf (visited on 12/12/2022).

- LIN, TEDDY; SIVA SRIKUKENTHIRAN; ERIC MILLER; AMER SHALABY (2018). 'Subway user behaviour when affected by incidents in Toronto (SUBWAIT) survey A joint revealed preference and stated preference survey with a trip planner tool'. In: *Canadian Journal of Civil Engineering* 45.8, pp. 623–633. ISSN: 0315-1468. DOI: 10.1139/cjce-2017-0442.
- LIORIS, JENNIE; RAMTIN PEDARSANI; FATMA YILDIZ TASCIKARAOGLU; PRAVIN VARAIYA (2017). 'Platoons of connected vehicles can double throughput in urban roads'. In: *Transportation Research Part C: Emerging Technologies* 77, pp. 292–305. ISSN: 0968-090X. DOI: 10.1016/j.trc.2017.01.023.
- LIU, TAO; AVISHAI CEDER; JIHUI MA; WEI GUAN (2014). 'Synchronizing Public Transport Transfers by Using Intervehicle Communication Scheme'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2417.1, pp. 78–91. ISSN: 0361-1981. DOI: 10.3141/2417-09.
- LIU, YULIN; JONATHAN BUNKER; LUIS FERREIRA (2010). 'Transit Users' Route–Choice Modelling in Transit Assignment: A Review'. In: *Transport Reviews* 30.6, pp. 753–769. ISSN: 0144-1647. DOI: 10.1080/01441641003744261.
- LONG, SIHUI; XIAOJIE LUAN; FRANCESCO CORMAN (2021). 'Passenger-oriented rescheduling of trains and rolling stock for handling large passenger demand: linearized models with train capacity constraint'. In: *Transportmetrica B: Transport Dynamics* 9.1, pp. 641– 672. ISSN: 2168-0566. DOI: 10.1080/21680566.2021.1932638.
- LOPEZ, PABLO ALVAREZ; EVAMARIE WIESSNER; MICHAEL BEHRISCH; LAURA BIEKER-WALZ; JAKOB ERDMANN; YUN-PANG FLOTTEROD; ROBERT HILBRICH; LEONHARD LUCKEN; JOHANNES RUMMEL; PETER WAGNER (2018). 'Microscopic Traffic Simulation using SUMO'. In: 21st International Conference on Intelligent Transportation Systems (ITSC). IEEE, pp. 2575–2582. ISBN: 978-1-7281-0321-1. DOI: 10.1109/ITSC.2018. 8569938.
- LUO, CHUNLING; LEI XU (2021). 'Railway disruption management: Designing bus bridging services under uncertainty'. In: *Computers & Operations Research* 131, p. 105284. ISSN: 03050548. DOI: 10.1016/j.cor.2021.105284.
- MANDL, CHRISTOPH E. (1979). Applied Network Optimization. Academic Press Inc. ISBN: 0-12-468350-9.
- MANDL, CHRISTOPH E. (1980). 'Evaluation and optimization of urban public transportation networks'. In: *European Journal of Operational Research* 5.6, pp. 396–404. ISSN: 03772217. DOI: 10.1016/0377-2217(80)90126-5.
- MCMICHAEL, A. J. (2003). Climate change and human health: Risks and responses / editors, A.J. McMichael ... [et al.] Geneva: World Health Organization. ISBN: 924156248X.
- MEADOWS, DONELLA H. (1972). The limits to growth: A report for the Club of Rome's project on the predicament of mankind / Donella H. Meadows ... [et al.] New York: Universe Books. ISBN: 0876631650.
- MEOOOW (2022). k shortest paths and Eppstein's algorithm: Algorithm 1. URL: https: //codeforces.com/blog/entry/102085 (visited on 31/03/2023).

META (2023). facebook. URL: https://www.facebook.com/ (visited on 09/03/2023).

- MIETZSCH, OLIVER (2019). 'Fare reductions as an instrument to increase the attractiveness of public transport [Translated from German: Fahrpreissenkungen als Instrument zur Attrakt-ivitätssteigerung im OPNV]'. In: Verkehr und Technik (V + T) 9. DOI: 10.37307/j.1868-7911.2019.09.07.
- MINISTRY OF TRANSPORT, GOVERNMENT OF SINGAPORE (2022). Fares, Payment Structure, Journey Planning. URL: https://www.mot.gov.sg/what-we-do/publictransport/fares-payment-structure-journey-planning (visited on 31/12/2022).
- MO, BAICHUAN; HARIS N. KOUTSOPOULOS; ZUO-JUN MAX SHEN; JINHUA ZHAO (2023). 'Robust path recommendations during public transit disruptions under demand uncertainty'. In: *Transportation Research Part B: Methodological* 169, pp. 82–107. ISSN: 01912615. DOI: 10.1016/j.trb.2023.02.004. URL: https://www.sciencedirect. com/science/article/pii/S0191261523000176.
- MO, BAICHUAN; HARIS N. KOUTSOPOULOS; JINHUA ZHAO (2022). 'Inferring passenger responses to urban rail disruptions using smart card data: A probabilistic framework'. In: *Transportation Research Part E: Logistics and Transportation Review* 159, p. 102628. ISSN: 13665545. DOI: 10.1016/j.tre.2022.102628.
- MOBILITY DATA (2022). GTFS: Making Public Transit Data Universally Accessible. URL: https://gtfs.org/ (visited on 14/12/2022).
- MOBIMEO GMBH (2023). DB track agent [Translated from German: DB Streckenagent]. URL: https://mobimeo.com/en/db-streckenagent-use-case/ (visited on 13/02/2023).
- MOMENTIVE (2023). Sample Size Calculator. URL: https://www.surveymonkey.com/mp/ sample-size-calculator/ (visited on 25/04/2023).
- MONTRÉAL TRANSPORT COMPANY [TRANLATED FROM FRENCH: SOCIÉTÉ DE TRANS-PORT DE MONTRÉAL] (STM) (2023). *iBUS - Real time*. URL: https://www.stm.info/ en/about/major_projects/completed-major-projects/ibus-real-time (visited on 13/02/2023).
- MORALES, DIEGO; JUAN CARLOS MUÑOZ; PEDRO GAZMURI (2019). 'A stochastic model for bus injection in an unscheduled public transport service'. In: *Transportation Research Part C: Emerging Technologies*. ISSN: 0968-090X. DOI: 10.1016/j.trc.2019.05.029.
- MÜLLER-HANNEMANN, MATTHIAS; RALF RÜCKERT; SEBASTIAN S. SCHMIDT (2019).
 'Vehicle Capacity-Aware Rerouting of Passengers in Delay Management'. In: 19th Symposium on Algorithmic Approaches for Transportation Modelling, Optimization, and Systems (ATMOS 2019). Ed. by VALENTINA CACCHIANI; ALBERTO MARCHETTI-SPACCAMELA. Vol. 75. OpenAccess Series in Informatics (OASIcs). Dagstuhl, Germany: Schloss Dagstuhl–Leibniz-Zentrum fuer Informatik, 7:1–7:14. ISBN: 978-3-95977-128-3. DOI: 10.4230/OASI CS.ATMOS.2019.7.
- NATIONAL RAIL (UK) (2023). Alternative Routes During Disruption. URL: https://www.nationalrail.co.uk/stations_destinations/46655.aspx (visited on 13/02/2023).
- NESHELI, MAHMOOD MAHMOODI (2016). 'Developing Optimal Tactics and Strategies for Public-Transport Operations'. Dissertation. Auckland, New Zealand: University of Auckland. URL: https://researchspace.auckland.ac.nz/bitstream/2292/28793/6/whole. pdf (visited on 08/12/2022).

- NESHELI, MAHMOOD MAHMOODI; AVISHAI CEDER (2014). 'Optimal combinations of selected tactics for public-transport transfer synchronization'. In: *Transportation Research Part C: Emerging Technologies* 48, pp. 491–504. ISSN: 0968-090X. DOI: 10.1016/j.trc.2014. 09.013.
- NGUYEN, S.; S. PALLOTTINO (1988). 'Equilibrium traffic assignment for large scale transit networks'. In: *European Journal of Operational Research* 37.2, pp. 176–186. ISSN: 03772217. DOI: 10.1016/0377-2217(88)90327-X.
- OPENSTREETMAP (OSM) FOUNDATION (2023). OpenStreetMap. URL: https://www.openstreetmap.org (visited on 24/02/2023).
- OUTWATER, MAREN L.; GREG SPITZ; JOHN LOBB; MARGARET CAMPBELL; RAM M. PENDYALA; BHARGAVA SANA; WILLIAM WOODFORD; AECOM CONSULT et al. (2011). 'Project TCRP H-37: Characteristics of Premium Transit Services that Affect Mode Choice: Summary of Phase 1'. In: 90th Annual Meeting of the Transportation Research Board. URL: https://trid.trb.org/view/1092918 (visited on 17/04/2023).
- ÖZKAN, ERHUN; AMY R. WARD (2020). 'Dynamic Matching for Real-Time Ride Sharing'. In: Stochastic Systems 10.1, pp. 29–70. ISSN: 1946-5238. DOI: 10.1287/stsy.2019.0037.
- PANGILINAN, CHRISTOPHER; NIGEL WILSON; ANGELA MOORE (2008). 'Bus Supervision Deployment Strategies and Use of Real-Time Automatic Vehicle Location for Improved Bus Service Reliability'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2063.1, pp. 28–33. ISSN: 0361-1981. DOI: 10.3141/2063-04.
- PASSENGER FOCUS, ed. (2011). Information: Rail Passengers Needs During Unplanned Disruption-Qualitative Research. London, United Kingdom. URL: https://www.transportfocus. org.uk/publication/information-rail-passengers-needs-during-unplanneddisruption/ (visited on 12/12/2022).
- PENDER, BRENDAN; GRAHAM CURRIE; ALEXA DELBOSC; NIRAJAN SHIWAKOTI (2014).
 'Social Media Use during Unplanned Transit Network Disruptions: A Review of Literature'.
 In: Transport Reviews 34.4, pp. 501–521. ISSN: 0144-1647. DOI: 10.1080/01441647.
 2014.915442.
- PETIT, ANTOINE; YANFENG OUYANG; CHAO LEI (2018). 'Dynamic bus substitution strategy for bunching intervention'. In: *Transportation Research Part B: Methodological* 115, pp. 1–16. ISSN: 01912615. DOI: 10.1016/j.trb.2018.06.001.
- PIGEON, CAROLINE; ALINE ALAUZET; LAURENCE PAIRE-FICOUT (2021). 'Factors of acceptability, acceptance and usage for non-rail autonomous public transport vehicles: A systematic literature review'. In: *Transportation Research Part F: Traffic Psychology and Behaviour* 81, pp. 251–270. ISSN: 13698478. DOI: 10.1016/j.trf.2021.06.008.
- POINSIGNON, FRANÇOIS; LEI CHEN; SIDA JIANG; KUN GAO; HUGO BADIA; ERIK JENELIUS (2022). 'Autonomous Vehicle Fleets for Public Transport: Scenarios and Comparisons'. In: *Green Energy and Intelligent Transportation*, p. 100019. ISSN: 27731537. DOI: 10.1016/j.geits.2022.100019.
- PRASHKER, JOSEPH N. (1979). 'Direct analysis of the perceived importance of attributes of reliability of travel modes in urban travel'. In: *Transportation* 8.4, pp. 329–346. ISSN: 0049-4488. DOI: 10.1007/BF00167987.
- PTV GROUP (2023). PTV Visum 2022 Manual. URL: https://www.myptv.com/en (visited on 02/04/2023).

- PYTHON SOFTWARE FOUNDATION (2022). python. URL: https://www.python.org/ (visited on 01/12/2022).
- RAHIMI, EHSAN; ALI SHAMSHIRIPOUR; RAMIN SHABANPOUR; ABOLFAZL MOHAMMA-DIAN; JOSHUA AULD (2019). 'Analysis of transit users' waiting tolerance in response to unplanned service disruptions'. In: *Transportation Research Part D: Transport and Environment*. ISSN: 13619209. DOI: 10.1016/j.trd.2019.10.011.
- RAHIMI, EHSAN; ALI SHAMSHIRIPOUR; RAMIN SHABANPOUR; ABOLFAZL MOHAMMA-DIAN; JOSHUA AULD (2020). 'Analysis of Transit Users' Response Behavior in Case of Unplanned Service Disruptions'. In: *Transportation Research Record: Journal of the Transportation Research Board* 2674.3, pp. 258–271. ISSN: 0361-1981. DOI: 10.1177/0361198120911921.
- RAHIMI, MAHSA; ZAHRA GHANDEHARIOUN; ANASTASIOS KOUVELAS; FRANCESCO COR-MAN (2021). 'Multi-modal management actions for public transport disruptions: an agentbased simulation'. In: 2021 7th International Conference on Models and Technologies for Intelligent Transportation Systems (MT-ITS). IEEE, pp. 1–6. ISBN: 978-1-7281-8995-6. DOI: 10.1109/MT-ITS49943.2021.9529305.
- RAHIMI SIEGRIST, MAHSA; BEDA BÜCHEL; FRANCESCO CORMAN (2022). 'Evaluating the Effects of Disruptions on the Behavior of Travelers in a Multimodal Network Utilizing Agent-Based Simulation'. In: *Transportation Research Record: Journal of the Transportation Research Board*, p. 036119812210940. ISSN: 0361-1981. DOI: 10.1177/ 03611981221094005.
- RAHIMI SIEGRIST, MAHSA; FRANCESCO CORMAN (2021). 'Modeling and Quantifying Interaction of Information and Capacity in Public Transport Disruptions'. In: *Journal of Advanced Transportation* 2021, pp. 1–21. ISSN: 0197-6729. DOI: 10.1155/2021/5398316.
- RAU, ANDREAS; MADHUR JAIN; MENG XIE; TERON NGUYEN; TAO LIU; XIAODONG LIU; Z. UL-ABEDIN (2019). 'Planning and Design of a New Dynamic Autonomous Public Transport System: The DART System in Singapore'. In: 26th ITS World Congress. URL: h ttps://www.researchgate.net/profile/Teron-Nguyen/publication/337730728_ Planning_and_Design_of_a_New_Dynamic_Autonomous_Public_Transport_Syste m_The_DART_System_in_Singapore/links/5de73df8299bf10bc33d6a70/Planningand-Design-of-a-New-Dynamic-Autonomous-Public-Transport-System-The-DART-System-in-Singapore.pdf (visited on 08/12/2022).
- RAU, ANDREAS; LIANGYUAN TIAN; MADHUR JAIN; MENG XIE; TAO LIU; YUAN ZHOU (2018). 'Dynamic Autonomous Road Transit (DART) For Use-case Capacity More Than Bus'. In: Proceedings mobil.TUM2018: Urban Mobility - Shaping the Future Together: International Scientific Conference on Mobility and Transport. URL: https://www.resea rchgate.net/publication/325788529_Dynamic_Autonomous_Road_Transit_DART_ For_Use-case_Capacity_More_Than_Bus (visited on 26/03/2019).
- RESEARCH SOCIETY FOR ROAD AND TRAFFIC ENGINEERING [TRANSLATED FROM GERMAN: FORSCHUNGSGESELLSCHAFT FÜR STRASSEN- UND VERKEHRSWESEN] (FGSV) (2006). Notes on microscopic traffic flow simulation: fundamentels and applications [Translated from German: Hinweise zur mikroskopischen Verkehrsflusssimulation: Grundlagen und Anwendungen]. Köln. URL: https://www.fgsv-verlag.de/verkehrsflusssimulation (visited on 25/04/2023).

- REVKIN, ANDREW C. (2016). 'New Coverage of Coal's Link to Global Warming, in 1912'. In: New York Times. URL: https://archive.nytimes.com/dotearth.blogs.nytimes. com/2016/10/21/coals-link-to-global-warming-explained-in-1912/ (visited on 18/12/2022).
- RODRIGUEZ, JOSEPH; HARIS N. KOUTSOPOULOS; SHENHAO WANG; JINHUA ZHAO (2023). 'Cooperative bus holding and stop-skipping: A deep reinforcement learning framework'. In: *Transportation Research Part C: Emerging Technologies* 155, p. 104308. ISSN: 0968-090X. DOI: 10.1016/j.trc.2023.104308.
- ROTHE, IRENE; HERBERT SÜSSE; KLAUS VOSS (1996). 'The method of normalization to determine invariants'. In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 18.4, pp. 366–376. ISSN: 01628828. DOI: 10.1109/34.491618.
- RÜCKERT, RALF; MARTIN LEMNIAN; CHRISTOPH BLENDINGER; STEFFEN RECHNER; MATTHIAS MÜLLER-HANNEMANN (2017). 'PANDA: a software tool for improved train dispatching with focus on passenger flows'. In: *Public Transport* 9.1-2, pp. 307–324. ISSN: 1613-7159. DOI: 10.1007/s12469-016-0140-0.
- SÁEZ, DORIS; CRISTIÁN E. CORTÉS; FREDDY MILLA; ALFREDO NÚÑEZ; ALEJANDRO TIRACHINI; MARCELA RIQUELME (2012). 'Hybrid predictive control strategy for a public transport system with uncertain demand'. In: *Transportmetrica* 8.1, pp. 61–86. ISSN: 1812-8602. DOI: 10.1080/18128601003615535.
- SALONEN, ARTO O. (2018). 'Passenger's subjective traffic safety, in-vehicle security and emergency management in the driverless shuttle bus in Finland'. In: *Transport Policy* 61, pp. 106–110. ISSN: 0967070X. DOI: 10.1016/j.tranpol.2017.10.011.
- SCHMÖCKER, JAN-DIRK; MICHAEL G. H. BELL; FUMITAKA KURAUCHI (2008). 'A quasidynamic capacity constrained frequency-based transit assignment model'. In: *Transportation Research Part B: Methodological* 42.10, pp. 925–945. ISSN: 01912615. DOI: 10.1016/j. trb.2008.02.001.
- SCHMÖCKER, JAN-DIRK; SHOSHANA COOPER; WILLIAM ADENEY (2005). 'Metro Service Delay Recovery'. In: Transportation Research Record: Journal of the Transportation Research Board 1930.1, pp. 30–37. ISSN: 0361-1981. DOI: 10.1177/0361198105193000104.
- SCHMÖCKER, JAN-DIRK; ACHILLE FONZONE; HIROSHI SHIMAMOTO; FUMITAKA KUR-AUCHI; MICHAEL G. H. BELL (2011). 'Frequency-based transit assignment considering seat capacities'. In: *Transportation Research Part B: Methodological* 45.2, pp. 392–408. ISSN: 01912615. DOI: 10.1016/j.trb.2010.07.002.
- SCHRÖDER, DANIEL; LUKAS KIRN; JULIA KINIGADNER; ALLISTER LODER; PHILIPP BLUM; YIHAN XU; MARKUS LIENKAMP (2022). 'Ending the myth of mobility at zero costs: An external cost analysis'. In: *Research in Transportation Economics*, p. 101246. ISSN: 07398859. DOI: 10.1016/j.retrec.2022.101246.
- SETHURAMAN, GANESH; XIAODONG LIU; FREDERIK R. BACHMANN; MENG XIE; AY-BIKE ONGEL; FRITZ BUSCH (2019). 'Effects of Bus Platooning in an Urban Environment'.
 In: 2019 IEEE Intelligent Transportation Systems Conference (ITSC). IEEE, pp. 974–980. ISBN: 978-1-5386-7024-8. DOI: 10.1109/ITSC.2019.8917041.
- SHEEHAN, BARRY; FINBARR MURPHY; MARTIN MULLINS; CIAN RYAN (2019). 'Connected and autonomous vehicles: A cyber-risk classification framework'. In: *Transportation*

Research Part A: Policy and Practice 124, pp. 523-536. ISSN: 0965-8564. DOI: 10.1016/j.tra.2018.06.033.

- SINCLAIR, KARINE; JEAN-FRANCOIS CORDEAU; GILBERT LAPORTE (2014). 'Improvements to a large neighborhood search heuristic for an integrated aircraft and passenger recovery problem'. In: *European Journal of Operational Research* 233.1, pp. 234–245. ISSN: 03772217. DOI: 10.1016/j.ejor.2013.08.034.
- SINGAPORE DEPARTMENT OF STATISTICS (2022). Empowering You with Trusted Data. URL: https://www.singstat.gov.sg/ (visited on 27/12/2022).
- SOHN, KEEMIN; JIYOUNG YUN (2009). 'Separation of car-dependent commuters from normalchoice riders in mode-choice analysis'. In: *Transportation* 36.4, pp. 423–436. ISSN: 0049-4488. DOI: 10.1007/s11116-009-9209-9.
- SOMMER, CARSTEN; RAMÓN BRIEGEL; JONAS HARZ; MARISSA REISERER (2022). 'Strategies for meeting climate protection targets in transport Integrated push and pull strategy required at all planning levels [Translated from German: Strategien zur Einhaltung der Klimaschutzziele im Verkehr Integrierte Push-und Pull-Strategie auf allen Planungsebenen erforderlich]'. In: Der Nahverkehr. URL: https://www.uni-kassel.de/fb14bau/index.php?eID= dumpFile&t=f&f=2504&token=2ce7ee90b87237b9f9c8f14609629e3f4dad0a41 (visited on 02/06/2022).
- SPIESS, HEINZ; MICHAEL FLORIAN (1989). 'Optimal strategies: A new assignment model for transit networks'. In: *Transportation Research Part B: Methodological* 23.2, pp. 83–102. ISSN: 01912615. DOI: 10.1016/0191-2615(89)90034-9.
- SRYHENI, SAID (2022). Find All Simple Paths Between Two Vertices in a Graph. URL: ht tps://www.baeldung.com/cs/simple-paths-between-two-vertices (visited on 28/03/2023).
- STATE OF BERLIN (2022). Tickets & line networks: Tickets, fare zones and route plans of the public transport system in Berlin [Translated from German: Fahrkarten, Tickets & Liniennetze: Fahrkarten, Tarifzonen und Linienpläne des öffentlichen Nahverkehrs in Berlin]. URL: https://www.berlin.de/tourismus/infos/nahverkehr/1772016-1721041fahrkarten-tickets-liniennetze.html (visited on 31/12/2022).
- STATISTICS OFFICE BERLIN-BRANDENBURG [TRANSLATED FROM GERMAN: AMT FÜR STATISTIK BERLIN-BRANDENBURG] (2022). From numbers to information: Statistics Berlin-Brandenburg [Translated from German: Von der Zahl zur Information: Statistik Berlin-Brandenburg]. URL: https://www.statistik-berlin-brandenburg.de/ (visited on 27/12/2022).
- STRATHMAN, JAMES; THOMAS KIMPEL; KENNETH DUEKER; RICHARD GERHART; KENNETH TURNER; DAVID GRIFFIN; STEVE CALLAS (2001). 'Bus Transit Operations Control: Review and an Experiment Involving Tri-Met's Automated Bus Dispatching System'. In: Journal of Public Transportation 4.1. ISSN: 1077-291X. DOI: 10.5038/2375-0901.4.1.1.
- TAEIHAGH, ARAZ; HAZEL SI MIN LIM (2019). 'Governing autonomous vehicles: emerging responses for safety, liability, privacy, cybersecurity, and industry risks'. In: *Transport Reviews* 39.1, pp. 103–128. ISSN: 0144-1647. DOI: 10.1080/01441647.2018.1494640.
- TIRACHINI, ALEJANDRO; CRISTIÁN E. CORTÉS; SERGIO R. JARA-DÍAZ (2011). 'Optimal design and benefits of a short turning strategy for a bus corridor'. In: *Transportation* 38.1, pp. 169–189. ISSN: 0049-4488. DOI: 10.1007/s11116-010-9287-8.

- TOWER TRANSIT SINGAPORE (2020). *Tower Transit*. URL: https://towertransit.sg/ (visited on 09/03/2023).
- TRANSPORT FOR LONDON (TFL) (2023). A list of available TfL data feeds and guidelines for using them. URL: https://tfl.gov.uk/info-for/open-data-users/our-open-data#on-this-page-2 (visited on 13/02/2023).
- TRANSPORT FOR LONDON (TFL) CUSTOMER SERVICE AND OPERATIONAL PANEL (2017). TfL Customer Information Strategy and Programme. URL: https://content. tfl.gov.uk/csopp-20170302-part-1-item06-customer-information-strategyand-programme.pdf (visited on 13/02/2023).
- TRANSPORT FOR LONDON (TFL) SAFETY, ACCESSIBILITY AND SUSTAINABILITY PANEL (2016). TfL Customer Information Strategy. URL: https://content.tfl.gov. uk/sasp-20160310-p1-item13-customer-information-strategy.pdf (visited on 13/02/2023).
- TRANSPORT FOR NEW SOUTH WALES (TFNSW) (2023). Incident management framework. URL: https://www.transport.nsw.gov.au/railcorp/access/emergencypreparedness (visited on 17/02/2023).
- TSUCHIYA, R.; Y. SUGIYAMA; K. YAMAUCHI; K. FUJINAMI; R. ARISAWA; T. NAK-AGAWA (2006). 'Route-choice support system for passengers in the face of unexpected disturbance of train operations'. In: *Computers in railways X*. Ed. by J. ALLAN; C. A. BREBBIA; A. F. RUMSEY; G. SCIUTTO; S. SONE; C. J. GOODMAN. WIT Transactions on The Built Environment. Southampton: WIT, pp. 189–197. ISBN: 1845641779. DOI: 10.2495/CR060191.
- TUMCREATE LIMITED (2021). Towards the Ultimate Public Transport System. URL: https://www.tum-create.edu.sg/content/towards-ultimate-public-transport -system-0 (visited on 27/12/2022).
- UBER TECHNOLOGIES INC. (2022). How does Uber match riders with drivers? URL: https: //www.uber.com/us/en/marketplace/matching/ (visited on 20/12/2022).
- UL ABEDIN, ZAIN (2019). 'A Methodology to Design Multimodal Public Transit Networks -Procedures and Applications'. Dissertation. Munich: Technical University of Munich. URL: https://mediatum.ub.tum.de/1482714 (visited on 17/04/2023).
- UL ABEDIN, ZAIN; FRITZ BUSCH; DAVID WANG; ANDREAS RAU; DU BO (2018). 'Comparison of Public Transport Network Design Methodologies Using Solution-Quality Evaluation'. In: *Journal of Transportation Engineering, Part A: Systems* 144.8. DOI: 10.1061/ JTEPBS.0000159.
- VAN DER HURK, EVELIEN; LEO KROON; GÁBOR MARÓTI (2018). 'Passenger Advice and Rolling Stock Rescheduling Under Uncertainty for Disruption Management'. In: *Transportation Science* 52.6, pp. 1391–1411. ISSN: 0041-1655. DOI: 10.1287/trsc.2017.0759.
- VAN LIEROP, DEA; MADHAV G. BADAMI; AHMED M. EL-GENEIDY (2018). 'What influences satisfaction and loyalty in public transport? A review of the literature'. In: *Transport Reviews* 38.1, pp. 52–72. ISSN: 0144-1647. DOI: 10.1080/01441647.2017.1298683.
- VEELENTURF, LUCAS P.; LEO G. KROON; GÁBOR MARÓTI (2017). 'Passenger oriented railway disruption management by adapting timetables and rolling stock schedules'. In: *Transportation Research Part C: Emerging Technologies* 80, pp. 133–147. ISSN: 0968-090X. DOI: 10.1016/j.trc.2017.04.012.

- VOLVO AB (2019). NTU Singapore and Volvo unveil world's first full size, autonomous electric bus. URL: https://www.volvobuses.com/en/news/2019/mar/volvoand-singapore-university-ntu-unveil-world-first-full-size-autonomouselectric-bus.html (visited on 08/12/2022).
- VUCHIC, VUKAN (1969). 'Propagation of Schedule Disturbances in Line-Haul Passenger Transportation'. In: UITP Revue, pp. 281-284. URL: https://repository.upenn.edu/ ese_papers/746.
- WANG, JIADONG; ZHENZHOU YUAN; ZHICHAO CAO; ZIQI LU (2021). 'Optimal Bus Bridging Schedule with Transfer Passenger Demand during Disruptions of Urban Rail Transit'. In: Journal of Transportation Engineering, Part A: Systems 147.10, p. 04021071. ISSN: 2473-2907. DOI: 10.1061/JTEPBS.0000568.
- WANG, SHENHAO; JINHUA ZHAO (2019). 'Risk preference and adoption of autonomous vehicles'. In: *Transportation Research Part A: Policy and Practice* 126, pp. 215–229. ISSN: 0965-8564. DOI: 10.1016/j.tra.2019.06.007.
- WATERSON, B. J.; B. RAJBHANDARI; N. B. HOUNSELL (2003). 'Simulating the Impacts of Strong Bus Priority Measures'. In: *Journal of Transportation Engineering* 129.6, pp. 642–647. ISSN: 0733-947X. DOI: 10.1061/(ASCE)0733-947X(2003)129:6(642).
- WATKINS, KARI EDISON; BRIAN FERRIS; ALAN BORNING; G. SCOTT RUTHERFORD; DAVID LAYTON (2011). 'Where Is My Bus? Impact of mobile real-time information on the perceived and actual wait time of transit riders'. In: *Transportation Research Part A: Policy* and Practice 45.8, pp. 839–848. ISSN: 0965-8564. DOI: 10.1016/j.tra.2011.06.010.
- WEINSZIEHR, LEON; ANONTIOS TSAKARESTOS; FREDERIK R. BACHMANN; KLAUS BO-GENBERGER (2023). 'Detection of Bus Bunching through the Analysis of Prevalent Public Transport Control Data'. In: 11th Symposium of the European Association for Research in Transportation (hEART). URL: https://mediatum.ub.tum.de/node?id=1720493 (visited on 15/11/2023).
- WIKIPEDIA (2023). Box plot. URL: https://en.wikipedia.org/wiki/Box_plot# Alternative_forms (visited on 31/03/2023).
- WILKE, KIM (2023). 'Influencing Demand in Rail Passenger Transport in Incident Situations - Development of a Methodology for Modelling the Decision Behaviour of Passengers [Translated from German: Nachfragebeeinflussung im Schienenpersonenverkehr in Störfallsituationen - Entwicklung einer Methodik zur Modellierung des Entscheidungsverhaltens von Reisenden]'. Dissertation. Darmstadt: Technical University of Darmstadt [Translated from German: Technische Universität Darmstadt]. DOI: 10.26083/tuprints-00023110.
- WOLRD HEALTH ORGANIZATION (2021). Listings of WHO's response to COVID-19. URL: https://www.who.int/news/item/29-06-2020-covidtimeline (visited on 27/12/2022).
- WU, YUANYUAN; DAVID Z.W. WANG; FENG ZHU (2022). 'Influence of CAVs platooning on intersection capacity under mixed traffic'. In: *Physica A: Statistical Mechanics and its Applications* 593, p. 126989. ISSN: 03784371. DOI: 10.1016/j.physa.2022.126989.
- YAP, MENNO D.; NIELS VAN OORT; ROB VAN NES; BART VAN AREM (2018). 'Identification and quantification of link vulnerability in multi-level public transport networks: a passenger perspective'. In: *Transportation* 45.4, pp. 1161–1180. ISSN: 0049-4488. DOI: 10.1007/s11116-018-9892-5.

- YUEN, KUM FAI; LING QIAN CHOO; XUE LI; YIIK DIEW WONG; FEI MA; XUEQIN WANG (2022). 'A theoretical investigation of user acceptance of autonomous public transport'. In: *Transportation*, pp. 1–25. ISSN: 0049-4488. DOI: 10.1007/s11116-021-10253w.
- ZHU, YONGQIU; ROB M.P. GOVERDE (2019a). 'Dynamic Passenger Assignment for Major Railway Disruptions Considering Information Interventions'. In: Networks and Spatial Economics 19.4, pp. 1249–1279. ISSN: 1572-9427. DOI: 10.1007/s11067-019-09467-4.
- ZHU, YONGQIU; ROB M.P. GOVERDE (2019b). 'Railway timetable rescheduling with flexible stopping and flexible short-turning during disruptions'. In: *Transportation Research Part B: Methodological* 123, pp. 149–181. ISSN: 01912615. DOI: 10.1016/j.trb.2019.02.015.
- ZHU, YONGQIU; ROB M.P. GOVERDE (2020). 'Integrated timetable rescheduling and passenger reassignment during railway disruptions'. In: *Transportation Research Part B: Methodological* 140, pp. 282–314. ISSN: 01912615. DOI: 10.1016/j.trb.2020.09.001.

List of Abbreviations

AFaO AFC APC API ARSRU AVL	adapted fix and optimise 31 automatic fare collection 7, 79, 167, 168 automatic passenger count 7, 40, 79, 167 application programming interface 15, 39, 95, 144 advice and rolling stock rescheduling problem with uncertainty 31, 36, 161 automatic vehicle location 7, 19, 20
BMVBS	German Federal Ministry of Transport, Building, and Urban Development [Translated from German: Bundesministerium für Vekehr, Bauen und Stadtentwicklung] 39
BMVBW	German Federal Ministry of Transport, Building, and Housing [Translated from German: Bundesministerium für Vekehr, Bauen und Wohnen] 30
BMVI	German Federal Ministry of Transportation and digital Infrastructure [Trans- lated from German: Bundesministerium für Vekehr und digitale Infrastruk- tur] 28, 40
BVG	Berlin transport company [Translated from German: Berliner Verkehrsbe- triebe] 47, 57, 61
CAPR CR+HR	capacity-aware passenger redirection 23, 75, 165, 166, 175, 207 capacity reallocation in combination with heuristic redirection 102, 104, 106, 110, 111, 116, 137, 145–147, 153, 159, 209, 226, 232
CR+OR	capacity reallocation in combination with optimised redirection 102, 104, 106, 110, 111, 116, 137, 140, 145–147, 153, 159, 209, 210, 227, 233
DART	Dynamic Autonomous Road Transit 4, 18, 28, 60, 143–148, 154, 173, 175, 207
DB DLR	German rail [Translated from German: Deutsche Bahn] 25, 38 German Aerospace Center [Translated from German: Deutsches Zentrum für Luft- und Raumfahrt] 93
EFA	electronic timetable information [Translated from German: Elektronische Fahrplan Auskunft] 40
GPS GTFS	global positioning system 15, 19, 21 general transit feed specification 15, 78, 165

HAFAS	Hanover consulting timetable information system [Translated from German: Hannover Consulting Fahrplan Auskunftssystem] 40
HR	heuristic redirection 102, 104, 116, 135–138, 140, 145–147, 152, 158, 159, 173, 209, 224, 230
IC IPCC	incident channel 66, 67 Intergovernmental Panel on Climate Change 1
IQR ITCS	inter quartile range 125, 127 intermodal transport control system 19, 39, 78, 95, 165–167, 207
KVG	Kassel transport company [Translated from German: Kasseler Verkehrsgesellschaft] 148, 149
LR	PT line-rerouting 98, 99, 102–104, 114–116, 118–120, 122, 123, 127–130, 132–142, 144, 145, 147, 148, 154, 155, 163, 172, 206, 207, 209, 210, 230–236
LRT LS	light rail transit 53, 54 PT line-splitting 98, 99, 102–105, 108, 109, 112–116, 122, 124, 126, 127, 130, 131, 133–142, 144–147, 151, 153–155, 163, 172, 206, 207, 209, 224–229
MaaS MATSim MRT	Mobility-as-a-Service 38 Multi-Agent Transport Simulation 27, 31, 36
	mass rapid transit 52–55, 56
NTU NVV	Nanyang Technological University 61 North Hessian transport association [Translated from German: Nordhessis- cher Verkehrsverbund] 148, 149
NTU NVV OCC	Nanyang Technological University 61 North Hessian transport association [Translated from German: Nordhessis- cher Verkehrsverbund] 148, 149 operations control centre 4, 12, 16, 19–22, 24, 33, 34, 39–43, 46, 71, 74, 76–78, 80, 95, 155, 157, 160, 163–169, 173, 175, 205, 207, 212
NTU NVV OCC OD	Nanyang Technological University 61 North Hessian transport association [Translated from German: Nordhessis- cher Verkehrsverbund] 148, 149 operations control centre 4, 12, 16, 19–22, 24, 33, 34, 39–43, 46, 71, 74, 76–78, 80, 95, 155, 157, 160, 163–169, 173, 175, 205, 207, 212 origin-destination 7, 25, 29, 74–76, 79, 80, 82, 83, 85–87, 89–92, 95–98, 100, 101, 106, 107, 111, 125, 130, 135, 140–142, 144, 149, 151, 158–161, 164, 167, 174, 202, 203, 207, 209, 221–223
NTU NVV OCC OD OR	Nanyang Technological University 61 North Hessian transport association [Translated from German: Nordhessis- cher Verkehrsverbund] 148, 149 operations control centre 4, 12, 16, 19–22, 24, 33, 34, 39–43, 46, 71, 74, 76–78, 80, 95, 155, 157, 160, 163–169, 173, 175, 205, 207, 212 origin-destination 7, 25, 29, 74–76, 79, 80, 82, 83, 85–87, 89–92, 95–98, 100, 101, 106, 107, 111, 125, 130, 135, 140–142, 144, 149, 151, 158–161, 164, 167, 174, 202, 203, 207, 209, 221–223 optimised redirection 102–104, 106, 116, 135–138, 145–147, 152, 158, 159, 173, 209, 225, 231
NTU NVV OCC OD OR OSM	Nanyang Technological University 61 North Hessian transport association [Translated from German: Nordhessis- cher Verkehrsverbund] 148, 149 operations control centre 4, 12, 16, 19–22, 24, 33, 34, 39–43, 46, 71, 74, 76–78, 80, 95, 155, 157, 160, 163–169, 173, 175, 205, 207, 212 origin-destination 7, 25, 29, 74–76, 79, 80, 82, 83, 85–87, 89–92, 95–98, 100, 101, 106, 107, 111, 125, 130, 135, 140–142, 144, 149, 151, 158–161, 164, 167, 174, 202, 203, 207, 209, 221–223 optimised redirection 102–104, 106, 116, 135–138, 145–147, 152, 158, 159, 173, 209, 225, 231 Open Street Map 149

PDA PI	passenger demand assignment 7, 8 passenger information 10, 13–15, 23–27, 29–31, 38–40, 42, 43, 46, 48, 53, 54, 56, 66–68, 71–73, 75, 77, 80, 82, 91–93, 96, 100, 140, 161–165, 168, 172
PMD PROUD PSC PT	personal mobile device 49 Passenger ROUting during Distruption 168 path search channel 66 public transport v, 1–4, 7–21, 23–33, 36, 38–43, 46–50, 52–67, 71–81, 83, 85–87, 89–101, 103, 104, 107, 110, 111, 114, 116, 121, 127, 133–135, 138–140, 143–147, 149–152, 154–168, 171–175, 201–203, 205–207, 209, 211
ΡΤV	PTV Planning Transportation Traffic [Translated from German: Planen Transport und Verkehr GmbH] 93, 149
RAPTOR RATP	round-based public transit optimized router 31, 36 the autonomous administration of the Paris transport services [Translated from French: Régie Autonome des Transports Parisiens] 38
REPACK	REdirecting Passengers and reAllocating Capacities Knowingly 71–74, 76– 78, 83, 91–99, 102, 106, 107, 109, 111, 116, 117, 121, 123, 133, 138, 140, 148, 151, 152, 154, 155, 157–169, 171–176, 205, 207
SMS STM STN SUBT SUMO	short message service 15 Montréal transport company [Translated from French: Soviété de Transport de Montréal] 39 simple toy network 71, 72, 160, 205 suburban train 52, 54 Simulation of Urban MObility 28, 31, 36, 93, 95, 97, 144, 149
TfL TfNSW TraCl	Transport for London 38 Transport for New South Wales 38 traffic contol interface 95, 144
UK USA	United Kingdom 38 United States of America 29, 32
V2V VBB VDV	vehicle-to-vehicle 143 transport association Berlin/Brandenburg [Translated from German: Verkehrsverbund Berlin/Brandenburg] 40 association of German transport companies [Translated from German: Verb- and Deutscher Verkehrsunternehmen] 39, 78, 165

ZRK Kassel region special purpose association [Translated from German: Zweck-verband Raum Kassel] 148

List of Symbols

$B \\ b \in B_{sup}^{pl} \\ b \in B_{sup} \\ b \in B$	Set of all buses 87 Bus in set of support buses, assigned to PT support line $pl \in PL_{sup}$ 90 Bus in set of support buses 89–91, 202, 203 Bus in set of all buses 89, 90
C_l C_p	Remaining link capacity of link $l \in L$ 77, 85, 86 Remaining path capacity of path $p \in P$ 77, 83, 89
$\begin{split} G &= (N,L) \\ rg_{od}^{\hat{p}} \\ rg_{od}^{p} \\ sg_{od} \\ wg_{od}^{\hat{p}} \\ wg_{od}^{p} \end{split}$	Network graph existing of set of all nodes N and set of all links L 77, 80 Group of passengers, associated with od , being redirected onto their respective fastest path $\hat{p} \in P_{od}$ 82, 85, 87, 106, 140, 158, 159, 201 Group of passengers, associated with od , being redirected onto path $p \in P_{od}$ 82, 83, 85–87, 89–91, 100, 101, 106, 152, 158, 167, 202 Group of self-redirected passengers, associated with od 81, 82, 85, 89 Group of passengers, associated with od , waiting for the dissolution of the incident and staying on their respective $0 \in P_{od}$, while respective $rg_{od}^{\hat{p}}$ on $p \in P_{od}$ 82 Group of passengers, associated with od , waiting for the dissolution of the incident and staying on their respective original path $0 \in P_{od}$, while respective $rg_{od}^{\hat{p}}$ on the respective fastest path $\hat{p} \in P_{od}$ 82, 83, 85, 86, 89, 90, 100
\breve{h}_p	Longest headway among the PT lines which are serving of path $p \in P$ 81, 101, 103, 130, 203
$L_i \subset L$ $L_p \subset L$ L $l \in L_p$ $l \in L$	Set of links disrupted by an incident, subset of L 77 Set of links which are part of $p \in P$, subset of L 77 Set of all links 77, 201 Link in set of links which are part of path $p \in P$ 85, 86, 201 Link in set of all links 77, 85, 86, 201
$\begin{split} M &\in 0, 1^{P_{od}xL} \\ m_{pl} \end{split}$	Path-link matrix, in which the element m_{pl} is 1 if link $l \in L_p$ 85, 86, 201 Binary element of $M \in 0, 1^{P_{od}xL}$ which is 1 if link $l \in L_p$ 85, 86
$N_L \subset N$	Set of line nodes, PT lines serving corresponding stop nodes 77

$N_P \subset N$ N	Set of stop nodes, reflecting physical PT stops 77 Set of all nodes 77, 201
$\begin{array}{l} OD\\ OD_a \subset OD\\ od \in OD_a \end{array}$	Set of all OD pairs 80, 202 Set of affected OD pairs, subset of <i>OD</i> 75, 80, 82, 85–87 OD pair in the set of all affected OD pairs 75, 80–83, 85–87, 90–92, 95, 100, 101, 106, 111, 130, 140–142, 144, 151, 158, 159, 161, 164, 167, 206
$\begin{array}{l} 0 \in P_{od} \\ P_{od} \subset P \\ P \end{array}$	Original path of od 81–83, 85, 90, 95, 100, 201 Set of alternative paths, associated with od , subset of P 80, 83 Set of all paths 202
$\tilde{p} \in P_{od}$	Assigned path of rg_{od}^p associated with od 83, 87, 90, 91, 95, 100, 106, 130, 203
$\hat{p} \in P_{od}$	Fastest path among $p \in P_{od}$, associated with od 80–83, 86, 87, 100, 101, 130, 201
$p \in P_{od}$	Path in set of available alternative paths for $od 81-83$, 85-87, 89, 90, 106, 158, 201-203
$p \in P$	Path in set of all paths 77, 201
$PL_{ins} \subset PL$	Set of insufficient lines, subset of PL 86
$PL_{sup} \subset PL$	Set of support lines, subset of PL 86, 87, 89, 90, 202
PL	Set of all PT lines 86, 202
$pl \in PL_{ins}^{ps}$	PT line in the set of insufficient lines, associated with path segment ps , subset of PL_{ins} 89, 90, 159, 203
$pl \in PL_{sup}$	PT line in set of support lines $PL_{sup} \subset PL$ 90, 201, 202
$ps \in PS_p$	Path segment in set of path segments of path p with insufficient remaining capacity 89, 90, 159, 202, 203
$ps \in PS_p$	Path segment in set of path segments of path p 83, 89
TT^p_{od}	Total travel time of od , if assigned to $p \in P_{od}$ 85
$\Delta t t_u$	Delay of PT user $u \in U_a$ 76
t_i^e	End time of the incident 77, 81, 85, 203
t_i^s	Start time of the incident 77, 79, 81, 82, 85
U_{arr}^{b}	Arrival time at the origin of redirection 82 Original departure time of hus $h \in D$, at $m \in DI$, 00
ι_{dep}	Original departure time of bus $b \in D_{sup}$ at $pl \in FL_{sup}$ 90 Time constant to consider delays due to microd PT vehicles and interference.
ι_{con}	with other passengers [BACHMANN, DANDL et al., 2023] 81
t_i	Incident duration 102–105, 107–116, 118–120, 122–126, 128–132, 136–142, 144–148, 151, 153, 206, 207, 224–236
$t^p_{e,ps}$	End time of passenger redirection at $ps \in \bar{PS_p}$ of $p \in P_{od}$ 90
$t^p_{s,ps}$	Start time of passenger redirection at $ps\in ar{PS}_p$ of $p\in P_{od}$ 90
T^p_{od}	Redirection duration of passengers associated with od for path $p \in \mathit{P_{od}}$
	81–83, 85, 87, 92, 95, 100, 101, 106, 130, 144, 150, 155, 158, 160, 164, 203

$\begin{array}{c}tt_{0}^{od}\\tt_{\tilde{p}}^{od}\\tt_{pl}^{b}\\tt_{ps}^{b}\\tt_{u}^{b}\end{array}$	Travel time needed for the original path 0 by od 81, 85 Travel time needed for assigned path $\tilde{p} \in P_{od}$ by od 87 Travel time of bus $b \in B_{sup}$ from its original PT line route to the insufficient PT line route of $pl \in PL_{ins}^{ps}$ of $ps \in PS_p$ 89, 90 Travel time of bus $b \in B_{sup}$ along the insufficient PT line $pl \in PL_{ins}^{ps}$ to the start of path segment $ps \in PS_p$ 90 Travel time needed for path p 81–83, 85, 87 Waiting time of PT user $u \in U_{od}$ until the end of the incident (t_i^e) 85
$U_a \subset U$	Set of affected PT users, subset of <i>U</i> 74, 76, 101, 102, 104, 105, 107–116, 118–124, 126–129, 141, 142, 145–148, 152, 153, 206, 207, 209, 210, 224–236
$U_{ida} \subset U$	Set of indirectly affected PT users, subset of U 74, 85, 107, 108, 110, 115, 116, 152
$U_{od} \subset U$	Set of PT users associated with od , subset of U 85
$U_{red} \subset U$	Set of redirected PT users, subset of U 123, 126, 127, 129, 206
$U_{una} \subset U$ U	Set of unaffected PT users, subset of U 74, 152 Set of all PT users 101, 104, 105, 107–109, 111, 113–116, 118, 120–123
0	131, 132, 145–148, 152, 153, 203, 206
$u \in U_a$	PT user in set of all affected PT users 74, 76, 85, 125, 202
$u \in U_{ida}$	PT user in set of all indirectly affected PT users 76
$u \in U_{od}$ $u \in U$	PT user in set of all PT users associated with <i>od</i> 85, 203 PT user in set of all PT users 74, 131, 132, 206
$u \in U$	
x_{od}^p	Binary decision variable which is 1 if OD pair od is assigned to path $p \in P_{od}$ 86
η	Compliance rate, which expresses the share of passengers who comply with given path advice 82, 83, 85, 100, 102–106, 108, 110, 111, 113, 116, 120, 122, 123, 125, 127, 130–132, 140, 145–147, 152, 158, 160, 163, 206, 224–235
ψ	Factor by which \check{h}_p is multiplied to adjust redirection duration T_{od}^p 81, 83, 85, 101–105, 108, 109, 112–115, 118–120, 122–124, 126, 128–132, 134, 138, 141, 142, 144, 146, 148, 150, 151, 153, 155, 158, 160, 172, 174, 206, 207, 224–236
ξ	Self-redirection rate, which expresses the share of self-redirected passengers 80, 83, 85, 99, 102, 104, 105, 108–112, 114–116, 118, 119, 123–129, 131, 132, 138, 140, 144–148, 153, 160, 161, 206, 207, 209, 210, 224–236
ζ	Information rate, which expresses the share of passengers who receive path advice 82, 83, 85, 100, 102–106, 108, 110, 111, 113, 116, 120, 122, 123, 125, 127, 130–132, 140–142, 145–147, 152, 158, 163, 164, 206, 224–235

List of Figures

1.1	Thesis structure	5
2.1	Incident management procedure of an OCC [BACHMANN, BRIEM et al., 2022]	22
3.1	In the last 2 years, before the COVID-19 pandemic, which has been your primary	
~ ~	means of transport?	50
3.2	In the last 2 years, BEFORE the COVID-19 pandemic, how often do you take	Б О
2 2	Which of the following happened during the incident?	50
3.4	Which stage of your trip were you at when you found out about the incident?	53
3.5	Which of these channels did you obtain official information from?	54
3.6	How did you react to the incident?	56
3.7	What are your preferred communication channels during incidents?	57
3.8	What would have been the most important information for you during this	
	incident?	58
3.9	If the operator provides you with an alternative path within the PT network,	
0.10	would you take it?	59
3.10	How confident are you that an autonomous, i.e. driverless, vehicle (e.g. an	
	5 Neutral 10 Extremely confident)	60
3 11	Can you indicate how unpleasant or pleasant you feel now at this moment	00
0.11	when you think about the risk of a similar incident occurring again? (0 Un-	
	pleasant, 5 Neutral, 10 Pleasant)	62
3.12	Can you indicate how unpleasant or pleasant you feel now, at this moment,	
	when you think about the risk of a similar incident occurring with an autonom-	
	ous, i.e. driverless, vehicle? (0 Unpleasant, 5 Neutral, 10 Pleasant)	63
3.13	During such incidents, how willing are you to follow path instructions from the	C A
2 1 4	transport operator? (0 Not at all willing, 5 Neutral, 10 Extremely willing)	64
3.14	the autonomous i.e. driverless vehicle? (0 Not at all willing 5 Neutral	
	10 Extremely willing)	65
3.15	Correlation matrices	69
0.10		00
4.1	Simple Toy Network (STN) [BACHMANN, RAU et al., 2021]	72
4.2		76
4.3	Example for the used representation of the PT supply [BACHMANN, DANDL of al. 2023]	70
ΔΔ	Situation analysis process	70
т.т		19

4.5 4.6 4.7	Passenger redirection process	84 88 91
5.1 5.2 5.3	Simulation study procedure	94 97 99
5.4 5.5	$\xi = 0\%$ in Scenario LS ($\psi = -1$)	105
5.6	$\xi = 0\%$ in Scenario LS ($\psi = -1$)	108
5.7 5.8	Overall relative delays of $U_a \subset U$ with $\xi > 0\%$ in Scenario LS ($\psi = -1$) Overall relative delays of $U_a \subset U$ and U with $\eta \cdot \zeta$ in 10% steps with $t_i = 60$	112
5.9	minutes in Scenario LS ($\psi = -1$)	113
5.10	Overall relative and total delays of $U_a \subset U$ and U with $t_i = 60$ minutes and $\xi = 0\%$ in Scenario LR ($\psi = 0$)	114
5.11	Overall relative and total delays of $U_a \subset U$ and U with $t_i = 90$ minutes and $\xi = 0\%$ in Scenario LR ($\psi = 0$)	118
5.12	Overall relative delays of $U_a \subset U$ with $\zeta > 0/0$ in Scenario LK ($\psi = 0$) Overall relative delays of $U_a \subset U$ and U with $\eta \cdot \zeta$ in 10% steps with $t_i = 60$ minutes in Scenario LR ($\psi = 0$)	119
5.14	Overall relative waiting time and delay of $U_a \subset U$ and U with $\eta \cdot \zeta$ in 10% steps in Scenario LS with $t_i = 60$ minutes ($\psi = -1$)	122
5.15	steps in Scenario LR with $t_i = 60$ minutes ($\psi = 0$)	123 124
5.17 5.18	Delay distribution of $U_{red} \subset U$ and $U_a \subset U$ in Scenario LS $(\psi = -1)$ Delay distribution of $U_a \subset U$ in Scenario LR $(\psi = 0)$	126 128
5.19 5.20 5.21	Delay distribution of $U_{red} \subset U$ and $U_a \subset U$ in Scenario LR ($\psi = 0$) Average delays by departure time of $u \in U$ with $\xi = 0\%$ in Scenario LS ($\psi = -1$) Average delays by departure time of $u \in U$ with $\xi = 0\%$ in Scenario LR ($\psi = -1$)	129 131 132
5.21 5.22 5.23	Stop setup at Nodes 8 and 10 in Scenario LS	133 134
5.24 5.25	Passenger crowding at Nodes 8 and 10 with $t_i = 60$ minutes in both scenarios Passenger crowding at Nodes 8 and 10 with $t_i = 60$ minutes in Scenarios LS	136
5.26 5.27	PT line occupancies in Scenarios LS and LR $\ldots \ldots .$ Relative delays of $U_a \subset U$ comparing merely five $od \in OD_a$ to all $od \in OD_a$	137 139
5.28	informed with $\zeta = 100\%$ in Scenarios LS $(\psi = -1)$	141
	informed with $\zeta = 100\%$ in Scenarios LR ($\psi = 0$)	142
5.29 5.30	Mandl PT network served by the DART system with disposition measures Overall relative delays in Scenario LS in a bus and the DART system with	145
--------------	--	--------------
	$t_i = 60 \text{ minutes and } \xi = 10\% \ (\psi = -2) \ \dots \ $. 146
5.31	Overall relative delays in Scenario LR in a bus and the DART system with $t_i = 30$ minutes and $\xi = 10\%$ ($\psi = -2$)	148
5.32	PT network of the city of Kassel	150
5.33	Overall relative and total passenger delays with an incident duration (t_i) of 60 minutes and $\xi = 0\%$ in Scenario LS $(\psi = 0)$. 153
6.1	CAPR integrated in the ITCS architecture [BACHMANN, TSAKARESTOS et	1.00
62	al., 2023]	. 166
0.2	from BACHMANN, BRIEM et al. [2022]	. 169
B.1	How long have you been resided in the respective city or its suburban area?	212
B.2	How old are you?	. 213
в.з В.4	What is the highest level of education you have completed?	213
B.5	Which of these best describes your current labour status?	214
B.6	How do you usually find your transit route within the mass public transport	04 F
R 7	If the operator were able to give you a route suggestion tailored to your travel	. 215
D.1	needs and preferences, which information would you be willing to share?	216
B.8	How did you notice that there was an incident?	. 217
B.9	What information was given to you?	218
B.10	Which of these characteristics is most important to you in the alternative path	219
C_{2}	The original OD matrix of the Mandl network [MANDL, 1979]	. 220 221
C.3	Delay distribution of $U_a \subset U$ with outliers in Scenario LR ($\psi = 0$)	. 236

List of Tables

2.1 2.2 2.3	$\begin{array}{llllllllllllllllllllllllllllllllllll$	8 12 12
2.4	Most common alternatives at different stages [of a passenger's trip] [LIN, SHALABY et al., 2016]	14
2.5	Overview of academic investigations [BACHMANN, TSAKARESTOS et al., 2023]	35
2.5	Overview of academic investigations [BACHMANN, TSAKARESTOS et al., 2023] (continued)	36
2.5 2.6 2.6	Overview of academic investigations [BACHMANN, TSAKARESTOS et al., 2023] (continued)	37 44 45
3.1	Information about the PT user survey	48
5.1 5.2	Lowest relative delays of each case for $U_a \subset U$ with $\xi = 0\%$ in Scenario LS (HR: Heuristic Redirection; OR: Optimised Redirection; CR+HR: Capacity Reallocation with Heuristic Redirection; CR+OR: Capacity Reallocation with Optimised Redirection)	102 104
7.1	Published and planned studies	176
C.1 C.2 C.3 C.4 C.5 C.6 C.7 C.8 C.9	OD matrix for Mandl network, as used in this thesis	222 223 224 225 226 227 228 229 230
C.10 C.11	Relative delay of $U_a \subset U$; Scenario LR; Approach OR	231 232

- C.14 Lowest relative delays of each case for $U_a \subset U$ with $\xi = 30\%$ in Scenario LR $\,$. 235

Annexe

A Incident Examples in Detail

One example of an often occurring kind of incident is falling passengers. For instance, if a bus driver needs to break hard or takes a turn too rapidly, sometimes, passengers standing in the bus and not holding on tightly enough fall during the ride. In most cases, they do not severely hurt themselves. However, in any case, the bus driver needs to check whether the passenger needs medical attention and/or wants to file a complaint. If fallen passengers hurt themselves severely and they need medical attention, the bus needs to wait until the doctor arrives and takes care of the passengers. In such cases, the bus cannot continue as planned and the run might even have to be cancelled. This is even worse if the bus needs to wait at a stop that is consequently blocked.

Another example of an often occurring kind of incident is a traffic accident. In the event of a traffic accident on a road or at an intersection where bus lines operate the accident might block the way for the buses of the according PT line. Consequently, those have to wait or take a detour which takes additional travel time which results in delays for the PT service and the passengers. This kind of situation worsens if a bus is involved in an accident itself. In this case, the bus needs to wait until the police arrive and document the accident before the bus can continue its trip, provided the bus and driver are still fit for duty. In severe cases with a broken-down bus and injured people, the run needs to be cancelled, and subsequently runs need to be rerouted for a longer period.

Buses are rather flexible, and a lot of incident situations can be easily dissolved, or their impact at least mitigated by rerouting them. Rail-bound services such as trams and metro, however, are considerably more restricted and therefore limited in their responses to incidents. An incident occurs sometimes just due to a parking offender standing too close to tram tracks. The tram needs to wait until the parking car is removed which can take some time. Due to fewer rerouting options in railway networks, this situation easily causes queuing of trams. In rare cases, such an incident also occurs through the deployment of emergency forces such as ambulance, the fire department, or police, if these park their vehicles on tram tracks.

One last description of a typical incident is about signal failures in rail-bound PT services. For example, if a metro signal fails, metro trains need to stop and wait until the signal is fixed. Since metro trains usually operate at a high frequency and possess a high vehicle capacity and high occupancies, especially during peak hours, such an incident can easily cause severe disturbances, depending on how quickly the signal can be fixed. Knowledge about incidents

has been gained through the author's experiences and visits to OCCs [Bachmann, Briem et al., 2022].

B Further Survey Results

B.1 Sample Description



Figure B.1: How long have you been resided in the respective city or its suburban area?



Figure B.2: How old are you?



Figure B.3: What is your gender?



Figure B.4: What is the highest level of education you have completed?



Figure B.5: Which of these best describes your current labour status?



Figure B.6: How do you usually find your transit route within the mass public transport network?



Figure B.7: If the operator were able to give you a route suggestion tailored to your travel needs and preferences, which information would you be willing to share?



B.2 Experienced Incidents

Figure B.8: How did you notice that there was an incident?



Berlin (N=355) Singapore (N=121)

Figure B.9: What information was given to you?



B.3 Passengers' Preferences

Figure B.10: Which of these characteristics is most important to you in the alternative path?

C Further Information About the Case Studies

C.1 The Original Mandl Network



Figure C.1: The original Mandl network [Mandl, 1979]

	2	3	4	5	6	7	8	9	10	11	12	13	14	15
1	400	200	60	80	150	75	75	30	160	30	25	35	0	0
2		50	120	20	180	90	90	15	130	20	10	10	5	0
3			40	60	180	90	90	15	45	20	10	10	5	0
4				50	100	50	50	15	240	40	25	10	5	0
5					50	25	25	10	120	20	15	5	0	0
6						100	100	30	880	60	15	15	10	0
7							50	15	440	35	10	10	5	0
8								15	440	35	10	10	5	0
9									140	20	5	0	0	0
10										600	250	500	200	0
11											75	95	15	0
12												70	0	0
13													45	0
14														0
14														

Figure C.2: The original OD matrix of the Mandl network [Mandl, 1979]

C.2 Demand in the Mandl Network

Node-IDs	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
1	0	600	300	90	120	225	112	112	45	240	45	37	52	0	0
2	600	0	75	180	30	270	135	135	22	195	30	15	15	7	0
3	300	75	0	60	90	270	135	135	22	675	300	15	15	7	0
4	90	180	60	0	75	150	75	75	22	360	60	37	15	7	0
5	120	30	90	75	0	75	37	37	15	180	30	22	7	0	0
6	225	270	270	150	75	0	150	150	45	1320	90	22	22	15	0
7	112	135	135	75	37	150	0	75	22	660	52	15	15	7	0
8	112	135	135	75	37	150	75	0	22	660	525	15	15	7	0
9	45	22	22	22	15	45	22	22	0	210	300	7	150	0	0
10	240	195	67	360	180	1320	660	660	420	0	900	375	750	300	150
11	45	30	300	60	30	90	52	52	300	900	0	112	142	22	150
12	37	15	15	37	22	22	15	15	75	375	112	0	105	0	150
13	52	15	15	15	7	22	15	15	0	750	142	105	0	67	0
14	0	7	7	7	0	15	7	7	0	300	22	0	67	0	0
15	0	0	0	0	0	0	0	0	0	0	150	150	0	0	0

Table C.1: OD matrix for Mandl network, as used in this thesis

Node-IDs	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
1	0	x1.5	x1.5	x1.5	x1.5	x1.5									
2	x1.5	0	x1.5	x1.5	x1.5	x1.5	x1.5								
3	x1.5	x1.5	0	x1.5	x1.5	x1.5	x1.5	x1.5	x1.5	x15	x15	x1.5	x1.5	x1.5	x1.5
4	x1.5	x1.5	x1.5	0	x1.5	×1.5	x1.5	x1.5	x1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5
5	x1.5	x1.5	x1.5	x1.5	0	x1.5	x1.5	x1.5	x1.5	x1.5	x1.5	×1.5	x1.5	x1.5	x1.5
6	x1.5	x1.5	x1.5	x1.5	x1.5	0	x1.5	x1.5	x1.5	×1.5	x1.5	x1.5	x1.5	×1.5	x1.5
7	x1.5	x1.5	x1.5	x1.5	x1.5	x1.5	0	x1.5	x1.5	x1.5	x1.5	×1.5	x1.5	x1.5	x1.5
8	x1.5	0	x1.5	x1.5	x1.5	×1.5	x1.5	x1.5	x1.5						
9	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	0	×1.5	×15	×1.5	+100; ×1.5	×1.5	×1.5
10	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	x3	0	×1.5	×1.5	×1.5	×1.5	+100; ×1.5
11	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×15	×1.5	0	×1.5	×1.5	×1.5	+100; ×1.5
12	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	x15	×1.5	×1.5	0	×1.5	×1.5	+100; ×1.5
13	x1.5	x1.5	0	x1.5	x1.5										
14	x1.5	x1.5	x1.5	0	x1.5										
15	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	×1.5	+100; ×1.5	+100; ×1.5	×1.5	×1.5	0

Table C.2: Changes to the original OD matrix in Figure C.2

223

C.3 Results

PT line-splitting

			$\xi =$	0%			$\xi =$	10%			$\xi =$	30%	
ψ	t_i						η	$\cdot \zeta$					
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
	30 minutes	100.2	72.1	50.0	38.7	79.6	65.5	46.3	36.7	49.8	52.5	39.4	34.1
1	60 minutes	100.0	63.6	34.7	19.4	77.2	56.4	32.1	20.2	43.5	42.9	28.2	21.2
	90 minutes	100.1	61.2	30.9	15.3	76.7	54.0	28.5	15.7	42.2	40.5	24.7	17.4
	30 minutes	100.2	67.7	42.6	31.2	79.6	56.6	38.6	31.9	49.8	39.5	33.8	32.2
0	60 minutes	100.0	61.7	32.2	18.8	77.2	49.0	28.9	21.6	43.5	31.5	24.9	23.0
	90 minutes	100.1	59.8	29.1	14.2	76.7	47.0	25.7	17.8	42.2	28.9	21.7	20.0
	30 minutes	100.2	65.2	39.0	30.0	79.6	54.6	36.6	32.2	49.8	38.7	33.1	32.1
-1	60 minutes	100.0	60.2	30.8	19.6	77.2	47.9	28.1	22.0	43.5	31.2	24.7	23.2
	90 minutes	100.1	59.0	28.1	15.4	76.7	46.5	25.1	18.5	42.2	28.7	21.6	20.2
	30 minutes	100.2	64.2	37.8	31.9	79.6	53.3	35.7	32.5	49.8	38.5	33.0	32.4
-2	60 minutes	100.0	59.0	30.2	21.7	77.2	47.4	27.7	23.0	43.5	31.0	24.8	23.4
	90 minutes	100.1	58.6	27.8	16.8	76.7	46.2	24.8	18.7	42.2	28.5	21.5	20.3
	30 minutes	100.2	63.7	37.1	32.2	88.9	57.6	36.1	32.2	67.9	47.2	34.0	32.2
-3	60 minutes	100.0	58.7	29.9	22.3	87.5	52.2	28.6	22.7	64.0	40.9	26.2	23.3
	90 minutes	100.1	58.3	27.5	17.9	87.3	51.5	25.9	18.2	63.4	38.7	23.4	19.5

Table C.3: Relative delay of $U_a \subset U$; Scenario LS; Approach HR

			$\xi =$	0%			$\xi =$	10%			$\xi =$	30%	
ψ	t_i						η	$\cdot \zeta$					
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
	30 minutes	100.2	71.5	49.9	38.1	79.6	64.9	46.3	36.4	49.8	51.6	39.3	34.5
1	60 minutes	100.0	63.3	34.8	17.9	77.2	56.0	32.3	18.1	43.5	42.6	28.2	19.5
	90 minutes	100.1	61.1	30.8	12.4	76.7	54.0	28.5	13.3	42.2	40.3	24.7	15.3
	30 minutes	100.2	66.8	42.4	30.6	79.6	56.1	38.5	30.4	49.8	38.9	33.7	32.3
0	60 minutes	100.0	61.7	32.4	15.3	77.2	48.8	29.1	17.8	43.5	31.1	24.9	22.9
	90 minutes	100.1	59.7	29.1	10.7	76.7	46.8	25.6	13.9	42.2	28.7	21.6	19.7
	30 minutes	100.2	64.9	39.3	27.7	79.6	54.3	36.7	28.5	49.8	38.3	33.0	32.2
-1	60 minutes	100.0	59.8	30.9	14.6	77.2	47.7	27.9	17.0	43.5	30.9	24.8	23.0
	90 minutes	100.1	59.0	28.2	10.1	76.7	46.3	25.2	13.5	42.2	28.4	21.6	19.9
	30 minutes	100.2	64.7	39.2	28.6	79.6	53.6	36.9	28.8	49.8	38.5	33.6	33.1
-2	60 minutes	100.0	59.3	30.7	14.4	77.2	47.2	27.7	16.8	43.5	30.8	24.8	23.2
	90 minutes	100.1	58.5	27.8	10.0	76.7	45.9	25.0	13.4	42.2	28.4	21.6	19.9
	30 minutes	100.2	65.5	41.7	29.5	88.9	58.7	38.8	27.9	67.9	47.3	36.1	30.6
-3	60 minutes	100.0	58.2	30.0	14.6	87.5	52.2	28.7	15.3	64.0	40.6	26.3	17.4
	90 minutes	100.1	58.1	27.6	10.1	87.3	51.4	26.0	11.2	63.4	38.7	23.4	13.8

Table C.4: Relative delay of $U_a \subset U$; Scenario LS; Approach OR

			ξ =	= 0%			$\xi =$	= 10%			$\xi =$	= 30%	
ψ	t_i						η	$\cdot \zeta$					
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
	30 minutes	NA	NA	NA	36.8	NA	NA	NA	34.8	NA	NA	NA	NA
1	60 minutes	NA	NA	NA	18.2	NA	NA	NA	17.9	NA	NA	NA	18.6
	90 minutes	NA	NA	NA	13.3	NA	NA	NA	13.9	NA	NA	NA	15.6
	30 minutes	NA	NA	NA	29.5	NA	NA	NA	29.1	NA	NA	NA	NA
0	60 minutes	NA	NA	NA	17.9	NA	NA	NA	18.7	NA	NA	NA	18.6
	90 minutes	NA	NA	NA	13.4	NA	NA	NA	16.3	NA	NA	NA	16.0
	30 minutes	NA	NA	NA	27.8	NA	NA	NA	NA	NA	NA	NA	NA
-1	60 minutes	NA	NA	NA	18.2	NA	NA	NA	18.7	NA	NA	NA	18.7
	90 minutes	NA	NA	NA	13.8	NA	NA	NA	16.4	NA	NA	NA	15.8
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
-2	60 minutes	NA	NA	NA	18.9	NA	NA	NA	18.9	NA	NA	NA	NA
	90 minutes	NA	NA	NA	14.1	NA	NA	NA	16.5	NA	NA	NA	NA
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
-3	60 minutes	NA	NA	NA	18.9	NA	NA	NA	18.9	NA	NA	NA	NA
	90 minutes	NA	NA	NA	15.4	NA	NA	NA	15.9	NA	NA	NA	NA

Table C.5: Relative delay of $U_a \subset U$; Scenario LS; Approach CR+HR

			ξ =	= 0%			ξ =	= 10%			ξ =	= 30%	
ψ	t_i						η	$1\cdot\zeta$					
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
	30 minutes	NA	NA	NA	36.3	NA	NA	NA	NA	NA	NA	NA	NA
1	60 minutes	NA	NA	NA	18.2	NA	NA	NA	18.3	NA	NA	NA	NA
	90 minutes	NA	NA	NA	13.5	NA	NA	NA	14.3	NA	NA	NA	16.3
	30 minutes	NA	NA	NA	29.8	NA	NA	NA	29.1	NA	NA	NA	NA
0	60 minutes	NA	NA	NA	17.9	NA	NA	NA	18.7	NA	NA	NA	NA
	90 minutes	NA	NA	NA	14.9	NA	NA	NA	16.3	NA	NA	NA	15.4
	30 minutes	NA	NA	NA	27.9	NA	NA	NA	NA	NA	NA	NA	NA
-1	60 minutes	NA	NA	NA	18.2	NA	NA	NA	18.7	NA	NA	NA	NA
	90 minutes	NA	NA	NA	14.0	NA	NA	NA	16.4	NA	NA	NA	15.8
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
-2	60 minutes	NA	NA	NA	18.9	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	14.8	NA	NA	NA	16.5	NA	NA	NA	NA
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
-3	60 minutes	NA	NA	NA	18.8	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	15.9	NA	NA	NA	15.9	NA	NA	NA	NA

Table C.6: Relative delay of $U_a \subset U$; Scenario LS; Approach CR+OR

a/1	+		η	$\cdot \zeta$	
ψ	ι_i	0%	35%	70%	100%
	30 minutes	HR, OR: 79.6	OR: 64.9	HR, OR: 46.3	CR+HR: 34.8
1	60 minutes	HR, OR: 77.2	OR: 56.0	HR: 32.1	CR+HR: 17.9
	90 minutes	HR, OR: 76.7	HR, OR: 54.0	HR, OR: 28.5	OR: 13.3
	30 minutos		OP: 56.1		CR+HR,
0	50 minutes	TIIX, OIX. 79.0	ON: 50.1	011. 30.3	CR+OR: 29.1
0	60 minutes	HR, OR: 77.2	OR: 48.8	HR: 28.9	OR: 17.8
	90 minutes	HR, OR: 76.7	OR: 46.8	OR: 25.6	OR: 13.9
	30 minutes	HR, OR: 79.6	OR: 54.3	HR: 36.6	OR: 28.5
-1	60 minutes	HR, OR: 77.2	OR: 47.7	OR: 27.9	OR: 17.0
	90 minutes	HR, OR: 76.7	OR: 46.3	HR: 25.1	OR: 13.5
	30 minutes	HR, OR: 79.6	HR: 53.3	HR: 35.7	OR: 28.8
-2	60 minutes	HR, OR: 77.2	OR: 47.2	HR, OR: 27.7	OR: 16.8
	90 minutes	HR, OR: 76.7	OR: 45.9	HR: 24.8	OR: 13.4
	30 minutes	HR, OR: 88.9	HR: 57.6	HR: 36.1	OR: 27.9
-3	60 minutes	HR, OR: 87.5	HR, OR: 52.2	HR: 28.6	OR: 15.3
	90 minutes	HR, OR: 87.3	OR: 51.4	HR: 25.9	OR: 11.2

Annexe

Table C.7: Lowest relative delays of each case for $U_a \subset U$ with $\xi = 10\%$ in Scenario LS

a/ 1	+			$\eta\cdot\zeta$	
Ψ	ι_i	0%	35%	70%	100%
	30 minutes	HR, OR: 49.8	OR: 51.6	OR: 39.3	HR: 34.1
1	60 minutes	HR, OR: 43.5	OR: 42.6	HR, OR: 28.2	CR+HR: 18.6
	90 minutes	HR, OR: 42.2	OR: 40.3	HR, OR: 24.7	OR: 15.3
	30 minutes	HR, OR: 49.8	OR: 38.9	OR: 33.7	HR: 32.2
0	60 minutes	HR, OR: 43.5	OR: 31.1	HR, OR: 24.9	CR+HR: 18.6
	90 minutes	HR, OR: 42.2	OR: 28.7	OR: 21.6	CR+OR: 15.4
	30 minutes	HR, OR: 49.8	OR: 38.3	OR: 33.0	HR: 32.1
-1	60 minutes	HR, OR: 43.5	OR: 30.9	HR: 24.7	CR+HR: 18.7
	90 minutes	HR, OR: 42.2	OR: 28.4	HR, OR: 21.6	CR+HR, CR+OR: 15.8
	30 minutes	HR, OR: 49.8	HR, OR: 38.5	HR: 33.0	HR: 32.4
-2	60 minutes	HR, OR: 43.5	OR: 30.8	HR, OR: 24.8	OR: 23.2
	90 minutes	HR, OR: 42.2	OR: 28.4	HR: 21.5	OR: 19.9
	30 minutes	HR, OR: 67.9	HR: 47.2	HR: 34.0	OR: 30.6
-3	60 minutes	HR, OR: 64.0	OR: 40.6	HR: 26.2	OR: 17.4
	90 minutes	HR, OR: 63.4	HR, OR: 38.7	HR, OR: 23.4	OR: 13.8

Table C.8: Lowest relative delays of each case for $U_a \subset U$ with $\xi = 30\%$ in Scenario LS

PT line-rerouting

			$\xi =$	0%			$\xi =$	10%			$\xi =$	30%	
ψ	t_i						η	$\cdot \zeta$					
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
	30 minutes	100.0	93.5	87.0	80.8	97.9	91.1	86.3	80.7	93.8	89.2	84.4	80.3
1	60 minutes	100.0	89.9	78.4	70.1	95.9	87.7	77.6	70.0	91.2	84.2	75.8	69.8
	90 minutes	100.0	85.5	71.4	60.8	95.9	82.6	71.0	60.8	87.2	76.9	68.8	60.7
	30 minutes	100.0	93.1	85.7	79.2	97.9	92.1	85.4	79.2	93.8	89.3	84.3	79.2
0	60 minutes	100.0	89.4	78.6	69.5	95.9	85.9	77.8	69.5	91.2	83.8	75.6	69.4
	90 minutes	100.0	84.5	71.7	60.5	95.9	82.4	70.3	60.5	87.2	77.1	68.6	60.5
	30 minutes	100.0	93.4	86.1	79.2	97.9	92.5	85.4	79.2	93.8	89.4	84.1	79.2
-1	60 minutes	100.0	89.1	77.9	69.4	95.9	85.9	77.3	69.4	91.2	83.6	75.9	69.4
	90 minutes	100.0	85.0	71.6	60.5	95.9	82.2	71.2	60.5	87.2	77.1	68.4	60.5
	30 minutes	100.0	93.4	86.1	79.2	97.9	92.5	85.4	79.2	93.8	89.4	84.1	79.2
-2	60 minutes	100.0	89.1	77.9	69.4	95.9	85.9	77.3	69.4	91.2	83.6	75.9	69.4
	90 minutes	100.0	85.0	71.6	60.5	95.9	82.2	71.2	60.5	87.2	77.1	68.4	60.5
	30 minutes	100.0	93.4	86.1	79.2	97.9	92.5	85.4	79.2	93.8	89.4	84.1	79.2
-3	60 minutes	100.0	89.1	77.9	69.4	95.9	85.9	77.3	69.4	91.2	83.6	75.9	69.4
	90 minutes	100.0	85.0	71.6	60.5	95.9	82.2	71.2	60.5	87.2	77.1	68.4	60.5

Table C.9: Relative delay of $U_a \subset U$; Scenario LR; Approach HR

			$\xi =$	0%			$\xi =$	10%			$\xi =$	30%	
ψ	t_i						η	$\cdot \zeta$					
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
1	30 minutes	100.0	93.7	86.9	83.9	97.9	91.6	86.3	82.7	93.8	89.3	84.0	80.5
	60 minutes	100.0	89.3	79.4	71.9	95.9	87.8	77.5	70.9	91.2	84.1	76.5	69.8
	90 minutes	100.0	85.8	71.9	61.8	95.9	82.2	71.5	61.0	87.2	77.6	68.4	60.7
	30 minutes	100.0	93.3	86.3	84.6	97.9	92.7	85.9	82.5	93.8	88.6	83.6	80.2
0	60 minutes	100.0	89.1	78.0	71.7	95.9	86.3	77.6	70.8	91.2	83.8	76.2	69.4
	90 minutes	100.0	84.8	71.6	62.0	95.9	82.4	71.5	61.3	87.2	77.4	68.9	60.5
	30 minutes	100.0	93.6	86.4	85.1	97.9	92.2	85.1	83.2	93.8	88.7	82.7	80.1
-1	60 minutes	100.0	88.9	79.2	71.6	95.9	86.5	77.3	70.8	91.2	83.4	76.6	69.4
	90 minutes	100.0	84.5	72.6	62.0	95.9	82.1	70.9	61.3	87.2	77.6	68.9	60.5
	30 minutes	100.0	93.6	86.3	85.1	97.9	92.2	85.1	83.2	93.8	88.7	82.7	80.1
-2	60 minutes	100.0	88.9	79.2	71.7	95.9	86.5	77.3	70.8	91.2	83.4	76.6	69.4
	90 minutes	100.0	84.5	72.6	62.0	95.9	82.1	70.9	61.3	87.2	77.6	68.9	60.5
	30 minutes	100.0	93.6	86.3	85.1	97.9	92.2	85.1	83.2	93.8	88.7	82.7	80.1
-3	60 minutes	100.0	88.9	79.2	71.6	95.9	86.5	77.3	70.8	91.2	83.4	76.6	69.4
	90 minutes	100.0	84.5	72.6	62.0	95.9	82.1	70.9	61.3	87.2	77.6	68.9	60.5

Table C.10: Relative delay of $U_a \subset U$; Scenario LR; Approach OR

	t_i		ξ=	= 0%			$\xi =$	= 10%			$\xi =$	= 30%	
ψ			$\eta \cdot \zeta$										
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
1	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
0	30 minutes	NA	NA	NA	78.0	NA	NA	NA	77.7	NA	NA	NA	NA
	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	30 minutes	NA	NA	NA	77.9	NA	NA	NA	77.7	NA	NA	NA	NA
-1	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	30 minutes	NA	NA	NA	77.9	NA	NA	NA	77.7	NA	NA	NA	NA
-2	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
-3	30 minutes	NA	NA	NA	77.9	NA	NA	NA	77.7	NA	NA	NA	NA
	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Table C.11: Relative delay of $U_a \subset U$; Scenario LR; Approach CR+HR

			ξ =	= 0%			ξ =	= 10%			ξ=	= 30%	
$ \psi $	t_i						$\overline{\eta}$	$1 \cdot \zeta$					
		0%	35%	70%	100%	0%	35%	70%	100%	0%	35%	70%	100%
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
1	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
0	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	32.4
	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	32.4
-1	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	32.4
-2	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	30 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	32.4
-3	60 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	90 minutes	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Table C.12: Relative delay of $U_a \subset U$; Scenario LR; Approach CR+OR

a/1	+		$\eta \cdot \zeta$						
Ψ	ι_i	0%	35%	70%	100%				
1	30 minutes	HR, OR: 97.9	HR: 91.1	HR, OR: 86.3	HR: 80.7				
	60 minutes	HR, OR: 95.9	HR: 87.7	OR: 77.5	HR: 70.0				
	90 minutes	HR, OR: 95.9	OR: 82.2	HR: 71.0	HR: 60.8				
0	30 minutes	HR, OR: 97.9	HR: 92.1	HR: 85.4	CR+HR: 77.7				
	60 minutes	HR, OR: 95.9	HR: 85.9	OR: 77.6	HR: 69.5				
	90 minutes	HR, OR: 95.9	HR, OR: 82.4	HR: 70.3	HR: 60.5				
	30 minutes	HR, OR: 97.9	OR: 92.2	OR: 85.1	CR+HR: 77.7				
-1	60 minutes	HR, OR: 95.9	HR: 85.9	HR, OR: 77.3	HR: 69.4				
	90 minutes	HR, OR: 95.9	OR: 82.1	OR: 70.9	HR: 60.5				
	30 minutes	HR, OR: 97.9	OR: 92.2	OR: 85.1	CR+HR: 77.7				
-2	60 minutes	HR, OR: 95.9	HR: 85.9	HR, OR: 77.3	HR: 69.4				
	90 minutes	HR, OR: 95.9	OR: 82.1	OR: 70.9	HR: 60.5				
	30 minutes	HR, OR: 97.9	OR: 92.2	OR: 85.1	CR+HR: 77.7				
-3	60 minutes	HR, OR: 95.9	HR: 85.9	HR, OR: 77.3	HR: 69.4				
	90 minutes	HR, OR: 95.9	OR: 82.1	OR: 70.9	HR: 60.5				

Annexe

Table C.13: Lowest relative delays of each case for $U_a \subset U$ with $\xi = 10\%$ in Scenario LR

a/1	+		$\eta \cdot \zeta$							
Ψ	ι_i	0%	35%	70%	100%					
1	30 minutes	HR, OR: 93.8	HR: 89.2	OR: 84.0	HR: 80.3					
	60 minutes	HR, OR: 91.2	OR: 84.1	HR: 75.8	HR, OR: 69.8					
	90 minutes	HR, OR: 87.2	HR: 76.9	OR: 68.4	HR, OR: 60.7					
	30 minutes	HR, OR: 93.8	OR: 88.6	OR: 83.6	HR: 79.2					
0	60 minutes	HR, OR: 91.2	HR, OR: 83.8	HR: 75.6	HR, OR: 69.4					
	90 minutes	HR, OR: 87.2	HR: 77.1	HR: 68.6	HR, OR: 60.5					
	30 minutes	HR, OR: 93.8	OR: 88.7	OR: 82.7	HR: 79.2					
-1	60 minutes	HR, OR: 91.2	OR: 83.4	HR: 75.9	HR, OR: 69.4					
	90 minutes	HR, OR: 87.2	HR: 77.1	HR: 68.4	HR, OR: 60.5					
	30 minutes	HR, OR: 93.8	OR: 88.7	OR: 82.7	HR: 79.2					
-2	60 minutes	HR, OR: 91.2	OR: 83.4	HR: 75.9	HR, OR: 69.4					
	90 minutes	HR, OR: 87.2	HR: 77.1	HR: 68.4	HR, OR: 60.5					
	30 minutes	HR, OR: 93.8	OR: 88.7	OR: 82.7	HR: 79.2					
-3	60 minutes	HR, OR: 91.2	OR: 83.4	HR: 75.9	HR, OR: 69.4					
	90 minutes	HR, OR: 87.2	HR: 77.1	HR: 68.4	HR, OR: 60.5					

Table C.14: Lowest relative delays of each case for $U_a \subset U$ with $\xi = 30\%$ in Scenario LR



Delay Distribution with Outliers



No Redirection
Heuristic Redirection (HR)
Optimised Redirection (OR)

:

:

1

1

Figure C.3: Delay distribution of $U_a \subset U$ with outliers in Scenario LR ($\psi = 0$)

Delay [minutes]